#### **CWI Tracts**

#### **Managing Editors**

K.R. Apt (CWI, Amsterdam) M. Hazewinkel (CWI, Amsterdam) J.M. Schumacher (CWI, Amsterdam) N.M. Temme (CWI, Amsterdam)

#### **Executive Editor**

M. Bakker (CWI Amsterdam, e-mail: Miente.Bakker@cwi.nl)

#### **Editorial Board**

W. Albers (Enschede)
M.S. Keane (Amsterdam)
J.K. Lenstra (Eindhoven)
P.W.H. Lemmens (Utrecht)
M. van der Put (Groningen)
A.J. van der Schaft (Enschede)
H.J. Sips (Delft, Amsterdam)
M.N. Spijker (Leiden)
H.C. Tijms (Amsterdam)

#### CWI

P.O. Box 94079, 1090 GB Amsterdam, The Netherlands Telephone + 31 - 20 592 9333 Telefax + 31 - 20 592 4199 URL http://www.cwi.nl

CWI is the nationally funded Dutch institute for research in Mathematics and Computer Science.

Nonlinear  $\mathcal{H}_{\infty}$  control: the singular case

W.C.A. Maas

1991 Mathematics Subject Classification: 93B36, 93C10, 70H20 ISBN 90 6196 468 7 NUGI-code: 811

Copyright ©1996, Stichting Mathematisch Centrum, Amsterdam Printed in the Netherlands

# Preface

Nonlinear  $\mathcal{H}_{\infty}$  control has been an important research topic during the last 5 years. Motivated by the state-space solution to the linear  $\mathcal{H}_{\infty}$  problem derived at the end of the eighties several authors have contributed to a nonlinear extension of these ideas. In most of this literature the so called regular nonlinear  $\mathcal{H}_{\infty}$  control problem is considered. Fundamental to this theory is a certain regularity assumption. When this assumption is violated we talk about the singular nonlinear  $\mathcal{H}_{\infty}$  control problem. In this book two approaches to this more general problem are presented. This monograph tries to bring the reader up to date with respect to the state-space solution of the nonlinear  $\mathcal{H}_{\infty}$  control problem, with emphasis on the singular case.

The required background is some knowledge of (nonlinear) control theory as well as some understanding of the state-space solution to the linear  $\mathcal{H}_{\infty}$  control problem. We tried to keep this book accessable for both researchers in the related fields as to graduate students by briefly recapitulating most of the important theory and providing the reader with an extensive reference list.

This book is the result of my work as a Ph.D. student at the Department of Applied Mathematics of the University of Twente in The Netherlands during the last 4 years. Needless to say several people have either directly or indirectly contributed to this book. Without claiming to be exhaustive I want to mention some of them explicitly. I would like to thank Arjan van der Schaft who gave me a perfect introduction to the area of nonlinear  $\mathcal{H}_{\infty}$  control. Henk Nijmeijer I thank for sparking my interest in nonlinear control theory. Many thanks also to Morten Dalsmo, Harry Trentelman and Carsten Scherer for there contributions to this monograph.

Nieuwegein, Oktober 1996,

Aloys Maas

.

# Contents

	Pref	face		i
1	Intr	oductio	n	1
	1.1	A brie	f history	1
	1.2	Two aj	pproaches to the singular $\mathcal{H}_{\infty}$ problem $\ldots$	3
	1.3	Organi	ization of this book	4
	1.4	Notati	on	7
2	The	L <sub>2</sub> -gair	n, and the nonlinear $\mathcal{H}_{\!\infty}$ control problem	9
	2.1	The $L_2$	2-gain of a nonlinear system	10
	2.2	The ${\cal H}$	$f_{\infty}$ control problem for a nonlinear system $\ldots$	19
		2.2.1	The regular state feedback $\mathcal{H}_{\infty}$ problem for nonlinear	
			systems	21
		2.2.2	The regular measurement feedback $\mathcal{H}_{\infty}$ problem for non-	
			linear systems	27
3	The	singula	r $\mathcal{H}_{\infty}$ problem: a cheap control approach	29
	3.1	Linear	disturbance attenuation	30
	3.2	Singul	ar nonlinear state feedback $\mathcal{H}_{\infty}$ control $\ldots$	32
		3.2.1	Singular $\mathcal{H}_{\infty}$ control $\ldots$	34
		3.2.2	Regularized $\mathcal{H}_{\infty}$ control $\ldots$	44
		3.2.3	The nonlinear system versus its linearization	48
	3.3	Singul	ar nonlinear measurement feedback $\mathcal{H}_{\infty}$ control $\ldots$	53
4	The	$\mathcal{H}_{\!\infty}$ alr	nost disturbance decoupling problem	61
	4.1	Almos	t disturbance decoupling for affine nonlinear	
		system	18	61

5	The	singula	r $\mathcal{H}_{\!\infty}$ problem: a geometric approach	<b>79</b>
	5.1	Singul	ar linear $\mathcal{H}_{\infty}$ control by state feedback $\ldots \ldots \ldots$	80
		5.1.1	State transformation for linear systems with $K = 0$	81
			Conditioned invariance	81
			State transformation	82
		5.1.2	State transformation for linear system with $K \neq 0$	83
		5.1.3	The quadratic matrix inequality	86
		5.1.4	The state feedback $\mathcal{H}_{\infty}$ control problem $\ldots \ldots \ldots$	87
		5.1.5	Factorization approach	89
	5.2	Singul	ar $\mathcal{H}_{\infty}$ problem for general nonlinear systems	91
		5.2.1	State transformation for affine nonlinear systems with-	
			out direct feedthrough from inputs to outputs	92
			Conditioned invariance	92
			State transformation	93
		5.2.2	State transformation for general nonlinear systems	97
		5.2.3	Special cases	103
			Regular $\mathcal{H}_{\infty}$ problem $\ldots \ldots \ldots \ldots \ldots$	103
			Full singular $\mathcal{H}_{\infty}$ problem $\ldots \ldots \ldots \ldots$	104
		5.2.4	The nonlinear state feedback $\mathcal{H}_{\infty}$ problem: sufficient	
			conditions	105
		5.2.5	The regular $\mathcal{H}_{\infty}$ subproblem	115
		5.2.6	The $\mathcal{H}_{\infty}$ almost disturbance decoupling problem $\ldots$	120
		5.2.7	Factorization approach	124
6	Exa	mnles		133
Ŭ	6.1	Tracki	ng of the orientation of a rigid body	133
		6.1.1	Cheap control approach	136
		6.1.2	Geometric approach	142
	6.2	Inverte	ed pendulum on a cart	150
		6.2.1	Cheap control approach	152
		6.2.2	Geometric approach	154
		6.2.3	Comparison of the two controllers	159
_				<b>_</b>
7	Rob	ust stab	pilization under gain-bounded uncertainties	165
	7.1	Param	eter uncertainty	165
	7.2	Multip	Discative perturbations	173

\_\_\_\_\_

iv

Contents
----------

8	Conclusions           8.1         Summary	<b>177</b> 177 180		
Α	Notions from differential geometry			
	Bibliography	187		
	Index	195		

v

# **Chapter 1**

# Introduction

This book deals with the singular  $\mathcal{H}_{\infty}$  problem for nonlinear systems. In this first chapter a brief overview is given of the state space approach to  $\mathcal{H}_{\infty}$  theory both for linear and nonlinear systems. In particular we pay attention to the two approaches used to solve the linear singular  $\mathcal{H}_{\infty}$  problem. Finally, the organization of the book is clarified and some standard notations are recalled or introduced.

## **1.1 A brief history**

System and control theory is a relatively new research field. The essential feature of system theory is the concept of system. A system is described as a process seen as a part of reality. The interconnection of this process with the outer world is described by external signals. This description has proved to be useful to model and control all kinds of processes.

One of the most powerful techniques in system and control theory is undoubtedly feedback. Roughly speaking a feedback is a mechanism which corrects the inputs of the process based on knowledge of the outputs. A fundamental problem of theoretical and practical interest in control theory is the design of feedbacks (controllers) that lead to a desirable performance not only for our nominal process, but also for the process under all kinds of disturbances. Over the years several approaches to this problem have been developed. One approach that was studied in the sixties and the seventies is known as the linear quadratic Gaussian (LQG) design method. In this method some statistical (Gaussian) properties are ascribed to the disturbances. Another approach which originates back to the beginning of the eighties ([Za 81]) is called  $\mathcal{H}_{\infty}$  optimization. This method can be viewed as a worst-case design methodology in the frequency domain. In fact  $\mathcal{H}_{\infty}$  stands for the space of complex functions which are bounded and analytic in the closed right half of the complex plane. This space is called Hardy space, after the British mathematician Godfrey H. Hardy (1877-1947). The  $\mathcal{H}_{\infty}$  optimal control theory was developed in response to the need for a synthesis procedure which explicitly incorporates modelling errors and external disturbances whose statistical nature is unknown.

Although originally this  $\mathcal{H}_{\infty}$  optimization method was formulated in the frequency domain it can also be described in the time domain. The  $\mathcal{H}_{\infty}$  norm from the exogenous disturbance inputs to the to-be-controlled variables in the frequency domain, used to describe the control objectives, is equal to the  $L_2$ induced norm for the time-domain versions, under the constraint of internal stability. A breakthrough in the  $\mathcal{H}_{\infty}$  theory for linear systems was the state-space solution to the  $\mathcal{H}_{\infty}$  problem (see for instance [DGKF 89], [KPR 88], [Sch 91], [St 92]). The state of a system summarizes the information about the system and the external signals acting on the system insofar as relevant for the future behavior. In the state space approach to system theory we distinguish two different classes of feedbacks, namely state feedback and measurement feedback. State feedback can be used when the state can be measured completely. In other cases we can only use measurement feedback based on a possibly disturbed function of the state. Both the state feedback solution and the measurement feedback solution to the  $\mathcal{H}_{\infty}$  problem are described using Riccati equations. The tools used to derive these results are familiar from the linear quadratic (LQ) and the linear quadratic Gaussian (LQG) problem. These tools are related to the notion of dissipativity (see [Wi 71]). The property of finite  $L_2$ -induced norm of a stable system, also called finite  $L_2$ -gain, can be characterized as the dissipativity of the system with respect to a certain supply rate.

The suboptimal  $\mathcal{H}_{\infty}$  problem can also be formulated as a two player, zero sum linear quadratic differential game, where the disturbances are considered as the maximizing player whose goal it is to maximize a certain cost criterium, while the controls denote the minimizing player whose goal it is to minimize the same cost criterion ([BB 90], [BO 82]).

Both the theory of dissipative systems and the theory of differential games have been also developed for nonlinear systems ([Wi 72], [BO 82]). Essential in these theories are Hamilton-Jacobi equations which extend the Riccati equations used in the linear theory. Therefore it was not surprising that after the state-space solution of the  $\mathcal{H}_{\infty}$  problem for linear systems had been found ([DGKF 89]) the approach was sought to be extended to nonlinear systems (see [vdS 91], [vdS 92], [IsAs 92], [BHW 93]). The solution of the nonlinear state feedback  $\mathcal{H}_{\infty}$  problem was described using a Hamilton-Jacobi inequality. The nonlinear measurement feedback  $\mathcal{H}_{\infty}$  problem is up to now not completely understood. Nevertheless sufficient conditions for the existence of controllers of a specific form solving the regular measurement feedback  $\mathcal{H}_{\infty}$  problem have been derived ([vdS 93], [IsAs 92]). Also necessary conditions for solvability of the problem have been given.

Most of the results are concerned with the *regular*  $\mathcal{H}_{\infty}$  problem for nonlinear systems which are affine in the inputs and the disturbances. Recently an extension of some of these results to general nonlinear systems have been made ([IsKa 95], see also [BHW 93]). The regularity of the  $\mathcal{H}_{\infty}$  problem is concerned with certain rank conditions on the feed through matrices. When these regularity assumptions are violated we talk about the *singular*  $\mathcal{H}_{\infty}$  problem. Singular  $\mathcal{H}_{\infty}$  problems naturally arise when considering certain robustness problems such as parameter uncertainty and multiplicative uncertainty ([HiPr 86], [St 92]).

For linear systems the solution of the regular  $\mathcal{H}_{\infty}$  problem has been extended to the singular case in two different ways. The two approaches are the cheap control approach (see citekpz87, [Pe 87a], [Pe 87b], [KPR 88], [ZK 88], [KPZ 90]) and the geometric approach ([StTr 90], [Sch 91], [St 91], [St 92]), which will be briefly described in the next section.

Clearly one of the open problems in the nonlinear  $\mathcal{H}_{\infty}$  theory is the singular nonlinear  $\mathcal{H}_{\infty}$  problem as considered in the present book. The aim of this monograph is to treat the singular nonlinear  $\mathcal{H}_{\infty}$  control problem by generalizing the concepts and ideas used in the two approaches to the singular  $\mathcal{H}_{\infty}$  problem for linear systems.

## **1.2** Two approaches to the singular $\mathcal{H}_{\infty}$ problem

Here we briefly describe the two different methods for solving the singular  $\mathcal{H}_{\infty}$  problem for linear systems.

In the first approach the singular state feedback  $\mathcal{H}_{\infty}$  problem is solved using a cheap control approach. This approach is based on regularization of the system such that the regularity assumptions are met, and the solution of the regular  $\mathcal{H}_{\infty}$  problem can be applied. The method is similar to the regularization of the singular linear optimal control problem, commonly called 'cheap control' approach. This disturbance attenuation approach was studied in the end of the eighties (see [KPZ 87], [Pe 87a], [Pe 87b], [KPR 88], [ZK 88], [KPZ 90]). The existence of a state feedback solution to the problem is characterized by

the solvability of a parameterized Riccati equality. Similar to the regular case an explicit state feedback can be given which solves the singular  $\mathcal{H}_{\infty}$  problem once a solution to the parameterized Riccati equality is found.

A second approach uses ideas from the geometric approach to linear system theory (see [Wo 79], [BaMa 92]). The notion of strongly controllable subspace ([Ha 83]) is used to decompose the state of the system. Correspondingly the singular  $\mathcal{H}_{\infty}$  problem can be decomposed into a regular  $\mathcal{H}_{\infty}$  subproblem and an almost disturbance decoupling problem with stability ([Wi 81], [Wi 82], [Tr 86]). Using these ideas the solvability of the singular  $\mathcal{H}_{\infty}$  problem can be characterized by Linear Matrix Inequalities (LMI) ([StTr 90], [St 91], [St 92], [Sch 91]). The tools used in this approach are similar to the geometric tools used to solve the singular LQ problem for linear systems.

### **1.3 Organization of this book**

The main part of this book is concerned with the singular nonlinear state feedback  $\mathcal{H}_{\infty}$  problem. The two approaches to solve the singular  $\mathcal{H}_{\infty}$  problem, the cheap control approach and the geometric approach, are considered in the Chapters 3 and 5 respectively. An overview of all the chapters follows:

#### Chapter 2: The $L_2$ -gain and the nonlinear $\mathcal{H}_{\infty}$ control problem

In this chapter we consider general nonlinear systems. Most of the material covered in this chapter is known. The notion of  $L_2$ -gain is introduced. Several characterizations of the finite  $L_2$ -gain property are given. Relevant results from the theory on dissipative systems, and about invariant manifolds and Hamiltonian systems are recapitulated.

In the second part of this chapter the  $\mathcal{H}_{\infty}$  problem is defined for general nonlinear systems. Also some results on the regular state feedback  $\mathcal{H}_{\infty}$  problem are recalled. A brief exposition of a way to solve the regular measurement feedback  $\mathcal{H}_{\infty}$  problem is given. The method for finding a compensator which solves the  $\mathcal{H}_{\infty}$  problem is provided by application of the the worst case certainty equivalence principle. This principle consists in first solving the state feedback problem and then replacing the actual state by the estimate of the state corresponding to the worst possible disturbance which is compatible with the applied input and the resulting output. This method will be explained in more detail at the end of Chapter 3 where it is used to construct a compensator which solves the singular nonlinear  $\mathcal{H}_{\infty}$  problem.

#### Chapter 3: The singular $\mathcal{H}_{\infty}$ problem: a cheap control approach

The singular nonlinear  $\mathcal{H}_{\infty}$  problem will be solved for nonlinear systems which are affine in the inputs and the disturbances. The results are an extension of the linear results derived using the cheap control approach for linear systems ([KPZ 87], [Pe 87a], [Pe 87b], [KPR 88], [ZK 88], [KPZ 90]). Part of these results have already appeared in [MvdS 94] and [MvdS 96].

In the first section the linear results are briefly recapitulated. The main part of this chapter is concerned with the singular nonlinear state feedback  $\mathcal{H}_{\infty}$  problem. A sufficient, and under an extra assumption also necessary, condition for the existence of a state feedback which solves the  $L_2$ -gain problem is given in terms of the solvability of a parameterized Hamilton-Jacobi inequality. This Hamilton-Jacobi inequality corresponds to a regular  $\mathcal{H}_{\infty}$  state feedback problem for a regularized version of the system. Another way to consider this singular problem is by linking it to the same problem for its linearization. It is shown that the state feedback  $\mathcal{H}_{\infty}$  control problem for the linearization is solvable if and only if the state feedback  $\mathcal{H}_{\infty}$  control problem for the nonlinear system is locally solvable. Finally in the third section we describe a way to solve the singular nonlinear measurement feedback  $\mathcal{H}_{\infty}$  problem using the worst case certainty equivalence principle.

#### Chapter 4: The $\mathcal{H}_{\infty}$ almost disturbance decoupling problem

In Chapter 4 the almost disturbance decoupling problem is considered for affine nonlinear systems without direct feedthrough from the inputs to the to-be-controlled outputs. These results are an extension of the results considered in the publications [MT 95] and [MRST 94] for single-input single-output (SISO) systems. This chapter is instrumental for the next chapter where the singular  $\mathcal{H}_{\infty}$  problem is decomposed into a regular  $\mathcal{H}_{\infty}$  problem and an almost disturbance decoupling problem. The  $\mathcal{H}_{\infty}$  almost disturbance decoupling problem considered in this chapter is a special case of a singular  $\mathcal{H}_{\infty}$  problem, since solvability of the  $\mathcal{H}_{\infty}$  almost disturbance decoupling problem implies that for every disturbance attenuation level  $\gamma > 0$  we can solve the singular  $\mathcal{H}_{\infty}$  problem.

#### Chapter 5: The singular $\mathcal{H}_{\infty}$ problem: a geometric approach

In this chapter we partially extend the results from the geometric approach to the linear singular  $\mathcal{H}_{\infty}$  problem to nonlinear systems. In the first section a recapitulation of the linear results is given. In order to extend these results to

nonlinear systems we need to introduce some notions from geometric nonlinear system theory. The maximal conditioned invariant distribution containing the input vector fields is introduced. This distribution constitutes a nonlinear extension of the strongly controllable subspace ([HaSi 83], [Ha 83]) used in the linear  $\mathcal{H}_{\infty}$  theory ([StTr 90], [St 92]). Similar to the results for linear systems we can decompose the singular nonlinear  $\mathcal{H}_{\infty}$  problem into two problems, a regular  $\mathcal{H}_{\infty}$  problem for one subsystem and an  $\mathcal{H}_{\infty}$  almost disturbance decoupling problem for another subsystem. In this way sufficient conditions for the solvability of the singular problem can be given. The solvability of the regular  $\mathcal{H}_{\infty}$  subproblem will be shown to be also necessary for the solvability of the singular  $\mathcal{H}_{\infty}$  problem. For a special class of nonlinear systems we derive necessary and sufficient conditions.

Finally a factorization idea is explored to derive conditions for solvability of the singular  $\mathcal{H}_{\infty}$  problem. In this method we reduce the nonlinear  $\mathcal{H}_{\infty}$  control problem to the nonlinear  $\mathcal{H}_{\infty}$  control problem for an auxiliary system, which is easier to solve.

#### **Chapter 6: Examples**

In this chapter two examples are considered to clarify and illustrate the ideas and results from the Chapters 3, 4 and 5. First we consider the model of the orientation of a rigid body. For this rigid body model we solve the tracking problem in two different ways. First the cheap control approach is used. This method comes down to the search for a solution to the parameterized Hamilton-Jacobi inequality. The second solution is constructed using the geometric approach from Chapter 5.

As a second example the nonlinear model of the inverted pendulum on a cart is considered. We pay attention to the selection of the parameter in the cheap control approach. Finally the two feedbacks constructed using the two methods are compared.

#### Chapter 7: Robust stabilization under gain-bounded uncertainties

Some applications of the singular  $\mathcal{H}_{\infty}$  theory are explained. These applications concern robustness problems like robust stabilization under parameter uncertainties and multiplicative perturbations.

#### **Chapter 8: Conclusions**

In this chapter we recapitulate the two methods for solving the singular state feedback  $\mathcal{H}_{\infty}$  problem for nonlinear systems. We also point out the merits and the drawbacks of the two methods. Finally some remarks are made about open problems that may be investigated in the future.

#### **Appendix A: Notions from differential geometry**

In the appendix some notions and concepts from differential geometry which are used in this monograph are briefly recapitulated.

## **1.4** Notation

Throughout this monograph we use a fairly standard notation. We denote for  $x \in \mathbb{R}^n$  by  $||x||^2$  or  $x^T x$  the squared *n*-dimensional Euclidean norm. We say that  $z: (0, \infty) \to \mathbb{R}^m$  is in  $L_2(t_0, t_1)$  if

$$\int_{t_0}^{t_1} \|z(\tau)\|^2 \,\mathrm{d}\tau < \infty.$$

A function is  $C^k$  if the function is k times continuously differentiable. A function is called smooth if it is  $C^{\infty}$ . By

 $V_x(x)$ 

we denote the *n*-dimensional row-vector of partial derivatives of a differentiable function  $V : \mathbb{R}^n \to \mathbb{R}$ , and by

 $V_{\rm r}^T(x)$ 

we denote the transposed column-vector. The Hessian matrix of the function V will be denoted by

 $V_{xx}(x)$ .

Consider

 $\dot{x} = f(x, d)$ 

where  $d \in \mathbb{R}^q$ , and x are local coordinates for a smooth state space manifold *M*. We denote by  $x(t_1) = \varphi(t_1, t_0, x_0, d)$  the state vector at time  $t_1$  with initial condition  $x(t_0) = x_0$  and input  $d : [t_0, t_1] \to \mathbb{R}^q$ .

The set of complex numbers is denoted as $\mathbb{C}$ . By $\mathbb{C}^-$ , $\mathbb{C}^0$ and $\mathbb{C}^+$ we r	nean
the subsets of $\ensuremath{\mathbb{C}}$ that have nonzero negative, zero, and nonzero positive real	l part
respectively. For an $n \times n$ matrix A we denote the set of eigenvalues by $\sigma$	(A).
By $\overline{\sigma}(A)$ we denote the maximal eigenvalue of the matrix A.	
We end definitions, remarks, lemmas, theorems, etc., by	
while we end the proofs by	

## **Chapter 2**

# The $L_2$ -gain, and the nonlinear $\mathcal{H}_{\infty}$ control problem

Originally the  $\mathcal{H}_{\infty}$  control problem was formulated as a design problem in the frequency domain ([Za 81]). However it can be also naturally formulated in the time-domain because for asymptotically stable linear systems the  $\mathcal{H}_{\infty}$ -norm of the transfer matrix from inputs to outputs is equal to the  $L_2$ -induced norm from the input time functions to the output time functions. This  $L_2$ -induced norm for a stable linear system is commonly called the  $L_2$ -gain of the system. Also for non-linear systems this  $L_2$ -gain can be defined and plays a crucial role in the studies of the so-called nonlinear  $\mathcal{H}_{\infty}$  optimal control problem ([BHW 93], [IsAs 92], [vdS 91], [vdS 92]); probably the terminology nonlinear  $L_2$ -gain optimal control problem would be more appropriate. Earlier the notion of finite  $L_2$ -gain of nonlinear systems was studied in the context of input-output stability and dissipativity for nonlinear systems ([HiMo 76], [Wi 72]). Later these results were extended by using a geometric approach based on invariant manifolds of Hamiltonian systems ([vdS 91], [vdS 92]).

In the first section of this chapter we will recapitulate those ingredients from these theories which are important for this book. In the second section the non-linear  $\mathcal{H}_{\infty}$  control problem is introduced and under a regularity assumption the known solution ([vdS 92], [vdS 93]) to the state feedback  $\mathcal{H}_{\infty}$  problem is recalled for general nonlinear systems.

## 2.1 The $L_2$ -gain of a nonlinear system

Consider nonlinear systems of the following state space form

$$\dot{x} = f(x, d)$$
  
 $z_j = h_j(x), \qquad j = 1, ..., p$ 
(2.1)

where  $x = (x_1, ..., x_n)$  are local coordinates for a smooth  $(C^{\infty})$  state space manifold denoted by M, the disturbances are denoted by  $d \in \mathbb{R}^q$  and  $z \in \mathbb{R}^p$ are the outputs. The functions f and h are  $C^k$ , with  $k \ge 1$  (at least continuously differentiable). Furthermore we assume that the system has an equilibrium in (x, d) = (0, 0), i.e., f(0, 0) = 0, and without loss of generality  $h_j(0) = 0$ , j = 1, ..., p.

For simplicity of notation we shall abbreviate the outputs as

$$z = h(x)$$

Following the literature we define the  $L_2$ -gain of the nonlinear system (2.1) in the following way ([Vi 93], [vdS 93]).

**Definition 2.1** Let  $\gamma$  be a fixed non-negative constant. The system (2.1) is said to have  $L_2$ -gain less than or equal to  $\gamma$  if for all  $x \in M$  there exists a constant  $K(x), 0 \le K(x) < \infty$ , with K(0) = 0, such that the following inequality holds

$$\int_0^t \|z(\tau)\|^2 \mathrm{d}\tau \le \gamma^2 \int_0^t \|d(\tau)\|^2 \mathrm{d}\tau + K(x_0)$$
(2.2)

for all disturbances  $d \in L_2(0, t)$  and all  $t \in [0, T]$ , with [0, T) any open interval in which the corresponding solutions of the differential equation  $\dot{x} = f(x, d)$ exist, with  $z(\tau) = h(\varphi(\tau, 0, x_0, d))$  denoting the output of (2.1) resulting from *d* for initial state  $x(0) = x_0$ .

The system has  $L_2$ -gain less than  $\gamma$  if there exists some  $0 \leq \tilde{\gamma} < \gamma$  such that the system (2.1) has  $L_2$ -gain less than or equal to  $\tilde{\gamma}$ . The  $L_2$ -gain is equal to  $\gamma$  if it has  $L_2$ -gain less than or equal to  $\gamma$  and not less than  $\gamma$ .

We can also define the finite  $L_2$ -gain property in another way. Define the *available storage* ([Wi 72]) as

$$V^{a}(x) = -\inf_{d} \frac{1}{2} \int_{0}^{t} \left( \gamma^{2} \| d(\tau) \|^{2} - \| z(\tau) \|^{2} \right) \mathrm{d}\tau$$

where the infimum is taken over all  $d \in L_2(0, t)$  and all  $t \ge 0$ , and where  $z(\tau)$  denotes the response of the system (2.1) to a disturbance d and an initial condition x(0) = x.

Now condition (2.2) is equivalent to saying that  $V^a$  is finite for every  $x \in M$ and  $V^a(0) = 0$ . A third equivalent way to define the finite  $L_2$ -gain property is by stating that the system (2.1) has  $L_2$ -gain less than or equal to  $\gamma$  if the system (2.1) is *dissipative* with respect to the *supply rate*  $s(d, z) = \frac{1}{2}\gamma^2 ||d||^2 - \frac{1}{2}||z||^2$ , in the sense that there exists a solution  $V \ge 0$  (called a *storage function*) to the *integral dissipation inequality* 

$$V(x(t_1)) - V(x(t_0)) \le \frac{1}{2} \int_{t_0}^{t_1} (\gamma^2 \| d(\tau) \|^2 - \| z(\tau) \|^2) d\tau, \quad V(0) = 0 \quad (2.3)$$

for all  $t_1 \ge t_0$  and all  $d \in L_2(t_0, t_1)$ . If  $V^a$  is finite for all x then it follows (see [Wi 72]) that  $V^a$  satisfies (2.3), while it is clear that  $V^a(x) \ge 0$  and  $V^a(0) = 0$ . Furthermore  $V^a$  is the minimal function  $V : M \to \mathbb{R}^+$  which satisfies this integral dissipation inequality, i.e., the minimal *storage function*. Correspondingly  $2V^a(x)$  is the minimal constant K(x) for which the inequality (2.2) holds.

If there exists a *continuously differentiable* solution V to the integral dissipation inequality (2.3) then this V is immediately seen to be a solution to the *differential dissipation inequality* 

$$V_x(x)f(x,d) \le \frac{1}{2}\gamma^2 \|d\|^2 - \frac{1}{2}\|z\|^2, \quad V(0) = 0$$
(2.4)

for all  $d \in \mathbb{R}^q$  and all  $x \in M$ . Since the Hessian at x = 0, d = 0 of

$$V_x(x) f(x, d) - \frac{1}{2}\gamma^2 ||d||^2 + \frac{1}{2} ||z||^2$$

with respect to d is equal to  $-\gamma^2 I$  it follows that at least locally near the origin we can find the worst case disturbance with respect to the inequality (2.4) given by

$$d_{\max}(x) = \arg\max_{d} \left( V_x(x) f(x, d) - \frac{1}{2}\gamma^2 \|d\|^2 + \frac{1}{2}h^T(x)h(x) \right).$$
(2.5)

Locally the differential dissipation inequality (2.4) is equivalent to the *Hamilton-Jacobi inequality* 

$$V_x(x)f(x, d_{\max}(x)) - \frac{1}{2}\gamma^2 \|d_{\max}(x)\|^2 + \frac{1}{2}h^T(x)h(x) \le 0, \quad V(0) = 0.$$
(2.6)

Summarizing:

**Theorem 2.2** The system (2.1) has  $L_2$ -gain less than or equal to  $\gamma$  if and only if there exists a solution  $V : M \to \mathbb{R}^+$  to the integral dissipation inequality (2.3) for all  $t_1 \ge t_0$ , all  $d \in L_2(t_0, t_1)$  and all  $x \in M$ .

Further, there exists a non-negative  $C^1$ -solution to the integral dissipation inequality (2.3) if and only if there exists a non-negative  $C^1$ -solution to the differential dissipation inequality (2.4) for all  $d \in \mathbb{R}^q$ .

There exists a local non-negative  $C^1$ -solution to the differential dissipation inequality (2.4) if and only if locally there exists a non-negative  $C^1$ -solution to the Hamilton-Jacobi inequality (2.6).

**Remark 2.3** When the system (2.1) is affine in d

$$\dot{x} = f(x) + \sum_{j=1}^{m} g_j(x) d_j = f(x) + g(x) d$$
  

$$z = h(x)$$
(2.7)

with g(x) an  $n \times m$  matrix, then the worst case disturbance  $d_{\max}(x)$  is globally defined and given by

$$d_{\max}(x) = \frac{1}{\gamma^2} g^T(x) V_x^T(x),$$

and hence the differential dissipation inequality for the system (2.7) is (globally) equivalent to the Hamilton-Jacobi inequality

$$V_x(x)f(x) + \frac{1}{2}\frac{1}{\gamma^2}V_x(x)g(x)g^T(x)V_x^T(x) + \frac{1}{2}h^T(x)h(x) \le 0, \quad V(0) = 0.$$

Another way to approach the finite  $L_2$ -gain property is by considering the Hamiltonian system corresponding to the finite  $L_2$ -gain for the system (2.1). Before doing so we define stability of the equilibrium for these systems. Two notions of stability will be used throughout this monograph.

**Definition 2.4** The equilibrium point x = 0 is said to be a *stable* equilibrium for the system (2.1), with d = 0, if for every neighborhood W of 0 there exists a neighborhood  $\tilde{W}$  of the origin such that for every initial condition  $x_0 \in \tilde{W}$  the solution  $\varphi(t, 0, x_0, 0)$  (d = 0) belongs to W for all times  $t \ge 0$ , i.e.,

$$\forall \varepsilon > 0, \ \exists \delta(\varepsilon) > 0 \text{ such that } \|x_0\| < \delta(\varepsilon) \Rightarrow \|\varphi(t, 0, x_0, 0)\| < \varepsilon, \quad \forall t \ge 0.$$

12

**Definition 2.5** The equilibrium point x = 0 is said to be a *locally asymptotically stable* equilibrium for the system (2.1), with d = 0, if 0 is stable and there exists a neighborhood  $\overline{W}$  of the origin such that all solutions  $\varphi(t, 0, x_0, 0)$  with  $x_0 \in \overline{W}$  converge to 0 if  $t \to \infty$ , i.e. there exists a  $\mu > 0$  such that

$$\|x_0\| < \mu \Rightarrow \lim_{t \to \infty} \varphi(t, 0, x_0, 0) = 0.$$
(2.8)

The equilibrium is said to be *globally asymptotically stable* if it is stable and all solutions  $\varphi(t, 0, x_0, 0)$  converge to 0.

Sometimes we will just say that  $\dot{x} = f(x, 0)$  is locally or globally asymptotically stable.

There is a close connection between the property of finite  $L_2$ -gain and asymptotic stability.

**Definition 2.6** The system (2.1) is called *zero-state observable* if for any trajectory such that  $d(t) \equiv 0$ ,  $z(t) \equiv 0$  implies  $x(t) \equiv 0$ .

**Theorem 2.7** ([HiMo 76], [vdS 92]) Assume the system (2.1) is zero-state observable. Suppose there exists a  $C^1$ -solution  $V \ge 0$  to either (2.3), (2.4) or (2.6). Then V(x) > 0,  $x \ne 0$ , and the free system  $\dot{x} = f(x, 0)$  is locally asymptotically stable. Furthermore, assume that V is proper (i.e., for each c > 0 the set  $\{x \in M | 0 \le V(x) \le c\}$  is compact), then  $\dot{x} = f(x, 0)$  is globally asymptotically stable.

The finite  $L_2$ -gain property can be also considered from a Hamiltonian viewpoint. I mention the main results without proofs. Details and proofs can be found in [AbMa 78], [vdS 91] and [vdS 92].

Consider the system (2.1), where the functions f,h are  $C^k$ -functions, with  $k \ge 2$ . Define the *pre-Hamiltonian*  $K_{\gamma} : T^*M \times \mathbb{R}^q \to \mathbb{R}$  as

$$K_{\gamma}(x, p, d) := p^{T} f(x, d) - \frac{1}{2} \gamma^{2} ||d||^{2} + \frac{1}{2} h^{T}(x) h(x)$$

with (x, p) the natural coordinates for the 2*n*-dimensional cotangent bundle  $T^*M$  (state x and co-state p). We note that (2.4) can be rewritten as

$$K_{\gamma}(x, V_x^T(x), d) \le 0,$$
  $V(0) = 0.$ 

The maximizing disturbance  $d^*(x, p) = \arg \max_d K_{\gamma}(x, p, d)$  can be substituted into the pre-Hamiltonian  $K_{\gamma}$ , which yields the *Hamiltonian*  $C^k$ -function  $H_{\gamma}: T^*M \to \mathbb{R}$  given by

$$H_{\gamma}(x, p) := K_{\gamma}(x, p, d^*(x, p)).$$

Note that by definition of  $d_{\max}(x)$ , see (2.5),  $d^*(x, V_x^T(x)) = d_{\max}(x)$  and hence the Hamilton-Jacobi inequality (2.6) can be written as

$$H_{\nu}(x, V_{r}^{T}(x)) \leq 0, \qquad V(0) = 0.$$

Corresponding to the Hamiltonian function  $H_{\gamma}$  we consider the  $C^{k-1}$  Hamiltonian vector field  $X_{H_{\gamma}}$  on  $T^*M$  given by

$$\dot{x}_i = \frac{\partial H_{\gamma}}{\partial p_i}(x, p) \dot{p}_i = -\frac{\partial H_{\gamma}}{\partial x_i}(x, p),$$
  $i = 1, ..., n$ 

with equilibrium (x, p) = (0, 0). Then  $V : M \to \mathbb{R}$  is a (not necessarily non-negative) solution to

$$H_{\gamma}(x, V_x^T(x)) = 0, \qquad V(0) = 0, \qquad V_x(0) = 0$$
 (2.9)

if and only if the *n*-dimensional submanifold

$$N = \{(x, p) \in T^*M | p^T = V_x(x)\}$$

is an *invariant manifold* of  $X_{H_{\gamma}}$  through (0, 0), i.e.,  $X_{H_{\gamma}}(x, p)$  is tangent to N at every point  $(x, p) \in N$ .

Now suppose that the linearization of  $X_{H_{\gamma}}$  at (0, 0) given by the *Hamiltonian matrix* 

$$DX_{H_{\gamma}}(0,0) = \begin{bmatrix} \frac{\partial^2 H_{\gamma}}{\partial x \partial p} & \frac{\partial^2 H_{\gamma}}{\partial p^2} \\ -\frac{\partial^2 H_{\gamma}}{\partial x^2} & -\frac{\partial^2 H_{\gamma}}{\partial p \partial x} \end{bmatrix} (0,0)$$

has no eigenvalues on the imaginary axis, i.e., the vector field  $X_{H_{\gamma}}$  is hyperbolic.

From the fact that  $DX_{H_{\gamma}}(0, 0)$  is a Hamiltonian matrix it follows that it has *n* eigenvalues in  $\mathbb{C}^-$  and *n* eigenvalues in  $\mathbb{C}^+$ , and therefore there exists an *n*dimensional *stable invariant manifold*  $N^-$  consisting of all points in  $T^*M$  converging to (0, 0) along the flow of  $X_{H_{\gamma}}$ . Furthermore  $N^-$  is tangent at (0, 0) to the stable eigenspace of  $DX_{H_{\gamma}}(0, 0)$  and  $N^-$  generates a  $C^k$  solution  $V^-$  of the Hamilton-Jacobi equality (2.9) if and only if  $N^-$  is *projectable* on M, i.e.,  $N^-$  is diffeomorphic to M under the  $C^{k-1}$  projection  $\pi : T^*M \to M$ , where  $\pi$  denotes the *canonical projection*  $\pi : (x, p) \mapsto x$ .

This solution  $V^-$  is not necessary non-negative. However when we additionally assume that the vector field f(x, 0) of the system (2.1) is globally asymptotically stable then  $V^-$  is nonnegative. Furthermore the minimal storage function  $V^-$  is equal to the available storage  $V^a$ . Summarizing:

**Theorem 2.8** Consider the system (2.1), with f,  $h C^k$  functions, with  $k \ge 2$ . Assume that the  $C^{k-1}$  vector field  $X_{H_{\gamma}}$  is hyperbolic in (0,0) and that  $N^-$  is projectable on M. Then there exists a unique  $C^k$  function  $V^- : M \to \mathbb{R}$  such that:

(i)

$$N^{-} = \left\{ (x, p) \in T^*M | p^T = V_x^{-}(x) \right\};$$

(ii)

 $H_{\gamma}(x, (V_x^-)^T(x)) = 0, \quad V(0) = 0, \quad V_x(0) = 0;$ 

(iii)  $\dot{x} = f(x, d^*(x, (V_x^-)^T(x)))$  is globally asymptotically stable on M.

Moreover if we also assume that f(x, 0) is globally asymptotically stable then also:

- (*iv*)  $V^- \ge 0$ , and  $V^- = V^a$ ;
- (v) the system (2.1) has  $L_2$ -gain less than or equal to  $\gamma$ .

The assumption about the *global* projectability of the submanifold  $N^-$  is hard to verify. By considering the linearization of (2.1) at the origin we can give an easily verifiable condition for the *local* projectability of the submanifold  $N^-$ .

Local instead of global projectability in Proposition 2.8 will lead to a local solution  $V^-$  and therefore to the notion of local  $L_2$ -gain of a system (2.1).

**Definition 2.9** Let  $\gamma$  be a fixed non-negative constant. The system (2.1) is said to have *locally L<sub>2</sub>-gain less than or equal to*  $\gamma$  if there exist a neighborhood

 $W \subset M$  of the origin and constants K(x),  $0 \leq K(x) < \infty$ , with K(0) = 0, such that

$$\int_0^t \|z(\tau)\|^2 \mathrm{d}\tau \le \gamma^2 \int_0^t \|d(\tau)\|^2 \mathrm{d}\tau + K(x_0)$$

for all  $t \in [0, T]$ , with [0, T) any open interval in which the corresponding solutions of the differential equation  $\dot{x} = f(x, d)$  exists, and for all  $d \in L_2(0, t)$  and  $x(0) \in W$  such that the state trajectories do not leave the neighborhood W.

The system has *locally*  $L_2$ -gain less than  $\gamma$  if there exists some  $0 \leq \tilde{\gamma} < \gamma$  such that the system (2.1) has locally  $L_2$ -gain less than or equal to  $\tilde{\gamma}$ . The *local*  $L_2$ -gain is equal to  $\gamma$  if it has local  $L_2$ -gain less than or equal to  $\gamma$  and not less than  $\gamma$  (Note that the local  $L_2$ -gain need not be defined).

The linearization of (2.1) at the origin is given by

$$\dot{\bar{x}} = F\bar{x} + G\bar{d}$$

$$\bar{z} = H\bar{x}$$
(2.10)

where  $\bar{x} \in \mathbb{R}^n$ ,  $\bar{d} \in \mathbb{R}^q$ ,  $\bar{z} \in \mathbb{R}^p$  and the matrices *F*, *G* and *H* are defined as:

$$F = \frac{\partial f}{\partial x}(0,0), \quad G = \frac{\partial f}{\partial d}(0,0), \quad H = \frac{\partial h}{\partial x}(0).$$

Then it can be shown that the Hamiltonian matrix  $DX_{H_{\gamma}}(0, 0)$  can be written in terms of the matrices F, G, and H.

**Lemma 2.10** The Hamiltonian matrix  $DX_{H_{y}}(0,0)$  is given by

$$DX_{H_{\gamma}}(0,0) = \begin{pmatrix} F & \frac{1}{\gamma^2} G G^T \\ -H^T H & -F^T \end{pmatrix}.$$
 (2.11)

**Proof** Since  $d^*(x, p)$  is the solution of  $\frac{\partial K_y}{\partial d} = 0$ 

$$p^{T}\frac{\partial f}{\partial d}(x, d^{*}(x, p)) = \gamma^{2} \left( d^{*}(x, p) \right)^{T}, \qquad \forall x, p.$$

Differentiation of this equality with respect to x in p = 0 leads to

$$\frac{\partial d^*}{\partial x}(x,0) = 0, \qquad \forall x$$

and the derivative with respect to p in (x, p) = (0, 0) is given by

$$\frac{\partial d^*}{\partial p}(0,0) = \frac{1}{\gamma^2} \left(\frac{\partial f}{\partial d}(0,0)\right)^T.$$

Hence

$$\begin{aligned} \frac{\partial H_{\gamma}}{\partial p}(x,p) &= f^{T}(x,d^{*}(x,p)) + p^{T}\frac{\partial f}{\partial d}(x,d^{*}(x,p))\frac{\partial d^{*}}{\partial p}(x,p) \\ &-\gamma^{2}\left(d^{*}(x,p)\right)^{T}\frac{\partial d^{*}}{\partial p}(x,p) \\ &= f^{T}(x,d^{*}(x,p)), \\ \frac{\partial H_{\gamma}}{\partial x}(x,p) &= p^{T}\frac{\partial f}{\partial x}(x,d^{*}(x,p)) + p^{T}\frac{\partial f}{\partial x}(x,d^{*}(x,p))\frac{\partial d^{*}}{\partial x}(x,p) \\ &-\gamma^{2}\left(d^{*}(x,p)\right)^{T}\frac{\partial d^{*}}{\partial x}(x,p) + h^{T}(x)h(x) \\ &= p^{T}\frac{\partial f}{\partial x}(x,d^{*}(x,p)) + h^{T}(x)h(x). \end{aligned}$$

Furthermore

$$\begin{aligned} \frac{\partial^2 H_{\gamma}}{\partial x \partial p}(0,0) &= \frac{\partial f}{\partial x}(0,d^*(0,0)) = \frac{\partial f}{\partial x}(0,0) = F, \\ \frac{\partial^2 H_{\gamma}}{\partial p^2}(0,0) &= \frac{\partial f}{\partial d}(0,d^*(0,0))\frac{\partial d^*}{\partial p}(0,0) \\ &= \frac{1}{\gamma^2}\frac{\partial f}{\partial d}(0,0)\left(\frac{\partial f}{\partial d}(0,0)\right)^T = \frac{1}{\gamma^2}GG^T, \\ \frac{\partial^2 H_{\gamma}}{\partial x \partial p}(0,0) &= \left(\frac{\partial f}{\partial x}(0,d^*(0,0))\right)^T = F^T, \\ \frac{\partial^2 H_{\gamma}}{\partial x^2}(0,0) &= \left(\frac{\partial h}{\partial x}(0)\right)^T \left(\frac{\partial h}{\partial x}(0)\right) = H^T H. \end{aligned}$$

So the first condition in Theorem 2.8 about the hyperbolicity of the vector field  $X_{H_{\gamma}}$  simply amounts to checking that the Hamiltonian matrix (2.11) does not have purely imaginary eigenvalues.

For the linearized system (2.10) the following result is well known.

**Theorem 2.11** Consider the linearized system (2.10). Assume (F, G) is stabilizable and the Hamiltonian matrix (2.11) does not have purely imaginary eigenvalues. Then there exists a unique symmetric solution P which satisfies:

(i) the algebraic Riccati equation

$$F^{T}P + PF + \frac{1}{\gamma^{2}}PGG^{T}P + H^{T}H = 0;$$
 (2.12)

(ii)

$$\sigma\left(F + \frac{1}{\gamma^2}GG^TP\right) \subset \mathbb{C}^-; \tag{2.13}$$

(iii)

span 
$$\begin{pmatrix} I_n \\ P \end{pmatrix}$$
 = stable eigenspace of  $\begin{pmatrix} F & \frac{1}{\gamma^2}GG^T \\ -H^TH & -F^T \end{pmatrix}$ .

If we also assume that F is asymptotically stable then moreover:

- (*iv*)  $P \geq 0$ ;
- (v) the system (2.10) has  $L_2$ -gain less than  $\gamma$ .

Thus if (F, G) is stabilizable then the stable eigenspace of  $DX_{H_{\gamma}}(0, 0)$  can be parameterized by the  $\bar{x}$  coordinates and since  $N^-$  is tangent at (0, 0) to this stable eigenspace there exists a neighborhood W of the origin in M where the stable invariant manifold  $N^-$  is locally projectable, i.e.,  $N^- \cap T^*W$  is projectable on W. Now we can derive the following local version of Theorem 2.8.

**Theorem 2.12** Consider the system (2.1), with f,  $h C^k$  functions, with  $k \ge 2$ , and its linearization (2.10). Assume that the matrix (2.11) has no eigenvalues on the imaginary axis and that (F, G) is stabilizable. Then there exists a neighborhood W of 0 in M and a unique  $C^k$  function  $V^-: W \to \mathbb{R}$  such that:

(i)

$$N^{-} = \left\{ (x, p) \in T^{*}M | p^{T} = V_{x}^{-}(x), x \in W \right\};$$

(ii)

 $H_{\gamma}(x, (V_x^-)^T(x)) = 0, \quad V(0) = 0, \quad \forall x \in W;$ 

(iii)  $\dot{x} = f(x, d^*(x, (V_x^-)^T(x)))$  is (locally) asymptotically stable on W.

Moreover if we assume that  $\dot{x} = f(x, 0)$  is asymptotically stable on a possibly smaller neighborhood W (this is implied by F asymptotically stable), then also on W:

- (*iv*)  $V^- \ge 0$  and  $V^- = V^a$ ;
- (v) the system (2.1) has locally  $L_2$ -gain less than  $\gamma$ .

If the nonlinear system (2.1) has local  $L_2$ -gain less than (or equal to)  $\gamma$  it is easy to prove that also the linearization (2.10) has  $L_2$ -gain less than (or equal to)  $\gamma$  (see [vdS 92]). Furthermore if the linearization has  $L_2$ -gain less than  $\gamma$ and is asymptotically stable then the matrix (2.11) has no eigenvalues on the imaginary axis ([Sch 91]).

Together with Theorem 2.11 and Theorem 2.12 this leads to the following corollary.

**Corollary 2.13** Consider the nonlinear system (2.1) together with its linearization (2.10). Assume F is asymptotically stable. Then the following statements are equivalent:

- (i) the linear system (2.10) has  $L_2$ -gain less than  $\gamma$ ;
- (ii) there exists a symmetric solution  $P \ge 0$  to (2.12), (2.13);
- (iii) the nonlinear system (2.1) has locally  $L_2$ -gain less than  $\gamma$ .

## **2.2** The $\mathcal{H}_{\infty}$ control problem for a nonlinear system

Consider nonlinear systems in state space form

$$\dot{x} = f(x, u, d)$$
  
 $y = g(x, d)$  (2.14)  
 $z = h(x, u)$ 

with two sets of inputs u and d, and two sets of outputs y and z. Here  $u \in \mathbb{R}^m$  denotes the vector of *control inputs*,  $d \in \mathbb{R}^q$  are the *exogenous inputs* (disturbances to be rejected and/or reference signals to be tracked),  $y \in \mathbb{R}^r$  are the *measured outputs* and  $z \in \mathbb{R}^p$  are the *to-be-controlled outputs* which could be tracking errors or cost variables. Finally the state x are local coordinates for a smooth

*n*-dimensional manifold *M*. The functions f, g, h are  $C^k$ , with  $k \ge 2$ . The system (2.14) is assumed to have an equilibrium at the origin, i.e., f(0, 0, 0) = 0, and without loss of generality also g(0, 0) = 0 and h(0, 0) = 0.

The main topic of this book will be the state feedback  $\mathcal{H}_{\infty}$  problem. In this problem it is assumed that g(x, d) = x and the goal is to construct a static state feedback such that the closed-loop system with (2.14) has  $L_2$ -gain less than (or equal to) a certain constant  $\gamma$ .

**Definition 2.14** Nonlinear state feedback  $L_2$ -gain optimal control problem: find, if existing, the smallest value  $\gamma^* \ge 0$  such that for any  $\gamma > \gamma^*$  there exists a state feedback

$$u = l(x),$$
  $l(0) = 0$  (2.15)

such that the  $L_2$ -gain of the closed loop system (2.14), (2.15) has  $L_2$ -gain less than or equal to  $\gamma$  (from d to z).

**Definition 2.15** Nonlinear state feedback  $\mathcal{H}_{\infty}$  optimal control problem: find, if existing, the smallest value  $\gamma^* \geq 0$  such that for any  $\gamma > \gamma^*$  there exists a state feedback

$$u = l(x), \qquad \qquad l(0) = 0$$

such that the closed loop system has  $L_2$ -gain less than or equal to  $\gamma$ , and the origin is locally asymptotically stable.

In the definition of the  $L_2$ -gain problem only the  $L_2$ -gain is considered, a priori without stability considerations. The main reason for introducing these two problems is that we want to consider internal stability separately. As seen in Theorem 2.7 certain stability properties can be derived from the finite  $L_2$ -gain property.

In Chapter 3 also the measurement feedback case shall be considered briefly. In this problem it is assumed that we can only measure a function of both state and disturbance, i.e., y = g(x, d).

**Definition 2.16** Nonlinear measurement feedback  $\mathcal{H}_{\infty}$  optimal control problem: find, if existing, the smallest value  $\gamma^* \ge 0$  such that for any  $\gamma > \gamma^*$  there exists a compensator

$$\xi = k(\xi, y)$$
  
 $u = m(\xi, y)$ 
(2.16)

with k(0, 0) = 0 and m(0, 0) = 0, such that the closed loop system (2.14), (2.16) has  $L_2$ -gain (from d to z) less than or equal to  $\gamma$ , and the origin is locally asymptotically stable.

In most of the literature about nonlinear  $\mathcal{H}_{\infty}$  theory it is assumed that the mapping *h* is in some sense injective from *u* to *z*. The  $\mathcal{H}_{\infty}$  problem under this regularity assumption is often referred to as the *regular*  $\mathcal{H}_{\infty}$  problem (see e.g. [BHW 93], [IsAs 92], [IsKa 95], [vdS 91], [vdS 92], [vdS 93]). The main aim of this monograph is to drop this regularity assumption. When this regularity assumption is violated the  $\mathcal{H}_{\infty}$  problem is called *singular*. Sometimes to emphasize the more general character of this  $\mathcal{H}_{\infty}$  problem it is called the *general*  $\mathcal{H}_{\infty}$  problem.

In the next subsection some of the results derived for the regular  $\mathcal{H}_{\infty}$  problem are recapitulated.

Also the linearization of the nonlinear system (2.14) around the origin shall be considered, denoted as

$$\dot{\bar{x}} = F\bar{x} + G\bar{u} + E\bar{d} \bar{y} = M\bar{x} + N\bar{d}$$

$$\bar{z} = H\bar{x} + K\bar{u}$$

$$(2.17)$$

where  $\bar{u} \in \mathbb{R}^m$ ,  $\bar{x} \in \mathbb{R}^n$ ,  $\bar{d} \in \mathbb{R}^q$ ,  $\bar{y} \in \mathbb{R}^r$ ,  $\bar{z} \in \mathbb{R}^p$  and the matrices *F*, *G*, *E*, *M*, *N*, *H* and *K* are defined as:

$$F = \frac{\partial f}{\partial x}(0,0,0), \quad G = \frac{\partial f}{\partial u}(0,0,0), \quad E = \frac{\partial f}{\partial d}(0,0,0),$$
$$M = \frac{\partial g}{\partial x}(0,0), \quad N = \frac{\partial g}{\partial d}(0,0), \quad H = \frac{\partial h}{\partial x}(0,0), \quad K = \frac{\partial h}{\partial u}(0,0).$$

The regular  $\mathcal{H}_{\infty}$  problem corresponds to the matrix K having full column rank, i.e., rank K = m.

# 2.2.1 The regular state feedback $\mathcal{H}_{\infty}$ problem for nonlinear systems

In this subsection we consider systems of the form (2.14) with g(x, d) = x and which satisfy the following regularity assumption

Assumption 1 The derivative of h with respect to u has full column rank at the origin, i.e.,

$$\operatorname{rank}\left(\frac{\partial h}{\partial u}(0,0)\right) = m.$$

We seek for a nonlinear static state feedback

$$u = l(x),$$
  $l(0) = 0$  (2.18)

such that the closed-loop of (2.14) with (2.18) has  $L_2$ -gain less than or equal to  $\gamma$  from d to z.

This  $L_2$ -gain optimal control problem can be viewed as a two player, zero sum differential game, where u is called the minimizing player which goal it is to minimize the cost criterium

$$\frac{1}{2} \int_0^t \left( \|h(x(\tau), u(\tau))\|^2 - \gamma^2 \|d(\tau)\|^2 \right) d\tau$$

for every *t*, and *d* is called the maximizing player whose goal it is to maximize the same cost criterium. The pre-Hamiltonian function associated with this game for the system (2.14) is a function  $K_{\gamma}: T^*M \times \mathbb{R}^q \times \mathbb{R}^m \to \mathbb{R}$  defined as

$$K_{\gamma}(x, p, d, u) := p^{T} f(x, u, d) + \frac{1}{2} \|h(x, u)\|^{2} - \frac{1}{2} \gamma^{2} \|d\|^{2}.$$

Under the regularity Assumption 1 this pre-Hamiltonian  $K_{\gamma}$  has a unique saddle point with respect to u and d in a neighborhood of the origin, i.e., there exist unique functions  $d^* = d^*(x, p)$  and  $u^* = u^*(x, p)$  defined around (x, p) =(0, 0), satisfying

$$\frac{\partial K_{\gamma}}{\partial d}(x, p, d^*(x, p), u^*(x, p)) = 0$$
  
$$\frac{\partial K_{\gamma}}{\partial u}(x, p, d^*(x, p), u^*(x, p)) = 0$$

with  $d^*(0, 0) = 0$  and  $u^*(0, 0) = 0$ , satisfying the saddle point condition

$$K_{\gamma}(x, p, d, u^{*}) \le K_{\gamma}(x, p, d^{*}, u^{*}) \le K_{\gamma}(x, p, d^{*}, u)$$
(2.19)

for every (x, p, d, u) around the origin. The existence of these functions can easily be seen by calculating the Hessian of  $K_{\gamma}$  with respect to d and u at the origin. This Hessian equals

$$\left(\begin{array}{cc} -\gamma^2 I & 0\\ 0 & \left(\frac{\partial h}{\partial u}(0,0)\right)^T \left(\frac{\partial h}{\partial u}(0,0)\right) \end{array}\right).$$

Because of Assumption 1 local existence of the functions  $d^*(x, p)$  and  $u^*(x, p)$  now follows from the Implicit Function Theorem.

Substituting the saddle point  $d^*$ ,  $u^*$  into the pre-Hamiltonian  $K_{\gamma}$  leads to the Hamiltonian  $H_{\gamma}: T^*M \to \mathbb{R}$  defined as

$$H_{\gamma}(x, p) := K_{\gamma}(x, p, d^*(x, p), u^*(x, p)).$$
(2.20)

The following result can be easily deduced from Theorem 2.2 and the saddle point condition (2.19).

**Theorem 2.17** Consider the system (2.14) with y = x, under the Assumption 1. Let  $\gamma > 0$ . Suppose there exists a local  $C^s$  ( $k \ge s \ge 1$ ) solution  $V \ge 0$  to

$$H_{\gamma}(x, V_x^T(x)) \le 0 \tag{2.21}$$

with V(0) = 0. Then the  $C^{s-1}$  state feedback

$$u = u^*(x, V_x^T(x))$$
(2.22)

locally solves the state feedback  $L_2$ -gain control problem, with constant  $\gamma$ , for the system (2.14).

Conversely, suppose there exists a state feedback (2.18) which solves the state feedback  $L_2$ -gain optimal control problem in the sense that there exists a differentiable solution  $V \ge 0$  to the Hamilton-Jacobi inequality (2.6) for the closed-loop system (2.14), (2.18), then V is also a solution of (2.21) and hence also the feedback (2.22) leads to a closed-loop system which has  $L_2$ -gain less than or equal to  $\gamma$ .

**Proof** The first part of the theorem follows from Theorem 2.2 and (2.19). For the converse statement we know there exists a solution V to

$$K_{\gamma}(x, V_x^T(x), d, l(x)) \leq 0$$

for all d. Then by taking u = l(x) and  $p^T = V_x(x)$  in the saddle point condition (2.19) it follows that

$$H_{\gamma}(x, V_x^T(x)) \leq K_{\gamma}(x, V_x^T(x), d^*(x, V_x^T(x)), l(x)) \leq 0.$$

**Remark 2.18** Consider the following system which is affine in the inputs u and the disturbances d

$$\dot{x} = f(x) + \sum_{j=1}^{m} g_j(x) u_j + \sum_{i=1}^{q} e_i(x) d_i z = h(x) + \sum_{i=1}^{m} k_j(x) u_j$$

In shorthand notation we will write this system as

with g(x), e(x), k(x) matrices consisting of the corresponding columns. For the system (2.23) the saddle point solution  $d^*(x, p)$  and  $u^*(x, p)$  is given by

$$d^{*}(x, p) = \frac{1}{\gamma^{2}} e^{T}(x) p$$
  
$$u^{*}(x, p) = -\left(k^{T}(x)k(x)\right)^{-1} \left(g^{T}(x)p + k^{T}(x)h(x)\right)$$

where the maximizing d is globally defined, and the minimizing u is globally defined provided the matrix  $k^{T}(x)k(x)$  is non-singular for all x. The Hamilton-Jacobi inequality (2.21) for the system (2.23) is then given by

$$V_{x}(x) f(x) + \frac{1}{2}h^{T}(x)h(x) + \frac{1}{2}\frac{1}{\gamma^{2}}V_{x}(x)e(x)e^{T}(x)V_{x}^{T}(x)$$
  
$$-\frac{1}{2}\left(V_{x}(x)g(x) + h^{T}(x)k(x)\right)\left(k^{T}(x)k(x)\right)^{-1}\left(g^{T}(x)V_{x}^{T}(x) + k^{T}(x)h(x)\right)$$
  
$$\leq 0$$
  
with  $V(0) = 0.$ 

(0) 01

We see that V solving the Hamilton-Jacobi inequality (2.21) is a candidate *Lyapunov function* for the closed loop system, and from Theorem 2.7 we deduce the following stability result:

**Theorem 2.19** Assume that the closed loop system (2.14), (2.22) is zero-state observable. Suppose there exists a  $C^1$  solution  $V \ge 0$  to (2.21). Then V > 0,  $x \ne 0$ , and the free system  $\dot{x} = f(x, u^*(x, V_x^T(x)), 0)$  is locally asymptotically stable, and globally asymptotically stable if V is proper.

Using the Hamiltonian and the machinery explained in the last section we can easily derive the following results.

**Theorem 2.20** Define the  $C^k$  Hamiltonian

$$H_{\gamma}^{l}(x, p) = K_{\gamma}(x, p, d_{l}^{*}(x, p), l(x))$$

$$= p^{T} f(x, l(x), d_{l}^{*}(x, p)) + \frac{1}{2}h^{T}(x, l(x))h(x, l(x))$$

$$-\frac{1}{2}\gamma^{2} \|d_{l}^{*}(x, p)\|^{2}$$
(2.24)

where

$$d_l^*(x, p) = \arg\max_d K_{\gamma}(x, p, d, l(x))$$

Assume that the  $C^{k-1}$  vector field  $X_{H_{\gamma}}$  is hyperbolic at (0, 0) and that the stable invariant manifold  $N^-$  of  $X_{H_{\gamma}}$  is projectable on M. Then there exists a unique  $C^k$  function  $V^-: M \to \mathbb{R}$  such that:

(i)

$$N^{-} = \{(x, p) \in T^*M | p^T = V_x^{-}(x)\};$$

(ii)

$$H_{\gamma}(x, (V_x^-)^T(x)) = 0, \qquad V(0) = 0, \qquad V_x(0) = 0;$$

(iii)  $\dot{x} = f(x, l(x), d_l^*(x, (V_x^-)^T(x)))$  is globally asymptotically stable on M.

Moreover if we additionally assume that  $\dot{x} = f(x, l(x), 0)$  is globally asymptotically stable then:

(*iv*)  $V^- \ge 0$ ;

(v) the closed-loop system (2.14), (2.18) has  $L_2$ -gain less than  $\gamma$ .

L		

Consider now the linearization of the nonlinear system (2.14), (2.17), with  $\bar{y} = \bar{x}$ . The next result is well known (see [St 92], [Sch 91]).

**Theorem 2.21** Consider the linearized system (2.17), with  $\bar{y} = \bar{x}$ , and assume the system (F, G, H, K) has no invariant zeros on the imaginary axis, i.e.,

rank 
$$\begin{pmatrix} F - j\omega I & G \\ H & K \end{pmatrix} = n + m, \quad \forall \omega \in \mathbb{R}.$$
 (2.25)

Then the following statements are equivalent:

(i) there exists a feedback law  $\bar{u} = L\bar{x}$  such that after applying this feedback to the system (2.17) the closed-loop system has  $L_2$ -gain less than  $\gamma$  and is asymptotically stable; (ii) there exists a solution  $P \ge 0$  to

$$F^{T}P + PF + \frac{1}{\gamma^{2}}PEE^{T}P$$

$$- (PG + H^{T}K)(K^{T}K)^{-1}(K^{T}H + G^{T}P) + H^{T}H = 0$$
(2.26)

satisfying

$$\sigma\left(F - G\left(K^{T}K\right)^{-1}\left(K^{T}H + G^{T}P\right) + \frac{1}{\gamma^{2}}EE^{T}P\right) \subset \mathbb{C}^{-}.$$
 (2.27)

The Hamiltonian matrix corresponding to the Hamiltonian (2.20), denoted by  $DX_{H_v}(0, 0)$ , is given by

$$DX_{H_{\gamma}}(0,0) = \begin{pmatrix} F - G(K^{T}K)^{-1}K^{T}H & -G(K^{T}K)^{-1}G^{T} + \frac{1}{\gamma^{2}}EE^{T} \\ -H^{T}(I - K(K^{T}K)^{-1}K^{T})H & -F^{T} + H^{T}K(K^{T}K)^{-1}G^{T} \end{pmatrix}.$$

Similarly as before, this matrix can be used to prove local results by considering the linearization.

**Theorem 2.22** Consider the system (2.14), with y = x, and its linearization given by (2.17), with  $\bar{y} = \bar{x}$ . Assume that (2.25) is satisfied. Then the following statements are equivalent:

- (i) there exists a feedback law  $\bar{u} = L\bar{x}$  such that after applying this feedback to the system (2.17) the closed-loop system has  $L_2$ -gain less than  $\gamma$  and is asymptotically stable;
- (ii) there exists a solution  $P \ge 0$  to (2.26) satisfying (2.27);
- (iii) there exists a neighborhood  $W \subset M$  of 0 and a nonlinear state feedback u = l(x) defined on W, such that F + GL, with  $L = \frac{\partial l}{\partial x}(0)$ , is asymptotically stable and the closed-loop system (2.14), (2.18) has locally  $L_2$ -gain less than  $\gamma$  on W (i.e., the nonlinear state feedback  $\mathcal{H}_{\infty}$  problem is solvable on W).
Finally we note that if the system (2.14) is not influenced by disturbances then the  $\mathcal{H}_{\infty}$  problem turns into an optimal control problem in the sense that the Hamiltonian  $H_{\gamma}$  does not depend on  $\gamma$ , and is equal to the (optimal) Hamiltonian for the optimal control problem of minimizing

$$\frac{1}{2}\int_0^\infty \|h(x(t))\|^2\,\mathrm{d}t.$$

In this case feedback solutions can be found by applying nonlinear optimal control theory ([LeMa 67]).

## 2.2.2 The regular measurement feedback $\mathcal{H}_{\infty}$ problem for nonlinear systems

Consider systems of the form

$$\dot{x} = f(x, u, d_1)$$
  
 $y = g(x) + d_2$  (2.28)  
 $z = h(x, u)$ 

satisfying the regularity Assumption 1. We want to construct a compensator of the form

$$\begin{aligned} \xi &= k(\xi, y) \\ u &= m(\xi, y) \end{aligned}$$

which solves the measurement feedback  $\mathcal{H}_{\infty}$  problem.

A useful method for finding such a compensator is the worst case certainty equivalence principle. This principle consists in solving first the state feedback problem and then replacing the actual state x in the feedback by the estimated state corresponding to the worst possible disturbance which is compatible with the applied input and the resulting output (see [DBB 93], [BO 82], [BB 90]).

The state feedback problem for the system (2.28), with y = x, is solved in Subsection 2.2.1. To construct the estimation of the state corresponding to the worst possible disturbance compatible with the applied input and the resulting output we have to solve a maximization problem with constraints. This maximization problem is hard to solve for general nonlinear systems. The special structure of the measurement equation of the system (2.28) however makes it possible to rewrite this constrained maximization problem as an unconstrained problem ([vdS 93]).

More details will be given in Chapter 3 where this method is used to construct a controller using a regularized version of the affine nonlinear system.

### Chapter 3

# The singular $\mathcal{H}_{\infty}$ problem: a cheap control approach

In this chapter the singular  $\mathcal{H}_{\infty}$  problem is considered from a cheap control point of view. In a sense this comes down to regularizing the singular  $\mathcal{H}_{\infty}$  problem. For linear systems this way of solving the state feedback  $\mathcal{H}_{\infty}$  problem has been studied in the end of the eighties ([KPZ 87], [Pe 87a], [Pe 87b], [KPR 88], [ZK 88], [KPZ 90]). The basic tool in this approach is a parameterized algebraic Riccati equation. This approach can be extended to nonlinear systems. We prove that the solvability of the singular state feedback  $\mathcal{H}_{\infty}$  problem can be characterized by the solvability of a parameterized Hamilton-Jacobi inequality. This Hamilton-Jacobi inequality also corresponds to the regular  $\mathcal{H}_{\infty}$  problem for a regularized version of our system. This second viewpoint leads to a slightly more conservative feedback for the singular  $\mathcal{H}_{\infty}$  problem.

First the results for linear systems shall be recapitulated. After that the main part of this chapter is devoted to the nonlinear extension of this theory in which also the connection with the  $\mathcal{H}_{\infty}$  problem for the linearization of the nonlinear system shall be explained. Finally the measurement feedback  $\mathcal{H}_{\infty}$  problem is considered in the last section. In this section we use the more conservative feedback corresponding to the regular  $\mathcal{H}_{\infty}$  problem for a regularized version of our nonlinear system in order to be able to apply the worst case certainty equivalence principle.

#### 3.1 Linear disturbance attenuation

We consider linear systems of the form (2.17), with  $\bar{y} = \bar{x}$ ,

$$\dot{\bar{x}} = F\bar{x} + G\bar{u} + E\bar{d} \bar{z} = H\bar{x} + K\bar{u}$$

$$(3.1)$$

Here we assume that there exists no direct feedthrough from the disturbances  $\bar{d}$  to the to-be-controlled variables  $\bar{z}$ . The theory, however, can be directly extended to systems with a direct feedthrough from  $\bar{d}$  to  $\bar{z}$ .

The following matrices are defined.

**Definition 3.1** Suppose rank(K) =  $m_1 \le m$ . Let  $U \in \mathbb{R}^{p \times m_1}$  and  $\Pi \in \mathbb{R}^{m_1 \times m_2}$  be any matrices such that:

- $\operatorname{rank}(U) = \operatorname{rank}(\Pi) = m_1;$
- $K = U\Pi;$
- $\Pi \Pi^T = I_{m_1}$ .

Let  $\Phi \in \mathbb{R}^{(m-m_1) \times m}$  be such that:

- $\Phi\Pi^T = 0;$
- $\Phi \Phi^T = I_{m-m_1}$  ( $\Phi$  is void if  $m_1 = m$ ).

Define

$$\Sigma := \Pi^T (U^T U)^{-1} \Pi$$
$$= \Pi^T (\Pi K^T K \Pi^T)^{-1} \Pi.$$

It should be noted that matrices U,  $\Pi$  and  $\Phi$  as in Definition 3.1 always exist.

Now the aim is to find a feedback  $\bar{u} = L\bar{x}$  for the system (3.1) such that the resulting closed-loop system has  $L_2$ -gain from  $\bar{d}$  to  $\bar{z}$  less then or equal to a constant  $\gamma$ , and such that the closed-loop system F + GL is asymptotically stable. The solvability of this problem can be characterized by the following property in terms of a parameterized algebraic Riccati equation. **Definition 3.2** Let  $\Phi$  and  $\Sigma$  be as in Definition 3.1. The system (3.1) is said to satisfy the *Algebraic Riccati Equation (ARE) with constant*  $\gamma$  if there exists a positive definite matrix Q and an  $\varepsilon > 0$  such that there exists a positive definite solution P to

$$\left(F - G\Sigma K^{T} H\right)^{T} P + P\left(F - G\Sigma K^{T} H\right) + \frac{1}{\gamma^{2}} P E E^{T} P \qquad (3.2)$$

$$-PG\Sigma G^{T}P - \frac{1}{\varepsilon}PG\Phi^{T}\Phi G^{T}P + H^{T}(I - K\Sigma K^{T})H + \varepsilon Q = 0.$$

The following lemma shows that the existence of a positive definite solution P of ARE (3.2) does not depend on the particular choice of Q.

**Lemma 3.3** ([ZK 88]) Suppose there exists a positive definite matrix  $Q \in \mathbb{R}^{n \times n}$ and a constant  $\varepsilon > 0$  such that the algebraic Riccati equation (3.2) has a positive definite solution. Then given any positive definite  $\tilde{Q} \in \mathbb{R}^{n \times n}$  there exists a constant  $\varepsilon^* > 0$  such that the ARE (3.2) with Q replaced by  $\tilde{Q}$  has a positive definite solution P for all  $\varepsilon \in (0, \varepsilon^*]$ .

For the solvability of the singular  $\mathcal{H}_{\infty}$  problem for the system (3.1) the following can be stated ([ZK 88]).

**Theorem 3.4** Consider the system (3.1). Let  $\gamma > 0$ . Then the following statements are equivalent:

- (i) there exists a linear feedback  $\bar{u} = L\bar{x}$  such that F + GL is asymptotically stable and the closed-loop system (3.1) with this feedback has  $L_2$ -gain less than  $\gamma$ ;
- (ii) the system (3.1) satisfies the Algebraic Riccati Equation with constant  $\gamma$  (3.2).

Moreover if P > 0 is a solution of the ARE (3.2) for some Q > 0 and constant  $\varepsilon > 0$  then the feedback

$$L = -\left(\frac{1}{2\varepsilon}\Phi^{T}\Phi + \Sigma\right)G^{T}P - \Sigma K^{T}H$$
(3.3)

leads to a closed-loop system which has  $L_2$ -gain from  $\overline{d}$  to  $\overline{z}$  less than  $\gamma$ , and F + GL is stable.

Furthermore, if there exists a positive definite solution P of (3.2), then there also exists a stabilizing solution of (3.2) (see [KPZ 90]).

**Theorem 3.5** Suppose for Q > 0 there exists an  $\varepsilon^* > 0$  such that (3.2) has a positive definite solution  $P_{\varepsilon}$  for every  $\varepsilon \in (0, \varepsilon^*]$ . Then for every  $\varepsilon \in (0, \varepsilon^*)$  there also exists a stabilizing solution  $\tilde{P}_{\varepsilon} > 0$  for (3.2), i.e., there exists a solution  $\tilde{P}_{\varepsilon} > 0$  to (3.2) with the additional property that

$$F - G\Sigma K^{T} H + \frac{1}{\gamma^{2}} K K^{T} \tilde{P}_{\varepsilon} - G\Sigma G^{T} \tilde{P}_{\varepsilon} - \frac{1}{\varepsilon} G \Phi^{T} \Phi G^{T} \tilde{P}_{\varepsilon}$$

is asymptotically stable.

#### **3.2** Singular nonlinear state feedback $\mathcal{H}_{\infty}$ control

To keep the ideas behind the results transparent and to avoid complex notation we will restrict attention to nonlinear systems of the form (2.14), with y = x, which are affine in the disturbances d and the inputs u,

$$\Sigma \begin{cases} \dot{x} = f(x) + g(x)u + e(x)d \\ z = h(x) + k(x)u \end{cases}$$
(3.4)

We try to extend the linear results described in the last section to solve the singular state feedback  $\mathcal{H}_{\infty}$  problem. So we want to find a nonlinear static feedback

$$u = l(x),$$
  $l(0) = 0$  (3.5)

such that the resulting closed-loop system has  $L_2$ -gain less than or equal to  $\gamma$ , i.e., cf. Definition 2.1: for every  $x \in M$  there exists a constant K(x),  $0 \leq K(x) < \infty$ , with K(0) = 0, such that

$$\int_0^t \|z(x(\tau))\|^2 d\tau \le \gamma^2 \int_0^t \|d(\tau)\|^2 d\tau + K(x(0))$$

for all  $d \in L_2[0, t]$  and all  $t \ge 0$ , with  $x(\tau)$  denoting the response of (3.4), (3.5) for initial condition x(0) = x.

We make the following constant rank assumption.

Assumption 2 There exists a neighborhood W of the origin such that

rank 
$$k(x) = m_1 \le m$$
,  $\forall x \in W$ .

#### 32

From Assumption 2 it follows that at least locally we can make similar definitions as in the linear case (see Definition 3.1).

**Definition 3.6** Define an  $(m \times (m - m_1))$ -matrix  $\beta_2(x)$  such that:

- im  $\beta_2(x) = \ker k(x);$
- $\beta_2^T(x)\beta_2(x) = I_{m-m_1};$

on a neighborhood  $\overline{W}$  of the origin, which may be smaller than W. Then define an  $(m \times m_1)$ -matrix  $\beta_1(x)$  such that:

- $\beta_1^T(x)\beta_2(x) = 0;$
- $\beta_1^T(x)\beta_1(x) = I_{m_1}$ .

For notational reasons also introduce

$$\psi(x) = \beta_1(x) \left( \beta_1^T(x) k^T(x) k(x) \beta_1(x) \right)^{-1} \beta_1^T(x).$$

Note that matrices  $\beta_1(x)$  and  $\beta_2(x)$  as in Definition 3.6 always exist. Furthermore note that from Definition 3.6 it follows that

$$\begin{pmatrix} \beta_1(x) & \beta_2(x) \end{pmatrix}^{-1} = \begin{pmatrix} \beta_1^T(x) \\ \beta_2^T(x) \end{pmatrix}.$$
 (3.6)

**Remark 3.7** In Definition 3.6 the matrices are defined in a way which is slightly different from Definition 3.1. Nevertheless Definition 3.6 is an extension of Definition 3.1. This can be shown by comparing the present Definition 3.6 with the following straightforward extension of Definition 3.1 to the nonlinear setting:

Define a  $(p \times m_1)$ -matrix v(x), a  $(m_1 \times m)$ -matrix  $\beta_1^T(x)$  and a  $((m - m_1) \times m)$ -matrix  $\beta_2^T(x)$  such that:

- rank  $\nu(x) = \operatorname{rank} \beta_1^T(x) = m_1;$
- $k(x) = \nu(x)\beta_1^T(x);$
- $\beta_1^T(x)\beta_1(x) = I_{m_1};$

• 
$$\beta_2^T(x)\beta_1(x) = 0$$
,

•  $\beta_2^T(x)\beta_2(x) = I_{m-m_1}$ .

Let now  $\beta_1(x)$  and  $\beta_2(x)$  be defined according to Definition 3.6. Then define  $\nu(x)$  as

$$\nu(x) = k(x)\beta_1(x).$$

It follows that (using (3.6)):

$$k(x) = k(x)I_m$$
  
=  $k(x) \left( \begin{array}{cc} \beta_1(x) & \beta_2(x) \end{array} \right) \left( \begin{array}{c} \beta_1^T(x) \\ \beta_2^T(x) \end{array} \right)$   
=  $\left( \begin{array}{cc} k(x)\beta_1(x) & 0 \end{array} \right) \left( \begin{array}{c} \beta_1^T(x) \\ \beta_2^T(x) \end{array} \right)$   
=  $\nu(x)\beta_1^T(x).$ 

Hence  $\nu(x)$ ,  $\beta_1^T(x)$  and  $\beta_2^T(x)$  satisfy the alternative definition given in this remark.

Conversely, if we define v(x),  $\beta_1^T(x)$  and  $\beta_2^T(x)$  according to this alternative definition then  $v = \beta_2(x)w$  for some w implies that

$$k(x)v = v(x)\beta_1^T(x)v = v(x)\beta_1^T(x)\beta_2(x)w = 0.$$

Therefore  $v \in \ker k(x)$  and  $\operatorname{im} \beta_2(x) \subset \ker k(x)$ . Finally because both  $\operatorname{im} \beta_2(x)$  and  $\ker k(x)$  have dimension  $m - m_1$ , we have that  $\operatorname{im} \beta_2(x) = \ker k(x)$ .  $\Box$ 

**Remark 3.8** As can be seen from Remark 3.7 for a linear system the matrices  $\Pi$  and  $\Phi$  in Definition 3.1 correspond to the matrices  $\beta_1^T$ , respectively  $\beta_2^T$ .  $\Box$ 

#### **3.2.1** Singular $\mathcal{H}_{\infty}$ control

To clarify the definitions and the choice of our assumptions we use Definition 3.6 to rewrite the system  $\Sigma$ . We apply to  $\Sigma$  the preliminary feedback

$$u=\beta(x)v$$

where

$$\beta(x) = \left(\begin{array}{cc} \beta_1(x) & \beta_2(x) \end{array}\right), \qquad (3.7)$$

which leads to the transformed system

$$\dot{x} = f(x) + g_1(x)v_1 + g_2(x)v_2 + e(x)d z = h(x) + v(x)v_1$$
(3.8)

where  $v_1 = \beta_1^T(x)u$  can be seen as the regular part of the inputs and  $v_2 = \beta_2^T(x)u$  as the singular part. The matrices  $g_1(x)$  and  $g_2(x)$  are given by

$$g_1(x) = g(x)\beta_1(x)$$
  
 $g_2(x) = g(x)\beta_2(x)$  (3.9)

and  $\nu(x)$  has full column rank. In [MvdS 96] we started with systems which are already in the special form (3.8).

Now we search for a feedback of the form

$$v = \begin{pmatrix} v_1 \\ v_2 \end{pmatrix} = \begin{pmatrix} l_1(x) \\ l_2(x) \end{pmatrix} = l(x)$$
(3.10)

which solves the nonlinear state feedback  $\mathcal{H}_{\infty}$  control problem for the system (3.8). Because the matrix  $\beta(x)$  in (3.7) is invertible this state feedback  $\mathcal{H}_{\infty}$  control problem for the transformed system is solvable with constant  $\gamma$  if and only if the state feedback  $\mathcal{H}_{\infty}$  control problem for the original system  $\Sigma$  is solvable with the same constant  $\gamma$ . Moreover the feedback (3.10) solves the state feedback  $\mathcal{H}_{\infty}$  control problem with constant  $\gamma$  for the system (3.8) if and only if the feedback

$$u = \beta(x)l(x)$$

solves the state feedback  $\mathcal{H}_{\infty}$  control problem with constant  $\gamma$  for  $\Sigma$ .

So we try to find a feedback (3.10) such that the  $L_2$ -gain for the closed-loop system (3.8), (3.10) is less than (or equal to)  $\gamma$ .

The pre-Hamiltonian corresponding to this problem is given by

$$K_{\gamma}(x, p, d, v) = p^{T} (f + g_{1}v_{1} + g_{2}v_{2} + ed) + \frac{1}{2} (h + vv_{1})^{T} (h + vv_{1}) - \frac{1}{2}\gamma^{2} ||d||^{2}.$$

As in the regular  $\mathcal{H}_{\infty}$  problem considered in Subsection 2.2.1 the maximizing disturbance  $d^*$  and the minimizing (optimal) input  $v_1^*$  can be calculated (see also Remark 2.18) as:

$$d^{*}(x, p) = \frac{1}{\gamma^{2}} e^{T}(x) p;$$
  

$$v_{1}^{*}(x, p) = -\left(v^{T}(x)v(x)\right)^{-1} \left(g_{1}^{T}(x)p + v^{T}(x)h(x)\right).$$

For the singular part of the inputs  $v_2$  we take an arbitrary function of x and p

$$v_2(x, p) = \alpha(x, p).$$

Substitution of these input and disturbance functions into the pre-Hamiltonian  $K_{\gamma}$  leads to the Hamiltonian  $H_{\gamma}: T^*M \to \mathbb{R}$  given by

$$H_{\gamma}(x, p) = p^{T} \left( f(x) + g_{1}(x)v_{1}^{*}(x, p) + g_{2}(x)\alpha(x, p) + e(x)d^{*}(x, p) \right) + \frac{1}{2} \left( h(x) + v(x)v_{1}^{*}(x, p) \right)^{T} \left( h(x) + v(x)v_{1}^{*}(x, p) \right) - \frac{1}{2}\gamma^{2} \|d^{*}(x, p)\|^{2} = p^{T} \left( f(x) - g_{1}(x) \left( v^{T}(x)v(x) \right)^{-1} v^{T}(x)h(x) + g_{2}(x)\alpha(x, p) \right) + \frac{1}{2} \frac{1}{\gamma^{2}} p^{T} e(x)e^{T}(x)p - \frac{1}{2} p^{T} g_{1}(x) \left( v^{T}(x)v(x) \right)^{-1} g_{1}^{T}(x)p + \frac{1}{2} h^{T}(x) \left( I_{p} - v(x) \left( v^{T}(x)v(x) \right)^{-1} v^{T}(x) \right) h(x)$$

which in the original vector fields g(x) and k(x) can be rewritten as

$$\begin{split} H_{\gamma}(x, p) &= p^{T} \left( f(x) - g(x)\beta_{1}(x) \left( \beta_{1}^{T}(x)k^{T}(x)k(x)\beta_{1}(x) \right)^{-1} \beta_{1}^{T}(x)k^{T}(x)h(x) \right) \\ &+ p^{T}g(x)\beta_{2}(x)\alpha(x, p) + \frac{1}{2}\frac{1}{\gamma^{2}}p^{T}e(x)e^{T}(x)p \\ &- \frac{1}{2}p^{T}g(x)\beta_{1}(x) \left( \beta_{1}^{T}(x)k^{T}(x)k(x)\beta_{1}(x) \right)^{-1} \beta_{1}^{T}(x)g^{T}(x)p \\ &+ \frac{1}{2}h^{T}(x)h(x) \\ &- \frac{1}{2}h^{T}(x)k(x)\beta_{1}(x) \left( \beta_{1}^{T}(x)k^{T}(x)k(x)\beta_{1}(x) \right)^{-1} \beta_{1}^{T}(x)k^{T}(x)h(x) \\ &= p^{T} \left( f(x) - g(x)\psi(x)k^{T}(x)h(x) + g(x)\beta_{2}(x)\alpha(x, p) \right) \\ &+ \frac{1}{2}\frac{1}{\gamma^{2}}p^{T}e(x)e^{T}(x)p - \frac{1}{2}p^{T}g(x)\psi(x)g^{T}(x)p \\ &+ \frac{1}{2}h^{T}(x) \left( I_{p} - k(x)\psi(x)k^{T}(x) \right)h(x). \end{split}$$

Now the following result can be stated.

**Theorem 3.9** Consider the nonlinear system  $\Sigma$ . Let  $\gamma > 0$ . Suppose there exists a non-negative  $C^1$ -solution V to the Hamilton-Jacobi inequality

$$V_{x}(x) \left( f(x) - g(x)\psi(x)k^{T}(x)h(x) + g(x)\beta_{2}(x)\alpha(x, V_{x}^{T}(x)) \right) \\ + \frac{1}{2}V_{x}(x) \left( \frac{1}{\gamma^{2}}e(x)e^{T}(x) - g(x)\psi(x)g^{T}(x) \right) V_{x}^{T}(x) \qquad (3.11) \\ + \frac{1}{2}h^{T}(x) \left( I_{p} - k(x)\psi(x)k^{T}(x) \right) h(x) \leq 0$$

with V(0) = 0. Then the feedback

$$u = \varphi(x) = -\psi(x) \left( g^{T}(x) V_{x}^{T}(x) + k^{T}(x) h(x) \right) + \beta_{2}(x) \alpha(x, V_{x}^{T}(x))$$
(3.12)

locally solves the state feedback  $L_2$ -gain control problem with constant  $\gamma$  for the system  $\Sigma$ .

**Proof** This result follows by applying Theorem 2.2 to the system

$$\dot{x} = f(x) + g(x)\varphi(x) + e(x)d$$
  
$$z = h(x) + k(x)\varphi(x)$$

where  $\varphi(x)$  is taken as the feedback defined in (3.12).

As an alternative approach we could also look at the closed-loop system (3.8) together with the feedback

$$v = \begin{pmatrix} v_1 \\ v_2 \end{pmatrix} = \begin{pmatrix} -\left(v^T(x)v(x)\right)^{-1}\left(g_1^T(x)V_x^T(x) + v^T(x)h(x)\right) \\ \alpha(x, V_x^T(x)) \end{pmatrix}.$$

Again using Theorem 2.2 it follows that the closed-loop system has  $L_2$ -gain less than or equal to  $\gamma$ . Then also our original system  $\Sigma$  together with the feedback

$$u = (\beta_{1}(x) \ \beta_{2}(x)) v$$
  
=  $(\beta_{1}(x) \ \beta_{2}(x)) \left( -(v^{T}(x)v(x))^{-1}(g_{1}^{T}(x)V_{x}^{T}(x) + v^{T}(x)h(x)) \right)$   
=  $-\psi(x)(g^{T}(x)V_{x}^{T}(x) + k^{T}(x)h(x)) + \beta_{2}(x)\alpha(x, V_{x}^{T}(x))$ 

leads to a closed-loop system which has  $L_2$ -gain less than or equal to  $\gamma$ .

To compare this result with the results for linear systems we will make a somewhat arbitrary assumption with respect to the singular part of the feedback,  $\alpha(x, p)$ . From now on we take similar to the linear case

$$\alpha(x, p) = -\frac{1}{2\varepsilon}g_2^T(x)p = -\frac{1}{2\varepsilon}\beta_2^T(x)g^T(x)p.$$

Theorem 3.9 specialized for this specific choice of  $\alpha$  reads as:

**Corollary 3.10** Consider the nonlinear system  $\Sigma$ . Let  $\gamma > 0$ . Suppose there exists a constant  $\varepsilon > 0$  such that there exists a non-negative  $C^1$ -solution V to the Hamilton-Jacobi inequality

$$V_{x}(x) \left( f(x) - g(x)\psi(x)k^{T}(x)h(x) \right) + \frac{1}{2}\frac{1}{\gamma^{2}}V_{x}(x)e(x)e^{T}(x)V_{x}^{T}(x) -\frac{1}{2}V_{x}(x) \left( g(x)\psi(x)g^{T}(x) + \frac{1}{\varepsilon}g(x)\beta_{2}(x)\beta_{2}^{T}(x)g^{T}(x) \right) V_{x}^{T}(x)$$
(3.13)  
$$+\frac{1}{2}h^{T}(x) \left( I_{p} - k(x)\psi(x)k^{T}(x) \right) h(x) \leq 0$$

with V(0) = 0. Then the feedback

$$u = \varphi(x) = -\left(\frac{1}{2\varepsilon}\beta_2(x)\beta_2^T(x) + \psi(x)\right)g^T(x)V_x^T(x) -\psi(x)k^T(x)h(x)$$
(3.14)

locally solves the state feedback  $L_2$ -gain control problem with constant  $\gamma$  for the system  $\Sigma$ .

From this corollary we can easily derive the following result. This result is the nonlinear extension of the sufficiency part of Theorem 3.4 as far as only  $L_2$ -gains are considered.

**Corollary 3.11** Consider the nonlinear system  $\Sigma$ . Let  $\gamma > 0$ . Suppose there exist constants  $\varepsilon$ ,  $\mu > 0$  such that there exists a non-negative  $C^1$ -solution V to the Hamilton-Jacobi inequality

$$V_{x}(x) \left( f(x) - g(x)\psi(x)k^{T}(x)h(x) \right) \\ + \frac{1}{2}(\frac{1}{\gamma^{2}} + \mu)V_{x}(x)e(x)e^{T}(x)V_{x}^{T}(x) \\ - \frac{1}{2}V_{x}(x) \left( g(x)\psi(x)g^{T}(x) + \frac{1}{\varepsilon}g(x)\beta_{2}(x)\beta_{2}^{T}(x)g^{T}(x) \right) V_{x}^{T}(x) \qquad (3.15) \\ + \frac{1}{2}h^{T}(x) \left( I_{p} - k(x)\psi(x)k^{T}(x) \right)h(x) \leq 0$$

with V(0) = 0. Then there exists a constant  $\overline{\gamma} < \gamma$  such that the feedback (3.14) locally solves the state feedback  $L_2$ -gain control problem with constant  $\overline{\gamma}$  for the system  $\Sigma$ .

**Proof** From inequality (3.15) it follows that there exists a constant  $0 \le \tilde{\gamma} < \gamma$  such that (3.13) is satisfied with  $\gamma$  replaced by  $\tilde{\gamma}$ . For instance take  $\tilde{\gamma} = \frac{\gamma}{1+\mu\gamma}$ . Then by Theorem 3.9 the closed-loop system has  $L_2$ -gain less than or equal to  $\tilde{\gamma}$  and therefore less than  $\gamma$ .

**Remark 3.12** The Hamilton-Jacobi inequality (3.15) is the extension of the algebraic Riccati equation (3.1). For linear systems of the form (3.1) the inequality (3.15) comes down to the existence of positive constants  $\varepsilon$ ,  $\mu > 0$  and a positive definite solution *P* to

$$\begin{split} \tilde{Q} &:= P\left(F - G\Sigma K^{T} H\right) + \left(F - G\Sigma K^{T} H\right)^{T} P + \left(\frac{1}{\gamma^{2}} + \mu\right) P E E^{T} P \\ &- P G\Sigma G^{T} P - \frac{1}{\varepsilon} P G \Phi^{T} \Phi G^{T} P + H^{T} \left(I_{p} - K\Sigma K^{T}\right) H \\ &\leq 0. \end{split}$$

Then this P is also a solution to (3.2) for

$$Q = \frac{\mu}{\varepsilon} P E E^T P - \frac{1}{\varepsilon} \tilde{Q} > 0.$$

On the other hand if there exists a positive definite solution P to (3.2) for a certain Q > 0 then by Finsler's Theorem (see [KPZ 90], [Pe 87b] and [Ja 77] for a proof) we can choose  $\mu > 0$  such that

$$\mu P E E^T P < \varepsilon Q$$

and for this  $\mu$  *P* is a solution to (3.15).

In Corollaries 3.10 and 3.11 the feedback (3.14) consists of a singular and a regular part. The gain of the singular feedback part is parameterized by  $\varepsilon$ . We could of course also apply feedbacks of the form (3.14) where the gain  $\frac{1}{2\varepsilon}$  for the singular part is replaced by a larger gain  $\eta \ge \frac{1}{2\varepsilon}$ .

**Corollary 3.13** Consider the nonlinear system  $\Sigma$ . Let  $\gamma > 0$ . Suppose there exists a constant  $\varepsilon > 0$  such that there exists a non-negative  $C^1$ -solution V to the Hamilton-Jacobi inequality (3.13) with V(0) = 0. Then for  $\eta \ge \frac{1}{2\varepsilon}$  the state feedback  $L_2$ -gain control problem with constant  $\gamma$  is locally solvable for the system  $\Sigma$  by the feedback

$$u = \varphi(x) = -\left(\eta\beta_2(x)\beta_2^T(x) + \psi(x)\right)g^T(x)V_x^T(x) -\psi(x)k^T(x)h(x).$$
(3.16)

**Proof** The result follows by applying Theorem 2.2 to the closed-loop system of  $\Sigma$  with the feedback (3.16), together with the following inequality

$$V_{x}(x) (f(x) + g(x)\varphi(x)) + \frac{1}{2} \frac{1}{\gamma^{2}} V_{x}(x)e(x)e^{T}(x)V_{x}^{T}(x) + \frac{1}{2} (h(x) + k(x)\varphi(x))^{T} (h(x) + k(x)\varphi(x)) = V_{x}(x) (f(x) - g(x)\psi(x)k^{T}(x)h(x)) + \frac{1}{2} \frac{1}{\gamma^{2}} V_{x}(x)e(x)e^{T}(x)V_{x}^{T}(x) - \eta V_{x}(x)g(x)\beta_{2}(x)\beta_{2}^{T}(x)g^{T}(x)V_{x}^{T}(x) - \frac{1}{2} V_{x}(x)g(x)\psi(x)g^{T}(x)V_{x}^{T}(x) + \frac{1}{2} h^{T}(x) (I_{p} - k(x)\psi(x)k^{T}(x)) h(x) \leq V_{x}(x) (f(x) - g(x)\psi(x)k^{T}(x)h(x)) + \frac{1}{2} \frac{1}{\gamma^{2}} V_{x}(x)e(x)e^{T}(x)V_{x}^{T}(x) - \frac{1}{2\varepsilon} V_{x}(x)g(x)\beta_{2}(x)\beta_{2}^{T}(x)g^{T}(x)V_{x}^{T}(x) - \frac{1}{2\varepsilon} V_{x}(x)g(x)\beta_{2}(x)\beta_{2}^{T}(x)V_{x}^{T}(x) + \frac{1}{2} h^{T}(x) (I_{p} - k(x)\psi(x)k^{T}(x)) h(x) \leq 0$$

for all  $\eta \geq \frac{1}{2\varepsilon}$ .

**Corollary 3.14** Consider the nonlinear system  $\Sigma$ . Let  $\gamma > 0$ . Suppose there exist constants  $\varepsilon$ ,  $\mu > 0$  such that there exists a non-negative  $C^1$ -solution V to the Hamilton-Jacobi inequality (3.15). Then there exists a constant  $\overline{\gamma} < \gamma$  such that the feedback (3.16) locally solves the nonlinear state feedback  $L_2$ -gain control problem with constant  $\overline{\gamma}$  for the system  $\Sigma$  for every  $\eta \geq \frac{1}{2\varepsilon}$ .  $\Box$ 

**Proof** The proof follows easily along the lines of the proofs of Corollary 3.11 and Corollary 3.13.

The converse result of Corollary 3.10 is more involved. Assume there exists a feedback u = l(x), l(0) = 0, such that the closed-loop system of  $\Sigma$  with this feedback has  $L_2$ -gain less than (or equal to)  $\gamma$ . When does there exists a solution  $V \ge 0$  to the Hamilton-Jacobi inequality (3.13) or (3.15)?

We start by making the following assumption on the closed-loop system resulting from applying u = l(x), l(0) = 0.

Assumption 3 The  $L_2$ -gain from d to  $\beta_2^T(x)l(x)$  is finite, i.e., there exists a constant N > 0 such that for all  $x \in M$  there exists a constant  $\tilde{K}(x), 0 \leq \tilde{K}(x) < \infty$ , with  $\tilde{K}(0) = 0$  such that

$$\int_0^t \|\beta_2^T(x(\tau))l(x(\tau))\|^2 d\tau \le N \int_0^t \|d(\tau)\|^2 d\tau + \tilde{K}(x)$$

for all t > 0 and all  $d \in L_2(0, t)$ , where  $x(\tau)$  is the solution of the state equation of the closed loop system  $\Sigma$  with the feedback u = l(x).

Then the following converse result can be stated.

**Theorem 3.15** Consider the system  $\Sigma$ . Suppose that the feedback u = l(x) with l(0) = 0 solves the state feedback  $L_2$ -gain control problem with  $\overline{\gamma} < \gamma$  for the system  $\Sigma$  with a differentiable storage function K(x) and Assumption 3 is satisfied with a differentiable storage function  $\tilde{K}(x)$ . Then there exists a solution V of (3.15) for certain  $\varepsilon$ ,  $\mu > 0$ , and hence also the feedback (3.16) results in a closed-loop system which has  $L_2$ -gain less than  $\gamma$ .

**Proof** The system

$$\dot{x} = f(x) + g(x)l(x) + e(x)d$$
  
$$z = h(x) + k(x)l(x)$$

has  $L_2$ -gain less than  $\gamma$ . We rewrite this system as done in the beginning of this section (see system (3.8)), using the notation defined in (3.9). Then also the system

$$\dot{x} = f(x) + g_1(x)l_1(x) + g_2(x)l_2(x) + e(x)d$$
  
$$z = h(x) + u(x)l_1(x)$$

with  $l_1(x) = \beta_1^T(x)l(x)$  and  $l_2(x) = \beta_2^T(x)l(x)$  has  $L_2$ -gain less than  $\gamma$ . Hence there exists a constant  $\nu > 0$  such that for all  $x \in M$  there exists a differentiable function  $K(x), 0 \le K(x) < \infty$ , with K(0) = 0, such that

$$\int_0^t \|z(x(\tau))\|^2 \mathrm{d}\tau \le (\gamma^2 - \nu) \int_0^t \|d(\tau)\|^2 \mathrm{d}\tau + K(x(0))$$
(3.17)

holds for all  $d \in L_2[0, t]$  and for all  $t \ge 0$ . On the other hand from Assumption 3 it follows that there exists a constant N > 0 such that for every  $x \in M$  there exists a differentiable  $\tilde{K}(x), 0 \le \tilde{K}(x) < \infty$ , with  $\tilde{K}(x(0)) = 0$ , such that

$$\int_{0}^{t} \|\beta_{2}^{T}(x(\tau))l(x(\tau))\|^{2} d\tau = \int_{0}^{t} \|l_{2}(x(\tau))\|^{2} d\tau$$
  
$$\leq N \int_{0}^{t} \|d(\tau)\|^{2} d\tau + \tilde{K}(x(0)). \quad (3.18)$$

Combining the inequalities (3.17) and (3.18) one obtains

$$\int_0^t \left( \|z(x(\tau))\|^2 + \varepsilon \|l_2(x(\tau))\|^2 \right) d\tau$$
  

$$\leq (\gamma^2 - \beta) \int_0^t \|d(\tau)\|^2 d\tau + K(x(0)) + \varepsilon \tilde{K}(x(0))$$

for  $0 < \varepsilon < \nu/N$  and  $\beta > 0$  sufficiently small. According to Assumption 3 there exists for  $\varepsilon$ ,  $\mu > 0$  sufficiently small a solution  $V \ge 0$  (for instance  $V(x) = K(x) + \varepsilon \tilde{K}(x)$ ) to

$$V_{x}(x) (f(x) + g_{1}(x)l_{1}(x) + g_{2}(x)l_{2}(x)) + \frac{1}{2} \left(\frac{1}{\gamma^{2}} + \mu\right) V_{x}(x)e(x)e^{T}(x) V_{x}^{T}(x) + \frac{1}{2} ||z(x)||^{2} + \frac{1}{2}\varepsilon ||l_{2}(x)||^{2} \leq 0$$

with V(0) = 0.

Finally we use a completion of the squares argument

$$V_{x}(x) \left( f(x) - g_{1}(x) \left( v^{T}(x)v(x) \right)^{-1} \left( g_{1}^{T}(x) V_{x}^{T}(x) + u^{T}(x)h(x) \right) \right. \\ \left. - \frac{1}{\varepsilon} g_{2}(x) g_{2}^{T}(x) V_{x}^{T}(x) \right) \\ \leq V_{x}(x) \left( f(x) + g_{1}(x) l_{1}(x) + g_{2}(x) l_{2}(x) \right) \\ \left. + V_{x}(x) g_{1}(x) \left( - \left( v^{T}(x)v(x) \right)^{-1} \left( g_{1}^{T}(x) V_{x}^{T}(x) + v^{T}(x)h(x) \right) - l_{1}(x) \right) \right. \\ \left. + V_{x}(x) g_{2}(x) \left( - \frac{1}{\varepsilon} g_{2}^{T}(x) V_{x}^{T}(x) - l_{2}(x) \right) \right.$$

$$\leq -\frac{1}{2} \left( \frac{1}{\gamma^{2}} + \mu \right) V_{x}(x) e(x) e^{T}(x) V_{x}^{T}(x) - \frac{1}{2} h^{T}(x) h(x) - \frac{1}{2} \left\| \left( v^{T}(x) v(x) \right)^{\frac{1}{2}} l_{1}(x) + \left( v^{T}(x) v(x) \right)^{-\frac{1}{2}} \left( g_{1}^{T}(x) V_{x}^{T}(x) + v^{T}(x) h(x) \right) \right\|^{2} - \frac{1}{2} \left\| l_{2}(x) + \frac{1}{\varepsilon} g_{2}^{T}(x) V_{x}^{T}(x) \right\|^{2} - \frac{1}{2\varepsilon} V_{x}(x) g_{2}(x) g_{2}^{T}(x) V_{x}^{T}(x) - \frac{1}{2} V_{x}(x) g_{1}(x) \left( v^{T}(x) v(x) \right)^{-1} g_{1}^{T}(x) V_{x}^{T}(x) + \frac{1}{2} h^{T}(x) v(x) \left( v^{T}(x) v(x) \right)^{-1} v^{T}(x) h(x)$$

from which it can be concluded that

$$V_{x}(x) \left( f(x) - g_{1}(x) \left( \nu^{T}(x)\nu(x) \right)^{-1} \nu^{T}(x)h(x) \right) \\ + \frac{1}{2} \left( \frac{1}{\gamma^{2}} + \mu \right) V_{x}(x)e(x)e^{T}(x)V_{x}^{T}(x) \\ - \frac{1}{2} V_{x}(x) \left( g_{1}(x) \left( \nu^{T}(x)\nu(x) \right)^{-1} g_{1}^{T}(x) + \frac{1}{\varepsilon} g_{2}(x)g_{2}^{T}(x) \right) V_{x}^{T}(x) \\ + \frac{1}{2} h^{T}(x) \left( I_{p} - \nu(x) \left( \nu^{T}(x)\nu(x) \right)^{-1} \nu^{T}(x) \right) h(x) \leq 0.$$

Substitution of  $g_1$  and  $g_2$  now leads to

$$\begin{aligned} V_{x}(x) \left( f(x) - g(x)\beta_{1}(x) \left(\beta_{1}^{T}(x)k^{T}(x)k(x)\beta_{1}(x)\right)^{-1} \beta_{1}^{T}(x)k^{T}(x)h(x) \right) \\ &+ \frac{1}{2}h^{T}(x) \left( I_{p} - k(x)\beta_{1}(x) \left(\beta_{1}^{T}(x)k^{T}(x)k(x)\beta_{1}(x)\right) \beta_{1}^{T}(x)k^{T}(x) \right) h(x) \\ &+ \frac{1}{2}V_{x}(x) \left( \left(\frac{1}{\gamma^{2}} + \mu\right)e(x)e^{T}(x) - \frac{1}{\varepsilon}g(x)\beta_{2}(x)\beta_{2}^{T}(x)g^{T}(x) \right) V_{x}^{T}(x) \\ &- \frac{1}{2}V_{x}(x)g(x)\beta_{1}(x) \left(\beta_{1}^{T}(x)k^{T}(x)k(x)\beta_{1}(x)\right) \beta_{1}^{T}(x)g^{T}(x)V_{x}^{T}(x) \\ &= V_{x}(x) \left( f(x) - g(x)\psi(x)k^{T}(x)h(x) \right) \\ &+ \frac{1}{2}(\frac{1}{\gamma^{2}} + \mu)V_{x}(x)e(x)e^{T}(x)V_{x}^{T}(x) \\ &- \frac{1}{2}V_{x}(x)g(x) \left( \psi(x) + \frac{1}{\varepsilon}\beta_{2}(x)\beta_{2}^{T}(x) \right) g^{T}(x)V_{x}^{T}(x) \\ &+ \frac{1}{2}h^{T}(x) \left( I_{p} - k(x)\psi(x)k^{T}(x) \right) h(x) \leq 0 \end{aligned}$$

and the conclusion follows from Corollary 3.14.

**Remark 3.16** Corollary 3.14 together with Theorem 3.15 gives under Assumption 3 a necessary and sufficient condition for the solvability of the state feedback  $L_2$ -gain control problem with constant less than  $\gamma$ . The question arises under what conditions a feedback of the form (3.16) will satisfy Assumption 3. A partial answer to this question will be given in the next subsection (see Corollary 3.20).

**Remark 3.17** For linear systems Assumption 3 is already implied by the fact that the feedback is assumed to be internally stabilizing. Thus Theorem 3.15 is a generalization of Theorem 3.4 to the nonlinear setting.  $\Box$ 

#### **3.2.2** Regularized $\mathcal{H}_{\infty}$ control

A different but similar way to attack the singular  $\mathcal{H}_{\infty}$  problem is considered in this subsection. Let  $\varepsilon > 0$ . We consider the following *regularized* version of the system  $\Sigma$ :

$$\Sigma_r \begin{cases} \dot{x} = f(x) + g(x)u + e(x)d \\ z_r = \begin{pmatrix} h(x) + k(x)u \\ \sqrt{\varepsilon}\beta_2^T(x)u \end{pmatrix}$$
(3.19)

Using Assumption 3 we can state the following:

**Theorem 3.18** Let u = l(x), l(0) = 0, be a feedback for the system  $\Sigma$ . Then we have the following implications regarding the statements (i) and (ii) below. The statement (ii) implies (i), and under Assumption 3 the statements (i) and (ii) are equivalent.

- (i) The closed loop system  $\Sigma$  with static state feedback u = l(x) has  $L_2$ -gain less than  $\gamma$ .
- (ii) For  $\varepsilon$  sufficiently small the closed loop system  $\Sigma_r$  with static state feedback u = l(x) has  $L_2$ -gain less than  $\gamma$ .

**Proof** (*i*)  $\Rightarrow$  (*ii*) By Assumption 3 there exists a constant N > 0 such that for all  $x \in M$  there exists a constant  $\tilde{K}(x)$ ,  $0 \leq \tilde{K}(x) < \infty$ , with  $\tilde{K}(0) = 0$  such that

$$\int_0^t \|\beta_2^T(x(\tau))l(x(\tau))\|^2 d\tau \le N \int_0^t \|d(\tau)\|^2 d\tau + \tilde{K}(x(0))$$

for all t > 0 and all  $d \in L_2(0, t)$ .

From Definition 2.14 we know that there exists a constant  $\delta > 0$  such that for every  $x \in M$  there exists a constant K(x),  $0 \le K(x) < \infty$ , with K(0) = 0, such that

$$\int_0^t \|z(x(\tau))\|^2 \mathrm{d}\tau \le (\gamma^2 - \delta) \int_0^t \|d(\tau)\|^2 \mathrm{d}\tau + K(x(0))$$

for all  $d \in L_2(0, t)$  and all  $t \ge 0$ .

Now take  $\varepsilon > 0$  such that  $\varepsilon N < \delta$ . Then some  $\mu > 0$  can be found such that, with bias term  $M(x) := K(x) + \varepsilon \tilde{K}(x)$ , the following inequality holds

$$\int_{0}^{t} (\|z(x(\tau))\|^{2} + \varepsilon \|\beta_{2}^{T}(x(\tau))l(x(\tau))\|^{2}) d\tau$$

$$\leq (\gamma^{2} - \mu) \int_{0}^{t} \|d(\tau)\|^{2} d\tau + M(x(0))$$

for all t > 0 and all  $d \in L_2(0, t)$ . Hence u = l(x) combined with  $\Sigma_r$  leads to a closed loop system which has  $L_2$ -gain from d to  $z_r$  less than  $\gamma$ .

For proving  $(ii) \Rightarrow (i)$  we note that if u = l(x) solves the suboptimal problem for the system  $\Sigma_r$  then it also solves the suboptimal problem for  $\Sigma$  because

$$\int_0^t \|z_r(\tau)\|^2 \mathrm{d}\tau \geq \int_0^t \|z(\tau)\|^2 \mathrm{d}\tau.$$

Based on Theorem 3.18 we will search for a state feedback which makes the  $L_2$ -gain for the system  $\Sigma_r$  less than  $\gamma$ . Since  $\Sigma_r$  is a regular system we can find the min-max solution for this  $\mathcal{H}_{\infty}$  problem. The pre-Hamiltonian  $K_{\gamma}$ :  $T^*M \times \mathbb{R}^q \times \mathbb{R}^m$  corresponding to this problem is

$$K_{\gamma}(x, p, d, u) = p^{T} (f(x) + g(x)u + e(x)d) - \frac{1}{2}\gamma^{2} ||d||^{2} + \frac{1}{2} ||z||^{2} + \frac{1}{2}\varepsilon ||\beta_{2}^{T}(x)u||^{2}$$

which has saddle point solution:

$$d^{*} = \frac{1}{\gamma^{2}} e^{T}(x) p,$$
  

$$u^{*} = -\left(\frac{1}{\varepsilon} \beta_{2}(x) \beta_{2}^{T}(x) + \psi(x)\right) g^{T}(x) p - \sigma(x) k^{T}(x) h(x). \quad (3.20)$$

Substitution of this saddle point solution into the pre-Hamiltonian  $K_{\gamma}$  leads to the Hamiltonian  $H_{\gamma}(x, p) = K_{\gamma}(x, p, d^*(x, p), u^*(x, p))$  given as

$$H_{\gamma}(x, p) = p^{T} \left( f(x) - g(x)\psi(x)k^{T}(x)h(x) \right) + \frac{1}{2} \frac{1}{\gamma^{2}} p^{T} e(x)e^{T}(x)p \\ - \frac{1}{2} p^{T} g(x) \left( \psi(x) + \frac{1}{\varepsilon} \beta_{2}(x) \beta_{2}^{T}(x) \right) g^{T}(x)p \\ + \frac{1}{2} h^{T}(x) \left( I_{p} - k(x)\psi(x)k^{T}(x) \right) h(x).$$

**Theorem 3.19** Consider the nonlinear system  $\Sigma$ . Let  $\gamma > 0$ . Suppose there exists a  $C^1$  solution  $V \ge 0$  to the Hamilton-Jacobi inequality (3.13). Then the state feedback

$$u = l(x) = -\left(\frac{1}{\varepsilon}\beta_2(x)\beta_2^T(x) + \psi(x)\right)g^T(x)V_x^T(x) -\psi(x)k^T(x)h(x)$$
(3.21)

locally solves the state feedback  $L_2$ -gain problem with constant  $\gamma$  for the system  $\Sigma$ .

**Proof** The closed loop system  $\Sigma_r$ , (3.21) is given by

$$\dot{x} = f(x) - g(x) \left(\frac{1}{\varepsilon}\beta_2(x)\beta_2^T(x) + \psi(x)\right) g^T(x) V_x^T(x)$$
$$-g(x)\sigma(x)k^T(x)h(x) + e(x)d$$
$$z_r = \left(\begin{array}{c}h(x) + k(x)u\\-\frac{1}{\sqrt{\varepsilon}}\beta_2^T(x)g^T(x)V_x^T(x)\end{array}\right)$$

which has by Theorem 2.2 and Theorem 3.18  $L_2$ -gain less than or equal to  $\gamma$ .

Based on this Theorem we can give the following (partial) answer to the question posed in Remark 3.16.

**Corollary 3.20** Consider the system  $\Sigma$ . Assume there exists a solution  $V \ge 0$  to the Hamilton-Jacobi inequality (3.13). Then for any  $\eta \ge \frac{1}{\varepsilon}$  the feedback (3.16) satisfies Assumption 3.

**Proof** For a fixed  $\eta \geq \frac{1}{\varepsilon}$  look at the system

$$\dot{x} = f(x) + g(x)u + e(x)d,$$
  
$$z = \begin{pmatrix} h(x) + k(x)u \\ \frac{1}{\sqrt{\eta}}\beta_2^T(x)u \end{pmatrix}.$$

The Hamiltonian for the  $\mathcal{H}_{\!\infty}$  problem for this system is given by

$$H_{\gamma}(x, p) = p^{T} \left( f(x) - g(x)\psi(x)k^{T}(x)h(x) \right) + \frac{1}{2} \frac{1}{\gamma^{2}} p^{T} e(x)e^{T}(x)p \\ - \frac{1}{2} p^{T} g(x) \left( \psi(x) + \eta \beta_{2}(x)\beta_{2}^{T}(x) \right) g^{T}(x)p \\ + \frac{1}{2} h^{T}(x) \left( I_{p} - k(x)\psi(x)k^{T}(x) \right) h(x).$$

Hence for  $\eta \geq \frac{1}{\varepsilon}$ :

$$\begin{aligned} V_{x}(x) \left(f(x) - g(x)\psi(x)k^{T}(x)h(x)\right) &+ \frac{1}{2}\frac{1}{\gamma^{2}}V_{x}(x)e(x)e^{T}(x)V_{x}^{T}(x) \\ &- \frac{1}{2}V_{x}(x)g(x)\psi(x)g^{T}(x)V_{x}^{T}(x) \\ &- \frac{1}{2}\eta V_{x}(x)g(x)\beta_{2}(x)\beta_{2}^{T}(x)g^{T}(x)V_{x}^{T}(x) \\ &+ \frac{1}{2}h^{T}(x) \left(I_{p} - k(x)\psi(x)k^{T}(x)\right)h(x) \end{aligned}$$

$$\leq V_{x}(x) \left(f(x) - g(x)\psi(x)k^{T}(x)h(x)\right) + \frac{1}{2}\frac{1}{\gamma^{2}}V_{x}(x)e(x)e^{T}(x)V_{x}^{T}(x) \\ &- \frac{1}{2}V_{x}(x)g(x)\psi(x)g^{T}(x)V_{x}^{T}(x) \\ &- \frac{1}{2}\frac{1}{\varepsilon}V_{x}(x)g(x)\beta_{2}(x)\beta_{2}^{T}(x)g^{T}(x)V_{x}^{T}(x) \\ &+ \frac{1}{2}h^{T}(x) \left(I_{p} - k(x)\psi(x)k^{T}(x)\right)h(x) \end{aligned}$$

$$\leq 0. \end{aligned}$$

It follows that the gain from d to  $\sqrt{\eta}\beta_2^T(x)g^T(x)V_x^T(x)$  is less than or equal to  $\gamma$ . Hence the gain from d to  $\eta\beta_2^T(x)g^T(x)V_x^T(x)$  is less than or equal to  $\gamma\sqrt{\eta}$ .

Until now we have not considered the stability of the closed loop system. The following theorem can be easily obtained from Theorem 2.7.

**Theorem 3.21** Suppose there exists a solution  $V \ge 0$  to (3.13). Assume the closed-loop system of  $\Sigma$  with a feedback of the form (3.16) for  $\eta \ge \frac{1}{2\varepsilon}$  is zero-state observable. Then V(x) > 0 for  $x \ne 0$  and the closed loop system (with  $d(t) \equiv 0$ ) is locally asymptotically stable.

Assume additionally that V is proper, then the closed loop system is globally asymptotically stable.  $\hfill \Box$ 

#### 3.2.3 The nonlinear system versus its linearization

In this section we consider the link between the solvability of the state feedback  $L_2$ -gain and  $\mathcal{H}_{\infty}$  control problems (see Definitions 2.14 and 2.15) for the nonlinear system  $\Sigma$  and the solvability of these problems for its linearization. So we consider the linearization of the nonlinear system  $\Sigma$  around the equilibrium x = 0:

$$\dot{\bar{x}} = F\bar{x} + G\bar{u} + E\bar{d}$$

$$\ddot{z} = H\bar{x} + K\bar{u}$$
(3.22)

where  $\bar{u} \in \mathbb{R}^m$ ,  $\bar{x} \in \mathbb{R}^n$ ,  $\bar{d} \in \mathbb{R}^q$ ,  $\bar{z} \in \mathbb{R}^p$  and the matrices *F*, *G*, *K*, *H* and *M* are defined as:

$$F = \frac{\partial f}{\partial x}(0), \quad G = g(0), \quad E = e(0), \quad H = \frac{\partial h}{\partial x}(0), \quad K = k(0).$$

Straightforwardly from Chapter 2 the following results are obtained.

**Theorem 3.22** Suppose the  $L_2$ -gain of (3.4), (3.5) is less than  $\gamma$ , and assume F + GL with  $L = \frac{\partial l}{\partial x}(0)$  is asymptotically stable, then there exists a neighborhood W of 0 and a smooth function  $V \ge 0$  on W satisfying (3.15).

Alternatively, assume f + gl is globally asymptotically stable. Define the Hamiltonian

$$H_{\gamma}(x, p) = p^{T} (f(x) + g(x)l(x)) + \frac{1}{\gamma^{2}} p^{T} e(x)e^{T}(x)p + \frac{1}{2} (h(x) + k(x)l(x))^{T} (h(x) + k(x)l(x))$$

and suppose  $X_{H_{\gamma}}$  is hyperbolic at (0,0), and its stable invariant manifold is diffeomorphic to M under the canonical projection  $\pi : T^*M \to M$ . Then there exists a global solution  $V \ge 0$  to (3.15).

**Theorem 3.23** Let  $\gamma > 0$ . Suppose there exists a smooth feedback u = l(x), with l(0) = 0, for  $\Sigma$  such that the  $L_2$ -gain of the nonlinear system  $\Sigma$ , (3.5) is less than (or equal to)  $\gamma$ . Then the linear feedback  $\bar{u} = L\bar{x}$ , with  $L = \frac{\partial l}{\partial x}(0)$ , for (3.22) results in the linear closed loop system

$$\dot{\bar{x}} = (F + GL)\bar{x} + E\bar{d}$$
  
$$\bar{z} = (H + KL)\bar{x}$$
(3.23)

which also has  $L_2$ -gain less than (or equal to)  $\gamma$ .

**Proof** The linearization of  $\Sigma$  with (3.5) is equal to (3.23) ([vdS 92]).

For the linearization (3.22) the matrices described in Definition 3.1 are given by

$$U = v(0) = k(0)\beta_1(0), \quad \Sigma = \psi(0), \quad \Phi = \beta_2^T(0).$$

**Remark 3.24** If the state feedback  $L_2$ -gain control problem with gain less than or equal to  $\gamma$  is solvable for the nonlinear system in the sense that for some  $\varepsilon > 0$ there exists a differential non-negative solution V to (3.13) then we know from Corollary 3.10 that (3.14) is a solution of the state feedback  $L_2$ -gain control problem for the nonlinear system  $\Sigma$ . From Theorem 3.23 it follows that then the linear feedback

$$\bar{u} = -\left(\frac{1}{2\varepsilon}\beta_2(0)\beta_2^T(0) + \psi(0)\right)g^T(0)V_{xx}^T(0) - \psi(0)k^T(0)\frac{\partial h}{\partial x}(0)$$
$$= -\left(\frac{1}{2\varepsilon}\Phi^T\Phi + \Sigma\right)G^TP^T\bar{x} - \Sigma K^TH\bar{x}$$
(3.24)

solves the state feedback  $L_2$ -gain optimal control problem for the linearized system (3.22) and

$$P = V_{xx}(0)$$

is a solution of

$$(F - G\Sigma K^{T}H)^{T} P + P(F - G\Sigma K^{T}H) + \frac{1}{\gamma^{2}} PEE^{T}P$$
$$-PG\Sigma G^{T}P - \frac{1}{\varepsilon} PG\Phi^{T}\Phi G^{T}P + H^{T}(I - K\Sigma K^{T})H \leq 0$$

On the other hand when the state feedback  $L_2$ -gain control problem with gain less than  $\gamma$  is solvable in the sense that for some  $\varepsilon > 0$  and  $\mu > 0$  there exists a differentiable non-negative solution V to (3.15) then again the feedback

(3.14) solves the problem and the feedback (3.24) solves the state feedback  $\mathcal{H}_{\infty}$  control problem with gain less than  $\gamma$  for the linearization (3.22). Furthermore similar to Remark 3.12 there can be chosen a matrix Q such that  $P = V_{xx}(0)$  is a solution to (3.2).

Furthermore the following connection between the nonlinear system and its linearization can be stated.

**Theorem 3.25** Consider the linearized system (3.22). Let  $\gamma > 0$ . Suppose there exists a feedback  $\bar{u} = L\bar{x}$  such that the  $L_2$ -gain of the closed loop system (from  $\bar{d}$  to  $\bar{z}$ ) is less than  $\gamma$  and the closed loop system is asymptotically stable. Then there exists a neighborhood W of  $x_0$  and a smooth function  $V \ge 0$  defined on W such that V is a solution of the Hamilton-Jacobi inequality (3.15). Furthermore, for  $\eta \ge \frac{1}{2\varepsilon}$ , the feedback (3.16) locally solves the state feedback  $L_2$ -gain problem with constant less than  $\gamma$  for the system  $\Sigma$ .

**Proof** The  $L_2$ -gain of the closed loop linearized system is less than  $\gamma$ . By definition this means that there exists a constant  $\tilde{\gamma} < \gamma$  for which the closed-loop system has  $L_2$ -gain less than or equal to  $\tilde{\gamma}$ . This implies that for every  $\hat{\gamma}$  which satisfies  $\tilde{\gamma} < \hat{\gamma} < \gamma$  we have that the closed-loop system has  $L_2$ -gain less than  $\hat{\gamma}$ . Hence by the Theorems 3.4 and 3.5 there exists for every Q > 0 a constant  $\varepsilon^* > 0$  and a stabilizing solution  $P_{\varepsilon} > 0$  of the ARE

$$(F - G\Sigma K^{T}H)^{T} P + P(F - G\Sigma K^{T}H) + \left(\frac{1}{\gamma^{2}} + \nu\right) PEE^{T}P$$
$$-PG\Sigma G^{T}P - \frac{1}{\varepsilon}PG\Phi^{T}\Phi G^{T}P + H^{T}(I - K\Sigma K^{T})H + \varepsilon Q = 0$$

for some  $\kappa > 0$  and  $\varepsilon \in (0, \varepsilon^*]$ . Now take the Hamiltonian

$$H_{\gamma}(x, p) := p^{T} \left( f(x) - g(x)\psi(x)k^{T}(x)h(x) \right) \\ + \frac{1}{2} \left( \frac{1}{\gamma^{2}} + \kappa \right) p^{T} e(x)e^{T}(x)p \\ - \frac{1}{2}p^{T}g(x)\psi(x)g^{T}(x)p - \frac{1}{2\varepsilon}p^{T}g(x)\beta_{2}(x)\beta_{2}^{T}(x)g^{T}(x)p \\ + \frac{1}{2}h^{T}(x) \left( I_{p} - k(x)\psi(x)k^{T}(x) \right) h(x) + \varepsilon q(x)$$

where q is an arbitrary function  $q: M \to \mathbb{R}^+$  which satisfies

$$q(0) = 0$$
,  $\frac{\partial q}{\partial x}(0) = 0$ ,  $\frac{\partial^2 q}{\partial x^2}(0) = Q > 0$ .

Then the linearization of  $X_{H_{\gamma}}$  at (0, 0) is given by the Hamiltonian matrix

$$DX_{H_{\gamma}}(0,0) = \tag{3.25}$$

$$\begin{pmatrix} F - G\Sigma K^{T}H & (\frac{1}{\gamma^{2}} + \kappa)EE^{T} - G\Sigma G^{T} - \frac{1}{\varepsilon}G\Phi^{T}\Phi G^{T} \\ -H^{T}(I_{p} - K\Sigma K^{T})H - \varepsilon Q & -(F - G\Sigma K^{T}H)^{T} \end{pmatrix}.$$

Then  $P_{\varepsilon} = P_{\varepsilon}^{T}$  is a solution of (3.2) if and only if

$$\begin{pmatrix} F - G\Sigma K^{T}H & (\frac{1}{\gamma^{2}} + \kappa)EE^{T} - G\Sigma G^{T} - \frac{1}{\varepsilon}G\Phi^{T}\Phi G^{T} \\ -H^{T}(I_{p} - K\Sigma K^{T})H - \varepsilon Q & -(F - G\Sigma K^{T}H)^{T} \end{pmatrix} \begin{bmatrix} I \\ P_{\varepsilon} \end{bmatrix}$$
$$= \begin{bmatrix} I \\ P_{\varepsilon} \end{bmatrix} \left(F + (\frac{1}{\gamma^{2}} + \kappa)EE^{T}P_{\varepsilon} - G\Sigma G^{T}P_{\varepsilon} - \frac{1}{\varepsilon}G\Phi^{T}\Phi G^{T}P_{\varepsilon}\right)$$
and thus

and thus

span 
$$\begin{bmatrix} I \\ P_{\varepsilon} \end{bmatrix}$$
 = stable eigenspace of  
 $\begin{pmatrix} F - G\Sigma K^{T}H & (\frac{1}{\gamma^{2}} + \kappa)EE^{T} - G\Sigma G^{T} - \frac{1}{\varepsilon}G\Phi^{T}\Phi G^{T} \\ -H^{T}(I_{p} - K\Sigma K^{T})H - \varepsilon Q & -(F - G\Sigma K^{T}H)^{T} \end{pmatrix}$ 

for some  $\varepsilon$ . Thus the Hamiltonian matrix (3.25) does not have imaginary eigenvalues.

Then the stable invariant manifold  $N^-$  of  $X_{H_{\gamma}}$  through (0, 0) is *n*-dimensional and is tangent at (0, 0) to

span 
$$\begin{bmatrix} I \\ P_{\varepsilon} \end{bmatrix}$$

for  $\varepsilon \in (0, \varepsilon^*)$ .

Furthermore locally around 0 the manifold  $N^-$  is given as

$$N^{-} = \left\{ \left( x, \, p = \frac{\partial^{T} V_{\varepsilon}}{\partial x}(x) \right) \, \middle| \, x \text{ around } 0 \right\}$$

where  $V_{\varepsilon}$  is a (local) solution of the Hamilton-Jacobi equation

$$V_{x}(x) \left(f(x) - g(x)\psi(x)k^{T}(x)h(x)\right)$$
  
+  $\frac{1}{2}\left(\frac{1}{\gamma^{2}} + \kappa\right)V_{x}(x)e(x)e^{T}(x)V_{x}^{T}(x)$   
-  $\frac{1}{2}V_{x}(x)\left(g(x)\psi(x)g^{T}(x) + \frac{1}{\varepsilon}g(x)\beta_{2}(x)\beta_{2}^{T}(x)g^{T}(x)\right)V_{x}^{T}(x)$   
+  $\frac{1}{2}h^{T}(x)\left(I_{p} - k(x)\psi(x)k^{T}(x)\right)h(x) \leq 0$ 

with  $V_{xx}(0) = P$ . From Theorem 2.12  $V_{\varepsilon} \ge 0$  because

$$F - G\Sigma K^T H - G\Sigma G^T P_{\varepsilon} - \frac{1}{\varepsilon} G \Phi^T \Phi G^T P_{\varepsilon}$$

is asymptotically stable. Then the result follows from Corollary 3.14.

Finally summarizing some of these results we can state the following extension of Theorem 2.22.

**Theorem 3.26** Consider the nonlinear system  $\Sigma$  and its linearization (3.22). Then the following statements are equivalent:

- (i) there exists a linear feedback u
   = Lx
   such that after applying this feedback to the system Σ the closed loop system has L<sub>2</sub>-gain less than γ and F + GL is asymptotically stable;
- (ii) there exists a positive definite solution  $P_{\varepsilon}$  to the Algebraic Riccati Equation

$$(F - G\Sigma K^{T}H)^{T}P + P(F - G\Sigma K^{T}H) + \frac{1}{\gamma^{2}}PEE^{T}P$$
$$-PG\left(\Sigma + \frac{1}{\varepsilon}\Phi^{T}\Phi\right)G^{T}P + H^{T}(I - K\Sigma K^{T})H + \varepsilon Q = 0$$

for all Q > 0 and for some  $\varepsilon > 0$ , which also satisfies

$$\sigma\left(F + \frac{1}{\gamma^2} E E^T P - G \Sigma (G^T P + K^T H) - \frac{1}{\varepsilon} G \Phi^T \Phi G^T P\right) \subset \mathbb{C}^-;$$

(iii) there exists a neighborhood  $W \subset M$  of 0, and a nonlinear state feedback u = l(x) defined on W, such that F + GL, with  $L = \frac{\partial l}{\partial x}(0)$ , is asymptotically stable and the closed loop system of this nonlinear feedback and the system  $\Sigma$  has locally  $L_2$ -gain less than  $\gamma$  on W.

# 3.3 Singular nonlinear measurement feedback $\mathcal{H}_{\infty}$ control

It is possible to extend the state feedback results to the measurement feedback problem by using the worst case certainty equivalence principle. Consider systems of the form:

$$\Sigma^{m} \begin{cases} \dot{x} = f(x) + g(x)u + e(x)d_{1} \\ y = c(x) + d_{2} \\ z = h(x) + k(x)u \end{cases}$$
(3.26)

We search for a dynamic affine nonlinear compensator

$$\dot{\xi} = w(\xi) + p(\xi)y$$
  
 $u = q(\xi)$ 
(3.27)

with w(0) = 0 and q(0) = 0, such that the closed loop system has  $L_2$ -gain less than  $\gamma$ .

For application of this worst case certainty equivalence principle we again use a regularized version of our system  $\Sigma^m$ , given by

$$\Sigma_r^m \begin{cases} \dot{x} = f(x) + g(x)u + e(x)d_1 \\ y = c(x) + d_2 \\ \bar{z} = \begin{pmatrix} h(x) + k(x)u \\ \sqrt{\varepsilon}\beta_2^T(x)u \end{pmatrix} \end{cases}$$
(3.28)

To be able to derive a result similar to Theorem 3.18 we make the following assumption.

Assumption 4 Let (3.27) be a compensator for  $\Sigma^m$ . We assume that the  $L_2$ gain from  $d_1$ ,  $d_2$  to  $\beta_2^T(x)q(\xi)$  is finite, i.e., there exists a constant N > 0 such that for all  $x \in M$ ,  $\xi \in M_c$  there exists a constant  $\tilde{K}_c(x,\xi)$ ,  $0 \leq \tilde{K}_c(x,\xi) < \infty$ , with  $\tilde{K}(0,0) = 0$  such that

$$\int_{0}^{t} \|\beta_{2}^{T}(x(\tau))q(\xi(\tau))\|^{2} d\tau$$
  
$$\leq N \int_{0}^{t} (\|d_{1}(\tau)\|^{2} + \|d_{2}(\tau)\|^{2}) d\tau + \tilde{K}_{c}(x(0), \xi(0))$$

for all t > 0 and all  $d_1$ ,  $d_2 \in L_2(0, t)$ , where  $x(\tau)$  is the solution of the state equation of the closed loop system of  $\Sigma^m$  together with the compensator (3.27).

Then the following result can be stated.

**Theorem 3.27** Let (3.27) be a compensator for  $\Sigma^m$ . We have the following implications regarding the statements (i) and (ii) below. Statement (ii) implies (i), and under Assumption 4 the statements (i) and (ii) are equivalent.

- (i) The closed loop system consisting of  $\Sigma^m$  with the compensator (3.27) has  $L_2$ -gain less than  $\gamma$ .
- (ii) For  $\varepsilon$  sufficiently small the closed loop system consisting of  $\Sigma_r^m$  with the compensator (3.27) has  $L_2$ -gain less than  $\gamma$ .

The proof is similar to the proof of Theorem 3.18 and will be omitted.

Now we search our compensator among the set of compensators which result in a finite  $L_2$ -gain from  $d_1$ ,  $d_2$  to  $\beta_2^T(x)q(\xi)$ .

We will do so by applying the worst case certainty equivalence principle to the regularized system  $\Sigma_r^m$ . This principle consists in first solving the state feedback problem and then replacing the state by an estimated state. A sufficient condition for the worst case certainty equivalence principle to hold is that there exists a saddle point solution to this state feedback problem. This is the reason for considering a regularized version of the system  $\Sigma^m$ . Now a brief exposition of the construction of the compensator based on the worst-case certainty equivalence principle will be given (for more details see [BB 90], [vdS 93]).

We start by considering the  $\mathcal{H}_{\infty}$  problem on a finite time horizon, i.e., we consider the  $L_2$ -gain on some fixed finite interval  $[T_1, T_2]$ . This amounts to the max-min solution of the performance criterion

$$\frac{1}{2} \int_{T_1}^{T_2} \left( \varepsilon \| \beta_2^T(x) u \|^2 + (h(x) + k(x)u)^T (h(x) + k(x)u) - \gamma^2 d_1^T d_1 - \gamma^2 d_2^T d_2 \right) dt$$
(3.29)

where the control u(t),  $t \in [T_1, T_2]$ , is only allowed to depend on  $y(\tau)$  with  $T_1 \le \tau \le t$ . This problem can be split into two parts.

First we look at the state feedback  $\mathcal{H}_{\infty}$  control problem, considered in Subsection 3.2 for the infinite horizon case. This leads in the finite horizon case to the non-stationary Hamilton-Jacobi equation (compare with (3.15))

$$V_t(x,t) + V_x(x,t) \left( f(x) - g(x)\psi(x)k^T(x)h(x) \right)$$

54

$$+\frac{1}{2}\left(\frac{1}{\gamma^{2}}+\delta\right)V_{x}(x,t)e(x)e^{T}(x)V_{x}^{T}(x,t)$$
  
$$-\frac{1}{2}V_{x}(x,t)\left(g(x)\psi(x)g^{T}(x)+\frac{1}{\varepsilon}g(x)\beta_{2}(x)\beta_{2}^{T}(x)g^{T}(x)\right)V_{x}^{T}(x,t)$$
  
$$+\frac{1}{2}h^{T}(x)\left(I_{p}-k(x)\psi(x)k^{T}(x)\right)h(x) = 0$$
  
$$V(x,T_{2}) = 0$$

with resulting suboptimal state feedback

$$u(t) = -\left(\frac{1}{\varepsilon}\beta_2(x)\beta_2^T(x) + \psi(x)\right)g^T(x)V_x^T(x,t) - \psi(x)k^T(x)h(x).$$
(3.30)

Secondly, let  $\tau \in [T_1, T_2]$ , and let  $\bar{u}(t)$  and  $\bar{y}(t)$ ,  $t \in [T_1, \tau]$  be a given pair of inputs and corresponding measured output trajectories of the system  $\Sigma_r^m$ . Then we look for the maximizing solution  $x(T_1)$  and  $d_1(t)$ ,  $d_2(t)$  of the performance criterion

$$\frac{1}{2} \int_{T_1}^{\tau} \left( \varepsilon \| \beta_2^T(x) \bar{u} \|^2 + (h(x) + k(x) \bar{u})^T (h(x) + k(x) \bar{u}) - \gamma^2 d_1^T d_1 - \gamma^2 d_2^T d_2 \right) dt + V(x(\tau), \tau)$$
(3.31)

which satisfies the constraint that the measured output equals  $\bar{y}(t)$ . We assume that this maximization problem has a unique solution for every  $\tau \in [T_1, T_2]$ . Then we define for every  $\tau \in [T_1, T_2]$ 

$$\tilde{u}(\tau) = -\left(\frac{1}{\varepsilon}\beta_2(\hat{x}(\tau))\beta_2^T(\hat{x}(\tau)) + \psi(\hat{x}(\tau))\right)g^T(\hat{x}(\tau))V_x^T(\hat{x}(\tau),\tau) -\psi(\hat{x}(\tau))k^T(\hat{x}(\tau))h(\hat{x}(\tau))$$
(3.32)

where  $\hat{x}(\cdot)$  is the state trajectory corresponding to the maximizing solution of the performance criterion (3.31). Now  $\hat{x}(\cdot)$  depends on  $\bar{u}(\cdot)$  and by (3.32) we have defined a causal mapping from  $\bar{u}(t)$  to  $\tilde{u}(t)$ ,  $t \in [T_1, T_2]$ . Denote the fixed point of this mapping by  $\hat{u}(t)$ . This fixed point only depends on  $\bar{y}$ , in a causal way. Now

$$\hat{u}(t) = -\left(\frac{1}{\varepsilon}\beta_{2}(\hat{x}(t))\beta_{2}^{T}(\hat{x}(t)) + \psi(\hat{x}(t))\right)g^{T}(\hat{x}(t))V_{x}^{T}(\hat{x}(t), t) -\psi(\hat{x}(t))k^{T}(\hat{x}(t))h(\hat{x}(t))$$
(3.33)

is the solution of the considered optimization problem (3.29) (see [BB 90]). This is called the *worst case certainty equivalence solution*.

We still have to solve the maximization problem of the performance criterion (3.31) under the constraint that the measured output equals  $\bar{y}(t)$ . Since the disturbance  $d_2$  fully influences the observations y we can substitute  $d_2 = c(x) - \bar{y}$  into the performance criterion such that the constraint is automatically satisfied. Now we have simplified the constrained maximization problem to the unconstrained maximization of the performance criterium (3.31) with  $d_2 = c(x) - \bar{y}$ . This can be done in a classical way by first maximizing the criterion (3.31) under the constraint that the final value of the state  $x(\tau)$  equals x, and after that maximizing the solution of this problem with respect to x. The second maximization with respect to x is equal to the maximum of  $S(x, \tau) =$  $V(x, \tau) - W(x, \tau)$  where  $W \ge 0$  satisfies

$$W_{t}(x,t) + W_{x}(x,t) (f(x) + g(x)\bar{u}(t)) + \frac{1}{2} \frac{1}{\gamma^{2}} W_{x}(x,t) e(x) e^{T}(x) W_{x}^{T}(x,t) + \frac{1}{2} (h(x) + k(x)\bar{u}(t))^{T}(x) (h(x) + k(x)\bar{u}(t)) - \frac{1}{2} \gamma^{2} c^{T}(x) c(x)$$
(3.34)  
$$+ \gamma^{2} c^{T}(x) \bar{y}(t) - \frac{1}{2} \gamma^{2} \|\bar{y}(t)\|^{2} + \frac{1}{2} \varepsilon \|\beta_{2}^{T}(x)\bar{u}(t)\|^{2} = 0, W(x, T_{1}) = 0.$$

Assume that this maximum is determined by  $S_x(\hat{x}(t), t) = 0$  and that the Hessian is non-degenerate. Then the state equation for  $\hat{x}(\cdot)$  can be found by differentiation of this equality (see [vdS 93]).

The resulting compensator which solves the *infinite* horizon  $\mathcal{H}_{\infty}$  problem can be found by letting  $T_2 \to \infty$  while imposing that  $x(t) \to 0$  for  $t \to \infty$  and  $T_1 \to -\infty$  while  $x(t) \to 0$  for  $t \to -\infty$ .

A finite dimensional approximation to the constructed nonlinear controller which locally solves the  $\mathcal{H}_{\infty}$  problem is given by

$$\begin{split} \dot{\xi} &= f(\xi) - g(\xi) \left( \frac{1}{\varepsilon} \beta_2(\xi) \beta_2^T(\xi) + \sigma(\xi) \right) g^T(\xi) V_{\xi}^T(\xi) \\ &- g(\xi) \psi(\xi) k^T(\xi) h(\xi) + \frac{1}{\gamma^2} e(\xi) e^T(\xi) V_{\xi}^T(\xi) \\ &+ \gamma^2 \left[ W_{\xi\xi}(\xi) - V_{\xi\xi}(\xi) \right]^{-1} \frac{\partial c^T}{\partial \xi} (\xi) (y(t) - c(\xi)) \end{split}$$
(3.35)  
$$u &= - \left( \frac{1}{\varepsilon} \beta_2(\xi) \beta_2^T(\xi) + \psi(\xi) \right) g^T(\xi) V_{\xi}^T(\xi) - \psi(\xi) k^T(\xi) h(\xi) \end{split}$$

where  $V(\xi)$  is a solution of (3.15) with equality and  $W(\xi)$  is a solution of the stationary version of (3.34) ( $\bar{u}(t) = 0$ ;  $\bar{y}(t) = 0$ )

$$W_{x}(x) f(x) + \frac{1}{2} \frac{1}{\gamma^{2}} W_{x}(x) e(x) e^{T}(x) W_{x}^{T}(x) + \frac{1}{2} h^{T}(x) h(x) - \frac{1}{2} \gamma^{2} c^{T}(x) c(x) = 0$$
(3.36)  
$$W(0) = 0$$

such that:

$$f - g\left(\frac{1}{\varepsilon}\beta_{2}\beta_{2}^{T} + \psi\right)g^{T}V_{x}^{T} - g\psi k^{T}h + \frac{1}{\gamma^{2}}ee^{T}V_{x}^{T}$$
  
is exponentially stable;  
$$-\left(f + \frac{1}{\gamma^{2}}ee^{T}W_{x}^{T}\right)$$
 is exponentially stable; (3.37)

Hence we have the following result.

 $W_{xx}(x) > V_{xx}(x), \quad \forall x.$ 

**Theorem 3.28** Consider the system  $\Sigma^m$ , and suppose there exist a constant  $\varepsilon > 0$  and solutions  $V \ge 0$ ,  $W \ge 0$  to (3.15) with equality, respectively (3.36), satisfying (3.37). Then the closed loop system  $\Sigma^m$ , (3.35) has locally  $L_2$ -gain less than  $\gamma$ .

**Proof** The proof is based on linearization of the closed-loop system and combining the results about the so called central controller (see [DGKF 89]) and Theorem 2.22 (see [vdS 93]).

Also a converse result can be obtained, invoking Assumption 4.

**Theorem 3.29** Suppose the  $\mathcal{H}_{\infty}$  suboptimal control problem for  $\Sigma^m$  with constant  $\overline{\gamma} < \gamma$  is solvable by a compensator (3.27) satisfying Assumption 4 in the sense that there exists a constant  $\mu > 0$  and a  $C^1$ -solution  $V(x, \xi) \ge 0$  to the corresponding Hamilton-Jacobi inequality

$$V_{x}(x,\xi) (f(x) + g(x)q(\xi)) + V_{\xi}(x,\xi) (w(\xi) + p(\xi)c(x)) + \frac{1}{2} \left(\frac{1}{\gamma^{2}} + \mu\right) V_{x}(x,\xi)e(x)e^{T}(x)V_{x}^{T}(x,\xi) + \frac{1}{2} \left(\frac{1}{\gamma^{2}} + \mu\right) V_{\xi}(x,\xi)p(\xi)p^{T}(\xi)V_{\xi}(x,\xi)$$
(3.38)

$$+\frac{1}{2}(h(x) + k(x)q(\xi))^{T}(h(x) + k(x)q(\xi)) + \frac{1}{2}\varepsilon q^{T}(\xi)\beta_{2}(x)\beta_{2}^{T}(x)q(\xi) \leq 0,$$

$$V(0,0) = 0.$$

Furthermore assume that  $V_{\xi}(x, \xi) = 0$  has a  $C^1$ -solution  $\xi = F(x)$ , F(0) = 0, with  $F: M \to M_c$ . Then there exist nonnegative solutions P(x) and W(x) to the Hamilton-Jacobi inequality (3.15) respectively

$$W_{x}(x) f(x) + \frac{1}{2} \frac{1}{\tilde{\gamma}^{2}} W_{x}(x) e(x) e^{T}(x) W_{x}^{T}(x) \qquad (3.39)$$
  
+  $\frac{1}{2} h^{T}(x) h(x) - \frac{1}{2} \tilde{\gamma}^{2} c^{T}(x) c(x) \leq 0,$   
 $W(0) = 0,$ 

where  $\frac{1}{\tilde{\gamma}^2} = \frac{1}{\gamma^2} + \mu$  satisfying the coupling condition

$$V(x) \le W(x) \tag{3.40}$$

near 0.

Proof Define

$$P(x) := V(x, F(x)) \ge 0$$

then substitution of  $\xi = F(x)$  into (3.38) yields

$$P_{x}(x) (f(x) + g(x)q(F(x))) + \frac{1}{2}(\frac{1}{\gamma^{2}} + \mu)P_{x}(x)e(x)e^{T}(x)P_{x}^{T}(x) + \frac{1}{2}(h(x) + k(x)q(F(x)))^{T}(h(x) + k(x)q(F(x))) + \frac{1}{2}\varepsilon q^{T}(F(x))\beta_{2}(x)\beta_{2}^{T}(x)q(F(x)) = P_{x}(x) (f(x) - g(x)\psi(x)k^{T}(x)h(x)) + \frac{1}{2}(\frac{1}{\gamma^{2}} + \mu)P_{x}(x)e(x)e^{T}(x)P_{x}^{T}(x) - \frac{1}{2}P_{x}(x) \left(g(x)\psi(x)g^{T}(x) + \frac{1}{\varepsilon}g(x)\beta_{2}(x)\beta_{2}^{T}(x)g^{T}(x)\right)P_{x}^{T}(x) + \frac{1}{2}h^{T}(x) (I_{p} - k(x)\psi(x)k^{T}(x))h(x)$$

$$+ \frac{1}{2}\varepsilon \left\| \beta_{2}^{T}(x)q(F(x)) + \frac{1}{\varepsilon}\beta_{2}^{T}(x)g^{T}(x)P_{x}^{T}(x) \right\|^{2} \\ + \frac{1}{2} \left\| \left( v^{T}(x)v(x) \right)^{\frac{1}{2}}\beta_{1}^{T}(x)q(F(x)) \\ + \left( v^{T}(x)v(x) \right)^{-\frac{1}{2}} \left( \beta_{1}^{T}(x)g^{T}(x)P_{x}^{T}(x) + v^{T}(x)h(x) \right) \right\|^{2}$$

while P(0) = V(0, 0) = 0. Thus we see that P(x) is a solution to the Hamilton-Jacobi inequality (3.15). For a solution W to (3.39) we define

$$W(x) := V(x,0) \ge 0.$$

Substitution of  $\xi = 0$  into (3.38) and completion of the squares yields

$$W_{x}(x)f(x) + \frac{1}{2}\frac{1}{\tilde{\gamma}^{2}}W_{x}(x)e(x)e^{T}(x)W_{x}^{T}(x) + \frac{1}{2}h^{T}(x)h(x) \\ + \frac{1}{2}\frac{1}{\tilde{\gamma}^{2}}\left\|\tilde{\gamma}^{2}c(x) + p^{T}(0)V_{\xi}^{T}(x,0)\right\|^{2} - \frac{1}{2}\tilde{\gamma}^{2}c^{T}(x)c(x) \leq 0$$

with W(0) = 0, and consequently W is a solution of (3.39).

Finally since  $V(x, \xi) \ge 0$  and  $V_{\xi}(x, F(x)) = 0$  it necessarily follows that  $P(x) = V(x, F(x)) \le V(x, 0) = W(x)$ , at least for x near zero.

The measurement feedback  $\mathcal{H}_{\infty}$  control problem even in the regular case is not completely solved. As described in this subsection sufficient and necessary conditions for the singular case can be given but a full characterization of the problem has not been derived yet. In this subsection we only considered singularities with respect to the inputs. The  $\mathcal{H}_{\infty}$  problem with singular measurements in which the direct feedthrough from disturbances to measurements is not surjective is an open problem.

### **Chapter 4**

# The $\mathcal{H}_{\infty}$ almost disturbance decoupling problem

A special case of a  $\mathcal{H}_{\infty}$  optimal control problem is the almost disturbance decoupling problem (ADDP) in which the goal is to make the  $L_2$ -gain from disturbances to the to-be-controlled variables arbitrary small by applying a (possibly high) gain feedback. For single-input single-output affine nonlinear systems this problem is considered and sufficient conditions for the solvability are given in [MRST 94] and [MT 95]. In this chapter we make an extension of these results to the multi-input multi-output case. To be able to extend the SISO results we assume that a certain decoupling condition is satisfied.

The results of this chapter are instrumental for the next chapter where the singular  $\mathcal{H}_{\infty}$  problem is solved in a geometric way by decomposing via a state transformation the  $\mathcal{H}_{\infty}$  problem into a regular  $\mathcal{H}_{\infty}$  problem together with an  $\mathcal{H}_{\infty}$  almost disturbance decoupling problem.

## 4.1 Almost disturbance decoupling for affine nonlinear systems

Consider affine nonlinear MIMO systems with *m* inputs and *p* outputs  $(m \ge p)$  of the form

$$\dot{x} = f(x) + \sum_{j=1}^{m} g_j(x) u_j + \sum_{i=1}^{q} e_i(x) d_i$$
  

$$z = h(x)$$
(4.1)

where again  $x = (x_1, ..., x_n)$  are local coordinates for a smooth manifold M,  $z \in \mathbb{R}^p$ , and the functions  $f, g_j, e_i$  and h are smooth.

Parallel to the  $L_2$ -gain and  $\mathcal{H}_{\infty}$  problems defined in section 2.2 the following two problems are defined.

**Definition 4.1** The  $L_2$ -gain almost disturbance decoupling problem is solvable for (4.1) if there exists a smooth parameterized state feedback control

$$u = u(x, k)$$

such that there exists a constant K(x),  $0 \le K(x) < \infty$ , K(0) = 0, such that for every t,  $0 \le t \le T$ ,

$$\int_0^t z^T(\tau) z(\tau) \mathrm{d}\tau \leq \frac{1}{k} \int_0^t d^T(\tau) d(\tau) \mathrm{d}\tau + K(x_0)$$

for the closed-loop system with initial condition  $x(0) = x_0$ , and for every disturbance function  $d(\tau) \in L_2(0, t)$ , with [0, T) any open interval on which the corresponding solution exists.

**Definition 4.2** The  $\mathcal{H}_{\infty}$  almost disturbance decoupling problem is solvable for (4.1) if the  $L_2$ -gain almost disturbance decoupling problem is solvable by a feedback  $u = u(x, k), u(0, k) = 0, \forall k \in \mathbb{R}^+$ , while the origin is globally asymptotically stable for the closed-loop dynamics

$$\dot{x} = f(x) + g(x)u(x,k)$$

for every k (large enough).

Let us define the strong control and the disturbance characteristic indices for (4.1), which indicate the amount of differentiations before the specific output components are influenced by the control respectively the disturbance.

**Definition 4.3** The strong control characteristic indices  $\rho_1, \ldots, \rho_p$  for the system (4.1) are defined such that:

$L_{g_j}L_f^r h_l(x) = 0,$	$0 \le r \le \rho_l - 2, 1 \le j \le m, \forall x;$
$L_{g_i}L_f^{\rho_l-1}h_l(x)\neq 0,$	for some $j, \forall x$ .

If  $L_{g_i}L_f^i h_l(x) = 0$ ,  $\forall i, j \text{ and } \forall x \in \mathbb{R}^n$ , then  $\rho_l = \infty$ .
It should be noted that the strong characteristic indices do not always exists. This is due to the fact that we require the functions  $L_{g_j}L_f^{\rho_l-1}h_l$  to be unequal to zero for all x. We call this strong control characteristic indices  $\rho_1, \ldots, \rho_p$  well defined if  $\rho_l$  exists and is finite for all  $l = 1, \ldots, p$ .

**Definition 4.4** The *disturbance characteristic indices*  $v_1, \ldots, v_p$  for the system (4.1) are defined such that:

$$L_{e_i}L_f^r h_l(x) = 0, \qquad 0 \le r \le \nu_l - 2, \ 1 \le i \le q, \ \forall x;$$
  
$$L_{e_i}L_f^{\nu_l - 1} h_l(x) \ne 0, \qquad \text{for some } i, \text{ and for some } x.$$

If  $L_{e_i}L_f^j h_l(x) = 0$ ,  $\forall i, j \text{ and } \forall x \in \mathbb{R}^n$ , then  $\nu_l = \infty$ .

Assume that  $\rho_1, \ldots, \rho_p$  are well defined. We introduce the  $(p \times m)$  decoupling matrix A(x):

$$A(x) = \begin{pmatrix} L_{g_1} L_f^{\rho_1 - 1} h_1(x) & \cdots & L_{g_m} L_f^{\rho_1 - 1} h_1(x) \\ \vdots & & \vdots \\ L_{g_1} L_f^{\rho_p - 1} h_p(x) & \cdots & L_{g_m} L_f^{\rho_p - 1} h_p(x) \end{pmatrix}.$$
 (4.2)

Assume that this decoupling matrix has full row rank. Then we know that if  $v_i > \rho_i$  holds for all *i* the disturbance decoupling problem is solvable by a regular state feedback (see [Is 89], [NvdS 90], [MT 95]), or otherwise stated the  $L_2$ -gain from disturbance to output can be made zero by applying a proper state feedback. This problem is no longer solvable when for some *i* we have that  $v_i \le \rho_i$ . In this case it is interesting to look at the almost disturbance decoupling problem. In this chapter we consider the case where  $v_i \le \rho_i$  for all *i*. Because we assume that the decoupling matrix A(x) has full row rank it is not difficult to extend the results to the case where for some *i* it holds that  $v_i > \rho_i$ .

We describe the results only for the case where the number of inputs is equal to the number of outputs (m = p). For the situation m > p assuming that the strong controllability indices are well defined and the decoupling matrix A(x) has full row rank we can at least locally always select p independent columns of this matrix A(x) and apply the results using as decoupling matrix the  $(p \times p)$  matrix consisting of these p independent columns together with the corresponding inputs.

We start by the following straightforward extension of the results for SISO systems.

**Theorem 4.5** Let for the system (4.1) the following assumptions be satisfied:

- (*i*) m = p;
- (*ii*)  $\rho_1, \ldots, \rho_p$  are well defined;
- (iii) the decoupling matrix A(x) has full rank;
- (iv) the distribution  $G = \{g_1, \ldots, g_m\}$  is involutive;
- (v)  $d\left(L_{e_j}L_f^ih_l\right) \in \text{span}\left\{dh_l, d\left(L_fh_l\right), \dots, d\left(L_f^ih_l\right)\right\}, v_l 1 \le i \le \rho_l 1, 1 \le j \le q, 1 \le l \le p, \forall x \in \mathbb{R}^n;$
- (vi) the vector fields

$$\tilde{f} = f - gA^{-1} \begin{pmatrix} L_f^{\rho_1} h_1 \\ \vdots \\ L_f^{\rho_p} h_p \end{pmatrix}, \qquad \tilde{g} = gA^{-1}$$

are complete (see Appendix A).

Then the  $L_2$ -gain almost disturbance decoupling problem is solvable.

**Proof** The matrix A(x) can be written as

$$A(x) = \begin{pmatrix} dL_f^{\rho_1-1}h_1(x) \\ \vdots \\ dL_f^{\rho_p-1}h_p(x) \end{pmatrix} (g_1(x) \cdots g_m(x)).$$

Because m = p it follows from the non-singularity of this matrix that the *m*-vectors  $g_1(x), \ldots, g_m(x)$  are linearly independent. Hence the distribution G is constant dimensional.

Therefore by (iv), (vi) a global version of Frobenius' Theorem yields a global change of coordinates (see [MRS 89] and the references in there)

$$q_{11} = h_1(x),$$
  

$$\vdots$$
  

$$q_{1\rho_1} = L_f^{\rho_1 - 1} h_1(x),$$
  

$$q_{21} = h_2(x),$$

$$i$$

$$q_{2\rho_{2}} = L_{f}^{\rho_{2}-1}h_{2}(x),$$

$$i$$

$$(4.3)$$

$$i$$

$$q_{p1} = h_{p}(x),$$

$$i$$

$$q_{p\rho_{p}} = L_{f}^{\rho_{p}-1}h_{p}(x),$$

$$q_{\rho+1} = \varphi_{\rho+1}(x),$$

$$i$$

$$q_{n} = \varphi_{n}(x)$$

with  $\rho = \sum_{i=1}^{m}$  and  $\varphi_i(x)$ ,  $\rho + 1 \le i \le n$ ,  $\varphi_i(0) = 0$ , such that

$$\langle \mathrm{d}\varphi_i, G \rangle = 0$$

together with the state feedback

$$v = A(x)u + \begin{pmatrix} L_f^{\rho_1} h_1(x) \\ \vdots \\ L_f^{\rho_p} h_p(x) \end{pmatrix}$$
(4.4)

which globally transform (4.1) into

$$\dot{q}_{11} = q_{12} + \sum_{i=1}^{q} L_{e_i} h_1(x) d_i = q_{12} + W_{11}^T(q) d,$$
  

$$\vdots$$
  

$$\dot{q}_{1\rho_1} = v_1 + \sum_{i=1}^{q} L_{e_i} L_f^{\rho_1 - 1} h_1(x) d_i = v_1 + W_{1\rho_1}^T(q) d,$$
  

$$\dot{q}_{21} = q_{22} + \sum_{i=1}^{q} L_{e_i} h_2(x) d_i = q_{22} + W_{21}^T(q) d,$$
  

$$\vdots$$
  

$$\dot{q}_{2\rho_2} = v_2 + \sum_{i=1}^{q} L_{e_i} L_f^{\rho_2 - 1} h_2(x) d_i = v_2 + W_{2\rho_2}^T(q) d,$$

$$i \qquad (4.5)$$

$$i \qquad (4.7)$$

$$(4.7)$$

$$(4.7)$$

$$(4.7)$$

$$(4.7)$$

$$(4.7)$$

$$(4.7)$$

$$(4.7)$$

$$(4.7)$$

$$(4.7)$$

$$(4.7)$$

$$(4.7)$$

$$(4.7)$$

$$(4.7)$$

$$(4.7)$$

$$(4.7)$$

$$(4.7)$$

$$(4.7)$$

$$(4.7)$$

$$(4.7)$$

$$(4.7)$$

$$(4.7)$$

$$(4.7)$$

$$(4.7)$$

$$(4.7)$$

$$(4.7)$$

$$(4.7)$$

$$(4.7)$$

$$(4.7)$$

$$(4.7)$$

$$(4.7)$$

$$(4.7)$$

$$(4.7)$$

$$(4.7)$$

$$(4.7)$$

$$(4.7)$$

$$(4.7)$$

$$(4.7)$$

$$(4.7)$$

$$(4.7)$$

$$(4.7)$$

$$(4.7)$$

$$(4.7)$$

$$(4.7)$$

$$(4.7)$$

$$(4.7)$$

$$(4.7)$$

$$(4.7)$$

$$(4.7)$$

$$(4.7)$$

$$(4.7)$$

$$(4.7)$$

$$(4.7)$$

$$(4.7)$$

$$(4.7)$$

$$(4.7)$$

$$(4.7)$$

$$(4.7)$$

$$(4.7)$$

$$(4.7)$$

$$(4.7)$$

$$(4.7)$$

$$(4.7)$$

$$(4.7)$$

$$(4.7)$$

$$(4.7)$$

$$(4.7)$$

$$(4.7)$$

$$(4.7)$$

$$(4.7)$$

$$(4.7)$$

$$(4.7)$$

$$(4.7)$$

$$(4.7)$$

$$(4.7)$$

$$(4.7)$$

$$(4.7)$$

$$(4.7)$$

$$(4.7)$$

$$(4.7)$$

$$(4.7)$$

$$(4.7)$$

$$(4.7)$$

$$(4.7)$$

$$(4.7)$$

$$(4.7)$$

$$(4.7)$$

$$(4.7)$$

$$(4.7)$$

$$(4.7)$$

$$(4.7)$$

$$(4.7)$$

$$(4.7)$$

$$(4.7)$$

$$(4.7)$$

$$(4.7)$$

$$(4.7)$$

$$(4.7)$$

$$(4.7)$$

$$(4.7)$$

$$(4.7)$$

$$(4.7)$$

$$(4.7)$$

$$(4.7)$$

$$(4.7)$$

$$(4.7)$$

$$(4.7)$$

$$(4.7)$$

$$(4.7)$$

$$(4.7)$$

$$(4.7)$$

$$(4.7)$$

$$(4.7)$$

$$(4.7)$$

$$(4.7)$$

$$(4.7)$$

$$(4.7)$$

$$(4.7)$$

$$(4.7$$

with  $q_r = (q_{\rho+1}, \ldots, q_n)$  and  $d = (d_1, \ldots, d_q)$ . Because of condition ( $\nu$ ) the system (4.5) takes the following form (see [MRST 94])

$$\dot{q}_{11} = q_{12} + W_{11}^{T}(q_{11})d,$$

$$\vdots$$

$$\dot{q}_{1\rho_{1}} = v_{1} + W_{1\rho_{1}}^{T}(q_{11}, \dots, q_{1\rho_{1}})d,$$

$$\dot{q}_{21} = q_{22} + W_{21}^{T}(q_{21})d,$$

$$\vdots$$

$$\dot{q}_{2\rho_{2}} = v_{2} + W_{2\rho_{2}}^{T}(q_{21}, \dots, \rho_{2\rho_{2}})d,$$

$$\vdots$$

$$\dot{q}_{p1} = q_{p2} + W_{p1}^{T}(q_{p1})d,$$

$$\vdots$$

$$\dot{q}_{p\rho_{p}} = v_{p} + W_{p\rho_{p}}^{T}(q_{p1}, \dots, q_{p\rho_{p}})d,$$

$$\dot{q}_{r} = \varphi(q) + \Psi^{T}(q)d,$$

$$z_{l} = q_{l1}, \quad 1 \leq l \leq p.$$

$$(4.6)$$

Now the first  $\rho$  variables of the system are separated in *p* blocks for which we can apply the construction of the state feedback used in the proof in [MT 95] or [MRST 94].

In other words we can consider p systems

$$\Sigma_{i} \begin{cases} \dot{q}_{i1} = q_{i2} + W_{i1}^{T}(q_{i1})d, \\ \vdots \\ \dot{q}_{i\rho_{i}} = v_{i} + W_{i\rho_{i}}(q_{i1}, \dots, q_{i\rho_{i}})d, \\ z_{i} = q_{i1} \end{cases}$$
(4.7)

for  $1 \le i \le p$ .

Construct for each of these systems a state feedback control

$$v_i = q_{i,\rho_i+1}^*(q_{i1},\ldots,q_{i\rho_i},k)$$

such that

$$\int_0^T z_i^T(\tau) z_i(\tau) \mathrm{d}\tau \leq \frac{\rho_i}{k} \int_0^T d^T(\tau) d(\tau) \mathrm{d}\tau$$

for all  $T \ge 0$ . Then it follows that

$$\int_0^T z^T(\tau) z(\tau) \mathrm{d}\tau \leq \frac{\rho}{k} \int_0^T d^T(\tau) d(\tau) \mathrm{d}\tau.$$

Remark 4.6 The stability properties of the zero-dynamics

$$\dot{q}_r = \varphi(0, q_r) + \Psi^T(0, q_r)d$$

do not influence the  $L_2$ -gain result. In case we also require the closed-loop system to be internally stable then the stability of the zero-dynamics has to be taken into account as will be shown in Theorem 4.10.

As in the SISO case ([MRST 94], [MT 95]) also another set of sufficient conditions for the solvability of the  $L_2$ -gain almost disturbance decoupling problem can be formulated. **Theorem 4.7** Assume that for the system (4.1) the following is satisfied:

- (*i*) m = p;
- (*ii*)  $\rho_1, \ldots, \rho_p$  are well defined;
- (iii) the decoupling matrix A(x) has full rank;
- (iv) the vector fields

$$\tilde{f} = f - g A^{-1} \begin{pmatrix} L_f^{\rho_1} h_1 \\ \vdots \\ L_f^{\rho_p} h_p \end{pmatrix}, \qquad \qquad \tilde{g} = g A^{-1}$$

are complete.

Also assume that after applying the feedback

$$u = A^{-1}(x)v$$

to the system (4.1) the following conditions hold:

- (v)  $\mathcal{G}_{\rho-1} = \operatorname{span} \left\{ \tilde{g}_j, \operatorname{ad}_f \tilde{g}_j, \dots, \operatorname{ad}_f^{\rho_j-1} \tilde{g}_j; j = 1, \dots, m \right\}$  is involutive and has constant dimension  $\rho = \sum_{i=1}^m \rho_i$ ;
- (vi)  $\operatorname{ad}_{e_i} \mathcal{G}_j^l \subset \mathcal{G}_j^l$ , with  $\mathcal{G}_j^l = \operatorname{span} \left\{ \tilde{g}_l, \operatorname{ad}_f \tilde{g}_l, \ldots, \operatorname{ad}_f^j \tilde{g}_l \right\}, 1 \le i \le q,$  $0 \le j \le \rho_l - 2, 1 \le l \le m.$

Then the  $L_2$ -gain almost disturbance decoupling problem is solvable.  $\Box$ 

**Proof** We can globally define a change of coordinates (4.3) with  $\varphi_i(x)$ ,  $\rho + 1 \le i \le n$ ,  $\varphi_i(0) = 0$ , such that

$$\langle \mathrm{d}\varphi_i, \, \mathcal{G}_{\rho-1} \rangle = 0$$

together with the feedback (4.4) which globally transform (4.1) into

$$\dot{q}_{11} = q_{12} + \sum_{i=1}^{q} L_{e_i} h_1(x) d_i = q_{12} + W_{11}^T(q) d_i$$
  
:

$$\dot{q}_{1\rho_{1}} = v_{1} + \sum_{i=1}^{q} L_{e_{i}} L_{f}^{\rho_{1}-1} h_{1}(x) d_{i} = v_{1} + W_{1\rho_{1}}^{T}(q) d,$$
  
$$\dot{q}_{21} = q_{22} + \sum_{i=1}^{q} L_{e_{i}} h_{2}(x) d_{i} = q_{22} + W_{21}^{T}(q) d,$$

$$\dot{q}_{2\rho_2} = v_2 + \sum_{i=1}^q L_{e_i} L_{f_i}^{\rho_2 - 1} h_2(x) d_i = v_2 + W_{2\rho_2}^T(q) d,$$

$$\vdots$$

$$(4.8)$$

$$\dot{q}_{p1} = q_{p2} + \sum_{i=1}^{q} L_{e_i} h_p(x) d_i = q_{p2} + W_{p1}^T(q) d,$$

$$\begin{aligned} &\vdots \\ \dot{q}_{p\rho_p} &= v_p + \sum_{i=1}^q L_{e_i} L_f^{\rho_p - 1} h_p(x) d_i = v_p + W_{p\rho_p}^T(q) d, \\ &\dot{q}_r &= \varphi(q_{11}, q_{21}, \dots, q_{p1}, q_r) + \Psi^T(q) d, \\ &z_l &= q_{l1}, \quad 1 \le l \le p. \end{aligned}$$

In q-coordinates we have

$$\mathcal{G}_{\rho-1} = \operatorname{span}\left\{\frac{\partial}{\partial q_{i1}}, \ldots, \frac{\partial}{\partial q_{i\rho_i}}; 1 \le i \le p\right\}$$

and

$$G_j^l = \operatorname{span}\left\{\frac{\partial}{\partial q_{l(\rho_l-j)}}, \ldots, \frac{\partial}{\partial q_{l\rho_l}}\right\}$$

for  $0 \le j \le \rho_l - 1$ . Then conditions (*vi*) imply

$$W_{li}^{T}(q) = W_{li}^{T}(q_{l1}, \dots, q_{li}, q_{r})$$

$$1 \le i \le \rho_{l}, \ 1 \le l \le p.$$

$$\Psi^{T}(q) = \Psi^{T}(q_{11}, q_{21}, \dots, q_{p1}, q_{r})$$

And therefore (4.8) becomes

$$\dot{q}_{11} = q_{12} + W_{11}^T(q_{11}, q_r)d,$$
  
:

$$\begin{aligned} \dot{q}_{1\rho_{1}} &= v_{1} + W_{1\rho_{1}}^{T}(q_{11}, \dots, q_{1\rho_{1}}, q_{r})d, \\ \dot{q}_{21} &= q_{22} + W_{21}^{T}(q_{21}, q_{r})d, \\ \vdots \\ \dot{q}_{2\rho_{2}} &= v_{2} + W_{2\rho_{2}}^{T}(q_{21}, \dots, \rho_{2\rho_{2}}, q_{r})d, \\ \vdots \\ \dot{q}_{p1} &= q_{p2} + W_{p1}^{T}(q_{p1}, q_{r})d, \\ \vdots \\ \dot{q}_{p\rho_{p}} &= v_{p} + W_{p\rho_{p}}^{T}(q_{p1}, \dots, q_{p\rho_{p}}, q_{r})d, \\ \dot{q}_{r} &= \varphi(q_{11}, q_{21}, \dots, q_{p1}, q_{r}) + \Psi^{T}(q_{11}, q_{21}, \dots, q_{p1}, q_{r})d, \\ z_{l} &= q_{l1}, \quad 1 \leq l \leq p. \end{aligned}$$

$$(4.9)$$

Again the first  $\rho$  variables of the system have been separated in p blocks, for each of which we can apply the construction of the state feedback used in the proof of Theorem 4.5.

The conditions (v) in Theorem 4.5 and (vi) in Theorem 4.7 result in the strict block structure described in the proofs. On the other hand, a closer inspection of the proof shows that these conditions can be weakened. First we weaken the conditions of Theorem 4.5.

**Theorem 4.8** Assume that for the system (4.1) the following is satisfied:

- (*i*) m = p;
- (*ii*)  $\rho_1, \ldots, \rho_p$  are well defined;
- (iii) the decoupling matrix A(x) has full rank;
- (iv) the distribution  $G = \{g_1, \ldots, g_m\}$  is involutive;
- (v)  $d\left(L_{e_j}L_f^ih_l\right) \in \text{span}\left\{dh_s, d\left(L_fh_s\right), \dots, d\left(L_f^{\max(i,\rho_s-1)}h_s\right)\right\}, \text{ for any } s = 1, \dots, p, v_l 1 \le i \le \rho_l 1, 1 \le j \le q, 1 \le l \le p, \forall x \in \mathbb{R}^n;$

(vi) the vector fields

$$\tilde{f} = f - g A^{-1} \begin{pmatrix} L_f^{\rho_1} h_1 \\ \vdots \\ L_f^{\rho_p} h_p \end{pmatrix}, \qquad \tilde{g} = A^{-1} g$$

are complete.

Then the  $L_2$ -gain almost disturbance decoupling problem is solvable.  $\Box$ 

**Proof** As in the proof of Theorem 4.5 the distribution *G* is involutive and constant dimensional, and hence we can again define a change of coordinates (4.3) with  $\varphi_i(x)$ ,  $\rho + 1 \le i \le n$ ,  $\varphi_i(0) = 0$ , such that

$$\langle \mathrm{d}\varphi_i, G \rangle = 0.$$

Together with the state feedback (4.4) this change of coordinates globally transforms (4.1) into (4.5).

With condition (v) the system (4.1) with the feedback (4.4) is of the form

$$\dot{q}_{11} = q_{12} + W_{11}^{T}(\tilde{q}_{1})d,$$

$$\vdots$$

$$\dot{q}_{1\rho_{1}} = v_{1} + W_{1\rho_{1}}^{T}(\tilde{q}_{1}, \dots, \tilde{q}_{\rho_{1}})d,$$

$$\dot{q}_{21} = q_{22} + W_{21}^{T}(\tilde{q}_{1})d,$$

$$\vdots$$

$$\dot{q}_{2\rho_{2}} = v_{2} + W_{2\rho_{2}}^{T}(\tilde{q}_{1}, \dots, \tilde{q}_{\rho_{2}})d,$$

$$\vdots$$

$$\dot{q}_{p1} = q_{p2} + W_{p1}^{T}(\tilde{q}_{1})d,$$

$$\vdots$$

$$\dot{q}_{p\rho_{p}} = v_{p} + W_{p\rho_{p}}^{T}(\tilde{q}_{1}, \dots, \tilde{q}_{\rho_{p}})d,$$

$$\dot{q}_{r} = \varphi(q) + \Psi^{T}(q)d,$$

$$z_{l} = q_{l1}, \qquad 1 \le l \le p$$

$$(4.10)$$

where  $\tilde{q}_j$  is the set of coordinates given by

$$\tilde{q}_j = \{q_{kj}, \text{ for all } k = 1, \dots, p \text{ for which } j \le \rho_k\}.$$

Then we can construct a storage function and a static feedback for this system (4.10) which leads to closed loop system which has  $L_2$ -gain less than or equal to  $\frac{1}{k}$  by using the following algorithm:

Step 1 Define

$$q_{i2}^*(\tilde{q}_1, k) = -q_{i1} - \frac{1}{4}kq_{i1}W_{i1}^T(\tilde{q}_1)W_{i1}(\tilde{q}_1)$$
(4.11)

for  $i = 1, \ldots, p$ , and consider

$$V_1 = \frac{1}{2} \sum_{l=1}^p q_{l1}^2.$$

Its time derivative with respect to the closed loop system (4.10) using (4.11) is given by

$$\dot{V}_{1} = \sum_{l=1}^{p} \left( -q_{l1}^{2} - \frac{1}{4} k q_{l1}^{2} W_{l1}^{T} W_{l1} + q_{l1} W_{l1}^{T} d - \frac{1}{k} d^{T} d \right) + \frac{1}{k} d^{T} d \Big) = \sum_{l=1}^{p} \left( -q_{l1}^{2} - k \left\| \frac{1}{2} q_{l1}^{2} W_{l1} - \frac{1}{k} d \right\|^{2} + \frac{1}{k} d^{T} d \right) \le \sum_{l=1}^{p} \left( -q_{l1}^{2} + \frac{1}{k} d^{T} d \right).$$

Then define the index sets:

$$I_j = \{s | \rho_s > j\};$$
  

$$\mathcal{I}_j = \{s | \rho_s = j\};$$
  

$$\mathcal{K}_j = \{s | \rho_s \le j\}.$$

Finally we set

$$v_i = q_{i2}^*(\tilde{q}_1, k),$$
  
 $\hat{q}_{i1} = q_{i1}$ 

for all  $i \in \mathcal{J}_1$ .

**Step** s + 1 Assume that for a given index  $s, 1 \le s \le \max_i \rho_i$ , for the system

$$\begin{split} \dot{q}_{l1} &= q_{l2} + W_{l1}^{T}(\tilde{q}_{1})d \\ &\vdots \\ \dot{q}_{l\rho_{l}} &= v_{l} + W_{l\rho_{l}}^{T}(\tilde{q}_{1}, \dots, \tilde{q}_{\rho_{l}})d, \\ \dot{q}_{l1} &= q_{l2} + W_{l1}^{T}(\tilde{q}_{1})d \\ &\vdots \\ \dot{q}_{ls} &= v_{l} + W_{ls}^{T}(\tilde{q}_{1}, \dots, \tilde{q}_{s})d, \\ \end{split}$$

there exists for each  $l \in I_s$  s functions

$$q_{lj}^* = q_{lj}^*(\tilde{q}_1, \dots, \tilde{q}_{j-1}, k), \qquad 2 \le j \le s+1, \quad l \in I_s$$

with  $q_{lj}^*(0, \ldots, 0, k) = 0$  such that in the new coordinates

$$\bar{q}_{l1} = q_{l1} \bar{q}_{lj} = q_{lj} - q_{lj}^*(\tilde{q}_1, \dots, \tilde{q}_{j-1}, k), \qquad 2 \le j \le s, \quad l \in I_s$$

the function

$$V_s = \frac{1}{2} \sum_{l \in I_s} \sum_{j=1}^s \bar{q}_{lj}^2 + \frac{1}{2} \sum_{l \in \mathcal{K}_s} \sum_{j=1}^{\rho_l} \hat{q}_{lj}^2$$
(4.12)

has time derivative, with  $q_{l(s+1)} = q_{l(s+1)}^*$   $(l \in I_s)$ , satisfying the inequality

$$\dot{V}_{s} \leq -\sum_{l \in I_{s}} \sum_{j=1}^{s} \bar{q}_{lj}^{2} - \sum_{l \in \mathcal{K}_{s}} \sum_{j=1}^{\rho_{l}} \hat{q}_{lj}^{2} + \frac{c}{k} \|d\|^{2}$$

with

$$c = (\#I_s) \, s + \sum_{l \in \mathcal{K}_s} \rho_l$$

where  $#I_s$  indicates the number of indices in the set  $I_s$ . Then we consider the function

$$V_{s+1} = \frac{1}{2} \sum_{l \in I_s} \sum_{j=1}^{s+1} \bar{q}_{lj}^2 + \frac{1}{2} \sum_{l \in \mathcal{K}_s} \sum_{j=1}^{\rho_l} \hat{q}_{lj}^2$$

where

$$\bar{q}_{l(s+1)} = q_{l(s+1)} - q^*_{l(s+1)}(\tilde{q}_1, \dots, \tilde{q}_s, k), \qquad l \in I_s.$$

When  $q_{l(s+2)} = q_{l(s+2)}^*$  for all  $l \in I_s$  in (4.10), we have that

$$\dot{V}_{s+1} \leq -\sum_{l \in I_s} \sum_{j=1}^{s} \bar{q}_{lj}^2 - \sum_{l \in \mathcal{K}_s} \sum_{j=1}^{\rho_l} \hat{q}_{lj}^2 + \frac{c}{k} \|d\|^2$$

$$+ \sum_{l \in I_s} \bar{q}_{l(s+1)} \left[ \bar{q}_{ls} + W_{l(s+1)}^T d - \sum_{i=1}^{s} \frac{\partial q_{l(s+1)}^*}{\partial \tilde{q}_s} \dot{\tilde{q}}_s + q_{l(s+2)}^* \right]$$
(4.13)

where

$$\sum_{i=1}^{s} \frac{\partial q_{l(s+1)}^{*}}{\partial \tilde{q}_{s}} \tilde{q}_{s} = \sum_{j \in I_{s}} \sum_{i=1}^{s} \frac{\partial q_{l(s+1)}^{*}}{\partial q_{ji}} \left( q_{j(i+1)} + W_{ji}^{T} d \right)$$
$$+ \sum_{j \in \mathcal{H}_{s}} \sum_{i=1}^{\rho_{j}} \frac{\partial q_{l(s+1)}^{*}}{\partial q_{ji}} \left( q_{j(i+1)} + W_{ji}^{T} d \right)$$
$$+ \sum_{j \in \mathcal{H}_{s}} \frac{\partial q_{l(s+1)}^{*}}{\partial q_{j\rho_{j}}} \left( v_{j}(\tilde{q}_{1}, \dots, \tilde{q}_{\rho_{j}}, k) + W_{j\rho_{j}}^{T} d \right)$$

Now we define the following functions:

$$\begin{aligned} \alpha_{l}(\tilde{q}_{1}, \dots, \tilde{q}_{s+1}, k) &= \bar{q}_{ls} - \sum_{j \in I_{s}} \sum_{i=1}^{\rho_{j}} \frac{\partial q_{l(s+1)}^{*}}{\partial q_{ji}} q_{j(i+1)} \\ &- \sum_{j \in \mathcal{K}_{s}} \sum_{i=1}^{\rho_{j}} \frac{\partial q_{l(s+1)}^{*}}{\partial q_{ji}} q_{j(i+1)} \\ &- \sum_{j \in \mathcal{K}_{s}} \frac{\partial q_{l(s+1)}^{*}}{\partial q_{j\rho_{j}}} v_{j}(\tilde{q}_{1}, \dots, \tilde{q}_{\rho_{j}}, k); \end{aligned}$$

$$\beta_{l}(\tilde{q}_{1}, \dots, \tilde{q}_{s+1}) &= W_{l(s+1)} - \sum_{j \in I_{s}} \sum_{i=1}^{s} \frac{\partial q_{l(s+1)}^{*}}{\partial q_{ji}} W_{ji} \\ &- \sum_{j \in \mathcal{K}_{s}} \sum_{i=1}^{\rho_{j}} \frac{\partial q_{l(s+1)}^{*}}{\partial q_{ji}} W_{ji}; \end{aligned}$$

 $q_{l(s+2)}^{*}(\tilde{q}_{1},\ldots,\tilde{q}_{s+1},k) = -\alpha_{l} - \bar{q}_{l(s+1)} - \frac{1}{4}k\bar{q}_{l(s+1)}\beta_{l}^{T}\beta_{l};$ 

with  $l \in I_s$ . Then (4.13) becomes

$$\begin{split} \dot{V}_{s+1} &\leq -\sum_{l \in I_s} \sum_{j=1}^{s+1} \bar{q}_{lj}^2 - \sum_{l \in \mathcal{K}_s} \sum_{j=1}^{\rho_l} \hat{q}_{lj}^2 + \frac{c}{k} \|d\|^2 + \frac{\#I_s}{k} \|d\|^2 \\ &+ \sum_{l \in I_s} \bar{q}_{l(s+1)} \left[ \beta_l^T d - \frac{1}{4} k \bar{q}_{l(s+1)} \beta_l^T \beta_l \right] - \frac{\#I_s}{k} \|d\|^2 \\ &= -\sum_{l \in I_s} \sum_{j=1}^{s+1} \bar{q}_{lj}^2 - \sum_{l \in \mathcal{K}_s} \sum_{j=1}^{\rho_l} \hat{q}_{lj}^2 - \sum_{l \in I_s} k \left\| \frac{1}{2} \bar{q}_{l(s+1)} \beta_l - \frac{1}{k} d \right\|^2 \\ &+ \frac{c + \#I_s}{k} \|d\|^2 \\ &\leq -\sum_{l \in I_s} \sum_{j=1}^{s+1} \bar{q}_{lj}^2 - \sum_{l \in \mathcal{K}_s} \sum_{j=1}^{\rho_l} \hat{q}_{lj}^2 + \frac{c + \#I_s}{k} \|d\|^2. \end{split}$$

Define

$$\begin{array}{lll} \hat{q}_{li} &=& \bar{q}_{li}, & 1 \leq i \leq s+1 \\ v_l &=& q^*_{l(s+2)}(\tilde{q}_1, \dots, \tilde{q}_{s+1}, k) \end{array} \qquad l \in \mathcal{I}_{s+1}. \end{array}$$

Then from (4.13) we can conclude that

$$\dot{V}_{s+1} \leq -\sum_{l \in I_{s+1}} \sum_{j=1}^{s+1} \bar{q}_{lj}^2 - \sum_{l \in \mathcal{K}_{s+1}} \sum_{j=1}^{\rho_l} \hat{q}_{lj}^2 + \frac{c}{k} \|d\|^2$$

where

$$c = (\#I_s)s + \sum_{l \in \mathcal{K}_s} \rho_l + \#I_s$$
  
=  $(\#I_{s+1})(s+1) + \sum_{l \in \mathcal{K}_{s+1}} \rho_l.$ 

(end of step s + 1)

After  $\max_i \rho_i$  steps we have that

$$V_{\max_i \rho_i} = \frac{1}{2} \sum_{l=1}^{p} \sum_{j=1}^{\rho_l} \hat{q}_{lj}^2$$

while

$$\begin{split} \dot{V}_{\max_{i}\rho_{i}} &\leq -\sum_{l=1}^{p} \sum_{j=1}^{\rho_{l}} \hat{q}_{lj}^{2} + \frac{\rho}{k} \|d\|^{2} \\ &= -\sum_{l=1}^{p} q_{l1}^{2} - \sum_{l \in I_{1}} \sum_{j=1}^{\rho_{l}} \hat{q}_{lj}^{2} + \frac{\rho}{k} \|d\|^{2} \\ &\leq -\|z\|^{2} + \frac{\rho}{k} \|d\|^{2} \end{split}$$

which implies that

$$V_{\max_{i}\rho_{i}}(q(t)) - V_{\max_{i}\rho_{i}}(q(0)) \leq \int_{0}^{t} \left(-\|z(\tau)\|^{2} + \frac{\rho}{k}\|d(\tau)\|^{2}\right) \mathrm{d}\tau,$$

and because  $V_{\max_i \rho_i} \ge 0$ 

$$\int_0^t \|z(\tau)\|^2 \mathrm{d}\tau \le \int_0^t \|d(\tau)\|^2 \mathrm{d}\tau + V_{\max_i \rho_i}(q(0)).$$

Weakening of the conditions of Theorem 4.7 leads to:

**Theorem 4.9** Assume that for the system (4.1) the following is satisfied:

- (*i*) m = p;
- (ii)  $\rho_1, \ldots, \rho_p$  are well defined;
- (iii) the decoupling matrix A(x) has full rank;
- (iv) the vector fields

$$\tilde{f} = f - g A^{-1} \begin{pmatrix} L_f^{\rho_1} h_1 \\ \vdots \\ L_f^{\rho_p} h_p \end{pmatrix}, \qquad \qquad \tilde{g} = g A^{-1}$$

are complete.

Also assume that after applying the feedback

 $u = A^{-1}(x)v$ 

to the system (4.1) the following conditions hold:

- (v)  $\mathcal{G}_{\rho-1} = \operatorname{span} \left\{ g_j, \operatorname{ad}_f g_j, \dots, \operatorname{ad}_f^{\rho_j-1} g_j; j = 1, \dots, m \right\}$  is involutive and has constant dimension  $\rho = \sum_{i=1}^p \rho_i$ ;
- (vi)  $\operatorname{ad}_{e_i} \mathcal{G}_j \subset \mathcal{G}_j, \ 1 \leq i \leq q, \ 0 \leq j \leq \max_s \rho_s,$ with  $\mathcal{G}_j = \operatorname{span} \left\{ g_l, \operatorname{ad}_f g_l, \ldots, \operatorname{ad}_f^{j+\rho_l-\max_s \rho_s} g_l, \ 1 \leq l \leq m \right\}.$

Then the  $L_2$ -gain almost disturbance decoupling problem is solvable.

**Proof** It follows from (i)-(v) that we can globally define a change of coordinates as in the proof of Theorem 4.7 which leads to (4.5). In *q*-coordinates we have that

$$\mathcal{G}_j = \operatorname{span}\left\{\frac{\partial}{\partial q_{i(\max_s \rho_s - j)}}, \ldots, \frac{\partial}{\partial q_{i\rho_i}}; i = 1, \ldots, p\right\}$$

for  $0 \le j \le \max_s \rho_s - 1$ . Then condition (vi) leads to the system form

$$\begin{split} \dot{q}_{11} &= q_{12} + W_{11}^T(\tilde{q}_1, q_r)d, \\ \vdots \\ \dot{q}_{1\rho_1} &= v_1 + W_{1\rho_1}^T(\tilde{q}_1, \dots, \tilde{q}_{\rho_1}, q_r)d, \\ \dot{q}_{21} &= q_{22} + W_{21}^T(\tilde{q}_1, q_r)d, \\ \vdots \\ \dot{q}_{2\rho_2} &= v_2 + W_{2\rho_2}^T(\tilde{q}_1, \dots, \tilde{q}_{\rho_2}, q_r)d, \\ \vdots \\ \dot{q}_{p1} &= q_{p2} + W_{p1}^T(\tilde{q}_1, q_r)d, \\ \vdots \\ \dot{q}_{p\rho_p} &= v_p + W_{p\rho_p}^T(\tilde{q}_1, \dots, \tilde{q}_{\rho_p}, q_r)d, \\ \dot{q}_r &= \varphi(\tilde{q}_1, q_r) + \Psi^T(\tilde{q}_1, q_r)d, \\ z_l &= q_{l1}, \quad 1 \le l \le p \end{split}$$

where

 $\tilde{q}_j = \{q_{kj}, \text{ for all } k = 1, \dots, p \text{ for which } j \le \rho_k\}.$ 

Now the proof continues in the same way as the proof of Theorem 4.8 using the there described algorithm.

It should be remarked that several slightly different sets of conditions can be formulated replacing conditions (vi) in Theorem 4.9, all leading to a structure on which an algorithm similar to the one used in the proof of Theorem 4.8 can be applied.

Finally it can be shown that if we additionally assume the zero-dynamics to be independent of d, namely

$$\dot{q}_r = \varphi(0, q_r),$$

then asymptotic stability of the origin  $q_r = 0$  for the zero-dynamics implies that the  $\mathcal{H}_{\infty}$  almost disturbance decoupling problem is solvable.

**Theorem 4.10** If, in addition to the conditions (i)-(vi) of Theorem 4.7 or of Theorem 4.9, system (4.1) is such that also:

(vii) the zero dynamics are independent of d and globally asymptotically stable;

then the  $\mathcal{H}_{\infty}$  almost disturbance decoupling problem is solvable.

**Proof** Combine the SISO result from [MRST 94] with Theorem 4.7 respectively Theorem 4.9. Details are left to the reader.

### Chapter 5

# The singular $\mathcal{H}_{\infty}$ problem: a geometric approach

For linear systems a second approach to tackle the singular  $\mathcal{H}_{\infty}$  problem is discussed in [StTr 90], [St 92] and [Sch 91]. In this geometric approach the system is decomposed using a state transformation based on the strongly controllable subspace. For the decomposed system it is possible to split the singular  $\mathcal{H}_{\infty}$  problem into a regular state feedback  $\mathcal{H}_{\infty}$  problem for the first subsystem, having as inputs part of the states of the second subsystem, and an almost disturbance decoupling problem for the second subsystem to 'track' the constructed feedback solution of the regular state feedback problem of the first subsystem.

In this chapter we will show that a similar approach can be applied to the nonlinear singular  $\mathcal{H}_{\infty}$  problem. We first recapitulate the state feedback  $\mathcal{H}_{\infty}$  solution for linear systems. Instead of constructing the strongly controllable subspace as is done in [St 92] we extend the linear system by adding the inputs and the disturbances as extra state variables and construct for this extended system the minimal conditioned invariant subspace containing the input vector fields. The projection of this subspace onto the state space of the original linear system is equal to the strongly controllable subspace and thus is equivalent to the approach of [StTr 90], [St 92]. The advantage of the present construction, however, is that it admits a direct generalization to the nonlinear case, as will be explained in the second section. This nonlinear decomposition will be also used in Subsection 5.2.4 to derive sufficient conditions for the solvability of the nonlinear singular  $\mathcal{H}_{\infty}$  problem similar to the linear case. Part of these conditions are proved to be necessary for the solvability of the singular  $\mathcal{H}_{\infty}$  problem.

The almost disturbance decoupling problem for nonlinear systems is not

completely understood. In particular there are no necessary and sufficient (geometric) conditions known for the solvability of the almost disturbance decoupling problem. Therefore we were not able to derive a full characterization for the singular  $\mathcal{H}_{\infty}$  problem. However for special classes of systems necessary and sufficient conditions for the solvability of the singular  $\mathcal{H}_{\infty}$  problem will be derived.

Finally we apply a factorization idea to derive sufficient conditions. In this method the singular  $\mathcal{H}_{\infty}$  problem is reduced to the singular  $\mathcal{H}_{\infty}$  problem for an auxiliary system, which is easier solvable.

For affine nonlinear systems some of these ideas are also considered in the paper [As 94]. The decomposition in this paper, however, is not a generalization of the decomposition used in the linear case. Therefore even for linear systems this method will not lead to necessary and sufficient conditions.

#### 5.1 Singular linear $\mathcal{H}_{\infty}$ control by state feedback

In this section the state feedback  $\mathcal{H}_{\infty}$  results from [St 92], [StTr 90] are recapitulated for linear systems of the form

$$\overline{\Sigma} \begin{cases} \dot{\bar{x}} = F\bar{x} + G\bar{u} + E\bar{d} \\ \bar{z} = H\bar{x} + K\bar{u} \end{cases}$$
(5.1)

For details and proofs the reader is referred to [St 92], where also the extension to linear systems containing direct feedthrough terms from disturbances to outputs may be found. Instead of immediately introducing the strongly controllable subspace as in [St 92] we construct this subspace in two steps. First we extend the system  $\Sigma$  by including both the inputs and the disturbances as extra state components. For this extended system we calculate the minimal conditioned invariant subspace containing the image of the input matrix. In the second step we project this subspace onto the state space of the original system  $\overline{\Sigma}$ . In the first subsection we introduce the notion of conditioned invariance for linear systems without direct feedthrough from the inputs to the outputs are (the case K = 0). We decompose systems of this form using the minimal conditioned invariant subspace containing the image of the input matrix. In the second subsection we see that the extended system is of this special form. Furthermore the projection of the minimal conditioned invariant subspace including the input matrix for the extended system onto the state space of the original system  $\Sigma$  is equal to the strongly controllable subspace used in [St 92].

#### 5.1.1 State transformation for linear systems with K = 0

In this subsection our attention is restricted to systems of the form

$$\dot{\bar{x}} = F\bar{x} + G\bar{u} + E\bar{d}$$
  
$$\bar{z} = H\bar{x}$$
(5.2)

#### **Conditioned invariance**

Basic notions from the geometric approach to linear system theory are the notions of controlled invariance and conditioned invariance ([Ha 83], [HaSi 83], [Wo 79], [BaMa 92]). In this subsection our attention is focussed on the second notion.

**Definition 5.1** A subspace  $\mathcal{V}$  of  $\mathbb{R}^n$  is said to be an *invariant subspace* for the matrix F if

$$F\mathcal{V}\subset\mathcal{V}.$$

The subspace  $\mathcal{V}$  is said to be a *conditioned invariant subspace* with respect to the pair (F, H) if there exists a linear mapping B such that

$$(F+BH) \mathcal{V} \subset \mathcal{V}.$$

So a subspace  $\mathcal{V}$  is called conditioned invariant if there exists an output injection matrix B which renders the subspace  $\mathcal{V}$  invariant for the matrix F + BH. The property of conditioned invariance can also be characterized in the following way.

**Lemma 5.2** A subspace  $\mathcal{V}$  of  $\mathbb{R}^n$  is a conditioned invariant subspace for the pair (F, H) if and only if

$$F(\mathcal{V}\cap \ker H)\subset \mathcal{V}.$$

Now we introduce a subspace which will play a key role in this section.

**Definition 5.3** Consider the system (5.2). Let  $\mathcal{W}$  be a subspace of  $\mathbb{R}^n$ . Then there exists a *minimal conditioned invariant subspace* containing the subspace  $\mathcal{W}$ , denoted by  $\mathcal{S}^*(\mathcal{W})$ , defined as the smallest subspace  $\mathcal{V}$  of  $\mathbb{R}^n$  such that:

- (i)  $\mathcal{V}$  is conditioned invariant;
- (ii)  $\mathcal{W} \subset \mathcal{V}$ .

An explicit recursive algorithm to calculate  $S^*(W)$  can be given.

**Lemma 5.4** Consider the system (5.2). Let W be a subspace of  $\mathbb{R}^n$ . Then the minimal conditioned invariant subspace containing W,  $S^*(W)$ , is the limit of the increasing sequence of subspaces  $S_i(W)$  generated by the recursive algorithm

$$S_{1}(\mathcal{W}) := \mathcal{W}$$
  

$$S_{i+1}(\mathcal{W}) := \mathcal{W} + F(S_{i}(\mathcal{W}) \cap \ker H), \qquad i = 1, 2, \dots$$

#### **State transformation**

We transform the state of the system (5.2) by using the minimal conditioned invariant subspace containing im*G*,  $S^*(\text{im}G)$ . In order to do so we rewrite  $\mathbb{R}^n = \overline{X}_1 \oplus \overline{X}_2 \oplus \overline{X}_3$  with  $\overline{X}_2 = S^*(\text{im}G) \cap \ker H$ ,  $\overline{X}_2 \oplus \overline{X}_3 = S^*(\text{im}G)$  and  $\overline{X}_1$ arbitrary. Let  $(\xi_1, \ldots, \xi_{\nu}, \xi_{\nu+1}, \ldots, \xi_{\mu}, \xi_{\mu+1}, \ldots, \xi_n)$  be a basis of  $\mathbb{R}^n$  such that  $\xi_{\nu+1}, \ldots, \xi_{\mu}$  is a basis of  $S^*(\text{im}G) \cap \ker H$  and  $\xi_{\nu+1}, \ldots, \xi_n$  a basis of  $S^*(\text{im}G)$ .

Then the linear mappings F, G and H with respect to this basis have the following matrix form

$$F = \begin{pmatrix} F_{11} & 0 & F_{13} \\ F_{21} & F_{22} & F_{23} \\ F_{31} & F_{32} & F_{33} \end{pmatrix}, \quad G = \begin{pmatrix} 0 \\ G_2 \\ G_3 \end{pmatrix}, \quad H = \begin{pmatrix} H_1 & 0 & H_3 \end{pmatrix}.$$

If we accordingly decompose the state x and the matrix E as

$$\bar{x} = \begin{pmatrix} \bar{x}_1 \\ \bar{x}_2 \\ \bar{x}_3 \end{pmatrix}, \qquad E = \begin{pmatrix} E_1 \\ E_2 \\ E_3 \end{pmatrix}$$

ī

then in the new basis the system (5.2) takes the form

$$\dot{\bar{x}}_{1} = F_{11}\bar{x}_{1} + F_{13}\bar{x}_{3} + E_{1}\bar{d}$$

$$\begin{pmatrix} \dot{\bar{x}}_{2} \\ \dot{\bar{x}}_{3} \end{pmatrix} = \begin{pmatrix} F_{22} & F_{23} \\ F_{32} & F_{33} \end{pmatrix} \begin{pmatrix} \bar{x}_{2} \\ \bar{x}_{3} \end{pmatrix} + \begin{pmatrix} G_{2} \\ G_{3} \end{pmatrix} \bar{u}$$

$$(5.3)$$

$$+ \left(\begin{array}{c} F_{21} \\ F_{31} \end{array}\right) \bar{x}_1 + \left(\begin{array}{c} E_2 \\ E_3 \end{array}\right) \bar{d}$$
(5.4)

$$= H_1 \bar{x}_1 + H_3 \bar{x}_3 \tag{5.5}$$



Figure 5.1: system (5.2) after state transformation

This arrangement already suggests to write this system as an interconnection of two subsystems. The first subsystem  $\overline{\Sigma}_1$  is given by (5.3) and (5.5), and has as state space  $\overline{X}_1$ , input space  $\overline{X}_3$  and output space  $\mathbb{R}^p$ . The second subsystem  $\overline{\Sigma}_2$  is given by (5.4) having state space  $\overline{X}_2 \oplus \overline{X}_3$ , input space  $\mathbb{R}^m$  and output space  $\overline{X}_3$  (see Figure 5.1).

The system  $\overline{\Sigma}_1$  has the nice property that  $H_3$  has full column rank. Indeed, let  $\bar{x}_3$  be such that  $H_3\bar{x}_3 = 0$ . Then the vector  $\bar{x} = ( \begin{array}{cc} 0^T & 0^T & \bar{x}_3^T \end{array} )$  is an element of  $\bar{X}_2 = S^* \cap \ker H$ , and hence  $\bar{x}_3 = 0$ .

#### 5.1.2 State transformation for linear system with $K \neq 0$

Now we consider the system  $\overline{\Sigma}$ , with  $K \neq 0$ . We construct an extended system by adding the inputs  $\overline{u}$  and disturbances  $\overline{d}$  as extra state components:

$$\overline{\Sigma}_{e} \begin{cases} \dot{\bar{x}} = F\bar{x} + G\bar{u} + E\bar{d} \\ \dot{\bar{u}} = \bar{v} \\ \dot{\bar{d}} = \bar{w} \\ \bar{\bar{z}} = H\bar{x} + K\bar{u} \end{cases}$$
(5.6)

having state  $\bar{x}_e^T = (\ \bar{x}^T \ \bar{u}^T \ \bar{d}^T)$ . We can rewrite  $\overline{\Sigma}_e$  more compactly as:

$$\overline{\Sigma}_{e} \begin{cases} \dot{\bar{x}}_{e} = F_{e}\bar{x}_{e} + G_{e}\bar{v} + E_{e}\bar{w} \\ \bar{z} = H_{e}\bar{x}_{e} \end{cases}$$
(5.7)

where

$$F_{e} = \begin{pmatrix} F & G & E \\ 0 & 0 & 0 \\ 0 & 0 & 0 \end{pmatrix}, \quad G_{e} = \begin{pmatrix} 0 \\ I_{m} \\ 0 \end{pmatrix},$$
$$E_{e} = \begin{pmatrix} 0 \\ 0 \\ I_{q} \end{pmatrix}, \quad H_{e} = \begin{pmatrix} H & K & 0 \end{pmatrix}.$$

It is clear that  $\overline{\Sigma}_e$  is a system of the form considered in Subsection 5.1.1. Thus we can compute the minimal conditioned invariant subspace as in Subsection 5.1.1.

Now we calculate for the system  $\overline{\Sigma}_e$  the minimal conditioned invariant distribution containing  $\operatorname{im} G_e$ ,  $S_e^*(\operatorname{im} G_e)$ , using the algorithm from Lemma 5.4. Again this distribution is used to rewrite the state space  $\mathbb{R}^{n+m+q} = \overline{X}_{1e} \oplus \overline{X}_{2e} \oplus \overline{X}_{3e}$  with  $\overline{X}_{2e} = S_e^*(\operatorname{im} G_e) \cap \ker H_e$ ,  $\overline{X}_{2e} \oplus \overline{X}_{3e} = S_e^*(\operatorname{im} G_e)$  and  $\overline{X}_{1e}$  arbitrary. For the extended system  $\overline{\Sigma}_e$  the subspace  $\operatorname{im} G_e$  is given by

$$\operatorname{im} G_e = \operatorname{span} \{ \upsilon_1, \ldots, \upsilon_m \}$$

where  $v_1, \ldots, v_m$  is any basis of the input space  $\mathbb{R}^m$ . For later convenience we will choose  $v_1, \ldots, v_m$  such that  $v_1, \ldots, v_{m_1}$  is a basis of ker K. It follows from the algorithm and the special structure of the matrix  $F_e$  that the subspace  $\bar{X}_{1e}$  can be chosen such that

span 
$$\{\bar{d}_1,\ldots,\bar{d}_q\}\subset \bar{X}_{1e}$$
.

Finally let  $(\xi_1, \ldots, \xi_{\nu}, \xi_{\nu+1}, \ldots, \xi_{\mu}, \xi_{\mu+1}, \ldots, \xi_n)$  be a basis of  $\mathbb{R}^n$  such that:

$$\operatorname{span} \left\{ \xi_{\nu+1}, \ldots, \xi_{\mu} \right\} + \operatorname{span} \left\{ \upsilon_1, \ldots, \upsilon_{m_1} \right\} = \mathcal{S}_e^*(\operatorname{im} G_e) \cap \ker H_e;$$
  
$$\operatorname{span} \left\{ \xi_{\nu+1}, \ldots, \xi_n \right\} + \operatorname{span} \left\{ \upsilon_1, \ldots, \upsilon_m \right\} = \mathcal{S}_e^*(\operatorname{im} G_e).$$

Then the subspaces  $\bar{X}_{1e}$ ,  $\bar{X}_{2e}$  and  $\bar{X}_{3e}$  are given by:

$$\bar{X}_{1e} = \operatorname{span} \{ \xi_1, \dots, \xi_{\nu}, \bar{d}_1, \dots, \bar{d}_q \}; \bar{X}_{2e} = \operatorname{span} \{ \xi_{\nu+1}, \dots, \xi_{\mu}, \upsilon_1, \dots, \upsilon_{m_1} \}; \bar{X}_{3e} = \operatorname{span} \{ \xi_{\mu+1}, \dots, \xi_n, \upsilon_{m_1+1}, \dots, \upsilon_m \}.$$

In this new basis for the state space  $\mathbb{R}^n$  and the input space  $\mathbb{R}^m$  the mappings  $F_e$  and  $H_e$  have the following special form

$$F_{e} = \begin{pmatrix} F_{11} & 0 & F_{13} & 0 & G_{12} & E_{1} \\ F_{21} & F_{22} & F_{23} & G_{21} & G_{22} & E_{2} \\ F_{31} & F_{32} & F_{33} & G_{31} & G_{32} & E_{3} \\ \hline 0 & 0 & 0 & 0 \\ \hline 0 & 0 & 0 & 0 \\ \hline H_{e} = \begin{pmatrix} H_{1} & 0 & H_{3} & 0 & K_{2} & 0 \end{pmatrix} \end{pmatrix},$$

and the original system  $\overline{\Sigma}$  takes the form

$$\dot{\bar{x}}_{1} = F_{11}\bar{x}_{1} + (F_{13} \quad G_{12}) \begin{pmatrix} \bar{x}_{3} \\ \bar{u}_{2} \end{pmatrix} + E_{1}\bar{d}$$

$$\begin{pmatrix} \dot{\bar{x}}_{2} \\ \dot{\bar{x}}_{3} \end{pmatrix} = \begin{pmatrix} F_{22} \quad F_{23} \\ F_{32} \quad F_{33} \end{pmatrix} \begin{pmatrix} \bar{x}_{2} \\ \bar{x}_{3} \end{pmatrix} + \begin{pmatrix} G_{21} \\ G_{31} \end{pmatrix} \bar{u}_{1} + \begin{pmatrix} G_{22} \\ G_{32} \end{pmatrix} \bar{u}_{2}$$

$$+ \begin{pmatrix} F_{21} \\ F_{31} \end{pmatrix} \bar{x}_{1} + \begin{pmatrix} E_{2} \\ E_{3} \end{pmatrix} \bar{d}$$

$$(5.9)$$

$$\bar{z} = H_{1}\bar{x}_{1} + (H_{2} \quad K_{2}) \begin{pmatrix} \bar{x}_{3} \end{pmatrix}$$

$$\bar{z} = H_1 \bar{x}_1 + \left( \begin{array}{cc} H_3 & K_2 \end{array} \right) \left( \begin{array}{c} x_3 \\ \bar{u}_2 \end{array} \right)$$
(5.10)



Figure 5.2: system  $\overline{\Sigma}$  after input and state transformation

Again this system can be seen as an interconnection of two systems. The first subsystem  $\overline{\Sigma}_1$  is given by (5.8) and (5.10) with state  $\bar{x}_1$ , inputs  $(\bar{x}_3, \bar{u}_2, \bar{d})$  and outputs  $\bar{z}$ , and the second subsystem  $\overline{\Sigma}_2$  is given by (5.9) with state  $(\bar{x}_2, \bar{x}_3)$ , inputs  $(\bar{u}_1, \bar{x}_1, \bar{u}_2, \bar{d})$  and outputs  $\bar{x}_3$  (see Figure 5.2). As in 5.1.1 it follows that the matrix ( $H_3$   $K_2$ ) is injective.

#### 5.1.3 The quadratic matrix inequality

An important role in the linear theory for the singular  $\mathcal{H}_{\infty}$  problem is played by the *quadratic matrix inequality* 

$$F_{\gamma}(P) := \begin{pmatrix} F^T P + PF + \frac{1}{\gamma^2} PEE^T P + H^T H & H^T K + PG \\ K^T H + G^T P & K^T K \end{pmatrix} \ge 0.$$

Symmetric solutions P to this quadratic matrix inequality can be proved (see [St 92]) to be of the form

$$P = \left(\begin{array}{rrrr} P_1 & 0 & 0\\ 0 & 0 & 0\\ 0 & 0 & 0 \end{array}\right)$$

in the new basis. We define G(s) as the transfer matrix

$$\mathcal{G}(s) = H \left( sI - F \right)^{-1} G + K$$

and the  $(n \times (n + m))$  controllability pencil  $L_{\gamma}(P, s)$  by

$$L_{\gamma}(P,s) := \left( sI - F - \frac{1}{\gamma^2} EE^T P - G \right).$$

Then we recapitulate the following result ([St 92]).

**Theorem 5.5** The following statements are equivalent:

(i) there exists a symmetric solution  $P \ge 0$  to  $F_{\gamma}(P) \ge 0$  such that

rank 
$$F_{\gamma}(P) = \operatorname{rank}_{\mathbb{R}(s)} \mathcal{G}$$

and

$$\operatorname{rank}\begin{pmatrix} L_{\gamma}(P,s)\\ F_{\gamma}(P) \end{pmatrix} = n + \operatorname{rank}_{\mathbb{R}(s)} \mathcal{G} \qquad \forall s \in \mathbb{C}^0 \cup \mathbb{C}^+$$

where rank<sub> $\mathbb{R}(s)$ </sub>  $\mathcal{G}$  is the normal rank for the polynomial matrix  $\mathcal{G}(s)$ , with entries from  $\mathbb{R}(s)$ , defined by

$$\operatorname{rank}_{\mathbb{R}(s)} \mathcal{G} = \max \left\{ \operatorname{rank} \mathcal{G}(s) | s \in \mathbb{C} \right\};$$

(ii) there exists a symmetric solution  $P_1 \ge 0$  to

$$P_{1}F_{11} + F_{11}^{T}P_{1} + \frac{1}{\gamma^{2}}P_{1}E_{1}E_{1}^{T}P_{1} - P_{1}G_{12}\left(K_{2}^{T}K_{2}\right)^{-1}G_{12}^{T}P_{1} + H_{1}^{T}H_{1}$$
$$- \left(\begin{array}{c}H_{3}^{T}H_{1} + F_{13}^{T}P_{1}\\K_{2}^{T}H_{1} + G_{12}^{T}P_{1}\end{array}\right)^{T} \left(\begin{array}{c}H_{3}^{T}H_{3} & H_{3}^{T}K_{2}\\K_{2}^{T}H_{3} & K_{2}^{T}K_{2}\end{array}\right)^{-1} \left(\begin{array}{c}H_{3}^{T}H_{1} + F_{13}^{T}P_{1}\\K_{2}^{T}H_{1} + G_{12}^{T}P_{1}\end{array}\right)$$
$$= 0$$

satisfying

$$\sigma \left( F_{11} + \frac{1}{\gamma^2} E_1 E_1^T P_1 - \left( \begin{array}{c} F_{13}^T \\ G_{12}^T \end{array} \right)^T \left( \begin{array}{c} H_3^T H_3 & H_3^T K_2 \\ K_2^T H_3 & K_2^T K_2 \end{array} \right)^{-1} \left( \begin{array}{c} H_3^T H_1 + F_{13}^T P_1 \\ K_2^T H_1 + G_{12}^T P_1 \end{array} \right) \right) \subset \mathbb{C}^-.$$

It should be noted that statement (ii) in Theorem 5.5 is equivalent to the existence of a feedback law

$$\left(\begin{array}{c} \bar{x}_3\\ \bar{u}_2 \end{array}\right) = L\bar{x}_1$$

which solves the regular  $\mathcal{H}_{\infty}$  state feedback problem for the subsystem  $\overline{\Sigma}_1$  with inputs  $(\bar{x}_3, \bar{u}_2)$  (see Theorem 2.21).

#### 5.1.4 The state feedback $\mathcal{H}_{\infty}$ control problem

Solvability of the strict suboptimal  $\mathcal{H}_{\infty}$  problem for the linear system  $\overline{\Sigma}$  can be characterized as follows (compare with Theorem 2.21).

**Theorem 5.6** Consider  $\overline{\Sigma}$ . Let  $\gamma > 0$ . Assume the system (F, G, H, K) has no invariant zeros on the imaginary axis (see Theorem 2.21). Then the following statements are equivalent:

(i) there exists a feedback  $\bar{u} = L\bar{x}$  such that after applying this feedback to the system  $\overline{\Sigma}$  the closed loop system has  $L_2$ -gain less than  $\gamma$  and is asymptotically stable;

(ii) there exists a non-negative, symmetric solution P to the quadratic matrix inequality  $F_{\gamma}(P) \ge 0$  such that

rank 
$$F_{\gamma}(P) = \operatorname{rank}_{\mathbb{R}(s)} \mathcal{G}_{ci}$$

and

$$\operatorname{rank}\begin{pmatrix} L_{\gamma}(P,s)\\ F_{\gamma}(P) \end{pmatrix} = n + \operatorname{rank}_{\mathbb{R}(s)}\mathcal{G}_{ci}, \qquad \forall s \in \mathbb{C}^0 \cup \mathbb{C}^+.$$

Together with Theorem 5.5 this means that the solvability of the singular  $\mathcal{H}_{\infty}$  control problem is fully characterized by the solvability of the regular  $\mathcal{H}_{\infty}$  control problem for the subsystem  $\overline{\Sigma}_1$ . Before we give a sketch of the proof of Theorem 5.6 we first recapitulate the following result about the solvability of the almost disturbance decoupling problem with internal stability for linear systems of the form  $\overline{\Sigma}$  (see [Tr 86], [Wi 81], [Wi 82], [St 92]).

**Theorem 5.7** Consider the system  $\overline{\Sigma}$ . Let T and  $V_g$  denote the strongly controllable subspace and the stabilizable weakly unobservable subspace respectively. Then the following statements are equivalent:

(i)

$$\mathcal{V}_g + \mathcal{T} = \mathbb{R}^n;$$

(ii)

$$\operatorname{rank}\begin{pmatrix} sI-F & -G\\ H & K \end{pmatrix} = n + \operatorname{rank}(H & K), \qquad \forall s \in \mathbb{C}^0 \cup \mathbb{C}^+;$$

(iii) for all  $\varepsilon > 0$  there exists a feedback  $\overline{u} = L\overline{x}$  for  $\overline{\Sigma}$  such that the closedloop system has  $L_2$ -gain less than or equal to  $\varepsilon$  and is asymptotically stable (the almost disturbance decoupling problem with internal stability is solvable for  $\overline{\Sigma}$ ).

Sketch of the proof of Theorem 5.6 For details we refer to [St 92].

- (*ii*)  $\Rightarrow$  (*i*) Statement (*ii*) implies that the regular  $\mathcal{H}_{\infty}$  problem for the subsystem  $\overline{\Sigma}_1$  is solvable (Theorem 5.5 and Theorem 2.21). Then the state  $\overline{x}_3$  can be transformed by subtracting the optimal solution of the regular subproblem. So when this new variable is equal to zero we have perfect tracking of the solution for  $\Sigma_1$ . Now we transform the state of the system  $\overline{\Sigma}_2$  such that part of the state is equal to the tracking error between  $\overline{x}_3$  and the  $\overline{x}_3$ -feedback solution of the regular  $\mathcal{H}_{\infty}$  problem for the subsystem  $\overline{\Sigma}_1$ . This transformed system will be denoted by  $\overline{\Sigma}_2^T$ . Both  $\overline{\Sigma}_2$  and  $\overline{\Sigma}_2^T$  are strongly controllable which means that the state space is equal to the strongly controllable subspace. By Theorem 5.7 this implies that the almost disturbance decoupling problem with stability is solvable for  $\overline{\Sigma}_2^T$ . Combining these two results leads to the solvability of the  $\mathcal{H}_{\infty}$  problem of the system  $\overline{\Sigma}$ .
- (*i*)  $\Rightarrow$  (*ii*) From statement (*i*) it follows that (*F*, *G*) is stabilizable and for the system  $\overline{\Sigma}$  there exists a  $\delta < \gamma$  such that for all  $\overline{d} \in L_2(0, \infty)$  there exists an  $\overline{u} \in L_2(0, \infty)$  such that  $\overline{x} \in L_2(0, \infty)$  and

$$\int_0^\infty \|\bar{z}(\tau)\|^2 \mathrm{d}\tau \le \delta^2 \int_0^\infty \|\bar{d}(\tau)\|^2 \mathrm{d}\tau \tag{5.11}$$

then it follows that  $\bar{x}_1 \in L_2(0, \infty)$  and  $\bar{x}_3 \in L_2(0, \infty)$ . Together with (5.11) this implies that the regular  $\mathcal{H}_{\infty}$  problem for the subsystem  $\overline{\Sigma}_1$  is solvable (details see [St 92]), which via Theorem 2.21 and 5.5 leads to the statement *(ii)*.

#### 5.1.5 Factorization approach

The sufficiency part of the result in Theorem 5.6 can also be proved in a different way. Factorize the matrix  $F_{\gamma}(P)$  corresponding to the quadratic matrix inequality

$$F_{\gamma}(P) = \begin{pmatrix} H_P^T \\ K_P^T \end{pmatrix} \begin{pmatrix} H_P & K_P \end{pmatrix}$$

for certain matrices  $H_P$  and  $K_P$ . Define the new system

$$\overline{\Sigma}_{P} \begin{cases} \dot{\bar{x}} = \left(F + \frac{1}{\gamma^{2}} E E^{T} P\right) \bar{x} + G \bar{u} + E \bar{d}_{p} \\ \bar{z}_{p} = H_{P} \bar{x} + K_{P} \bar{u} \end{cases}$$
(5.12)

where  $\bar{d}_p = \bar{d} - \frac{1}{v^2} E^T P \bar{x}$ . Then the following result can be proved ([St 92]).

**Lemma 5.8** Consider  $\overline{\Sigma}$  and  $\overline{\Sigma}_P$ . Let P satisfy the condition (ii) of Theorem 5.6 and let L be a linear map  $L : \mathbb{R}^m \to \mathbb{R}^n$ . Then the following statements are equivalent:

- (i) the closed loop system consisting of  $\overline{\Sigma}$  with the feedback  $\overline{u} = L\overline{x}$  has  $L_2$ -gain from  $\overline{d}$  to  $\overline{z}$  less than  $\gamma$  and is asymptotically stable;
- (ii) the closed loop system consisting of  $\overline{\Sigma}_P$  with the feedback  $\overline{u} = L\overline{x}$  has  $L_2$ -gain from  $\overline{d}_p$  to  $\overline{z}_p$  less than  $\gamma$  and is asymptotically stable.

Sketch of the proof The  $L_2$ -gain part of the result can be proved using the following completion of the squares argument:

$$\begin{split} \|\bar{z}_{p}\|^{2} &- \gamma^{2} \|\bar{d}_{p}\|^{2} \\ &= \bar{x}^{T} H_{P}^{T} H_{P} \bar{x} + 2 \bar{x}^{T} H_{P}^{T} K_{P} \bar{u} + \bar{u}^{T} K_{P}^{T} K_{P} \bar{u} - \gamma^{2} \|\bar{d}\|^{2} + 2 \bar{x}^{T} P E \bar{d} \\ &- \frac{1}{\gamma^{2}} \bar{x}^{T} P E E^{T} P \bar{x} \\ &= \left( \bar{x}^{T} \quad \bar{u}^{T} \right) F_{\gamma}(P) \left( \frac{\bar{x}}{\bar{u}} \right) - \gamma^{2} \|\bar{d}\|^{2} + 2 \bar{x}^{T} P E \bar{d} - \frac{1}{\gamma^{2}} \bar{x}^{T} P E E^{T} P \bar{x} \\ &= 2 \bar{x}^{T} P E \bar{d} + \bar{x}^{T} H^{T} H \bar{x} + 2 \bar{x}^{T} H^{T} K \bar{u} + 2 \bar{x}^{T} P G \bar{u} + \bar{u}^{T} K^{T} K \bar{u} \\ &= \|\bar{z}\|^{2} - \gamma^{2} \|\bar{d}\|^{2} + 2 \bar{x}^{T} P \left( F \bar{x} + G \bar{u} + E \bar{d} \right) \\ &= \|\bar{z}\|^{2} - \gamma^{2} \|\bar{d}\|^{2} + 2 \bar{x}^{T} P \bar{x}. \end{split}$$

Now we take the integral from 0 to  $\infty$  of both sides of this equality:

$$\int_0^\infty \|\bar{z}_p(\tau)\|^2 - \gamma^2 \|\bar{d}_p(\tau)\|^2 d\tau$$
  
= 
$$\int_0^\infty \|\bar{z}(\tau)\|^2 - \gamma^2 \|\bar{d}(\tau)\|^2 d\tau + \lim_{t \to \infty} 2\bar{x}^T(t) P\bar{x}(t) - 2\bar{x}^T(0) P\bar{x}(0).$$

For  $\bar{x}(0) = 0$  and assuming that the system is asymptotically stable, implying that  $\lim_{t\to\infty} \bar{x}(t) = 0$ , then it follows that the  $L_2$ -gains from disturbance to output of the systems  $\overline{\Sigma}$  and  $\overline{\Sigma}_P$  are equal. Therefore application of a feedback  $\bar{u} = L\bar{x}$  to both systems leads to the same  $L_2$ -gain.

Condition (ii) from Theorem 5.6 can be rewritten as

$$\operatorname{rank}\left(\begin{array}{cc} sI - F - \frac{1}{\gamma^2} EE^T P & -G\\ H_P & K_P \end{array}\right) = n + \operatorname{rank}\left(\begin{array}{cc} H_P & K_P \end{array}\right), \quad \forall s \in \mathbb{C}^0 \cup \mathbb{C}^+$$

which implies solvability of the almost disturbance decoupling problem for the system  $\overline{\Sigma}_P$  (see Theorem 5.7).

# 5.2 Singular $\mathcal{H}_{\infty}$ problem for general nonlinear systems

In this section we try to extend the linear result explained in the previous section to general nonlinear systems of the form

$$\Sigma \begin{cases} \dot{x} = f(x, u, d) \\ y = x \\ z = h(x, u) \end{cases}$$
(5.13)

In the first two subsections we define a state transformation for these systems based on a minimal conditioned invariant distribution. Similar to the linear case (Section 5.1) we first define the notion of conditioned invariance for affine non-linear systems without a direct feedthrough from the inputs to the outputs. As in the linear case this step is instrumental in the second subsection where we construct an extended system from the system  $\Sigma$  by adding the inputs and the disturbances as extra state components. This extended system is affine and does not have a direct feedthrough from the new defined inputs to the outputs, which makes it possible to apply the results from the first subsection.

The rest of the chapter is concerned with finding sufficient and necessary conditions for the solvability of the singular  $\mathcal{H}_{\infty}$  problem. These conditions are in terms of the solvability of a regular  $\mathcal{H}_{\infty}$  problem for a subsystem of  $\Sigma$  and an  $\mathcal{H}_{\infty}$  almost disturbance decoupling problem. Sometimes it will be necessary to make extra assumptions on the structure of the system  $\Sigma$  in order to derive such conditions. For instance in Subsection 5.2.6 where necessary and sufficient conditions are derived and in Subsection 5.2.7 to apply a factorization approach.

## 5.2.1 State transformation for affine nonlinear systems without direct feedthrough from inputs to outputs

First we restrict our attention to nonlinear systems of the form

$$\dot{x} = f(x) + \sum_{j=1}^{m} g_j(x) u_j + \sum_{i=1}^{q} e_i(x) d_i z_l = h_l(x),$$
 (5.14)

#### **Conditioned invariance**

Also for nonlinear systems the notions of invariance, controlled invariance and conditioned invariance can be defined (see [IKGM 81], [MRS 94], [vdS 85], [vdS 87]). In this book the following definitions are used

**Definition 5.9** A distribution  $\mathcal{D}$  is said to be an *invariant distribution* under the dynamics (5.14) if

$$\begin{bmatrix} f, \mathcal{D} \end{bmatrix} \subset \mathcal{D}, \\ \begin{bmatrix} g_j, \mathcal{D} \end{bmatrix} \subset \mathcal{D}, \qquad j = 1, \dots, m.$$

Define

$$\ker \mathrm{d}h = \bigcap_{l=1}^p \ker \mathrm{d}h_l.$$

**Definition 5.10** A distribution  $\mathcal{D}$  is said to be a *conditioned invariant distribution* under the dynamics (5.14) if

$$\begin{bmatrix} f, \mathcal{D} \cap \ker dh \end{bmatrix} \subset \mathcal{D},$$
$$\begin{bmatrix} g_j, \mathcal{D} \cap \ker dh \end{bmatrix} \subset \mathcal{D}, \qquad j = 1, \dots, m.$$

For linear systems the second conditions are automatically satisfied because in that case the vectors  $g_j$  are constant and the distribution is generated by constant vectors from a linear subspace while the Lie bracket of constant vector fields is zero. It should be noted that these definitions of invariance and conditioned invariance are independent from the disturbance vector fields. Therefore for the calculation of these distributions we can consider the system (5.14) without the

disturbances. The definition of the notion of conditioned invariant distribution is an extension of the characterization of a conditioned invariant subspace in Lemma 5.2.

Similar to the linear case we can define for any distribution  $\mathcal{W}$  the *minimal involutive conditioned invariant distribution* containing  $\mathcal{W}$ . An explicit algorithm to calculate this distribution is given by

**Lemma 5.11** Consider the system (5.14). Then the minimal involutive conditioned invariant distribution containing W, denoted  $S^*(W)$ , is given by

$$\mathcal{S}^*(\mathcal{W}) = \bigcup_{k \ge 1} \overline{\mathcal{S}}_k(\mathcal{W})$$

where  $\overline{D}$  indicates the involutive closure of the distribution D, and the increasing sequence  $S_k(W)$  is generated by the following recursive algorithm

$$S_{1}(\mathcal{W}) = \mathcal{W}$$

$$S_{k+1}(\mathcal{W}) = \overline{S}_{k}(\mathcal{W}) + [f, \overline{S}_{k}(\mathcal{W}) \cap \ker dh]$$

$$+ \sum_{j=1}^{m} [g_{j}, \overline{S}_{k}(\mathcal{W}) \cap \ker dh] \qquad k = 1, 2, \dots$$

**Remark 5.12** For constructing just the minimal conditioned invariant distribution containing  $\mathcal{W}$  instead of the minimal *involutive* conditioned invariant distributions we can use the recursive algorithm without involutive closures.

#### State transformation

Inspired by the linear theory we will use the minimal involutive conditioned invariant distribution which contains the input vector fields  $g_j$ , which we will refer to as  $S^*$ , to construct a state transformation for the system (2.27). This distribution is according to Lemma 5.11 generated by the algorithm:

S\* algorithm

$$S_1 = \operatorname{span}\{g_1,\ldots,g_m\}$$

$$S_{k+1} = \overline{S}_k + [f, \overline{S}_k \cap \ker dh] + \sum_{j=1}^m [g_j, \overline{S}_k \cap \ker dh] \qquad k = 1, 2, \dots$$
$$S^* = \bigcup_{k \ge 1} \overline{S}_k$$

By construction  $S^*$  is involutive and conditioned invariant, i.e.,

$$\begin{bmatrix} f, \mathcal{S}^* \cap \ker dh \end{bmatrix} \subset \mathcal{S}^*, \tag{5.15}$$
$$\begin{bmatrix} g_j, \mathcal{S}^* \cap \ker dh \end{bmatrix} \subset \mathcal{S}^*, \qquad j = 1, \dots, m. \tag{5.16}$$

and is the minimal conditioned invariant involutive distribution which contains the input vector fields  $g_i$ .

If the distributions  $S^*$  and  $S^* \cap \ker dh$  are constant dimensional then we can make the following decomposition of the state space of (5.14). Define  $X_2 :=$  $S^* \cap \ker dh$ ,  $X_2 \oplus X_3 = S^*$  and  $X_1$  arbitrary such that  $X_1 \oplus X_2 \oplus X_3 = M$ .

Using Frobenius' Theorem ([NvdS 90], [Is 89]) it follows that similar to the linear case there exist local coordinates  $(\xi_1, \ldots, \xi_\nu, \xi_{\nu+1}, \ldots, \xi_\mu, \xi_{\mu+1}, \ldots, \xi_n)$  such that:

$$\begin{aligned} \mathcal{X}_1 &= \operatorname{span}\left\{\frac{\partial}{\partial \xi_1}, \dots, \frac{\partial}{\partial \xi_{\nu}}\right\}; \\ \mathcal{X}_2 &= \operatorname{span}\left\{\frac{\partial}{\partial \xi_{\nu+1}}, \dots, \frac{\partial}{\partial \xi_{\mu}}\right\}; \\ \mathcal{X}_3 &= \operatorname{span}\left\{\frac{\partial}{\partial \xi_{\mu+1}}, \dots, \frac{\partial}{\partial \xi_n}\right\}. \end{aligned}$$

In these new coordinates the system (5.14) takes the following form

$$\begin{aligned} \dot{x}_1 &= f_1(x_1, x_3) &+ \sum_{i=1}^{q} e_{i1}(x_1, x_2, x_3) d_i \\ \dot{x}_2 &= f_2(x_1, x_2, x_3) &+ \sum_{j=1}^{m} g_{j2}(x_1, x_2, x_3) u_j &+ \sum_{i=1}^{q} e_{i2}(x_1, x_2, x_3) d_i \\ \dot{x}_3 &= f_3(x_1, x_2, x_3) &+ \sum_{j=1}^{m} g_{j3}(x_1, x_2, x_3) u_j &+ \sum_{i=1}^{q} e_{i3}(x_1, x_2, x_3) d_i \\ z_l &= h(x_1, x_3), & l = 1, \dots, p \end{aligned}$$

or in shorthand notation

$$\dot{x}_1 = f_1(x_1, x_3) + e_1(x_1, x_2, x_3)d$$
 (5.17)

$$\dot{x}_2 = f_2(x_1, x_2, x_3) + g_2(x_1, x_2, x_3)u + e_2(x_1, x_2, x_3)d$$
 (5.18)

$$\dot{x}_3 = f_3(x_1, x_2, x_3) + g_3(x_1, x_2, x_3)u + e_3(x_1, x_2, x_3)d$$
 (5.19)

$$z = h(x_1, x_3) (5.20)$$

Remark 5.13 The above form can be easily verified by using the implications:

$$\begin{bmatrix} f, X_2 \end{bmatrix} \subset X_2 \oplus X_3 \quad \Rightarrow \quad \frac{\partial f_1}{\partial x_2} \equiv 0; \\ g_j \in X_2 \oplus X_3 \quad \Rightarrow \quad g_{j1} \equiv 0, \qquad \qquad j = 1, \dots, m; \\ X_2 \subset \ker dh \quad \Rightarrow \quad \frac{\partial h_l}{\partial x_2} \equiv 0, \qquad \qquad l = 1, \dots, p.$$

As in the linear case the system (5.14) can be viewed as an interconnection of two general nonlinear systems. The first subsystem, denoted by  $\Sigma_1$ , has state  $x_1$ , inputs  $(x_2, x_3, d)$  and outputs  $(z, x_1)$  and is given by (5.17), (5.20), and the second subsystem  $\Sigma_2$  has state  $(x_2, x_3)$ , inputs  $(x_1, u, d)$  and outputs  $(x_2, x_3)$ and is given by (5.18), (5.19) (see Figure 5.3).



Figure 5.3: system (5.14) after state transformation

It should be noted that compared with the decomposition of the linear system (5.2) there exists an extra connection, namely the state components  $x_2$  from the subsystem  $\Sigma_2$  influence the subsystem  $\Sigma_1$ . This is due to the fact that for nonlinear systems the disturbance vector fields are state dependent while they are not taken into account in the construction of the distribution  $S^*$ . A way to eliminate this extra interconnection is to use a stronger notion of invariance and conditioned invariance. This comes down to including the Lie-brackets with the disturbance vector fields  $e_i$  in the  $S^*$ -algorithm at the expense of a possibly larger distribution  $S^*$ . For the final state transformation of the system  $\Sigma$  this extension is not necessary as will be shown in the next subsection.

Comparing with the linear case we deduce that also in the nonlinear case the mapping h has an injectivity property.

**Lemma 5.14** Assume that  $S^*$  and  $S^* \cap \ker$  dh are constant dimensional. Then the Jacobian of the mapping h from  $x_3$  to z has full column rank and thus the mapping h from  $x_3$  to z is locally injective around every point  $p \in M$ .  $\Box$ 

**Proof** As shown earlier there exist coordinates  $(\xi_1, \ldots, \xi_{\nu}, \xi_{\nu+1}, \ldots, \xi_{\mu}, \xi_{\mu+1}, \ldots, \xi_n)$  such that:

$$\begin{aligned} \chi_2(p) &= \mathcal{S}^*(p) \cap \ker dh(p) \\ &= \operatorname{span} \left\{ \frac{\partial}{\partial \xi_{\nu+1}} \bigg|_p, \dots, \frac{\partial}{\partial \xi_{\mu}} \bigg|_p \right\}; \\ \chi_2(p) \oplus \chi_3(p) &= \mathcal{S}^*(p) \\ &= \operatorname{span} \left\{ \frac{\partial}{\partial \xi_{\nu+1}} \bigg|_p, \dots, \frac{\partial}{\partial \xi_n} \bigg|_p \right\}. \end{aligned}$$

By definition of  $X_2$  and  $X_3$ 

$$\frac{\partial h}{\partial \xi_j}\Big|_p = \mathrm{d}h|_p \;\; \frac{\partial}{\partial \xi_j}\Big|_p \neq 0, \qquad \qquad j = \mu + 1, \dots, n.$$

Suppose now there exist constants  $\alpha_{\mu+1}, \ldots, \alpha_n$  such that

$$\begin{aligned} \alpha_{\mu+1} \frac{\partial h}{\partial \xi_{\mu+1}} \bigg|_{p} + \dots + \alpha_{n} \frac{\partial h}{\partial \xi_{n}} \bigg|_{p} \\ &= dh|_{p} \left( \alpha_{\mu+1} \frac{\partial}{\partial \xi_{\mu+1}} \bigg|_{p} + \dots + \alpha_{n} \frac{\partial}{\partial \xi_{n}} \bigg|_{p} \right) = 0 \end{aligned}$$

then

$$\alpha_{\mu+1} \left. \frac{\partial}{\partial \xi_{\mu+1}} \right|_p + \dots + \alpha_n \left. \frac{\partial}{\partial \xi_n} \right|_p \in \ker \mathrm{d}h(p)$$

which in turn implies that there exist  $\alpha_1, \ldots, \alpha_{\mu}$  such that

$$\alpha_{\mu+1} \left. \frac{\partial}{\partial \xi_{\mu+1}} \right|_p + \dots + \alpha_n \left. \frac{\partial}{\partial \xi_n} \right|_p = \alpha_1 \left. \frac{\partial}{\partial \xi_1} \right|_p + \dots + \alpha_\mu \left. \frac{\partial}{\partial \xi_\mu} \right|_p$$

and because the vectors  $\{\frac{\partial}{\partial \xi_1}|_p, \ldots, \frac{\partial}{\partial \xi_n}|_p\}$  are independent it follows that all  $\alpha_i$  are equal to zero. Hence  $\frac{\partial h}{\partial x_3}|_p$  has full column rank, and by the Inverse function theorem this implies that h from  $x_3$  to z is locally injective.

#### 5.2.2 State transformation for general nonlinear systems

In this subsection we will use the results from Subsection 5.2.1 to define a state transformation for the original system  $\Sigma$ .

In order to do so we define similar to the linear case, an extended system  $\Sigma_e$  by adding the inputs u and the disturbances d to the state:

$$\Sigma_e \begin{cases} \dot{x} = f(x, u, d) \\ \dot{u} = v \\ \dot{d} = w \\ z = h(x, u) \end{cases}$$
(5.21)

The extended system is affine in the new inputs v and the new disturbances w. We rewrite the system  $\Sigma_e$  with state  $x_e^T = (\begin{array}{cc} x^T & u^T & d^T \end{array})$  as

$$\Sigma_{e} \begin{cases} \dot{x}_{e} = f_{e}(x_{e}) + \sum_{j=1}^{m} g_{je}(x_{e})v_{j} + \sum_{i=1}^{q} e_{ie}(x_{e})w_{i} \\ z = h_{e}(x_{e}) \end{cases}$$
(5.22)

where the vector fields are given by:

$$f_e(x_e) = \begin{pmatrix} f(x, u, d) \\ 0 \\ 0 \end{pmatrix}, \qquad g_{je}(x_e) = Id_{n+j},$$
$$e_{ie}(x_e) = Id_{n+m+i}, \qquad h_e(x_e) = h(x, u)$$

with  $Id_r$  the r-th identity vector, i.e., the (n + m + q)-vector with all elements zero except an element equal to one on the r-th position.

The system  $\Sigma_e$  is clearly of the form (5.14). So we can apply the state transformation based on the minimal conditioned invariant distribution containing the input vector fields as described in Subsection 5.2.1. The system  $\Sigma_e$  has a special structure which results in some nice properties for this state transformation. We start by making the following assumption

Assumption 5 The  $(p \times m)$ -matrix

$$\frac{\partial h}{\partial u}(x,u)$$

has constant column rank equal to  $m - m_1$  at least locally around the origin.  $\Box$ 

First we construct the minimal conditioned invariant distribution containing the input vector fields, denoted by  $S_e^*$ , by applying the  $S^*$ -algorithm described in Subsection 5.2.1. The distributions defined in the previous subsection for the system  $\Sigma$  applied to the system  $\Sigma_e$  will be marked by an extra subscript *e*, i.e.,  $\chi_{1e}$  etc..

**Lemma 5.15** Consider the system  $\Sigma_e$ . Assume Assumption 5 is satisfied and  $S_e^*$  and  $S_e^* \cap \ker$  dh are constant dimensional. Then the state transformation defined in Subsection 5.2.1 has the following properties:

*(i)* 

$$\mathcal{G}_e := \operatorname{span}\left\{\frac{\partial}{\partial u_1}, \ldots, \frac{\partial}{\partial u_m}\right\} \subset \mathcal{S}_e^* = \mathcal{X}_{2e} \oplus \mathcal{X}_{3e};$$

(ii)  $X_{1e}$  can be chosen such that

$$\operatorname{span}\left\{\frac{\partial}{\partial d_1},\ldots,\frac{\partial}{\partial d_q}\right\}\subset X_{1e};$$

(iii) a state dependent input transformation of the form  $u = \beta(x, \upsilon)$  can be found such that in the new coordinates  $\upsilon = (\upsilon_1, \ldots, \upsilon_{m_1}, \upsilon_{m_1+1}, \ldots, \upsilon_m)$  the distributions  $\chi_{2e}$  and  $\chi_{3e}$  can be chosen as follows:

$$\operatorname{span}\left\{\frac{\partial}{\partial \upsilon_{1}}, \ldots, \frac{\partial}{\partial \upsilon_{m_{1}}}\right\} \subset S_{e}^{*} \cap \ker dh = X_{2e};$$
$$\operatorname{span}\left\{\frac{\partial}{\partial \upsilon_{m_{1}+1}}, \ldots, \frac{\partial}{\partial \upsilon_{m}}\right\} \subset X_{3e}.$$

**Proof** Using the  $S^*$ -algorithm for  $\Sigma_e$  we see that

$$S_{1e} = \operatorname{span}\left\{\frac{\partial}{\partial u_1}, \ldots, \frac{\partial}{\partial u_m}\right\} = \overline{S}_{1e}$$

and because  $S^* = \bigcup_{l \ge 1} \overline{S}_{le}$  property (*i*) is proved. Because of Assumption 5 we can find a state dependent input transformation  $u = \beta(x, v)$  in such a way that for the first  $m_1 (\le m)$  components of the new  $v = (v_1, \ldots, v_m)$  the following holds

$$\frac{\partial}{\partial v_j} \in \overline{\mathcal{S}}_{1e} \cap \ker dh \iff \frac{\partial}{\partial v_j} \in \ker dh \iff \frac{\partial h_l}{\partial v_j} \equiv 0, \quad l = 1, \dots, p.$$
(5.23)
Hence *(iii)* is proved. Furthermore

$$S_{2e} = \overline{S}_{1e} + \left[ f(x, u, d) \frac{\partial}{\partial x}, \overline{S}_{1e} \cap \ker dh \right] + \sum_{j=1}^{m} \left[ \frac{\partial}{\partial u_j}, \overline{S}_{1e} \cap \ker dh \right]$$
  
$$= \overline{S}_{1e} + \left[ f(x, u, d) \frac{\partial}{\partial x}, \overline{S}_{1e} \cap \ker dh \right]$$
  
$$= \operatorname{span} \left\{ \frac{\partial}{\partial v_1}, \dots, \frac{\partial}{\partial v_m} \right\} + \sum_{j=1}^{m_1} \left[ f(x, \beta(x, v), d) \frac{\partial}{\partial x}, \frac{\partial}{\partial v_j} \right].$$

It can be easily seen that the Lie-bracket

$$\left[k(x,\upsilon,d)\frac{\partial}{\partial x},\frac{\partial}{\partial \upsilon_j}\right]$$

for an arbitrary function k is always of the form  $\tilde{k}(x, \upsilon, d)\frac{\partial}{\partial x}$ . Therefore the distributions  $S_{2e}$  and  $\overline{S}_{2e}$  only contains the vector fields  $\frac{\partial}{\partial u_1}, \ldots, \frac{\partial}{\partial u_m}$  and vector fields of the form

$$\tilde{k}(x, u, d) \frac{\partial}{\partial x}$$

The same can be proved for  $S_{3e}$ ,  $\overline{S}_{3e}$ , etc.. So  $X_{1e}$  can be chosen such that *(ii)* holds.

We already assumed that the distributions  $\mathcal{S}_e^*$  and  $\mathcal{S}_e^* \cap \ker dh$  are constant dimensional. Furthermore we want to be able to project these distributions on the state space M of the original system  $\Sigma$ , i.e., a distribution  $\mathcal{D}$  on  $M \times \mathbb{R}^m \times \mathbb{R}^q$  $\mathbb{R}^q$  is called *projectable* on M by the canonical projection  $\pi : M \times \mathbb{R}^m \times \mathbb{R}^q \to$ M given by  $\pi : x_e \mapsto x \ (x_e := (x, u, d))$  if for all  $\kappa_1, \kappa_2 \in M \times \mathbb{R}^m \times \mathbb{R}^q$  we have

$$\pi(\kappa_1) = \pi(\kappa_2) \Rightarrow \frac{\partial \pi}{\partial x_e}(\kappa_1) \left(\mathcal{D}(\kappa_1)\right) = \frac{\partial \pi}{\partial x_e}(\kappa_2) \left(\mathcal{D}(\kappa_2)\right).$$

The next lemma gives conditions for the projectability of these two distributions.

**Lemma 5.16** Assume that the distributions  $S_e^*$  and  $S_e^* \cap \ker dh$  are constant dimensional. The distributions  $S_e^*$  and  $S_e^* \cap \ker dh$  are projectable on M (by the canonical projection  $\pi : M \times \mathbb{R}^m \times \mathbb{R}^q \to M$  given by  $\pi(x, u, d) = x$ ) if

and only if the following inclusions hold:

$$\begin{bmatrix} \frac{\partial}{\partial d_j}, S_e^* \end{bmatrix} \subset S_e^*, \qquad j = 1, \dots, q; \qquad (5.24)$$
$$\begin{bmatrix} \frac{\partial}{\partial v_i}, S_e^* \cap \ker dh \end{bmatrix} \subset S_e^* \cap \ker dh + \mathcal{G}_e, \quad i = m_1 + 1, \dots, m.(5.25)$$

**Proof** From [MRS 94] it follows that we have to prove that:

$$\begin{bmatrix} \frac{\partial}{\partial u_j}, Q \end{bmatrix} \subset Q + \mathcal{G}_e + \mathcal{D}_e, \qquad j = 1, \dots, m; \qquad (5.26)$$

$$\left[\frac{\partial}{\partial d_i}, Q\right] \subset Q + \mathcal{G}_e + \mathcal{D}_e, \qquad i = 1, \dots, q; \qquad (5.27)$$

holds for  $Q = S_e^*$  and  $Q = S_e^* \cap \ker dh$ , where the distribution  $\mathcal{D}_e$  is defined as

$$\mathcal{D}_e = \operatorname{span}\left\{\frac{\partial}{\partial d_1}, \ldots, \frac{\partial}{\partial d_q}\right\}.$$

Since  $S_e^*$  is involutive and  $G_e \subset S_e^*$ , (5.26) is satisfied for  $Q = S_e^*$ . Furthermore from the proof of Lemma 5.15 it follows that  $S_e^*$  is spanned by the vector fields  $\frac{\partial}{\partial u_1}, \ldots, \frac{\partial}{\partial u_m}$  and by vector fields of the form  $\tilde{k}(x, u, d) \frac{\partial}{\partial x}$  for arbitrary functions  $\tilde{k}$ . Then it follows that

$$\left[\frac{\partial}{\partial d_i}, X\right] \notin \mathcal{D}_e$$

for all vector fields  $X \in S_e^*$ . Hence (5.24) holds if and only if (5.27) is satisfied for  $Q = S_e^*$ .

The vector fields  $\frac{\partial}{\partial v_i}$  for  $i = 1, ..., m_1$  are elements of  $\mathcal{S}_e^* \cap \ker dh$ . The involutivity of  $\mathcal{S}_e^* \cap \ker dh$  together with (5.25) implies that (5.26) holds for  $Q = \mathcal{S}_e^* \cap \ker dh$ . By (5.24):

$$\left[\frac{\partial}{\partial d_j}, \, \mathcal{S}_e^* \cap \ker \, \mathrm{d}h\right] \subset \, \mathcal{S}_e^*, \qquad \qquad j = 1, \dots, q. \tag{5.28}$$

Take an arbitrary vector field X from  $S_e^* \cap \ker dh$ . Then

$$\sum_{i=1}^{n} \frac{\partial h_l}{\partial x_i} X_i + \sum_{r=1}^{m} \frac{\partial h_l}{\partial u_r} X_{n+r} = 0, \qquad l = 1, \dots, p$$

and it follows that

$$dh_{l}\left[\frac{\partial}{\partial d_{j}}, X\right] = dh_{l}\left(\sum_{i=1}^{n} \frac{\partial X_{i}}{\partial d_{j}} \frac{\partial}{\partial x_{i}} + \sum_{r=1}^{m} \frac{\partial X_{r+n}}{\partial d_{j}} \frac{\partial}{\partial u_{r}}\right)$$

$$= \sum_{i=1}^{n} \frac{\partial h_{l}}{\partial x_{i}} \frac{\partial X_{i}}{\partial d_{j}} + \sum_{r=1}^{m} \frac{\partial h_{l}}{\partial u_{r}} \frac{\partial X_{r+n}}{\partial d_{j}}$$

$$= \frac{\partial}{\partial d_{j}}\left(\sum_{i=1}^{n} \frac{\partial h_{l}}{\partial x_{i}} X_{i} + \sum_{r=1}^{m} \frac{\partial h_{l}}{\partial u_{r}} X_{r+n}\right)$$

$$= 0 \qquad \qquad l = 1, \dots, p$$

or in coordinate free notation

$$dh_l\left[\frac{\partial}{\partial d_j}, X\right] = \frac{\partial}{\partial d_j} \left(X(h_l)\right) - X\left(\frac{\partial}{\partial d_j}(h_l)\right) = 0 \qquad l = 1, \dots, p$$

because  $X \in \ker dh_l$  and  $h_l$  does not depend on d. Hence

$$\left[\frac{\partial}{\partial d_j}, \, \mathcal{S}_e^* \cap \ker \, \mathrm{d}h\right] \subset \ker \, \mathrm{d}h$$

and together with (5.28) this proves (5.27) for  $S_e^* \cap \ker dh$ .

Assume that the distributions  $S_e^*$  and  $S_e^* \cap \ker dh$  are constant dimensional and that the conditions (5.24) and (5.25) from Lemma 5.16 are satisfied. Then the distributions  $S_e^*$  and  $S_e^* \cap \ker dh$  are projectable on the state manifold M by the canonical projection  $\pi : M \times \mathbb{R}^m \times \mathbb{R}^q \to M$  onto involutive distributions on M. Therefore we can find local coordinates written as  $(\xi_1, \ldots, \xi_\nu, \xi_{\nu+1}, \ldots, \xi_\mu,$  $\xi_{\mu+1}, \ldots, \xi_n)$  for M such that:

$$\pi_* \left( \mathcal{S}_e^* \cap \ker dh \right) = \operatorname{span} \left\{ \frac{\partial}{\partial \xi_{\nu+1}}, \dots, \frac{\partial}{\partial \xi_{\mu}} \right\};$$
$$\pi_* \mathcal{S}_e^* = \operatorname{span} \left\{ \frac{\partial}{\partial \xi_{\nu+1}}, \dots, \frac{\partial}{\partial \xi_n} \right\}.$$

Furthermore by condition (5.24) there exists a basis of  $\mathcal{S}_e^*$  which does not depend on the disturbances d. Then we transform the inputs u by a state dependent transformation  $u = \beta(x, v)$  such that for the first  $m_1$  components of the new coordinates  $v = (v_1, \ldots, v_{m_1}, v_{m_1+1}, \ldots, v_m)$  we have that

$$\operatorname{span}\left\{\frac{\partial}{\partial \upsilon_1},\ldots,\frac{\partial}{\partial \upsilon_{m_1}}\right\} = \mathcal{S}_e^* \cap \ker \mathrm{d} h \cap \mathcal{G}_e.$$

Then the distributions  $X_{1e}$ ,  $X_{2e}$  and  $X_{3e}$  can be chosen in the following way:

$$\begin{aligned} \mathcal{X}_{1e} &= \operatorname{span} \left\{ \frac{\partial}{\partial \xi_1}, \dots, \frac{\partial}{\partial \xi_{\nu}}, \frac{\partial}{\partial d_1}, \dots, \frac{\partial}{\partial d_q} \right\}; \\ \mathcal{X}_{2e} &= \operatorname{span} \left\{ \frac{\partial}{\partial \xi_{\nu+1}}, \dots, \frac{\partial}{\partial \xi_{\mu}}, \frac{\partial}{\partial \upsilon_1}, \dots, \frac{\partial}{\partial \upsilon_{m_1}} \right\}; \\ \mathcal{X}_{3e} &= \operatorname{span} \left\{ \frac{\partial}{\partial \xi_{\mu+1}}, \dots, \frac{\partial}{\partial \xi_n}, \frac{\partial}{\partial \upsilon_{m_1+1}}, \dots, \frac{\partial}{\partial \upsilon_m} \right\} \end{aligned}$$

In these new coordinates the vector field  $f_e(x_e)$  and the function  $h_e(x_e)$  have the following special structure

$$f_e(x_e) = \begin{pmatrix} f_1(x_1, x_3, u_2, d) \\ f_2(x_1, x_2, x_3, u_1, u_2, d) \\ f_3(x_1, x_2, x_3, u_1, u_2, d) \\ \hline 0 \\ \hline 0 \\ \hline 0 \\ \hline 0 \\ \hline \end{pmatrix}, \quad h_e(x_e) = h(x_1, x_3, u_2, d)$$

The projections of the distributions  $X_{1e}$ ,  $X_{2e}$  and  $X_{3e}$  on the state space M will be indicated by  $X_1^P$ ,  $X_2^P$  and  $X_3^P$ . In these new coordinates for the state space  $M = X_1^P \oplus X_2^P \oplus X_3^P$  and the input space  $\mathbb{R}^m$  the original system  $\Sigma$  takes the following form

$$\Sigma \begin{cases} \dot{x}_1 = f_1(x_1, x_3, u_2, d) \\ \dot{x}_2 = f_2(x_1, x_2, x_3, u_1, u_2, d) \\ \dot{x}_3 = f_3(x_1, x_2, x_3, u_1, u_2, d) \\ z = h(x_1, x_3, u_2) \end{cases}$$
(5.29)

where the derivative of h with respect to  $(x_3, u_2)$  has full column rank (see Lemma 5.14).

Again the system  $\Sigma$  can be viewed as the interconnection of two general nonlinear systems  $\Sigma_1$  and  $\Sigma_2$ . The first subsystem  $\Sigma_1$  with state  $x_1$ , inputs  $(x_3, u_2, d)$  and output  $(z, x_1)$  and another system  $\Sigma_2$  with state  $(x_2, x_3)$ , inputs  $(x_1, u_1, u_2, d)$  and output  $x_3$  (see Figure 5.4).

$$\Sigma_{1} \begin{cases} \dot{x}_{1} = f_{1}(x_{1}, x_{3}, u_{2}, d) \\ z = h(x_{1}, x_{3}, u_{2}) \end{cases}$$
$$\Sigma_{2} \begin{cases} \dot{x}_{2} = f_{2}(x_{1}, x_{2}, x_{3}, u_{1}, u_{2}, d) \\ \dot{x}_{3} = f_{3}(x_{1}, x_{2}, x_{3}, u_{1}, u_{2}, d) \end{cases}$$



Figure 5.4: system  $\Sigma$  after input and state transformation

#### 5.2.3 Special cases

This subsection will be concerned with the two extreme cases of the general  $\mathcal{H}_{\infty}$  problem for general nonlinear systems considered in this section.

#### Regular $\mathcal{H}_{\infty}$ problem

First of all we will look at the regular  $\mathcal{H}_{\infty}$  problem. So we will consider systems  $\Sigma$  which satisfy Assumption 1. These systems clearly satisfy Assumption 5, with  $m_1$  equal to zero. Therefore there is no need to transform the inputs.

The extended system  $\Sigma_e$  in this case satisfies

$$G_e \cap \ker \mathrm{d}h = 0.$$

Hence the  $S^*$ -algorithm applied to  $\Sigma_e$  stops after one step and locally around the origin we have that:

$$\mathcal{S}_{e}^{*} = \mathcal{G}_{e} = \operatorname{span}\left\{\frac{\partial}{\partial u_{1}}, \dots, \frac{\partial}{\partial u_{m}}\right\};$$
$$\mathcal{S}_{e}^{*} \cap \ker dh = 0.$$

These distributions are clearly projectable. There is no need for a transformation of the inputs and the conditions (5.24) and (5.25) for the projectability are satisfied:

$$\begin{bmatrix} \frac{\partial}{\partial d_j}, \mathcal{G}_e \end{bmatrix} = 0 \in \mathcal{G}_e, \qquad j = 1, \dots, q;$$
$$\begin{bmatrix} \frac{\partial}{\partial u_i}, 0 \end{bmatrix} = 0 \in \mathcal{G}_e, \qquad i = 1, \dots, m.$$

The distributions  $X_{1e}$ ,  $X_{2e}$  and  $X_{3e}$  for the system  $\Sigma_e$  are given by:

$$\begin{aligned} \mathcal{X}_{1e} &= \operatorname{span}\left\{\frac{\partial}{\partial x_1}, \dots, \frac{\partial}{\partial x_n}, \frac{\partial}{\partial d_1}, \dots, \frac{\partial}{\partial d_q}\right\}; \\ \mathcal{X}_{2e} &= 0; \\ \mathcal{X}_{3e} &= \operatorname{span}\left\{\frac{\partial}{\partial u_1}, \dots, \frac{\partial}{\partial u_m}\right\}. \end{aligned}$$

So the projected spaces are  $\mathcal{X}_1^P = M$ ,  $\mathcal{X}_2^P = 0$  and  $\mathcal{X}_3^P = 0$ , and the regular subsystem  $\Sigma_1$  is equal to the complete system  $\Sigma$ .

#### Full singular $\mathcal{H}_{\infty}$ problem

In the full singular case we assume that the output mapping h does not depend on the inputs u. So h(x, u) = h(x) and automatically Assumption 5 is satisfied with  $m_1 = m$ . In this special case we can see that for the extended system  $\Sigma_e$ it follows that

$$G_e \subset S_e^* \cap \ker \mathrm{d}h$$

and the projectability condition (5.25) is void. Again no transformation of the input is necessary.

The remaining assumptions are that the distributions  $S_e$  and  $S_e \cap \ker dh$ are constant dimensional and that condition (5.24) is satisfied, which implies projectability of the distributions onto the state space of the original system  $\Sigma$ . Then we can find local coordinates  $(\xi_1, \ldots, \xi_\nu, \xi_{\nu+1}, \ldots, \xi_\mu, \xi_{\mu+1}, \ldots, \xi_n)$ such that the distributions  $X_{1e}$ ,  $X_{2e}$  and  $X_{3e}$  for the extended system  $\Sigma_e$  are given by:

$$\begin{aligned} \mathcal{X}_{1e} &= \operatorname{span} \left\{ \frac{\partial}{\partial \xi_1}, \dots, \frac{\partial}{\partial \xi_{\nu}}, \frac{\partial}{\partial d_1}, \dots, \frac{\partial}{\partial d_q} \right\}; \\ \mathcal{X}_{2e} &= \operatorname{span} \left\{ \frac{\partial}{\partial \xi_{\nu+1}}, \dots, \frac{\partial}{\partial \xi_{\mu}}, \frac{\partial}{\partial u_1}, \dots, \frac{\partial}{\partial u_m} \right\}; \\ \mathcal{X}_{3e} &= \operatorname{span} \left\{ \frac{\partial}{\partial \xi_{\mu+1}}, \dots, \frac{\partial}{\partial \xi_n} \right\}. \end{aligned}$$

The system  $\Sigma$  after the state transformation is of the form

$$\Sigma \begin{cases} \dot{x}_1 = f_1(x_1, x_3, d) \\ \dot{x}_2 = f_2(x_1, x_2, x_3, u, d) \\ \dot{x}_3 = f_3(x_1, x_2, x_3, u, d) \\ z = h(x_1, x_3) \end{cases}$$

In this case there is no direct influence of the inputs on the regular subsystem  $\Sigma_1$ .

# 5.2.4 The nonlinear state feedback $\mathcal{H}_{\infty}$ problem: sufficient conditions

Using the transformations described in the previous subsections the nonlinear system  $\Sigma$  can be seen as the interconnection of two subsystems  $\Sigma_1$  and  $\Sigma_2$ . Inspired by the linear theory described in the beginning of this chapter we will split the singular  $\mathcal{H}_{\infty}$  problem for the system  $\Sigma$  into two problems. The first part is a  $\mathcal{H}_{\infty}$  problem for the subsystem  $\Sigma_1$  with inputs  $(x_3, u_2)$ , outputs z and disturbances d. Because the Jacobian of the mapping h from the inputs  $(x_3, u_2)$  to the outputs z has full column rank this problem is regular. So for this problem we can use the regular  $\mathcal{H}_{\infty}$  theory described in Chapter 2. The solution of this regular state feedback  $\mathcal{H}_{\infty}$  problem consists of a trajectory of part of the state of the subsystem  $\Sigma_2$ , namely  $x_3$ , and an optimal state feedback for the inputs  $u_2$ . In the second step we take the inputs  $u_2$  equal to the feedback solution of the regular subproblem and try to track the state components  $x_3$  along the solution of the regular problem by applying a suitable feedback for  $u_1$  to the subsystem  $\Sigma_2$ .

From Chapter 2 we know that the regular strictly suboptimal  $\mathcal{H}_{\infty}$  problem for the subsystem  $\Sigma_1$  with inputs  $(x_3, u_2)$ , disturbances d and outputs z is solvable if there exists for some  $\overline{\gamma} < \gamma$  a local solution  $V \ge 0$  to the Hamilton-Jacobi inequality

$$H_{\overline{\gamma}}(x_1, V_{x_1}^T(x_1)) \le 0, \qquad V(0) = 0$$

where the Hamiltonian  $H_{\overline{\gamma}}: T^* \mathcal{X}_1^P \to \mathbb{R}$  is given by

$$H_{\overline{\gamma}}(x_1, p_1) = K_{\overline{\gamma}}(x_1, p_1, d^*(x_1, p_1), x_3^*(x_1, p_1), u_2^*(x_1, p_1))$$

with  $K_{\gamma}: T^* \mathcal{X}_1^P \times \mathbb{R}^q \times \mathcal{X}_3^P \times \mathbb{R}^{m-m_1} \to \mathbb{R}$  the pre-Hamiltonian corresponding to this problem, that is

$$K_{\overline{\gamma}}(x_1, p_1, d, x_3, u_2) = p_1^T f_1(x_1, x_3, u_2, d) + \frac{1}{2} \|h(x_1, x_3, u_2)\|^2 - \frac{1}{2} \overline{\gamma}^2 \|d\|^2$$
  
and  $x_3^*(x_1, p_1), u_2^*(x_1, p_1)$  and  $d^*(x_1, p_1)$  is the unique solution of

$$\frac{\partial K_{\overline{\gamma}}}{\partial d}(x_1, p_1, d^*(x_1, p_1), x_3^*(x_1, p_1), u_2^*(x_1, p_1)) = 0$$
  
$$\frac{\partial K_{\overline{\gamma}}}{\partial x_3}(x_1, p_1, d^*(x_1, p_1), x_3^*(x_1, p_1), u_2^*(x_1, p_1)) = 0$$
  
$$\frac{\partial K_{\overline{\gamma}}}{\partial u_2}(x_1, p_1, d^*(x_1, p_1), x_3^*(x_1, p_1), u_2^*(x_1, p_1)) = 0$$

with  $x_3^*(0,0) = 0$ ,  $u_2^*(0,0) = 0$  and  $d^*(0,0) = 0$ .

Then the feedback

$$x_3 = x_3^*(x_1, V_{x_1}^T(x_1)),$$
  

$$u_2 = u_2^*(x_1, V_{x_1}^T(x_1))$$

applied to the subsystem  $\Sigma_1$  leads to a closed loop system which has  $L_2$ -gain less than or equal to  $\overline{\gamma} < \gamma$ .

In the second step we choose the inputs  $u_2$  equal to the optimal one

$$u_2 = u_2^*(x_1, V_{x_1}^T(x_1))$$

and we try to track the optimal state trajectory calculated in the first step for the state components  $x_3$ . In order to do so we introduce a new state variable for  $\Sigma_2$  equal to the tracking error

$$q_3 := x_3 - x_3^*(x_1, V_{x_1}^T(x_1)).$$
(5.30)

This state transformation leads to the following form of the system  $\Sigma$ , which we will refer to as  $\Sigma^{tr}$ :

$$\Sigma^{\text{tr}} \begin{cases} \dot{x}_1 = f_1(x_1, q_3 + x_3^*, u_2^*, d) \\ \dot{x}_2 = f_2(x_1, x_2, q_3 + x_3^*, u_1, u_2^* d) \\ \dot{q}_3 = \tilde{f}_3(x_1, x_2, q_3 + x_3^*, u_1, u_2^*, d) \\ z = h(x_1, q_3 + x_3^*, u_2^*) \end{cases}$$

where

$$\begin{aligned} x_3^* &= x_3^*(x_1, V_{x_1}^T(x_1)); \\ u_2^* &= u_2^*(x_1, V_{x_1}^T(x_1)); \\ \tilde{f}_3(x_1, x_2, x_3, u_1, u_2, d) &= f_3(x_1, x_2, x_3, u_1, u_2, d) \\ &- \frac{\mathrm{d}x_3^*}{\mathrm{d}x_1}(x_1, V_{x_1}^T(x_1)) f_1(x_1, x_3, u_2, d) \end{aligned}$$

The transformed system  $\Sigma^{tr}$  is still the interconnection of two subsystems namely (see Figure 5.5):

$$\Sigma_{1}^{\text{tr}} \begin{cases} \dot{x}_{1} = f_{1}(x_{1}, q_{3} + x_{3}^{*}, u_{2}^{*}, d) \\ z = h(x_{1}, q_{3} + x_{3}^{*}, u_{2}^{*}) \end{cases}$$
  
$$\Sigma_{2}^{\text{tr}} \begin{cases} \dot{x}_{2} = f_{2}(x_{1}, x_{2}, q_{3} + x_{3}^{*}, u_{1}, u_{2}^{*}d) \\ \dot{q}_{3} = \tilde{f}_{3}(x_{1}, x_{2}, q_{3} + x_{3}^{*}, u_{1}, u_{2}^{*}, d) \end{cases}$$



Figure 5.5: system  $\Sigma^{tr}$ 

It should be noted that  $q_3 = 0$  is the a solution of the regular  $\mathcal{H}_{\infty}$  state feedback problem for the subsystem  $\Sigma_1$ , with inputs  $q_3$ .

Parallel to the linear theory we want to solve the  $\mathcal{H}_{\infty}$  almost disturbance decoupling problem for the transformed subsystem  $\Sigma_2^{\text{tr}}$  with disturbances  $(x_1, d)$ , inputs  $u_1$  and outputs  $q_3$ . Because we use  $x_1$  as a disturbance input for the system  $\Sigma_2^{\text{tr}}$  we have to make an extra assumption on the input-to-state stability of the system  $\Sigma_2^{\text{tr}}$ .

Assumption 6 Assume that the  $L_2$ -gain from  $q_3$  and d to  $x_1$  for the system  $\Sigma_1^{\text{tr}}$  is finite, or otherwise stated, there exist constants  $M_1$ ,  $M_2 > 0$  and for every  $x_1$  there exists a constant  $M(x_1)$ ,  $0 \le M(x_1) < \infty$ , M(0) = 0, such that

$$\int_0^t \|x_1(\tau)\|^2 \mathrm{d}\tau \le M_1 \int_0^t \|q_3(\tau)\|^2 \mathrm{d}\tau + M_2 \int_0^t \|d(\tau)\|^2 \mathrm{d}\tau + M(x_1(0))$$
(5.31)

for all  $d, q_3 \in L_2(0, t)$  and for all  $t \ge 0$ .

Also other assumptions can be made to ensure input-to-state stability (see for instance [SW 94] for necessary and sufficient conditions for input-to-state stability).

**Theorem 5.17** Consider the system  $\Sigma$ . Suppose that for the extended system  $\Sigma_e$  the distributions  $S_e^*$  and  $S_e^* \cap \ker dh$  are constant dimensional, that Assumption 5 is satisfied and that (5.24) and (5.25) hold. Assume there exists for some  $\overline{\gamma} < \gamma$  a solution  $V \ge 0$  of

$$H_{\overline{\gamma}}(x_1, V_{x_1}^I(x_1)) \le 0 \tag{5.32}$$

with V(0) = 0. Additionally assume that Assumption 6 holds. Then there exists a constant  $k^*$  such that the nonlinear state feedback  $L_2$ -gain control problem with constant  $\overline{\gamma} < \gamma$  is solvable locally for  $\Sigma$  by the feedback

$$u_1 = \tilde{\beta}(x_1, x_2, x_3, k) = \beta(x_2, x_3 - x_3^*(x_1, V_{x_1}^T(x_1)), k)$$
  
$$u_2 = u_2^*(x_1, V_{x_1}^T(x_1))$$

for  $k > k^*$  if the parameterized feedback

$$u_1 = \beta(x_2, q_3, k)$$

solves the  $L_2$ -gain almost disturbance decoupling problem with constant  $\frac{1}{k}$  for  $\Sigma_2^{\text{tr}}$ .

Before we prove this theorem we will state the following observation.

**Lemma 5.18** Consider the system  $\Sigma$ . Suppose that for the extended system  $\Sigma_e$  the distributions  $S_e^*$  and  $S_e^* \cap \ker$  dh are constant dimensional, that Assumption 5 is satisfied and that (5.24) and (5.25) hold. Assume there exists for some  $\overline{\gamma} < \gamma$  a solution  $V \ge 0$  of

$$H_{\overline{\gamma}}(x_1, V_{x_1}^T(x_1)) \le 0$$
 (5.33)

with V(0) = 0. Then locally around the origin there exists constants  $N, \varepsilon > 0$ and a constant  $R(x_1), 0 \le R(x_1) < \infty$ , with R(0) = 0 such that

$$\int_0^t \|z(\tau)\|^2 \mathrm{d}\tau \le N \int_0^t \|q_3(\tau)\|^2 \mathrm{d}\tau + (\gamma^2 - \varepsilon) \int_0^t \|d(\tau)\|^2 \mathrm{d}\tau + R(x_1(0))$$
(5.34)

for all  $d, q_3 \in L_2(0, t)$  and  $x_1(0)$  such that the resulting state trajectories stay in a neighborhood of the origin.

**Proof** Consider the pre-Hamiltonian

$$L(x_1, p, d, q_3) := p^T f_1(x_1, q_3 + x_3^*, u_2^*, d) + \frac{1}{2} h^T(x_1, q_3 + x_3^*, u_2^*) h(x_1, q_3 + x_3^*, u_2^*) - \frac{1}{2} N ||q_3||^2 - \frac{1}{2} \overline{\gamma}^2 ||d||^2.$$

The Hessian of L with respect to d and  $q_3$  in  $(x_1, p, d, q_3) = (0, 0, 0, 0)$  is given by

$$\left(\begin{array}{cc} -\overline{\gamma}^2 I & 0\\ 0 & -NI + \left(\frac{\partial h}{\partial x_3}(0,0,0)\right)^T \left(\frac{\partial h}{\partial x_3}(0,0,0)\right) \end{array}\right),$$

where I is the identity matrix with dimension equal to the dimension of  $q_3$ . For N sufficiently large, i.e., larger than the largest eigenvalue of the matrix

$$\left(\frac{\partial h}{\partial x_3}(0,0,0)\right)^T \left(\frac{\partial h}{\partial x_3}(0,0,0)\right),$$

this matrix is negative definite. Since

$$L(x_1, p, d, q_3) = K_{\overline{\gamma}}(x_1, p, d, q_3 + x_3^*(x_1, p), u_3^*(x_1, p)) - \frac{1}{2}N ||q_3||^2$$

it follows that the following equalities have a unique solution given by  $q_3 = 0$ ,  $d = d^*(x_1, p)$ 

$$\frac{\partial L}{\partial q_3} = \frac{K_{\overline{\gamma}}}{\partial x_3} - N(x_3 - x_3^*(x_1, p))^T = 0$$
$$\frac{\partial L}{\partial d} = \frac{K_{\overline{\gamma}}}{\partial d} = 0$$

because  $\frac{\partial K_{\overline{Y}}}{\partial x_3} = 0$  and  $\frac{\partial K_{\overline{Y}}}{\partial d} = 0$  has a unique solution equal to  $x_3 = x_3^*(x_1, p)$ ( $q_3 = 0$ ) and  $d = d^*(x_1, p)$  when  $u_2 = u_2^*(x_1, p)$ . Thus  $q_3 = 0$ ,  $d = d^*(x_1, p)$  is the maximum of the pre-Hamiltonian L for N sufficiently large and

$$L(x_1, p, d, q_3) \leq K_{\overline{\gamma}}(x_1, p, d^*(x_1, p), x_3^*(x_1, p), u_2^*(x_1, p)).$$

Hence  $V \ge 0$  is a non-negative solution of the Hamilton-Jacobi inequality

$$L(x_1, V_{x_1}^T(x_1), d, q_3) \le 0.$$

This proves the inequality (5.34) for:

$$N > \overline{\sigma} \left( \left( \frac{\partial h}{\partial x_3}(0,0,0) \right)^T \left( \frac{\partial h}{\partial x_3}(0,0,0) \right) \right);$$
  

$$\varepsilon < \gamma^2 - \overline{\gamma}^2;$$
  

$$R(x_1) \geq 2V(x_1);$$

where  $\overline{\sigma}(P)$  denotes the largest eigenvalue of the symmetric matrix P.

**Proof of Theorem 5.17** The distributions  $S_e^*$  and  $S_e^* \cap \ker dh$  for the extended system  $\Sigma_e$  are constant dimensional and involutive, Assumption 5 is satisfied and the conditions (5.24) and (5.25) hold. Therefore we can apply the state

transformation described in Subsection 5.2.2. The solution of the Hamilton-Jacobi inequality (5.32) is used to transform the state  $x_3$  according to (5.30) which results in the interconnected systems  $\Sigma_1^{\text{tr}}$  and  $\Sigma_2^{\text{tr}}$ . Now we take  $u_2 = u_2^*(x_1, V_{x_1}^T(x_1))$ . We assumed that the feedback  $u_1 = \beta(x_2, q_3, k)$  solves the  $\mathcal{H}_{\infty}$  almost disturbance decoupling problem (see Definition 4.2) for  $\Sigma_2^{\text{tr}}$  with output  $q_3$  and disturbances  $x_1$  and d. Hence there exists a constant  $K(x_2, q_3)$ ,  $0 \le K(x_2, q_3) < \infty$ , K(0, 0) = 0, such that

$$\int_0^t \|q_3(\tau)\|^2 \mathrm{d}\tau \le \frac{1}{k} \int_0^t \left( \|d(\tau)\|^2 + \|x_1(\tau)\|^2 \right) \mathrm{d}\tau + K(x_2(0), q_3(0)) \quad (5.35)$$

for all  $d, x_1 \in L_2(0, t)$  and all t. Let  $z_{q_3}$  denote the output of the subsystem  $\Sigma_1^{\text{tr}}$  resulting from disturbance d, initial state  $x_1(0)$  and input  $q_3$ . Then from Assumption 6 and Lemma 5.18 it follows that there exist constants  $N, \varepsilon, M_1, M_2 > 0$ , and constants  $R(x_1), M(x_1), 0 \le R(x_1), M(x_1) < \infty, R(0) = 0, M(0) = 0$ , such that locally:

$$\int_{0}^{t} \|z_{q_{3}}(\tau)\|^{2} d\tau \leq N \int_{0}^{t} \|q_{3}(\tau)\|^{2} d\tau \qquad (5.36)$$

$$+ (\gamma^{2} - \varepsilon) \int_{0}^{t} \|d(\tau)\|^{2} d\tau + R(x_{1}(0));$$

$$\int_{0}^{t} \|x_{1}(\tau)\|^{2} d\tau \leq M_{1} \int_{0}^{t} \|q_{3}(\tau)\|^{2} d\tau \qquad (5.37)$$

$$+ M_{2} \int_{0}^{t} \|d(\tau)\|^{2} d\tau + M(x_{1}(0)).$$

Combining the inequalities (5.35) and (5.37), letting  $k > M_1$ , results in

$$\int_0^t \|q_3(\tau)\|^2 d\tau \leq \frac{1+M_2}{k-M_1} \int_0^t \|d(\tau)\|^2 d\tau + \frac{k}{k-M_1} \left( M(x_1(0)) + K(x_2(0), q_3(0)) \right)$$

which together with (5.36) leads to

$$\int_{0}^{t} \|z_{q_{3}}(\tau)\|^{2} \mathrm{d}\tau \leq \left( N\left(\frac{1+M_{2}}{k-M_{1}}\right) + \gamma^{2} - \varepsilon \right) \int_{0}^{t} \|d(\tau)\|^{2} \mathrm{d}\tau \qquad (5.38) + \frac{Nk}{k-M_{1}} \left( M(x_{1}(0)) + K(x_{2}(0), q_{3}(0)) \right) + R(x_{1}(0)).$$

Therefore if we choose

$$k^* \geq \frac{N}{\varepsilon}(1+M_2) + M_1 > M_1$$

then the  $L_2$ -gain from d to  $z_{q_3}$  is less than or equal to  $\gamma$ .

Theorem 5.17 only states an  $L_2$ -gain result. For local stability of the closed-loop system we have to make extra assumptions.

**Theorem 5.19** Assume all assumptions from Theorem 5.17 are satisfied. Additionally assume that Assumption 6 is satisfied with a proper, differentiable function M. Then there exists a constant  $k^*$  such that the nonlinear state feedback  $L_2$ -gain control problem with constant  $\overline{\gamma} < \gamma$  is locally solvable for  $\Sigma$  by the feedback

$$u_1 = \tilde{\beta}(x_1, x_2, x_3, k) = \beta(x_2, x_3 - x_3^*(x_1, V_{x_1}^T(x_1)), k)$$
  
$$u_2 = u_2^*(x_1, V_{x_1}^T(x_1))$$

for  $k > k^*$ , and the closed-loop system is stable if the parameterized feedback

$$u_1 = \beta(x_2, q_3, k)$$

solves the  $L_2$ -gain almost disturbance decoupling problem with constant  $\frac{1}{k}$  for  $\Sigma_2^{tr}$ , with a proper, nonnegative  $C^1$  storage function.

**Proof** We take the disturbance *d* equal to zero. Then combining the inequalities (5.35) and (5.37) leads to the next two inequalities for  $x_1$  and  $q_3$ :

$$\int_0^t \|x_1(\tau)\|^2 d\tau \leq \frac{k}{k - M_1} (M_1 K(x_2(0), q_3(0)) + M(x_1(0))); (5.39)$$
  
$$\int_0^t \|q_3(\tau)\|^2 d\tau \leq \frac{k}{k - M_1} \left( K(x_2(0), q_3(0)) + \frac{1}{k} M(x_1(0)) \right).$$

Furthermore there exists a proper solution  $V \ge 0$  to

$$V(x_{2}(t_{1}), q_{3}(t_{1})) - V(x_{2}(t_{0}), q_{3}(t_{0}))$$

$$\leq \frac{1}{2} \int_{t_{0}}^{t_{1}} \left(\frac{1}{k} \|x_{1}(\tau)\|^{2} - \|q_{3}(\tau)\|^{2}\right) d\tau$$

$$\leq \frac{1}{2} \int_{t_{0}}^{t_{1}} \frac{1}{k} \|x_{1}(\tau)\|^{2} d\tau$$

for all  $t_0 \le t_1$ . Now we take  $t_0 = 0$  and  $t_1 = t$  and use the inequality (5.39) leading to the existence of a proper solution  $V \ge 0$  to

$$V(x_2(t), q_3(t)) - V(x_2(0), q_3(0)) \\ \leq \frac{1}{2} \frac{1}{k - M_1} \left( M_1 K(x_2(0), q_3(0)) + M(x_1(0)) \right)$$

for all t. So for all t the function  $V(x_2(t), q_3(t)) - V(x_2(0), q_3(0))$  is bounded by a function depending on the initial states. Because V is proper it follows that the  $x_2$  and  $q_3$  dynamics are stable. In the same way the stability of the  $x_1$ dynamics can be proved using the properness of M. The  $L_2$ -gain result follows from Theorem 5.17.

In Theorem 5.19 we only obtained a stability result and no asymptotic stability. An obvious way to ensure local asymptotic stability is provided by the following theorem.

**Theorem 5.20** Assume all assumptions from Theorem 5.17 are satisfied. Additionally assume that Assumption 6 is satisfied with a differentiable storage function  $M(x_1)$ . Then there exists a constant  $k^*$  such that the nonlinear state feedback  $\mathcal{H}_{\infty}$  control problem is locally solvable with constant  $\overline{\gamma} < \gamma$  for  $\Sigma$  by the feedback

$$u_{1} = \hat{\beta}(x_{1}, x_{2}, x_{3}, k) = \beta(x_{2}, x_{3} - x_{3}^{*}(x_{1}, V_{x_{1}}^{T}(x_{1})), k)$$
  

$$u_{2} = u_{2}^{*}(x_{1}, V_{x_{1}}^{T}(x_{1}))$$
(5.40)

for  $k > k^*$  if the parameterized feedback

$$u_1 = \beta(x_2, q_3, k)$$

solves the  $L_2$ -gain almost disturbance decoupling problem with constant  $\frac{1}{k}$  for  $\Sigma_2^{tr}$ , with a differentiable storage function, and the closed-loop system consisting of  $\Sigma$  together with the feedback (5.40) is zero-state observable.

**Proof** The proof for the  $L_2$ -gain result follows along the lines of the proof of Theorem 5.17 where we take for the storage function  $R(x_1)$  a differentiable function satisfying  $R(x_1) \ge 2V(x_1)$  (see the proof of Lemma 5.18). Furthermore there exists a differentiable function  $K(x_2, q_3)$  satisfying (5.35). Then the inequality (5.38) is satisfied with a differentiable storage function and the function

$$\frac{2Nk}{k-M_1} \left( M(x_1) + K(x_2, q_3) \right) + 2R(x_1)$$

is a solution of the Hamilton-Jacobi inequality for the closed loop system. Finally, Theorem 2.7 implies that the closed loop system is asymptotically stable.

In the linear case the  $\mathcal{H}_{\infty}$  almost disturbance decoupling problem for  $\Sigma_2^{tr}$  is known to be solvable because the state space of  $\Sigma_2^{tr}$  is equal to the strongly controllable subspace (see [Wi 81], [Wi 82], [Tr 86]). For nonlinear systems such a characterization of the solvability of the singular  $\mathcal{H}_{\infty}$  problem by the solvability of the regular  $\mathcal{H}_{\infty}$  problem for  $\Sigma_1$  can not be derived.

Therefore there is no need to decompose the singular  $\mathcal{H}_{\infty}$  problem in this way. We could also replace solving the  $L_2$ -gain almost disturbance decoupling problem for  $\Sigma_2^{\text{tr}}$  with disturbances  $x_1$  and d, and outputs  $q_3$  by solving the same problem for the complete system  $\Sigma^{\text{tr}}$  with as disturbances only d, and outputs  $q_3$ . In this way there is no need for an extra assumption like Assumption 6, and Theorem 5.17 can be reformulated as follows.

**Theorem 5.21** Consider the system  $\Sigma$ . Suppose that for the extended system  $\Sigma_e$  the distributions  $S_e^*$  and  $S_e^* \cap \ker$  dh are constant dimensional. Assume there exists for some  $\overline{\gamma} < \gamma$  a solution  $V \ge 0$  to the Hamilton-Jacobi inequality (5.32) with V(0) = 0.

Then there exists a constant  $k^*$  such that the nonlinear state feedback  $L_2$ -gain control problem with constant  $\overline{\gamma} < \gamma$  is solvable for  $\Sigma$  by the feedback

$$u_1 = \hat{\beta}(x_1, x_2, x_3, k) = \beta(x_1, x_2, x_3 - x_3^*(x_1, V_{x_1}^T(x_1)), k)$$
  

$$u_2 = u_2^*(x_1, V_{x_1}^T(x_1))$$

for  $k > k^*$  if the parameterized feedback

$$u_1 = \beta(x_1, x_2, q_3, k) \tag{5.41}$$

solves the  $L_2$ -gain almost disturbance decoupling problem with constant  $\frac{1}{k}$  for  $\Sigma^{\text{tr}}$ .

**Proof** As in the proof of Theorem 5.17 the assumptions make it possible to apply the described state transformations resulting in the interconnection of the systems  $\Sigma_1^{\text{tr}}$  and  $\Sigma_2^{\text{tr}}$ . From the solvability of the almost disturbance decoupling problem we have that for the closed loop system  $\Sigma^{\text{tr}}$ , (5.41) there exists a constant  $K(x_1, x_2, q_3)$ ,  $0 \le K(x_1, x_2, q_3) < \infty$ , K(0, 0, 0) = 0 such that

$$\int_0^t \|q_3(\tau)\|^2 \mathrm{d}\tau \le \frac{1}{k} \int_0^t \|d(\tau)\|^2 \mathrm{d}\tau + K(x_1(0), x_2(0), q_3(0))$$
(5.42)

for all  $d \in L_2(0, t)$  and all t. Let  $z_{q_3}$  denote the output of the subsystem  $\Sigma_1^{\text{tr}}$  resulting from disturbance d, initial state  $x_1(0)$  and input  $q_3$ .

Then from Lemma 5.18 it follows that there exist constants  $N, \varepsilon > 0$ , and a constant  $R(x_1), 0 \le R(x_1) < \infty, R(0) = 0$  such that locally

$$\int_0^t \|z_{q_3}(\tau)\|^2 \mathrm{d}\tau \le N \int_0^t \|q_3(\tau)\|^2 \mathrm{d}\tau + (\gamma^2 - \varepsilon) \int_0^t \|d(\tau)\|^2 \mathrm{d}\tau + R(x_1(0)).$$
(5.43)

Combining the inequalities (5.42) and (5.43), results in

$$\int_{0}^{t} \|z_{q_{3}}(\tau)\|^{2} \mathrm{d}\tau \leq \left(\frac{N}{k} + \gamma^{2} - \varepsilon\right) \int_{0}^{t} \|d(\tau)\|^{2} \mathrm{d}\tau + NK(x_{1}(0), x_{2}(0), q_{3}(0)) + R(x_{1}(0)).$$
(5.44)

Therefore if we choose

$$k^* \geq \frac{N}{\varepsilon}$$

then the  $L_2$ -gain from d to  $z_{q_3}$  is less than or equal to  $\gamma$ .

To solve the state feedback  $\mathcal{H}_{\!\infty}$  problem we can give the following sufficient conditions

**Theorem 5.22** Assume the assumptions in Theorem 5.21 are satisfied. Then there exists a constant  $k^*$  such that the nonlinear state feedback  $\mathcal{H}_{\infty}$  control problem with constant  $\overline{\gamma} < \gamma$  is solvable for  $\Sigma$  by the feedback

$$u_1 = \tilde{\beta}(x_1, x_2, x_3, k) = \beta(x_1, x_2, x_3 - x_3^*(x_1, V_{x_1}^T(x_1)), k)$$
  

$$u_2 = u_2^*(x_1, V_{x_1}^T(x_1))$$

for  $k > k^*$  if the parameterized feedback

$$u_1 = \beta(x_1, x_2, q_3, k)$$

solves the  $\mathcal{H}_{\infty}$  almost disturbance decoupling problem with constant  $\frac{1}{k}$  for  $\Sigma^{\text{tr}}$ .

**Proof** The  $L_2$ -gain result follows from Theorem 5.21 and closed loop stability is implied by the solvability of the  $\mathcal{H}_{\infty}$  almost disturbance decoupling problem for the complete system.

#### 5.2.5 The regular $\mathcal{H}_{\infty}$ subproblem

In the previous subsection only sufficient conditions for the solvability of the singular  $\mathcal{H}_{\infty}$  problem were given. It is interesting to look whether these conditions are also necessary. We will prove under an extra regularity assumption that the solvability of the regular  $\mathcal{H}_{\infty}$  subproblem for the subsystem  $\Sigma_1$  is indeed necessary. Again we assume that the distributions  $\mathcal{S}_e^*$  and  $\mathcal{S}_e^* \cap \ker dh$  are constant dimensional, that Assumption 5 is satisfied and that the conditions (5.24) and (5.25) hold in order to be able to define the state and input transformation defined in Subsection 5.2.2.

Suppose that there exists a feedback

$$u = \begin{pmatrix} u_1 \\ u_2 \end{pmatrix} = \begin{pmatrix} l_1(x_1, x_2, x_3) \\ l_2(x_1, x_2, x_3) \end{pmatrix} = l(x)$$
(5.45)

which solves the state feedback  $L_2$ -gain control problem with constant  $\gamma$  for the system  $\Sigma$ , with a differentiable, non-negative storage function V. Then this function V is a solution to the Hamilton-Jacobi inequality

$$V_{x_{1}}(x) f_{1}(x_{1}, x_{3}, l_{2}(x), d) + V_{x_{2}}(x) f_{2}(x_{1}, x_{2}, x_{3}, l_{1}(x), l_{2}(x), d) + V_{x_{3}}(x) f_{3}(x_{1}, x_{2}, x_{3}, l_{1}(x), l_{2}(x), d)$$
(5.46)  
$$+ \frac{1}{2} h^{T}(x_{1}, x_{3}, l_{2}(x)) h(x_{1}, x_{3}, l_{2}(x)) - \frac{1}{2} \gamma^{2} d^{T} d \leq 0$$

for all x and d. Then we look at solutions to the next two equations

$$V_{x_2}(x_1, x_2, x_3) = 0,$$
  $V_{x_3}(x_1, x_2, x_3) = 0.$ 

Assume there exists a differentiable solution to these equalities given by

$$\begin{pmatrix} x_2 \\ x_3 \end{pmatrix} = \begin{pmatrix} F_2(x_1) \\ F_3(x_1) \end{pmatrix} =: F(x_1), \qquad F(0) = 0.$$

**Remark 5.23** If the Hessian of V with respect to  $x_2$  and  $x_3$  is non-singular at the origin then at least locally existence and uniqueness of such a solution is assured by the Implicit Function Theorem.

**Theorem 5.24** Let  $\gamma > 0$ . Suppose that the distributions  $S_e^*$  and  $S_e^* \cap \ker dh$  are constant dimensional, Assumption 5 is satisfied and that both the conditions (5.24) and (5.25) hold. Suppose there exists a feedback (5.45) which solves the

state feedback  $L_2$ -gain control problem for the system  $\Sigma$ , with constant  $\gamma$  and a differentiable storage function  $V \ge 0$ . Additionally assume that the equations

$$V_{x_2}(x_1, x_2, x_3) = 0,$$
  $V_{x_3}(x_1, x_2, x_3) = 0$  (5.47)

have a differentiable solution

$$\begin{pmatrix} x_2 \\ x_3 \end{pmatrix} = \begin{pmatrix} F_2(x_1) \\ F_3(x_1) \end{pmatrix} = F(x_1), \qquad F(0) = 0.$$

Then the feedback

$$\begin{array}{rcl} x_3 &=& F_3(x_1) \\ u_2 &=& l_2(x_1, F_2(x_1), F_3(x_1)) \end{array} (5.48)$$

solves the state feedback  $L_2$ -gain control problem, with constant  $\gamma$ , for the system  $\Sigma_1$  with a differentiable storage function given by

$$P(x_1) = V(x_1, F_2(x_1), F_3(x_1)).$$

**Proof** Substitution of  $x_2 = F_2(x_1)$  and  $x_3 = F_3(x_1)$  in the inequality (5.46), using the equations (5.47), leads to the following inequality for  $P(x_1)$ :

$$P_{x_1}(x_1)f_1(x_1, x_3, u_2, d) + \frac{1}{2}h^T(x_1, x_3, u_2)h(x_1, x_3, u_2) - \frac{1}{2}\frac{1}{\gamma^2}d^Td \le 0$$
(5.49)

for all  $x_1$ , d, when  $x_3$  and  $u_2$  are given by (5.48).

This means that the feedback (5.48) solves the state feedback  $L_2$ -gain control problem, with constant  $\gamma$ , for  $\Sigma_1$  with storage function *P* (see Theorem 2.2).

A similar result can be proved for the solvability of the state feedback  $\mathcal{H}_{\infty}$  control problem.

**Theorem 5.25** Let  $\gamma > 0$ . Suppose that the distributions  $S_e^*$  and  $S_e^* \cap \ker dh$  are constant dimensional, Assumption 5 is satisfied and that both the conditions (5.24) and (5.25) hold. Suppose there exists a feedback (5.45) which solves the state feedback  $\mathcal{H}_{\infty}$  control problem for the system  $\Sigma$ , with constant  $\gamma$  and a

differentiable storage function V(x) > 0 ( $x \neq 0$ ). Additionally assume that the equations

$$V_{x_2}(x_1, x_2, x_3) = 0,$$
  $V_{x_3}(x_1, x_2, x_3) = 0$  (5.50)

have a differentiable solution

$$\begin{pmatrix} x_2 \\ x_3 \end{pmatrix} = \begin{pmatrix} F_2(x_1) \\ F_3(x_1) \end{pmatrix} = F(x_1), \qquad F(0) = 0.$$

Then the feedback (5.48) solves the state feedback  $\mathcal{H}_{\infty}$  control problem, with constant  $\gamma$ , for the system  $\Sigma_1$  with a differentiable storage function given by

$$P(x_1) = V(x_1, F_2(x_1), F_3(x_1)).$$

**Proof** The  $L_2$ -gain result follows from Theorem 5.24. Because V(x) > 0 for  $x \neq 0$  it follows that  $P(x_1) > 0$  for  $x_1 \neq 0$  which implies that the closed-loop system  $\Sigma_1$ , (5.48) is locally asymptotically stable.

The saddle point solution of the pre-Hamiltonian  $K_{\gamma}(x_1, p_1, d, x_3, u_2)$  corresponding to the  $L_2$ -gain problem for the subsystem  $\Sigma_1$  is  $x_3^* = x_3^*(x_1, p), u_2^* = u_2^*(x_1, p)$  and  $d^* = d^*(x_1, p)$ . This solution satisfies the saddle point condition

$$K_{\gamma}(x_1, p, d, x_3^*, u_2^*) \le K_{\gamma}(x_1, p, d^*, x_3^*, u_2^*) \le K_{\gamma}(x_1, p, d^*, x_3, u_2)$$

for every  $(x_1, p, d, x_3, u_2)$ . For the specific case that p is equal to  $P_{x_1}(x_1)$  and  $x_3$  equal to  $F_3(x_1)$  it follows from the right inequality of the saddle point condition that also the feedback

$$x_{3} = x_{3}^{*}(x_{1}, P_{x_{1}}^{I}(x_{1}))$$
  

$$u_{2} = u_{2}^{*}(x_{1}, P_{x_{1}}^{T}(x_{1}))$$
(5.51)

solves the state feedback  $L_2$ -gain control problem, with constant  $\gamma$ , for the system  $\Sigma_1$ . Summarizing:

**Theorem 5.26** Assume all conditions of Theorem 5.24 are satisfied. Then the feedback (5.51) solves the state feedback  $L_2$ -gain control problem, with constant  $\gamma$ , for the system  $\Sigma_1$  with differentiable storage function

$$P(x_1) = V(x_1, F_2(x_1), F_3(x_1)).$$

**Proof** The proof follows by using the saddle point condition as explained above, together with the inequality (5.49) derived from Theorem 5.24.

Until now little is known about the structure of the feedback solution (5.45) to the singular  $L_2$ -gain control problem. The following result gives, under the worst case disturbance, some idea of the structure of this feedback.

**Theorem 5.27** Assume all conditions of Theorem 5.24 are satisfied. Additionally assume that:

$$F_3(x_1) = x_3^*(x_1, P_{x_1}^T(x_1));$$
  
$$l_2(x_1, F_2(x_1), F_3(x_1)) = u_2^*(x_1, P_{x_1}^T(x_1)).$$

Furthermore assume that (5.46) holds with equality along  $(x_1, F_2(x_1), F_3(x_1))$ and that the Hessian of V with respect to  $x_2$  and  $x_3$  is non-singular. Then the feedback (5.45) satisfies the following equality on  $(x_1, F_2(x_1), F_3(x_1))$ 

$$\begin{pmatrix} F_{2x_1}(x) \\ F_{3x_1}(x) \end{pmatrix} f_1(x_1, x_3, l_2(x), d_{\max}(x)) = \begin{pmatrix} f_2(x_1, x_2, x_3, l_1(x), l_2(x), d_{\max}(x)) \\ f_3(x_1, x_2, x_3, l_1(x), l_2(x), d_{\max}(x)) \end{pmatrix}$$
(5.52)

where  $d_{\max}(x)$  is the worst case disturbance corresponding to the inequality (5.46), given by

$$d_{\max}(x) = \arg \max_{d} \left\{ \frac{1}{2} h^{T}(x_{1}, x_{3}, l_{2}(x))h(x_{1}, x_{3}, l_{2}(x)) + V_{x_{1}}(x)f_{1}(x_{1}, x_{3}, l_{2}(x), d) + V_{x_{2}}(x)f_{2}(x_{1}, x_{2}, x_{3}, l_{1}(x), l_{2}(x), d) + V_{x_{3}}(x)f_{3}(x_{1}, x_{2}, x_{3}, l_{1}(x), l_{2}(x), d) - \frac{1}{2}\gamma^{2}d^{T}d \right\}.$$

**Proof** (5.46) can be written as

$$V_{x_{1}}f_{1}(x_{1}, x_{3}, l_{2}, d_{\max}) + V_{x_{2}}f_{2}(x_{1}, x_{2}, x_{3}, l_{1}, l_{2}, d_{\max}) + V_{x_{3}}f_{3}(x_{1}, x_{2}, x_{3}, l_{1}, l_{2}, d_{\max}) + (5.53)$$

$$\frac{1}{2}h^{T}(x_{1}, x_{3}, l_{2})h(x_{1}, x_{3}, l_{2}) - \frac{1}{2}\gamma^{2}d_{\max}^{T}d_{\max} \leq 0$$

with equality along  $(x_1, F_2(x_1), F_3(x_1))$ . Differentiation with respect to  $x_2$  of the left side of this inequality leads to

$$V_{x_{1}x_{2}}f_{1} + V_{x_{1}}\frac{\partial f_{1}}{\partial u_{2}}\frac{\partial l_{2}}{\partial x_{2}} + V_{x_{1}}\frac{\partial f_{1}}{\partial d}\frac{\partial d_{\max}}{\partial x_{2}} + V_{x_{2}x_{2}}f_{2} + V_{x_{2}}\frac{\partial f_{2}}{\partial x_{2}} + V_{x_{2}}\frac{\partial f_{2}}{\partial u_{1}}\frac{\partial l_{1}}{\partial x_{2}} + V_{x_{2}}\frac{\partial f_{2}}{\partial u_{2}}\frac{\partial l_{2}}{\partial x_{2}} + V_{x_{2}}\frac{\partial f_{2}}{\partial d}\frac{\partial d_{\max}}{\partial x_{2}} + V_{x_{3}x_{2}}f_{3} + V_{x_{3}}\frac{\partial f_{3}}{\partial x_{2}} + V_{x_{3}}\frac{\partial f_{3}}{\partial u_{1}}\frac{\partial l_{1}}{\partial x_{2}} + V_{x_{3}}\frac{\partial f_{3}}{\partial u_{2}}\frac{\partial l_{2}}{\partial x_{2}} + V_{x_{3}}\frac{\partial f_{3}}{\partial d}\frac{\partial d_{\max}}{\partial x_{2}} + \frac{\partial d_{2}}{\partial x_{2}}\left(\frac{1}{2}h^{T}h\right) - \frac{\partial d_{2}}{\partial x_{2}}\left(\frac{1}{2}\gamma^{2}d_{\max}^{T}d_{\max}\right).$$

Hence along  $(x_1, F_2(x_1), F_3(x_1))$ , using the maximizing property of  $d_{\max}$ , and the minimizing property of  $l_2 (= u_2^*)$ , it follows that

$$V_{x_1x_2}f_1 + V_{x_2x_2}f_2 + V_{x_3x_2}f_3 = 0.$$

The same holds for differentiation with respect to  $x_3$  along  $(x_1, F_2(x_1), F_3(x_1))$ , where we use apart from the maximizing respectively minimizing properties of  $d_{\text{max}}$  and  $u_2^*$  also the minimizing property of  $F_3$  (=  $x_3^*$ ). Therefore along  $(x_1, F_2(x_1), F_3(x_1))$  also

$$V_{x_1x_3}f_1 + V_{x_2x_3}f_2 + V_{x_3x_3}f_3 = 0.$$

Furthermore the equalities

$$V_{x_2}(x_1, F_2(x_1), F_3(x_1)) = 0,$$
  
$$V_{x_3}(x_1, F_2(x_1), F_3(x_1)) = 0$$

hold and differentiation of these equalities along  $(x_1, F_2(x_1), F_3(x_1))$  with respect to  $x_1$  leads to

$$V_{x_1x_2} + V_{x_2x_2}F_{2x_1} + V_{x_3x_2}F_{3x_1} = 0,$$
  
$$V_{x_1x_3} + V_{x_2x_3}F_{2x_1} + V_{x_3x_3}F_{3x_1} = 0.$$

Hence in vector notation we have along  $(x_1, F_2(x_1), F_3(x_1))$ 

$$\begin{pmatrix} V_{x_1x_2} \\ V_{x_1x_3} \end{pmatrix} + \begin{pmatrix} V_{x_2x_2} & V_{x_3x_2} \\ V_{x_2x_3} & V_{x_3x_3} \end{pmatrix} \begin{pmatrix} F_{2x_1} \\ F_{3x_1} \end{pmatrix} = 0,$$
$$\begin{pmatrix} V_{x_1x_2} \\ V_{x_1x_3} \end{pmatrix} f_1 + \begin{pmatrix} V_{x_2x_2} & V_{x_3x_2} \\ V_{x_2x_3} & V_{x_3x_3} \end{pmatrix} \begin{pmatrix} f_2 \\ f_3 \end{pmatrix} = 0.$$

It follows from the non-singularity of the Hessian of V with respect to  $x_2$  and  $x_3$  that along  $(x_1, F_2(x_1), F_3(x_1))$ 

$$\begin{pmatrix} F_{2x_1}(x) \\ F_{3x_1}(x) \end{pmatrix} f_1(x_1, x_3, l_2(x), d_{\max}) = \begin{pmatrix} f_2(x_1, x_2, x_3, l_1(x), l_2(x), d_{\max}) \\ f_3(x_1, x_2, x_3, l_1(x), l_2(x), d_{\max}) \end{pmatrix}$$

Remark 5.28 Condition (5.52) is equivalent to the submanifold

$$S = \{(x_1, x_2, x_3) \in M | x_2 = F_2(x_1), x_3 = F_3(x_1)\}$$

being invariant under the dynamics

$$\dot{x}_1 = f_1(x_1, x_3, l_2, d_{\max}), \dot{x}_2 = f_2(x_1, x_2, x_3, l_1, l_2, d_{\max}), \dot{x}_3 = f_3(x_1, x_2, x_3, l_1, l_2, d_{\max}).$$

#### 5.2.6 The $\mathcal{H}_{\infty}$ almost disturbance decoupling problem

In this subsection we will restrict ourself to systems of a special form. One of the important restrictions is the triangular structure of the state dependency of the disturbance vector fields. These restrictions are imposed in order to be able to apply the sufficiency results for the solvability of the almost disturbance decoupling problem derived in Chapter 4.

For linear systems necessary and sufficient conditions have been derived for the solvability of the almost disturbance decoupling problem (see [Tr 86], [Wi 81], [Wi 82]). Until now geometric necessary and sufficient conditions for the solvability of the almost disturbance decoupling problem for nonlinear systems have not been derived, and in fact it seems unlikely that this is possible. As a consequence it is not possible to derive necessary and sufficient conditions for the solvability of the singular  $\mathcal{H}_{\infty}$  problem for general nonlinear systems similar to the linear case.

In this subsection we consider systems of the following form

$$\dot{\xi}_{11} = \xi_{12} + W_1^T(\tilde{\xi}_1, \xi_r, u_r)d,$$
  
:

$$\begin{split} \dot{\xi}_{1\rho_{1}} &= u_{1} + W_{1\rho_{1}}^{T}(\tilde{\xi}_{1}, \dots, \tilde{\xi}_{\rho_{1}}, \xi_{r}, u_{r})d, \\ \dot{\xi}_{21} &= \xi_{22} + W_{21}^{T}(\tilde{\xi}_{1}, \xi_{r}, u_{r})d, \\ \vdots \\ \dot{\xi}_{2\rho_{2}} &= u_{2} + W_{2\rho_{2}}^{T}(\tilde{\xi}_{1}, \dots, \tilde{\xi}_{\rho_{2}}, \xi_{r}, u_{r})d, \\ \vdots \\ \dot{\xi}_{k1} &= \xi_{k2} + W_{k1}^{T}(\tilde{\xi}_{1}, \xi_{r}, u_{r})d, \\ \vdots \\ \dot{\xi}_{k\rho_{k}} &= u_{k} + W_{k\rho_{k}}^{T}(\tilde{\xi}_{1}, \dots, \tilde{\xi}_{\rho_{k}}, \xi_{r}, u_{r})d, \\ \dot{\xi}_{r} &= f(\tilde{\xi}_{1}, \xi_{r}, u_{r}) + e(\tilde{\xi}_{1}, \xi_{r}, u_{r})d, \\ z &= h(\tilde{\xi}_{1}, \xi_{r}, u_{r}) \end{split}$$

where  $\rho = \sum_{i=1}^{k} \rho_i$  and  $\tilde{\xi}_j$  indicates the *j*-th state components of every block

$$\tilde{\xi}_j = \left\{ \xi_{lj}, \text{ for all } l = 1, \dots, k \text{ for which } j \le \rho_l \right\}.$$

So  $\tilde{\xi}_1 = (\xi_{11}, \xi_{21}, \dots, \xi_{k1})$  etc..

The state of the system (5.54) is given by  $(\tilde{\xi}_1, \ldots, \tilde{\xi}_k, \xi_r)$  and the inputs are  $(u_1, \ldots, u_k, u_r)$ . Furthermore the function *h* is assumed to be an injective mapping from  $(\tilde{\xi}_1, u_r)$  to *z*. For the extended version of the system (5.54) the minimal conditioned invariant distribution containing the input vector fields can be calculated. It is left to the reader to prove that indeed the distributions are projectable and that the projected distributions  $\chi_1^P, \chi_2^P$  and  $\chi_3^P$  are given by:

$$\begin{aligned} \mathcal{X}_1^P &= \operatorname{span}\left\{\frac{\partial}{\partial\xi_r}\right\}; \\ \mathcal{X}_2^P &= \operatorname{span}\left\{\frac{\partial}{\partial\tilde{\xi}_2}, \dots, \frac{\partial}{\partial\tilde{\xi}_{\max_{i=1}^k \rho_k}}\right\}; \\ \mathcal{X}_3^P &= \operatorname{span}\left\{\frac{\partial}{\partial\tilde{\xi}_1}\right\}. \end{aligned}$$

Thus the system (5.54) can be seen as an interconnection of two systems. The first subsystem  $\Sigma_1$  has states  $\xi_r$ , inputs  $(\tilde{\xi}_1, u_r, d)$  and outputs z. This subsystem is connected with a second subsystem, denoted by  $\Sigma_2$ , which has states  $(\tilde{\xi}_1, \ldots, \tilde{\xi}_k)$ , inputs  $(u_1, \ldots, u_k, u_r, d)$  and outputs  $\tilde{\xi}$  (see Figure 5.6).



Figure 5.6: system (5.54)

For the system (5.54) we can prove the following  $L_2$ -gain result (see also Theorem 5.21).

**Theorem 5.29** Let  $\gamma > 0$ . Assume there exists a smooth feedback

$$\xi_1 = \alpha(\xi_r)$$
  
 $u_r = \beta(\xi_r)$ 

that solves the state feedback  $L_2$ -gain control problem, with constant less than  $\gamma$ , for the subsystem  $\Sigma_1$  Additionally assume that the vector field given by the vector

$$\begin{pmatrix} \xi_{12} & \cdots & \xi_{1\rho_1} & L_f^{\rho_1}\alpha_1(\xi_r) & \xi_{22} & \cdots & \xi_{2\rho_2} & L_f^{\rho_2}\alpha_2(\xi_r) & \cdots \\ & \cdots & f^T(\tilde{\xi}_1, \xi_r, \beta(\xi_r)) \end{pmatrix}^T$$

is complete. Then the state feedback  $L_2$ -gain control problem for the system (5.54) is solvable, i.e., there exists a feedback such that the closed loop system consisting of the system (5.54) together with the feedback has  $L_2$ -gain less than  $\gamma$ .

Conversely, suppose that the feedback

$$u = \begin{pmatrix} u_{1} \\ \vdots \\ u_{k} \\ u_{r} \end{pmatrix} = \begin{pmatrix} l_{1}(\xi_{11}, \dots, \xi_{1\rho_{1}}, \dots, \xi_{k1}, \dots, \xi_{k\rho_{k}}, \xi_{r}) \\ \vdots \\ l_{k}(\xi_{11}, \dots, \xi_{1\rho_{1}}, \dots, \xi_{k1}, \dots, \xi_{k\rho_{k}}, \xi_{r}) \\ l_{r}(\xi_{11}, \dots, \xi_{1\rho_{1}}, \dots, \xi_{k1}, \dots, \xi_{k\rho_{k}}, \xi_{r}) \end{pmatrix}$$

solves the state feedback  $L_2$ -gain control problem, with constant less than  $\gamma$ , for the system (5.54) with a differentiable storage function  $V \ge 0$ . Additionally

assume that the equations

$$V_{\xi_{lj}}(\xi_{11},\ldots,\xi_{1\rho_1},\ldots,\xi_{k1},\ldots,\xi_{k\rho_k},\xi_r)=0, \qquad l=1,\ldots,k, \ j=1,\ldots,\rho_l$$

have a differentiable solution

$$\begin{pmatrix} \xi_{11} \\ \vdots \\ \xi_{1\rho_{1}} \\ \vdots \\ \xi_{k1} \\ \vdots \\ \xi_{k\rho_{k}} \end{pmatrix} = \begin{pmatrix} F_{11}(\xi_{r}) \\ \vdots \\ F_{1\rho_{1}}(\xi_{r}) \\ \vdots \\ F_{k1}(\xi_{r}) \\ \vdots \\ F_{k\rho_{k}}(\xi_{r}) \end{pmatrix} =: F(\xi_{r}), \qquad F(0) = 0.$$

Then the state feedback  $L_2$ -gain control problem, with constant less than  $\gamma$ , for the subsystem  $\Sigma_1$  is solvable with a differentiable storage function.

**Proof** Choose the inputs  $u_r$  equal to the solution of the state feedback problem of the regular subsystem  $\Sigma_1$ 

$$u_r = \beta(\xi_r),$$

and define the new output equation

$$\tilde{h} = \tilde{\xi}_1 - \alpha(\xi_r).$$

The system consisting of the dynamics described in (5.54) together with this new output equation and the inputs  $u_r$  chosen as before satisfies the conditions of Theorem 4.9. Therefore the  $L_2$ -gain almost disturbance decoupling problem is solvable for this system. Finally it follows from Theorem 5.21 that the singular  $L_2$ -gain problem for the system (5.54) is solvable.

The second part of the theorem follows from Theorem 5.24.

Thus for systems of the form (5.54) the solvability of the singular  $L_2$ -gain problem is characterized by the solvability of the regular  $\mathcal{H}_{\infty}$  problem for the subsystem  $\Sigma_1$ .

For stability results extra assumptions should be imposed; similar to the ones used for Theorem 5.22, Theorem 4.10 and Theorem 5.25. This is left to the reader.

#### 5.2.7 Factorization approach

As described in Subsection 5.1.5 there exists a second way to derive conditions for the solvability of the singular  $\mathcal{H}_{\infty}$  problem. Also for nonlinear systems this seems to be an elegant way to derive at least sufficient conditions for the solvability of the singular  $\mathcal{H}_{\infty}$  problem.

To be able to extend the linear ideas to nonlinear systems we restrict the class of systems to nonlinear systems that are affine in the disturbances

$$\Sigma \begin{cases} \dot{x} = f(x, u) + e(x, u)d \\ z = h(x, u) \end{cases}$$

The pre-Hamiltonian corresponding to the  $\mathcal{H}_{\infty}$  problem, with  $L_2$ -gain less than or equal to  $\gamma$ , for the system  $\Sigma$  is given by

$$K_{\gamma}(x, p, u, d) = p^{T} \left( f(x, u) + e(x, u)d \right) + \frac{1}{2}h^{T}(x, u)h(x, u) - \frac{1}{2}\gamma^{2}d^{T}d.$$

The maximizing disturbance with respect to this pre-Hamiltonian is

$$d_{\max}(x, p, u) = \arg\max_{d} K_{\gamma}(x, p, u, d) = \frac{1}{\gamma^2} e^T(x, u) p.$$

As an extension of the quadratic matrix inequality used in the linear theory we introduce the *nonlinear dissipation inequality* defined as

$$F_{\gamma}(x, V(x), u) = K_{\gamma}(x, V_{x}^{T}(x), d_{\max}(x, V_{x}^{T}(x), u), u)$$
  
$$= V_{x}(x) f(x, u) + \frac{1}{2} \frac{1}{\gamma^{2}} V_{x}(x) e(x, u) e^{T}(x, u) V_{x}^{T}(x)$$
  
$$+ \frac{1}{2} h^{T}(x, u) h(x, u) \ge 0.$$
(5.55)

**Remark 5.30** The relation of the nonlinear dissipation inequality with the quadratic matrix inequality can be clarified by considering the nonlinear dissipation inequality for nonlinear systems which are not only affine in the disturbances but also in the inputs

$$\dot{x} = f(x) + g(x)u + e(x)d z = h(x) + k(x)u (5.56)$$

Then the inequality (5.55) amounts to

$$F_{\gamma}(x, V(x), u) = V_{x}(x) (f(x) + g(x)u) + \frac{1}{2} \frac{1}{\gamma^{2}} V_{x}(x) e(x) e^{T}(x) V_{x}^{T}(x) + \frac{1}{2} (h(x) + k(x)u)^{T} (h(x) + k(x)u) \ge 0$$

which can equivalently be described as the positive semi-definiteness of the matrix

$$\begin{pmatrix} 2V_x(x)f(x) + \frac{1}{\gamma^2}V_x(x)e(x)e^T(x)V_x^T(x) + h^T(x)h(x) & V_x(x)g(x) + h^T(x)k(x) \\ g^T(x)V_x^T(x) + k^T(x)h(x) & k^T(x)k(x) \end{pmatrix}$$

This is clearly equivalent to the quadratic matrix inequality condition considered for linear systems in Subsection 5.1.3.  $\hfill \Box$ 

As in Subsection 5.1.5 we define an auxiliary system based on a solution  $V \ge 0$ , V(0) = 0, to the nonlinear dissipation inequality as

$$\Sigma_{P} \begin{cases} \dot{x} = \left( f(x, u) + \frac{1}{\gamma^{2}} e(x, u) e^{T}(x, u) V_{x}^{T}(x) \right) + e(x, u) d_{p} \\ z_{p} = h_{p}(x, u) \end{cases}$$
(5.57)

where the new output equation is defined by

$$\frac{1}{2}h_p^T(x, u)h_p(x, u) = F_{\gamma}(x, V(x), u) = V_x(x)f(x, u) + \frac{1}{2}\frac{1}{\gamma^2}V_x(x)e(x, u)e^T(x, u)V_x^T(x) + \frac{1}{2}h^T(x, u)h(x, u)$$

and  $d_p := d - \frac{1}{\gamma^2} e^T(x, u) V_x^T(x)$ . Of course we want the new output equation to be sufficiently smooth. A sufficient condition for the local existence of a smooth  $h_p$  is provided by a generalized version of Morse's Lemma (see for Morse's Lemma for instance [JJT 86]).

**Lemma 5.31** Let f be a smooth function from  $\mathbb{R}^n$  to  $\mathbb{R}$  with f(0) = 0,  $\frac{\partial f}{\partial x}(0) = 0$ , rank  $\frac{\partial^2 f}{(\partial x)^2} = k \le n$  around the origin and all eigenvalues are nonnegative. Then there exists local coordinates  $q_1, \ldots, q_n$  such that in the new coordinates f can be written as

$$f(q) = q_1^2 + \dots + q_k^2.$$

**Proof** First we take the Taylor expansion of f around the origin

$$f(x_1,\ldots,x_n) = \frac{1}{2} \sum_{i,j} x_i x_j \frac{\partial^2 f}{\partial x_i \partial x_j}(0) + \frac{1}{6} \sum_{i,j,l} x_i x_j x_l \frac{\partial^3 f}{\partial x_i \partial x_j \partial x_l}(0) + \ldots$$

Since rank  $\frac{\partial^2 f}{\partial x^2}(0) = k$  we can find, by a linear coordinate transform, new coordinates  $v_1, \ldots, v_n$  such that

$$f(\nu_1,\ldots,\nu_n)=\nu_1^2+\cdots+\nu_k^2+\varphi(\nu_1,\ldots,\nu_n)$$

where  $\varphi$  consists of third order terms and higher of  $\nu_1, \ldots, \nu_n$ , that is  $\varphi(0) = 0$ ,  $\frac{\partial \varphi}{\partial \nu}(0) = 0$ ,  $\frac{\partial^2 \varphi}{\partial \nu^2}(0) = 0$ . Hence

$$\frac{\partial^2 f}{\partial \nu^2}(\nu) = 2 \begin{pmatrix} I_k & 0\\ 0 & 0 \end{pmatrix} + \frac{\partial^2 \varphi}{\partial \nu^2}(\nu)$$

around 0 and hence, since rank  $\frac{\partial^2 f}{\partial x^2}(x) = k$  around the origin

$$\frac{\partial^2 \varphi}{\partial \nu_i \partial \nu_j}(\nu) = 0, \qquad \qquad i = k+1, \dots, n, \quad j = 1, \dots, n$$

around the origin.

Then

$$\frac{\partial \varphi}{\partial \nu_i}(\nu_1,\ldots,\nu_n) = \int_0^1 \sum_{j=1}^n \frac{\partial^2 \varphi}{\partial \nu_i \partial \nu_j}(t\nu_1,\ldots,t\nu_n)\nu_j dt = 0, \qquad i = k+1,\ldots,n$$

and thus  $\varphi$  does not depend on  $\nu_{k+1}, \ldots, \nu_n$  around 0. Hence

$$f(\nu_1,\ldots,\nu_n)=\nu_1^2+\cdots+\nu_k^2+\varphi(\nu_1,\ldots,\nu_k)$$

and f only depends on  $v_1, \ldots, v_k$ .

Now we can apply Morse's Lemma (see e.g. [JJT 86]) to the function f (seen as a function on  $\mathbb{R}^k$ , with rank of its Hessian equal to k), and we can find coordinates  $q_1, \ldots, q_k$ , and  $q_{k+1} = v_{k+1}, \ldots, q_n = v_n$  such that

$$f(q_1,\ldots,q_n)=\sum_{i=1}^k q_i^2.$$

From Lemma 5.31 it follows that a sufficient condition for the local existence of a smooth output function  $h_p(x, u)$  is that the Hessian of the nonlinear dissipation inequality  $F_{\gamma}(x, V(x), u)$  with respect to x and u has constant rank around (x, u) = (0, 0). Indeed in that case the dimension of  $h_p(x, u)$  can be

chosen equal to the rank of the Hessian of the nonlinear dissipation inequality  $F_{\gamma}(x, V_x(x), u)$  where the components of  $h_p$  are chosen equal to the (x, u)coordinate representation of the corresponding q components.

The auxiliary system  $\Sigma_P$  will appear to be helpful for solving the  $\mathcal{H}_{\infty}$  problem for the system  $\Sigma$ . Similar to the linear case we obtain the following theorem relating the  $L_2$ -gains of  $\Sigma$  and  $\Sigma_P$ .

**Theorem 5.32** Suppose there exists a solution  $V \ge 0$  to the nonlinear matrix inequality  $F_{\gamma}(x, V(x), u)$ . Then for all d

$$\frac{1}{2} \int_0^t \left( \|z_p(\tau)\|^2 - \gamma^2 \|d_p(\tau)\|^2 \right) d\tau$$

$$= \frac{1}{2} \int_0^t \left( \|z(\tau)\|^2 - \gamma^2 \|d(\tau)\|^2 \right) d\tau + V(x(t)) - V(x(0)).$$

-

**Proof** This follows easily by writing out  $z_p$  and  $d_p$  in

$$\begin{aligned} \|z_p\|^2 - \gamma^2 \|d_p\|^2 &= h_p^T(x, u)h_p(x, u) - \frac{1}{\gamma^2} V_x(x)e(x, u)e^T(x, u)V_x^T(x) \\ &- 2V_x(x)e(x, u)d - \gamma^2 d^T d \\ &= 2V_x(x)(f(x, u) + e(x, u)d) \\ &+ h^T(x, u)h(x, u) - \gamma^2 \|d\|^2 \\ &= 2\frac{d}{dt}V(x) + \|z\|^2 - \gamma^2 \|d\|^2. \end{aligned}$$

Taking the integral from 0 to t leads to the desired result

$$\int_0^t \left( \|z_p(\tau)\|^2 - \gamma^2 \|d_p(\tau)\|^2 \right) d\tau$$
  
=  $2 \left( V(x(t)) - V(x(0)) \right) + \int_0^t \left( \|z(\tau)\|^2 - \gamma^2 \|d(\tau)\|^2 \right) d\tau.$ 

From Theorem 5.32 we can derive the following partial generalization of Theorem 5.8.

**Corollary 5.33** Suppose there exists a solution  $V \ge 0$ , V(0) = 0, to the nonlinear matrix inequality  $F_{\gamma}(x, V(x), u) \ge 0$  and there exists a feedback which solves the state feedback  $\mathcal{H}_{\infty}$  control problem, with constant  $\gamma$ , for the system  $\Sigma_P$  with storage function K(x). Then we have the following inequality for the system  $\Sigma$ 

$$\int_0^\infty \|z(\tau)\|^2 \mathrm{d}\tau \le \gamma^2 \int_0^\infty \|d(\tau)\| \mathrm{d}\tau + V(x(0)) + K(x(0))$$
  
for all  $d \in L(0,\infty)$ .

**Proof** The  $\mathcal{H}_{\infty}$  problem for the system  $\Sigma_P$  is solvable. So there exists a feedback such that for the closed loop system the following inequality holds

$$\frac{1}{2}\int_0^t \|z_p(\tau)\|^2 \mathrm{d}\tau \le \gamma^2 \frac{1}{2}\int_0^t \|d_p(\tau)\| \mathrm{d}\tau + K(x(0)).$$

Using Theorem 5.32 this gives

$$\int_0^t \left( \|z(\tau)\|^2 - \gamma^2 \|d(\tau)\|^2 \right) d\tau$$
  
=  $\int_0^t \left( \|z_p(\tau)\|^2 - \gamma^2 \|d_p(\tau)\|^2 \right) d\tau + 2V(x(0)) - 2V(x(t))$   
 $\leq 2V(x(0)) - 2V(x(t)) + 2K(x(0)).$ 

Because of the asymptotic stability of the closed loop system  $x(t) \rightarrow 0$  for  $t \rightarrow \infty$ . Thus if we take the limit in (5.58) for t tending to infinity then  $V(x(t)) \rightarrow 0$  because V(0) = 0 and we obtain

$$\int_0^\infty \|z(\tau)\|^2 \mathrm{d}\tau \le \gamma^2 \int_0^\infty \|d(\tau)\| \mathrm{d}\tau + V(x(0)) + K(x(0))$$
  
for all  $d \in L(0,\infty)$ .

Let us define the pre-Hamiltonian for the system  $\Sigma_P$ 

$$K_{\gamma}^{P}(x, p, d_{p}, u) = p^{T} \left( f(x, u) + \frac{1}{\gamma^{2}} e(x, u) e^{T}(x, u) V_{x}^{T}(x) + e(x, u) d_{p} \right) \\ + \frac{1}{2} h_{p}^{T}(x, u) h_{p}(x, u) - \frac{1}{2} \gamma^{2} d_{p}^{T} d_{p}.$$

With respect to the pre-Hamiltonians of the systems  $\Sigma$  and  $\Sigma_P$  a similar result as in Theorem 5.32 can be derived.

**Theorem 5.34** Suppose there exists a solution  $V \ge 0$  to the nonlinear matrix inequality  $F_{\gamma}(x, V(x), u)$ . Then

$$K_{\gamma}(x, p + V_x^T(x), d, u) = K_{\gamma}^P(x, p, d_p, u), \qquad \forall x, p, d, u.$$

**Proof** Again the result follows by substitution of the definitions of the new output function  $h_p$  and the new disturbance  $d_p$ :

$$\begin{split} K_{\gamma}(x, p + V_{x}^{T}(x), d, u) &= p^{T}\left(f(x, u) + e(x, u)d\right) + V_{x}(x)\left(f(x, u) + e(x, u)d\right) \\ &+ \frac{1}{2}h^{T}(x, u)h(x, u) - \frac{1}{2}\gamma^{2}d^{T}d \\ &= p^{T}\left(f(x, u) + e(x, u)d\right) + V_{x}(x)e(x, u)d + \frac{1}{2}h_{p}^{T}(x, u)h_{p}(x, u) \\ &- \frac{1}{2}\frac{1}{\gamma^{2}}V_{x}(x)e(x, u)e^{T}(x, u)V_{x}^{T}(x) - \frac{1}{2}\gamma^{2}d^{T}d \\ &= p^{T}\left(f(x, u) + \frac{1}{\gamma^{2}}e(x, u)e^{T}(x, u)V_{x}^{T}(x) + e(x, u)d_{p}\right) \\ &+ \frac{1}{2}h_{p}^{T}(x, u)h_{p}(x, u) - \frac{1}{2}\gamma^{2}d_{p}^{T}d_{p} \\ &= K_{\gamma}^{P}(x, p, d_{p}, u). \end{split}$$

From this theorem we can easily derive some interesting relations between the solvability of the  $L_2$ -gain control problems for the system  $\Sigma$  and  $\Sigma_P$ .

**Theorem 5.35** Suppose there exists a solution  $V \ge 0$  to the nonlinear dissipation inequality  $F_{\gamma}(x, V(x), u)$ . Additionally assume there exists a feedback u = l(x) for the system  $\Sigma_P$  such that the closed loop system has  $L_2$ -gain less than or equal to  $\gamma$  in the sense that there exists a solution  $W \ge 0$  to the dissipation inequality

$$K_{\nu}^{P}(x, W_{x}^{T}(x), d_{p}, l(x)) \leq 0$$

for all  $d_p$ . Then the same feedback applied to the system  $\Sigma$  leads to a closed loop system which has  $L_2$ -gain less than or equal to  $\gamma$  because W + V is a solution to

$$K_{\gamma}(x, (W+V)_x^T(x), d, l(x)) \leq 0$$

for all d.

**Proof** Use Theorem 5.34 with  $p = W_x^T(x)$  and u = l(x).

Also a converse result can be stated.

**Theorem 5.36** Assume there exists a solution  $V \ge 0$  to the Hamilton-Jacobi equality

$$K_{\gamma}(x, V_{x}^{T}(x), d_{\max}(x, V_{x}^{T}(x), u^{*}(x)), u^{*}(x))$$

$$= V_{x}(x) f(x, u^{*}(x)) + \frac{1}{2} \frac{1}{\gamma^{2}} V_{x}(x) e(x, u^{*}(x)) e^{T}(x, u^{*}(x)) V_{x}^{T}(x)$$

$$+ \frac{1}{2} h^{T}(x, u^{*}(x)) h(x, u^{*}(x)) = 0$$

where  $u^*(x)$  is defined by

$$u^{*}(x) = \arg\min_{u} K_{\gamma}\left(x, V_{x}^{T}(x), d_{\max}(x, V_{x}^{T}(x), u), u\right).$$
(5.59)

Then the feedback  $u = u^*(x)$  solves the state feedback  $L_2$ -gain control problem, with constant  $\gamma$ , for the system  $\Sigma$ . Furthermore  $V \ge 0$  is a solution to the nonlinear dissipation inequality and the zero function  $W \equiv 0$  is a solution to

$$K_{\gamma}^{P}(x, W_{x}^{T}(x), d_{p}, u^{*}(x)) \leq 0$$

for all  $d_p$ , with equality for  $d_p = 0$ .

**Proof** Use Theorem 5.34 with  $p = W_x^T(x) = 0$  and  $u = u^*(x)$ , where  $u^*(x)$  is defined by (5.59).

Note that in Theorem 5.36 we do not assume that  $u^*(x)$  is a saddle point solution of the pre-Hamiltonian corresponding to the  $L_2$ -gain control problem for the system  $\Sigma$ .

Theorem 5.35 can be used to simplify the  $L_2$ -gain problem for the system  $\Sigma$  after application of the state and input transformation as described in Subsection 5.2.2

$$\Sigma \begin{cases} \dot{x}_1 = f_1(x_1, x_3, u_2) + e_1(x_1, x_3, u_2)d \\ \dot{x}_2 = f_2(x_1, x_2, x_3, u_1, u_2) + e_2(x_1, x_2, x_3, u_1, u_2)d \\ \dot{x}_3 = f_3(x_1, x_2, x_3, u_1, u_2) + e_3(x_1, x_2, x_3, u_1, u_2)d \\ z = h(x_1, x_3, u_2) \end{cases}$$

We look at a solution  $V(x_1) \ge 0$  to the Hamilton Jacobi equality for the regular  $\mathcal{H}_{\infty}$  subproblem for the system  $\Sigma_1$  with a constant  $\gamma$ 

$$V_{x_{1}}(x_{1}) f_{1}(x_{1}, x_{3}^{*}, u_{2}^{*}) + \frac{1}{2} h^{T}(x_{1}, x_{3}^{*}, u_{2}^{*}) h(x_{1}, x_{3}^{*}, u_{2}^{*})$$

$$+ \frac{1}{2} \frac{1}{\gamma^{2}} V_{x_{1}}(x_{1}) e_{1}(x_{1}, x_{3}^{*}, u_{2}^{*}) e_{1}^{T}(x_{1}, x_{3}^{*}, u_{2}^{*}) V_{x_{1}}^{T}(x_{1}) = 0$$
(5.60)

where  $x_3^* = x_3^*(x_1, V_{x_1}^T(x_1))$  and  $u_2^* = u_2^*(x_1, V_{x_1}^T(x_1))$ . Because of the saddle point condition with respect to  $x_3^*$  and  $u_2^*$  this solution V, which only depends on the state  $x_1$  of the regular subsystem, is also a solution to the nonlinear dissipation inequality for the system  $\Sigma$ . This solution V to the nonlinear dissipation inequality will be used to define the new system  $\Sigma_P$ 

$$\Sigma_{P} \begin{cases} \dot{x}_{1} = f_{1}(x_{1}, x_{3}, u_{2}) + \frac{1}{\gamma^{2}}e_{1}(x_{1}, x_{3}, u_{2})e_{1}^{T}(x_{1}, x_{3}, u_{2})V_{x_{1}}^{T}(x_{1}) \\ +e_{1}(x_{1}, x_{3}, u_{2})d_{p} \\ \dot{x}_{2} = f_{2}(x_{1}, x_{2}, x_{3}, u_{1}, u_{2}) \\ +\frac{1}{\gamma^{2}}e_{2}(x_{1}, x_{2}, x_{3}, u_{1}, u_{2})e_{1}^{T}(x_{1}, x_{3}, u_{2})V_{x_{1}}^{T}(x_{1}) \\ +e_{2}(x_{1}, x_{2}, x_{3}, u_{1}, u_{2})d_{p} \\ \dot{x}_{3} = f_{3}(x_{1}, x_{2}, x_{3}, u_{1}, u_{2})d_{p} \\ \dot{x}_{3} = f_{3}(x_{1}, x_{2}, x_{3}, u_{1}, u_{2})e_{1}^{T}(x_{1}, x_{3}, u_{2})V_{x_{1}}^{T}(x_{1}) \\ +e_{3}(x_{1}, x_{2}, x_{3}, u_{1}, u_{2})d_{p} \\ z = h_{p}(x_{1}, x_{3}, u_{2}) \end{cases}$$

with  $d_p = d - \frac{1}{\gamma^2} e_1^T(x_1, x_3, u_2) V_{x_1}^T(x_1)$  and the new output function  $h_p$  being a solution to of

$$\frac{1}{2}h_{p}^{T}(x_{1}, x_{3}, u_{2})h_{p}(x_{1}, x_{3}, u_{2}) \\
= V_{x_{1}}(x_{1})f_{1}(x_{1}, x_{3}, u_{2}) + \frac{1}{2}\frac{1}{\gamma^{2}}V_{x_{1}}(x_{1})e_{1}(x_{1}, x_{3}, u_{2})e_{1}^{T}(x_{1}, x_{3}, u_{2})V_{x_{1}}^{T}(x_{1}) \\
+ \frac{1}{2}h^{T}(x_{1}, x_{3}, u_{2})h(x_{1}, x_{3}, u_{2}).$$

From Theorem 5.35 we know that a sufficient condition for the solvability of the  $L_2$ -gain problem with constant  $\gamma$  for  $\Sigma$  is the solvability of the same problem for  $\Sigma_P$ . Presumably this problem is easier to solve for  $\Sigma_P$ . The following result follows directly from Theorem 5.35.

**Corollary 5.37** Assume there exists a solution  $V(x_1) \ge 0$  to (5.60). Consider the systems  $\Sigma$  and  $\Sigma_P$ . The following statements are equivalent:

- (i) the state feedback  $L_2$ -gain control problem with gain  $\gamma$  is solvable for  $\Sigma$  with a differentiable storage function W + V where  $W \ge 0$ ;
- (ii) the state feedback  $L_2$ -gain control problem with gain  $\gamma$  is solvable for  $\Sigma_P$  with a differentiable storage function  $W \ge 0$ .

**Remark 5.38** In special cases such as described in the paper [DaMa 96] it is not necessary to consider the Hamilton-Jacobi equality (5.60), but also solutions to the Hamilton-Jacobi inequality yield similar results.  $\Box$ 

In the linear case the conditions (*ii*) from Theorem 5.6 imply that the almost disturbance decoupling problem for  $\Sigma_P$  is solvable.

The output equation  $h_p$  is related to the tracking error between the solution of the regular  $\mathcal{H}_{\infty}$  problem for the subsystem  $\Sigma_1$  as defined in Subsection 5.2.2 and the actual  $x_3$ . From the uniqueness of the saddle point solutions it follows that locally around  $x_1 = 0$  we have that

$$h_p(x_1, x_3, u_2^*(x_1, V_{x_1}^T(x_1)) = 0 \iff x_3 = x_3^*(x_1, V_{x_1}^T(x_1)).$$

So this factorization idea is closely related to the approach described in Subsection 5.2.4.

### **Chapter 6**

## Examples

In this chapter the two different methods for solving the singular  $\mathcal{H}_{\infty}$  control problem which are discussed in this monograph are applied to two examples.

The first example is concerned with a tracking problem for a rigid body spinning around its center of mass. The rigid body model that is used includes three control torques and three torque disturbances. The model is rewritten using Euler-parameters (see [WeDe 91]). This example is mainly meant to clarify and illustrate the theory. Therefore no comparison is made between the performances of the two different controllers resulting from the cheap control and from the geometric approach. We remark that in [MD 95] the geometric approach to the singular  $\mathcal{H}_{\infty}$  problem has been extended to the more general case in which also the movement of the center of mass is taken into account.

As a second example the inverted pendulum on a cart is considered.

#### 6.1 Tracking of the orientation of a rigid body

The orientation of the rigid body with respect to the inertial frame (I-frame) is represented by  $R \in SO(3)$ . The frame fixed to the rigid body is called the B-frame. So *R* denote the orthonormal rotation matrix from the I-frame to the B-frame. SO(3) is the Special Orthonormal Group of order 3 which is represented by the set of all  $3 \times 3$  orthogonal rotation matrices. We will study the orientation of a rigid body with kinematic differential equations given by

$$R = RS(\omega) \tag{6.1}$$

where  $\omega$  is the angular velocity, in B-frame coordinates. The skew symmetric matrix  $S(\omega)$  is given by

$$S(\omega) = \begin{pmatrix} 0 & -\omega_3 & \omega_2 \\ \omega_3 & 0 & -\omega_1 \\ -\omega_2 & \omega_1 & 0 \end{pmatrix}$$

for  $\omega^T = (\omega_1 \ \omega_2 \ \omega_3)^T$ . The equations of motion of the rigid body are then given in the B-frame as

$$M\dot{\omega} = -S(\omega)M\omega + u + d \tag{6.2}$$

where  $M = M^T > 0$  is the 3 × 3 inertia matrix of the rigid body in B-frame coordinates,  $u = (\begin{array}{cc} u_1 & u_2 & u_3 \end{array})^T$  are the control inputs and the torque disturbances are given by  $d = (\begin{array}{cc} d_1 & d_2 & d_3 \end{array})^T$ .

The reference orientation in SO(3) is given by the desired frame, the D-frame, represented by  $R_d$ .  $R_d \in SO(3)$  is the orthogonal rotation matrix from the I-frame to the D-frame. The desired angular velocity,  ${}^D\omega_d$ , in D-frame co-ordinates is given by  $S({}^D\omega_d) = R_d^T \dot{R}_d$ 

or equivalently

$$\dot{R}_d = R_d S({}^D \omega_d). \tag{6.3}$$

We define

$$\tilde{R} := R_d^T R. \tag{6.4}$$

 $\bar{R}$  is the rotation matrix from the D-frame to the B-frame. The desired angular velocities in B-frame coordinates are then given by

$$\omega_d = \tilde{R}^T \,{}^D\!\omega_d$$

We define  $\tilde{\omega}$ 

$$\tilde{\omega} := \omega - \omega_d$$
.

Differentiation of equation (6.4) gives

$$\begin{split} \dot{\tilde{R}} &= \dot{R}_d^T R + R_d^T \dot{R} \\ &= -S({}^D \omega_d) \tilde{R} + \tilde{R} S(\omega) \\ &= -\tilde{R} \tilde{R}^T S({}^D \omega_d) \tilde{R} + \tilde{R} S(\omega) \\ &= -\tilde{R} S(\omega_d) + \tilde{R} S(\omega) \\ &= \tilde{R} S(\tilde{\omega}) \end{split}$$
since  $\tilde{R}^T S({}^D \omega_d) \tilde{R} = S(\omega_d)$ . The error kinematics are therefore given by

$$\tilde{R} = \tilde{R}S(\tilde{\omega}). \tag{6.5}$$

Our main goal is to keep  $(\tilde{R}, \tilde{\omega}) = (I, 0)$  in the presence of the torque disturbances d. To this end we define the penalty variable

$$\left(\begin{array}{cc} d_{SO(3)}(\tilde{R},I) & \tilde{\omega}^T \end{array}\right)^T \tag{6.6}$$

where  $d_{SO(3)}(\tilde{R}, I)$  (see [SBE 91]) is the geodesic metric on SO(3) which is given by:

$$d_{SO(3)}(\tilde{R}, I) = \frac{2}{\pi} \arccos\left(\frac{1}{2}\sqrt{1 + \operatorname{Tr}(\tilde{R})}\right) \in [0, 1]$$

for  $\tilde{R} \in SO(3)$ .

In the control law to be found it is desirable to have an attitude deviation vector of dimension three. To this end we use Euler parameters which are also known as unit quaternions. The rotation matrix  $\tilde{R} \in SO(3)$  is parameterized by a rotation  $\tilde{\phi}$  around the unit vector  $\tilde{k}$  so that

$$\tilde{R} = I + \sin(\tilde{\phi})S(\tilde{k}) + (1 - \cos(\tilde{\phi}))S^2(\tilde{k}).$$

It is seen that  $(\tilde{\phi}, \tilde{k})$  and  $(-\tilde{\phi}, -\tilde{k})$  correspond to the same rotation matrix  $\tilde{R}$ . The Euler parameters corresponding to  $\tilde{R}$  are given by [Hug 86]

$$\tilde{\epsilon} = \sin(\frac{\tilde{\phi}}{2})\tilde{k}, \quad \tilde{\eta} = \cos(\frac{\tilde{\phi}}{2}).$$

The Euler parameters satisfy the normalization equation

$$\tilde{\epsilon}^T \tilde{\epsilon} + \tilde{\eta}^2 = 1$$

which means that  $(\tilde{\epsilon}, \tilde{\eta}) \in S^3$ , where  $S^3$  is the unit sphere in  $\mathbb{R}^4$ . It follows that

$$\tilde{R} = (\tilde{\eta}^2 - \tilde{\epsilon}^T \tilde{\epsilon})I + \tilde{\epsilon}\tilde{\epsilon}^T + 2\tilde{\eta}S(\tilde{\epsilon})$$

and it is seen that  $(\tilde{\epsilon}, \tilde{\eta})$  and  $(-\tilde{\epsilon}, -\tilde{\eta})$  correspond to the same rotation matrix  $\tilde{R}$ . Note that  $\tilde{\epsilon} = 0 \Leftrightarrow \tilde{R} = I$ . Also,  $\tilde{\epsilon} = 0 \Leftrightarrow \tilde{\eta} \pm 1$  so that both  $\tilde{\epsilon} = 0, \tilde{\eta} = 1$  and  $\tilde{\epsilon} = 0, \tilde{\eta} = -1$  correspond to  $\tilde{R} = I$ .

The kinematical differential equations associated with

$$\tilde{R} = \tilde{R}S(\tilde{\omega})$$

are given by a vector field on  $S^3$  defined by

$$\begin{pmatrix} \dot{\tilde{\epsilon}} \\ \dot{\tilde{\eta}} \end{pmatrix} = \frac{1}{2} \begin{pmatrix} \tilde{\eta}I + S(\tilde{\epsilon}) \\ -\tilde{\epsilon}^T \end{pmatrix} \tilde{\omega}.$$

Let the unit quaternions  $(\tilde{\epsilon}, \tilde{\eta})$  represent  $\tilde{R}$ . The error kinematics corresponding to the equations (6.5) are then given by

$$\dot{\tilde{\epsilon}} = \frac{1}{2} (\tilde{\eta}I + S(\tilde{\epsilon}))\tilde{\omega} \dot{\tilde{\eta}} = -\frac{1}{2} \tilde{\epsilon}^T \tilde{\omega}$$
(6.7)

The penalty variable defined in (6.6) can be written in Euler-parameters as

$$\left(\begin{array}{cc} \frac{2}{\pi} \arccos |\tilde{\eta}| & \tilde{\omega}^T \end{array}\right)^T$$

But we want this penalty variable to be a smooth, at least continuously differentiable, function. Therefore we use the fact that

$$\frac{2}{\pi} \|\tilde{\epsilon}\| \le d_{SO(3)}(\tilde{R}, I) \le \|\tilde{\epsilon}\|$$

to define as new penalty variable

$$z = \left( \begin{array}{cc} \tilde{\epsilon}^T & \tilde{\omega}^T \end{array} \right)^T. \tag{6.8}$$

#### 6.1.1 Cheap control approach

In the previous section we have constructed the following model for the rigid body in Euler coordinates

$$\begin{aligned} \dot{\tilde{\epsilon}} &= \frac{1}{2} \left( \tilde{\eta} I + S(\tilde{\epsilon}) \right) \tilde{\omega} \\ \dot{\tilde{\eta}} &= -\frac{1}{2} \tilde{\epsilon}^T \tilde{\omega} \\ \dot{\tilde{\omega}} &= -M^{-1} S(\tilde{\omega}) M \left( \tilde{\omega} + \omega_d \right) - M^{-1} S(\omega_d) M \tilde{\omega} + M^{-1} v + M^{-1} d \\ z &= \begin{pmatrix} \tilde{\epsilon} \\ \tilde{\omega} \end{pmatrix} \end{aligned}$$
(6.9)

where

$$u = v + M\dot{\omega}_d + S(\omega_d)M\omega_d. \tag{6.10}$$

The model (6.9) will also be denoted by

$$\dot{x} = f(x, \omega_d) + gv + gd$$
  

$$z = h(x)$$
(6.11)

where  $x = \begin{pmatrix} \tilde{\epsilon}^T & \tilde{\eta}^T & \tilde{\omega}^T \end{pmatrix}^T$  is the state and f, g and h are given by

$$f(x, \omega_d) = \begin{pmatrix} \frac{1}{2} (\tilde{\eta}I + S(\tilde{\epsilon})) \tilde{\omega} \\ -\frac{1}{2} \tilde{\epsilon}^T \tilde{\omega} \\ -M^{-1} S(\tilde{\omega}) M (\tilde{\omega} + \omega_d) - M^{-1} S(\omega_d) M \tilde{\omega} \end{pmatrix},$$
$$g = \begin{pmatrix} 0 \\ 0 \\ M^{-1} \end{pmatrix}, \qquad h(x) = \begin{pmatrix} \tilde{\epsilon} \\ \tilde{\omega} \end{pmatrix}.$$

It should be noted that  $\omega_d$ , the desired angular velocity, is time-varying. The direct feedthrough term from the inputs  $v \in \mathbb{R}^3$  to the to-be-controlled outputs  $z \in \mathbb{R}^6$  is zero. Therefore Assumption 2 in Section 3.2 is clearly satisfied and  $m_1 = 0$ . The  $(3 \times 3)$ -matrix  $\beta_2(x)$  can be chosen equal to the identity matrix  $I_3$  and the matrix  $\beta_1(x)$  is void. This means that v is a singular input and the transformed system (3.8) is equal to the original system (6.9).

Our first aim is to find for some  $\gamma$ ,  $\varepsilon > 0$  a solution  $V \ge 0$  of the Hamilton-Jacobi inequality (3.13) which for this system is given by

$$H_{\gamma}(x, V_{x}^{T}(x))$$

$$= V_{x}(x)f(x, \omega_{d}) + \frac{1}{2}\left(\frac{1}{\gamma^{2}} - \frac{1}{\varepsilon}\right)V_{x}(x)gg^{T}V_{x}^{T}(x) + \frac{1}{2}h^{T}(x)h(x)$$

$$= L_{f}V(x) + \frac{1}{2}\left(\frac{1}{\gamma^{2}} - \frac{1}{\varepsilon}\right)\left(L_{g}V(x)\right)^{2} + \frac{1}{2}h^{T}(x)h(x) \leq 0$$
(6.12)

with as equilibrium condition V(0, 1, 0) = 0.

Motivated by the results in ([DaEg 95]) we try the following storage function

$$V(\tilde{\epsilon}, \tilde{\eta}, \tilde{\omega}) = aT(\tilde{\omega}) + bG(\tilde{\epsilon}, \tilde{\omega}) + cP(\tilde{\eta})$$
(6.13)

where  $T(\tilde{\omega})$  is the kinetic energy,  $G(\tilde{\epsilon}, \tilde{\omega})$  is a cross term and  $P(\tilde{\eta})$  is a nonnegative function respectively given by:

$$T(\tilde{\omega}) = \frac{1}{2}\tilde{\omega}^T M \tilde{\omega},$$
  

$$G(\tilde{\epsilon}, \tilde{\omega}) = \tilde{\omega}^T M \tilde{\epsilon},$$
  

$$P(\tilde{\eta}) = 2(1 - \tilde{\eta}).$$

Since  $|\tilde{\eta}| \le 1$  by definition,  $P(\tilde{\eta}) \ge 0$  and  $P(\tilde{\eta}) = 0$  iff  $|\tilde{\eta}| = 1$ . So V(0, 1, 0) = 0 and moreover since

$$P(\tilde{\eta}) = 2(1 - \tilde{\eta})$$
  
=  $(1 - \tilde{\eta})^2 + \tilde{\epsilon}^T \tilde{\epsilon}$   
 $\geq \tilde{\epsilon}^T \tilde{\epsilon}$ 

we have that

$$V(\tilde{\epsilon}, \tilde{\eta}, \tilde{\omega}) = \frac{1}{2} a \tilde{\omega}^T M \tilde{\omega} + b \tilde{\omega}^T M \tilde{\epsilon} + 2c (1 - \tilde{\eta})$$
  

$$\geq \frac{1}{2} a \tilde{\omega}^T M \tilde{\omega} + b \tilde{\omega}^T M \tilde{\epsilon} + c \tilde{\epsilon}^T \tilde{\epsilon}$$
  

$$= \frac{1}{2} y^T Q y$$

where

$$y = \begin{pmatrix} \tilde{\epsilon} \\ \tilde{\omega} \end{pmatrix}, \qquad Q = \begin{pmatrix} 2cI & bM \\ bM & aM \end{pmatrix}.$$

Since the matrix Q is positive definite iff

$$a > 0,$$
  

$$acI > \frac{1}{2}b^2M$$
(6.14)

these conditions (6.14) are sufficient for the function V to be non-negative. Straightforward calculations lead to

$$L_{f}V(\tilde{\epsilon},\tilde{\eta},\tilde{\omega}) = -a\tilde{\omega}^{T}S(\omega_{d})M\tilde{\omega} + \frac{1}{2}b\tilde{\omega}^{T}M(\tilde{\eta}I + S(\tilde{\epsilon}))\tilde{\omega} -b\tilde{\epsilon}^{T}S(\tilde{\omega})M(\tilde{\omega} + \omega_{d}) - b\tilde{\epsilon}^{T}S(\omega_{d})M\tilde{\omega} + c\tilde{\epsilon}^{T}\tilde{\omega} L_{g}V(\tilde{\epsilon},\tilde{\eta},\tilde{\omega}) = a\tilde{\omega}^{T} + b\tilde{\epsilon}$$

where we have used that  $\omega_d^T S(\omega_d) = 0$ . Inserting these inequalities in the Hamiltonian  $H_{\gamma}(x, V_x^T(x))$  gives

$$H_{\gamma}(x, V_{x}^{T}(x)) = -a\tilde{\omega}^{T}S(\omega_{d})M\tilde{\omega} + \frac{1}{2}b\tilde{\omega}^{T}M\left(\tilde{\eta}I + S(\tilde{\epsilon})\right)\tilde{\omega} -b\tilde{\epsilon}^{T}S(\tilde{\omega})M\left(\tilde{\omega} + \omega_{d}\right) - b\tilde{\epsilon}^{T}S(\omega_{d})M\tilde{\omega} + c\tilde{\epsilon}^{T}\tilde{\omega} + \frac{1}{2}a^{2}\left(\frac{1}{\gamma^{2}} - \frac{1}{\varepsilon}\right)\|\tilde{\omega}\|^{2} + \frac{1}{2}b^{2}\left(\frac{1}{\gamma^{2}} - \frac{1}{\varepsilon}\right)\|\tilde{\epsilon}\|^{2} + ab\left(\frac{1}{\gamma^{2}} - \frac{1}{\varepsilon}\right)\tilde{\epsilon}^{T}\tilde{\omega} + \frac{1}{2}\|\tilde{\omega}\| + \frac{1}{2}\|\tilde{\epsilon}\|.$$
(6.15)

By choosing

$$c = ab\left(\frac{1}{\varepsilon} - \frac{1}{\gamma^2}\right) \tag{6.16}$$

the two terms in  $H_{\gamma}$  which include the inner product  $\tilde{\epsilon}^T \tilde{\omega}$  are eliminated. We assume that the desired angular velocity is bounded, i.e.,  $\|\omega_d\| \le \rho_d$  for a certain constant  $\rho_d$ . Then using Schwarz inequality, and the equalities  $\|S(\tilde{\omega})\| = \|\tilde{\omega}\|$ ,  $\|\tilde{\epsilon}\| \le 1$  and  $\|\tilde{\eta}I + S(\tilde{\epsilon})\| = 1$  we see that:

$$\begin{aligned} \left| \tilde{\omega}^{T} S(\omega_{d}) M \tilde{\omega} \right| &\leq \|M\| \|\omega_{d}\| \|\tilde{\omega}\|^{2} \leq \|M\| \rho_{d} \|\tilde{\omega}\|^{2}; \\ \left| \tilde{\omega}^{T} M \left( \tilde{\eta} I + S(\tilde{\epsilon}) \right) \tilde{\omega} \right| &\leq \|M\| \|\tilde{\omega}\|^{2}; \\ \left| \tilde{\epsilon}^{T} S(\tilde{\omega}) M \left( \tilde{\omega} + \omega_{d} \right) \right| &\leq \|M\| \|\tilde{\omega}\| \|\tilde{\omega} + \omega_{d}\| \leq \|M\| \|\tilde{\omega}\|^{2} + \|M\| \rho_{d} \|\tilde{\omega}\|; \\ \left| \tilde{\epsilon}^{T} S(\omega_{d}) M \tilde{\omega} \right| &\leq \|M\| \rho_{d} \|\tilde{\omega}\|. \end{aligned}$$

Inserting these inequalities and the choice for the constant c into the Hamiltonian  $H_{\gamma}$  one obtains

$$\begin{split} H_{\gamma}(x, V_{x}^{T}(x)) &\leq a \|M\| \rho_{d} \|\tilde{\omega}\|^{2} + \frac{3}{2} b \|M\| \|\tilde{\omega}\|^{2} + 2b \|M\| \rho_{d} \|\tilde{\omega}\| \\ &+ \frac{1}{2} a^{2} \left( \frac{1}{\gamma^{2}} - \frac{1}{\varepsilon} \right) \|\tilde{\omega}\|^{2} + \frac{1}{2} b^{2} \left( \frac{1}{\gamma^{2}} - \frac{1}{\varepsilon} \right) \|\tilde{\epsilon}\|^{2} \\ &= \left( a \|M\| \rho_{d} + \frac{3}{2} b \|M\| + \frac{1}{2} a^{2} \left( \frac{1}{\gamma^{2}} - \frac{1}{\varepsilon} \right) + \frac{1}{2} \right) \|\tilde{\omega}\|^{2} \\ &+ 2b \|M\| \rho_{d} \|\tilde{\omega}\| \\ &+ \left( \frac{1}{2} b^{2} \left( \frac{1}{\gamma^{2}} - \frac{1}{\varepsilon} \right) + \frac{1}{2} \right) \|\tilde{\epsilon}\|^{2}. \end{split}$$

The terms including  $\gamma$  and  $\varepsilon$  are the only terms that can be made negative by choosing  $\varepsilon < \gamma^2$ . So following the lines of the regulation problem considered in the paper [DaEg 95] does not lead to global results for the considered tracking problem unless the desired angular velocity is equal to zero for all t ( $\rho_d = 0$ ). However it is possible to derive a so called semi-global result since for every constant N > 0

$$\|\tilde{\omega}\|^2 \le N \|\tilde{\omega}\| \tag{6.17}$$

holds for all  $\|\tilde{\omega}\| \leq N$ . Using the inequality (6.17) the Hamiltonian  $H_{\gamma}$  has the following upper bound on the neighborhood  $\|\tilde{\omega}\| \leq N$ 

$$\begin{aligned} H_{\gamma}(x, V_{x}^{T}(x)) &\leq \left(aN \|M\|\rho_{d} + \frac{3}{2}bN\|M\| + 2b\|M\|\rho_{d} \\ &+ \frac{1}{2}a^{2}N\left(\frac{1}{\gamma^{2}} - \frac{1}{\varepsilon}\right) + \frac{1}{2}N\right)\|\tilde{\omega}\| \\ &+ \left(\frac{1}{2}b^{2}\left(\frac{1}{\gamma^{2}} - \frac{1}{\varepsilon}\right) + \frac{1}{2}\right)\|\tilde{\epsilon}\|^{2} \\ &= \delta_{\tilde{\omega}}(a, b, N, \gamma, \varepsilon)\|\tilde{\omega}\| + \delta_{\tilde{\epsilon}}(b, \gamma, \varepsilon)\|\tilde{\epsilon}\|^{2}. \end{aligned}$$

We want to choose the constants  $\varepsilon$ , a and b in such a way that this upper bound is less than or equal to zero, i.e., such that both the functions  $\delta_{\tilde{\omega}}(a, b, N, \gamma, \varepsilon)$ and  $\delta_{\tilde{\epsilon}}(b, N, \gamma, \varepsilon)$  are less than or equal to zero. The choice of a and b should at the same time satisfy condition (6.14), with c defined in (6.16). First of all we have to choose  $\varepsilon < \gamma^2$ , otherwise we are not able to make  $\delta_{\tilde{\epsilon}} \le 0$ . Then both the functions  $\delta_{\tilde{\omega}}(a, b, N, \gamma, \varepsilon)$  and  $\delta_{\tilde{\epsilon}}(b, N, \gamma, \varepsilon)$  can be made smaller or equal to zero. When we choose

$$b^{2} \ge \frac{1}{\frac{1}{\varepsilon} - \frac{1}{\gamma^{2}}}$$
(6.18)

it follows that  $\delta_{\tilde{\epsilon}}(b, \gamma, \varepsilon) \leq 0$ . Furthermore the function  $\delta_{\tilde{\omega}}(a, b, N, \gamma, \varepsilon)$  can be rewritten as a second order function in *a* 

$$\delta_{\tilde{\omega}}(a, b, N, \gamma, \varepsilon) = \left(\frac{1}{2}N\left(\frac{1}{\gamma^2} - \frac{1}{\varepsilon}\right)\right)a^2 + (\|M\|\rho_d N)a + \left(\frac{3}{2}bN\|M\| + 2b\|M\|\rho_d + \frac{1}{2}N\right). \quad (6.19)$$

This function is for  $\varepsilon < \gamma^2$  convex and has two zeros given by

$$\frac{\|M\|\rho_d N \pm \sqrt{\|M\|^2 \rho_d^2 N^2 + 2N\left(\frac{1}{\varepsilon} - \frac{1}{\gamma^2}\right)\left(\frac{3}{2}bN\|M\| + 2b\|M\|\rho_d + \frac{1}{2}N\right)}}{N\left(\frac{1}{\varepsilon} - \frac{1}{\gamma^2}\right)}.$$

Only one of the two zeros is positive. Hence we choose the constant a greater or equal to

$$\frac{\|M\|\rho_{d}N + \sqrt{\|M\|^{2}\rho_{d}^{2}N^{2} + 2N\left(\frac{1}{\varepsilon} - \frac{1}{\gamma^{2}}\right)\left(\frac{3}{2}bN\|M\| + 2b\|M\|\rho_{d} + \frac{1}{2}N\right)}{N\left(\frac{1}{\varepsilon} - \frac{1}{\gamma^{2}}\right)}.$$
(6.20)

Note that *a* satisfies

$$a^2 \geq \frac{3b \|M\|}{\frac{1}{\varepsilon} - \frac{1}{\gamma^2}}.$$

Together with c as in (6.16) that leads to

$$ac = a^2 b\left(\frac{1}{\varepsilon} - \frac{1}{\gamma^2}\right) \ge 3b^2 \|M\|$$

which implies the second condition from (6.14). Hence the constants a and c defined in (6.20) and (6.16) satisfy the conditions (6.14) and so the corresponding V is non-negative.

Summarizing it follows that for every  $\gamma$ , N > 0 and for every  $\varepsilon < \gamma^2$  we have constructed a local solution  $V \ge 0$  to the Hamilton-Jacobi inequality (6.15) defined on the neighborhood  $\|\tilde{\omega}\| \le N$ . This storage function is given by (6.13) where the constants *a*, *b* and *c* should be chosen according to (6.20), (6.18) and (6.16).

Now the following result can be derived from Corollary 3.13.

**Theorem 6.1** Let  $\gamma$ , N > 0 and choose  $\varepsilon$ , b and a such that:

$$\varepsilon < \gamma^{2};$$

$$b \geq \sqrt{\frac{1}{\frac{1}{\varepsilon} - \frac{1}{\gamma^{2}}}};$$

$$a \geq \frac{\|M\|\rho_{d}N + \sqrt{\|M\|^{2}\rho_{d}^{2}N^{2} + 2N\left(\frac{1}{\varepsilon} - \frac{1}{\gamma^{2}}\right)\left(\frac{3}{2}bN\|M\| + 2b\|M\|\rho_{d} + \frac{1}{2}N\right)}}{N\left(\frac{1}{\varepsilon} - \frac{1}{\gamma^{2}}\right)}$$

Then the error model (6.9) combined with the state feedback

$$v = -\mu \left( a\tilde{\omega} + b\tilde{\epsilon} \right) \tag{6.21}$$

with  $\mu \geq \frac{1}{2\varepsilon}$  leads to a closed loop system which has local  $L_2$ -gain less than or equal to  $\gamma$  on the neighborhood  $\|\tilde{\omega}\| \leq N$  and which is locally asymptotically stable.

**Proof** The local  $L_2$ -gain result follows from Corollary 3.13 using the storage function constructed previously. Furthermore we see that  $z \equiv 0$  implies that  $(\tilde{\epsilon}, \tilde{\omega}) = (0, 0)$ . From the normalization equation it follows that  $\tilde{\eta} \pm 1$ . If we define  $(\tilde{\epsilon}, \tilde{\omega}, \tilde{\eta}) = (0, 1, 0)$  as our zero-state, the closed loop system

$$\begin{pmatrix} \dot{\tilde{\epsilon}} \\ \dot{\tilde{\eta}} \\ \dot{\tilde{\omega}} \end{pmatrix} = \begin{pmatrix} \frac{1}{2} (\tilde{\eta}I + S(\tilde{\epsilon}))\tilde{\omega} \\ -\frac{1}{2}\tilde{\epsilon}^{T}\tilde{\omega} \\ -M^{-1}S(\tilde{\omega})M(\tilde{\omega}\omega_{d}) - M^{-1}S(\omega_{d})M\tilde{\omega} - \mu M^{-1}(a\tilde{\omega} + b\tilde{\epsilon}) \end{pmatrix} \\ + \begin{pmatrix} 0 \\ 0 \\ M^{-1} \end{pmatrix} d \\ z = \begin{pmatrix} \tilde{\epsilon} \\ \tilde{\omega} \end{pmatrix}$$

is zero-state observable in a neighborhood of  $(\tilde{\epsilon}, \tilde{\eta}, \tilde{\omega}) = (0, 1, 0)$ . Then local asymptotic stability follows from Theorem 2.7 or 3.21.

For our original system (6.7), (6.2) this result can be reformulated.

**Corollary 6.2** Choose  $\gamma$ , N,  $\varepsilon$ , a and b in the same way as in Theorem 6.1. Then the  $\mathcal{H}_{\infty}$  problem with state feedback for the rigid spacecraft (6.7), (6.2) with penalty variable (6.8) is solved locally by the state feedback

$$u = M\dot{\omega}_d + S(\omega_d)M\omega_d - \mu \left(a\tilde{\omega} + b\tilde{\epsilon}\right)$$
(6.22)

for every 
$$\mu \geq \frac{1}{2\epsilon}$$
 on the neighborhood  $\|\tilde{\omega}\| \leq N$ .

The choice of  $\gamma$ ,  $\varepsilon$  and N clearly influences the lower bound for the constants a and b and therefore the gain of the feedback (6.21). When the neighborhood  $\|\tilde{\omega}\| \leq N$  is increased such that the solution of the  $\mathcal{H}_{\infty}$  problem applies on a larger area around the zero-state this results in a higher lower bound for a as can be seen in (6.20). Also a choice of the parameter  $\varepsilon$  close to  $\gamma^2$  or to zero will obviously lead to a higher choice for both a and b. Therefore the choice of the constants N,  $\mu$  and  $\varepsilon$  depend on the specific model at hand and the desired performance.

#### 6.1.2 Geometric approach

The model for the rigid spacecraft can also be rewritten in the following form, see (6.2) and (6.7)

$$\begin{aligned} \dot{\tilde{\epsilon}} &= \frac{1}{2} \left( \tilde{\eta} I + S(\tilde{\epsilon}) \right) \tilde{\omega} \\ \dot{\tilde{\eta}} &= -\frac{1}{2} \tilde{\epsilon}^T \tilde{\omega} \\ \dot{\tilde{\omega}} &= -M^{-1} \left( S(\tilde{\omega}) + S(\omega_d) \right) M \left( \tilde{\omega} + \omega_d \right) + M^{-1} v + M^{-1} d \\ z &= \begin{pmatrix} \tilde{\epsilon} \\ \tilde{\omega} \end{pmatrix} \end{aligned}$$
(6.23)

where

$$u = v + M\dot{\omega}_d. \tag{6.24}$$

Again for the model (6.23) we will also use the shorthand notation

$$\dot{x} = f(x, \omega_d) + gv + gd$$
  

$$z = h(x)$$
(6.25)

where  $x = \begin{pmatrix} \tilde{\epsilon}^T & \tilde{\eta}^T & \tilde{\omega}^T \end{pmatrix}^T$  is the state and f, G and h are given by

$$f(x,\omega_d) = \begin{pmatrix} \frac{1}{2} \left( \tilde{\eta} I + S(\tilde{\epsilon}) \right) \tilde{\omega} \\ -\frac{1}{2} \tilde{\epsilon}^T \tilde{\omega} \\ -M^{-1} \left( S(\tilde{\omega}) + S(\omega_d) \right) M \left( \tilde{\omega} + \omega_d \right) \end{pmatrix},$$

$$g = \begin{pmatrix} 0 \\ 0 \\ M^{-1} \end{pmatrix}, \qquad h(x) = \begin{pmatrix} \tilde{\epsilon} \\ \tilde{\omega} \end{pmatrix}.$$

Because the inputs  $v \in \mathbb{R}^3$  do not directly influence the to-be-controlled outputs z Assumption 5 (Subsection 5.2.2) is clearly satisfied, with  $m_1$  equal to zero (see also Subsection 5.2.7). The system (6.23) or (6.25) will be referred to as  $\Sigma$ . Because  $m_1 = 0$  there is no need to transform the inputs. For applying the state transformation described in Chapter 5 we construct the extended system  $\Sigma_e$ 

$$\Sigma_e \begin{cases} \dot{x}_e = f_e(x_e, \omega_d) + g_e v + e_e d \\ z = h_e(x_e) \end{cases}$$

where the 13 dimensional state vector  $x_e$  is equal to  $\begin{pmatrix} x^T & v^T & d^T \end{pmatrix}^T$  and the functions  $f_e(x_e)$ ,  $g_e$ ,  $e_e$  and  $h_e(x_e)$  are given by

$$f_e(x_e, \omega_d) = \begin{pmatrix} f(x, \omega_d) + gv + gd \\ 0 \\ 0 \end{pmatrix}, \quad g_e = \begin{pmatrix} 0 \\ I_3 \\ 0 \end{pmatrix},$$
$$e_e = \begin{pmatrix} 0 \\ 0 \\ I_3 \end{pmatrix}, \quad h_e(x_e) = h(x).$$

For this system  $\Sigma_e$  we want to calculate the minimal conditioned invariant distribution containing the input vector fields using the  $S^*$ -algorithm.

In the first step of this algorithm we obtain

$$S_{1e} = \operatorname{span}\left\{\frac{\partial}{\partial v_1}, \frac{\partial}{\partial v_2}, \frac{\partial}{\partial v_3}\right\}$$

whose involutive closure is equal to  $S_{1e}$  itself and is contained in ker dh. Therefore we calculate in the second step  $S_{2e}$  which is given by

$$S_{2e} = S_{1e} + [f_e(x_e), S_{1e}]$$
  
= span  $\left\{ \frac{\partial}{\partial v_1}, \frac{\partial}{\partial v_2}, \frac{\partial}{\partial v_3}, \frac{\partial}{\partial \tilde{\omega}_1}, \frac{\partial}{\partial \tilde{\omega}_2}, \frac{\partial}{\partial \tilde{\omega}_3} \right\}$   
=  $\overline{S}_{2e}$ 

and which clearly satisfies

$$S_{2e} \cap \ker \mathrm{d}h = S_{1e}.$$

This means that the algorithm stops after two steps and that the minimal conditioned invariant distribution containing the input vector fields for the system  $\Sigma_e$  is given by

$$\mathcal{S}_{e}^{*} = \operatorname{span}\left\{\frac{\partial}{\partial v_{1}}, \frac{\partial}{\partial v_{2}}, \frac{\partial}{\partial v_{3}}, \frac{\partial}{\partial \tilde{\omega}_{1}}, \frac{\partial}{\partial \tilde{\omega}_{2}}, \frac{\partial}{\partial \tilde{\omega}_{3}}\right\}.$$

This distribution also satisfies the projectability condition (5.24)

$$\left[\frac{\partial}{\partial d_i}, S_e^*\right] = 0 \in S_e^*, \qquad i = 1, 2, 3.$$

The distributions  $X_{1e}$ ,  $X_{2e}$  and  $X_{3e}$  are then given by:

$$\begin{aligned} \mathcal{X}_{1e} &= \operatorname{span} \left\{ \frac{\partial}{\partial \tilde{\epsilon}_{1}}, \frac{\partial}{\partial \tilde{\epsilon}_{2}}, \frac{\partial}{\partial \tilde{\epsilon}_{3}}, \frac{\partial}{\partial \tilde{\eta}}, \frac{\partial}{\partial d_{1}}, \frac{\partial}{\partial d_{2}}, \frac{\partial}{\partial d_{3}} \right\}; \\ \mathcal{X}_{2e} &= \operatorname{span} \left\{ \frac{\partial}{\partial v_{1}}, \frac{\partial}{\partial v_{2}}, \frac{\partial}{\partial v_{3}} \right\}; \\ \mathcal{X}_{3e} &= \operatorname{span} \left\{ \frac{\partial}{\partial \tilde{\omega}_{1}}, \frac{\partial}{\partial \tilde{\omega}_{2}}, \frac{\partial}{\partial \tilde{\omega}_{3}} \right\}. \end{aligned}$$

Then the projections of  $X_{1e}$ ,  $X_{2e}$  and  $X_{3e}$  onto the state space of the original system are:

$$\begin{aligned} \mathcal{X}_{1}^{P} &= \operatorname{span} \left\{ \frac{\partial}{\partial \tilde{\epsilon}_{1}}, \frac{\partial}{\partial \tilde{\epsilon}_{2}}, \frac{\partial}{\partial \tilde{\epsilon}_{3}}, \frac{\partial}{\partial \tilde{\eta}} \right\}; \\ \mathcal{X}_{2}^{P} &= 0; \\ \mathcal{X}_{3}^{P} &= \operatorname{span} \left\{ \frac{\partial}{\partial \tilde{\omega}_{1}}, \frac{\partial}{\partial \tilde{\omega}_{2}}, \frac{\partial}{\partial \tilde{\omega}_{3}} \right\}. \end{aligned}$$

The resulting regular subsystem  $\Sigma_1$  is

$$\Sigma_{1} \begin{cases} \dot{\tilde{\epsilon}} &= \frac{1}{2} \left( \tilde{\eta} I + S(\tilde{\epsilon}) \right) \tilde{\omega} \\ \dot{\tilde{\eta}} &= -\frac{1}{2} \tilde{\epsilon}^{T} \tilde{\omega} \\ z &= \begin{pmatrix} \tilde{\epsilon} \\ \tilde{\omega} \end{pmatrix} \end{cases}$$

Notice that there are no disturbances entering this subsystem. Therefore the regular  $\mathcal{H}_{\infty}$  problem for the system  $\Sigma_1$  reduces to an optimal control problem. The cost-criterium for this optimal control problem is

$$\frac{1}{2}\int_0^T \|z(t)\|^2 \mathrm{d}t,$$

which we minimize by considering  $\tilde{\omega}$  as inputs to this subsystem. The pre-Hamiltonian of this problem is given by

$$K(\tilde{\epsilon}, \tilde{\eta}, p_{\tilde{\epsilon}}, p_{\tilde{\eta}}, \tilde{\omega}) = \frac{1}{2} p_{\tilde{\epsilon}}^T \left( \tilde{\eta} I + S(\tilde{\epsilon}) \right) \tilde{\omega} - \frac{1}{2} p_{\tilde{\eta}} \tilde{\epsilon}^T \tilde{\omega} + \frac{1}{2} \tilde{\epsilon}^T \tilde{\epsilon} + \frac{1}{2} \tilde{\omega}^T \tilde{\omega}.$$

Hence the optimal (minimizing) control  $\tilde{\omega}^*$  is given by

$$\tilde{\omega}^* = \frac{1}{2} \tilde{\epsilon} p_{\tilde{\eta}} - \frac{1}{2} \left( \tilde{\eta} I - S(\tilde{\epsilon}) \right) p_{\tilde{\epsilon}}.$$
(6.26)

Substituting this optimal control  $\tilde{\omega}^*$  into the pre-Hamiltonian  $K(\tilde{\epsilon}, \tilde{\eta}, p_{\tilde{\epsilon}}, p_{\tilde{\eta}}, \tilde{\omega})$  we see that this optimal control problem can be solved by finding a non-negative solution *V* to the following Hamilton-Jacobi equality

$$-\frac{1}{8}V_{\tilde{\epsilon}}(\tilde{\epsilon},\tilde{\eta})(\tilde{\eta}I+S(\tilde{\epsilon}))(\tilde{\eta}I-S(\tilde{\epsilon}))V_{\tilde{\epsilon}}^{T}(\tilde{\epsilon},\tilde{\eta})-\frac{1}{8}V_{\tilde{\eta}}^{2}(\tilde{\xi},\tilde{\eta})\tilde{\epsilon}^{T}\tilde{\epsilon}$$

$$+\frac{1}{4}V_{\tilde{\eta}}(\tilde{\epsilon},\tilde{\eta})\tilde{\epsilon}^{T}(\tilde{\eta}I-S(\tilde{\epsilon}))V_{\tilde{\epsilon}}^{T}(\tilde{\epsilon},\tilde{\eta})+\frac{1}{2}\tilde{\epsilon}^{T}\tilde{\epsilon} = 0.$$
(6.27)

Such a solution is quite easy to give as can be seen in the next lemma.

Lemma 6.3 The function

$$V(\tilde{\epsilon}, \tilde{\eta}) = 2\left(1 - \tilde{\eta}\right) \tag{6.28}$$

is a non-negative solution to (6.27). The corresponding optimal control is given by  $\tilde{\omega}^*(\tilde{\epsilon}, \tilde{\eta}) = -\tilde{\epsilon}$ .

**Proof** By definition of the Euler parameters  $|\tilde{\eta}| \leq 1$ , and hence V is non-negative. Substitution of V into (6.27) leads to

$$-\frac{1}{8}(-2)^2\tilde{\epsilon}^T\tilde{\epsilon} + \frac{1}{2}\tilde{\epsilon}^T\tilde{\epsilon} = 0.$$

The optimal control follows by substituting  $p_{\tilde{\epsilon}} = V_{\tilde{\epsilon}}^T(\tilde{\epsilon}, \tilde{\eta}) = 0$ ,  $p_{\tilde{\eta}} = V_{\tilde{\eta}}^T(\tilde{\epsilon}, \tilde{\eta})$ into (6.26)

$$\tilde{\omega}^*(\tilde{\epsilon}, \tilde{\eta}) = \frac{1}{2}\tilde{\epsilon}(-2) = -\tilde{\epsilon}.$$

**Remark 6.4** The system  $\Sigma_1$  is lossless ([Wi 72]). In general a system

$$\dot{x} = f(x) + g(x)u z = \begin{pmatrix} h(x) \\ u \end{pmatrix}$$

with  $x \in M$ ,  $u \in \mathbb{R}^m$  and  $z \in \mathbb{R}^p$  is called *lossless* if there exists a non-negative function  $V: M \to \mathbb{R}$  such that:

$$V_x(x) f(x) = 0;$$
  

$$V_x(x)g(x) = h^T(x).$$

For a lossless system it is well known ([TaAr 81], [vdS 93]) that a solution to the optimal control problem with cost criterium

$$\int_0^t \|z(\tau)\|^2 \mathrm{d}\tau$$

is given by

$$u=-h(x).$$

Now we want to look at the  $\mathcal{H}_{\infty}$  problem for the complete system (6.23). Because V is a solution to the Hamilton-Jacobi equality we can use the results from Subsection 5.2.7. Otherwise stated V is a solution to the nonlinear matrix inequality which for the system (6.23) reads as

$$2V_{x}(x) (f(x, \omega_{d}) + gv) + \frac{1}{\gamma^{2}} V_{x}(x) gg^{T} V_{x}^{T}(x) + h^{T}(x) h(x) \ge 0.$$

This nonlinear dissipation inequality is used to define a new output function  $h_p$ . First we note that  $V_x(x)g \equiv 0$ . Then we consider the equation

$$h_p^T(x)h_p(x) = 2V_x(x)f(x,\omega_d) + h^T(x)h(x)$$
  
=  $2\tilde{\epsilon}^T\tilde{\omega} + \tilde{\epsilon}^T\tilde{\epsilon} + \tilde{\omega}^T\tilde{\omega}$   
=  $(\tilde{\omega} + \tilde{\epsilon})^T(\tilde{\omega} + \tilde{\epsilon}).$ 

A smooth solution  $h_p(x)$  is given by

$$h_p(x) = \tilde{\omega} + \tilde{\epsilon}.$$

This new output function leads to a new system  $\Sigma_P$  which in this case since the disturbances are absent, is equal to  $\Sigma$  except for the new output

$$z_p = h_p(x) = \tilde{\omega} + \tilde{\epsilon}. \tag{6.29}$$

Now the following result can be derived from Theorem 5.35.

**Corollary 6.5** Suppose the  $L_2$ -gain problem for the system  $\Sigma$  is solvable with  $L_2$ -gain less than or equal to  $\gamma$  in the sense that there exists a non-negative solution W to the dissipation inequality

$$W_x(x)\left(f(x,\omega_d)+gv+gd\right)+\frac{1}{2}z_p^Tz_p-\frac{1}{2}\gamma^2d^Td\leq 0.$$

Then the  $L_2$ -gain problem for the system  $\Sigma_P$  is solvable with  $L_2$ -gain less than or equal to  $\gamma$ .

Finally it is sufficient to prove that the  $\mathcal{H}_{\infty}$  problem for the system  $\Sigma_P$  is solvable. For the system  $\Sigma_P$  the  $\mathcal{H}_{\infty}$  almost disturbance decoupling problem is solvable. This can be proved by verifying the conditions from Theorem 4.7 for the system

$$\Sigma_{P} \begin{cases} \dot{\tilde{\epsilon}} = \frac{1}{2} \left( \tilde{\eta}I + S(\tilde{\epsilon}) \right) \tilde{\omega} \\ \dot{\tilde{\eta}} = -\frac{1}{2} \tilde{\epsilon}^{T} \tilde{\omega} \\ \dot{\tilde{\omega}} = -M^{-1} \left( S(\tilde{\omega}) + S(\omega_{d}) \right) M \left( \tilde{\omega} + \omega_{d} \right) + M^{-1} v + M^{-1} d \\ z_{p} = \tilde{\omega} + \tilde{\epsilon} \end{cases}$$

We have that m = p = 3, the strong control characteristic indices  $\rho_1$ ,  $\rho_2$ ,  $\rho_3$  are all equal to 1 and the decoupling matrix A(x) is equal to the positive definite, symmetric matrix  $M^{-1}$ . Furthermore the vector fields

$$\begin{pmatrix} \frac{1}{2} \left( \tilde{\eta}I + S(\tilde{\epsilon}) \right) \tilde{\omega} \\ -\frac{1}{2} \tilde{\epsilon}^{T} \tilde{\omega} \\ \frac{1}{2} \left( \tilde{\eta}I + S(\tilde{\epsilon}) \right) \tilde{\omega} \end{pmatrix}, \quad \begin{pmatrix} 0 \\ 0 \\ I_{3} \end{pmatrix}$$

are complete and the set  $G_0$  is equal to

$$\operatorname{span}\left\{\frac{\partial}{\partial \tilde{\omega}_{1}}, \frac{\partial}{\partial \tilde{\omega}_{2}}, \frac{\partial}{\partial \tilde{\omega}_{3}}\right\}$$

which is clearly involutive and of dimension 3. The last condition is void. So the  $L_2$ -gain almost disturbance decoupling problem for the system  $\Sigma_P$  is solvable. An actual feedback can be constructed following the proof of Theorem 4.5

and the SISO results derived in [MRST 94] and [MT 95]. Indeed, because the decoupling matrix is non-singular we can input-output decouple the system (see [Is 89], [NvdS 90]) by the feedback

$$v = -M \begin{pmatrix} L_f h_1(x) \\ L_f h_2(x) \\ L_f h_3(x) \end{pmatrix} + M\tilde{v}$$
  
$$= -M \left( \dot{\tilde{\omega}} + \dot{\tilde{\epsilon}} \right) + M\tilde{v} \qquad (6.30)$$
  
$$= (S(\tilde{\omega}) + S(\omega_d)) M (\tilde{\omega} + \omega_d) - \frac{1}{2} M (\tilde{\eta}I + S(\tilde{\epsilon})) \tilde{\omega} + M\tilde{v}.$$

This feedback together with a transformation of the state components  $\tilde{\omega}$  into

$$q = \tilde{\omega} + \tilde{\epsilon}$$

leads to the system

~

$$\begin{aligned}
\tilde{\epsilon} &= \frac{1}{2} \left( \tilde{\eta} I + S(\tilde{\epsilon}) \right) \tilde{\omega} \\
\tilde{\eta} &= -\frac{1}{2} \tilde{\epsilon}^T \tilde{\omega} \\
\dot{q} &= \tilde{\nu} + M^{-1} d \\
z_p &= q
\end{aligned}$$
(6.31)

and we take

$$\tilde{v}_i = -q_i - \frac{1}{4} k q_i \left( \left( M_{i*}^{-1} \right) \left( M_{i*}^{-1} \right)^T \right) \qquad i = 1, \dots, 3$$
(6.32)

where  $M_{i*}^{-1}$  indicates the *i*-th row of the matrix  $M^{-1}$ . Now define

$$W(q) = \frac{1}{4}q^T q.$$

Then from the following completion of the squares argument (see [MT 95])

$$\begin{split} \dot{W} &= \sum_{i=1}^{3} \left( -\frac{1}{2} q_{i}^{2} - \frac{1}{2} \frac{1}{4} k q_{i}^{2} \left( \left( M_{i*}^{-1} \right) \left( M_{i*}^{-1} \right)^{T} \right) + \frac{1}{2} q_{i} \left( M_{i*}^{-1} \right) d \right. \\ &- \frac{1}{2} \frac{1}{k} d^{T} d + \frac{1}{2} \frac{1}{k} d^{T} d \right) \\ &= \frac{1}{2} \sum_{i=1}^{3} \left( -q_{i}^{2} - k \left\| \frac{1}{2} q_{i} \left( M_{i*}^{-1} \right)^{T} - \frac{1}{k} d \right\|^{2} + \frac{1}{k} d^{T} d \right) \\ &\leq \sum_{i=1}^{3} \left( -\frac{1}{2} q_{i}^{2} + \frac{1}{2} \frac{1}{k} d^{T} d \right) \\ &= -\frac{1}{2} q^{T} q + \frac{1}{2} \frac{3}{k} d^{T} d \end{split}$$

it follows that the feedback (6.32) applied to the system  $\Sigma_P$  leads to a closed loop system which has  $L_2$ -gain less than or equal to  $\sqrt{\frac{3}{k}}$ . Since k can be chosen arbitrarily large the feedback

$$u = M\dot{\omega}_d + (S(\tilde{\omega}) + S(\omega_d)) M (\tilde{\omega} + \omega_d) - \frac{1}{2} M (\tilde{\eta}I + S(\tilde{\epsilon})) \tilde{\omega}$$
$$-Mq - \frac{1}{4} k M q N_M$$

where

$$N_{M} = \operatorname{diag}\left\{ \left( M_{1*}^{-1} \right) \left( M_{1*}^{-1} \right)^{T}, \dots, \left( M_{3*}^{-1} \right) \left( M_{3*}^{-1} \right)^{T} \right\}$$

solves the  $\mathcal{H}_{\infty}$  almost disturbance decoupling problem for the system (6.23) with  $L_2$ -gain  $\sqrt{\frac{3}{k}}$ , where k can be chosen arbitrary large. Summarizing:

**Theorem 6.6** Let  $\gamma$  be an arbitrary positive constant. Then the feedback

$$u = M\dot{\omega}_d + (S(\tilde{\omega}) + S(\omega_d)) M (\tilde{\omega} + \omega_d) - \frac{1}{2} M (\tilde{\eta}I + S(\tilde{\epsilon})) \tilde{\omega}$$
  
$$-Mq - \frac{3}{4} \frac{1}{\nu^2} Mq N_M$$
(6.33)

applied to the system (6.23) leads to a closed loop system which has  $L_2$ -gain less than or equal to  $\gamma$  and is locally asymptotically stable.

Proof Take as a storage function

$$V(x) + W(x) = 2(1 - \tilde{\eta}) + \frac{1}{4}(\tilde{\omega} + \tilde{\epsilon})^T(\tilde{\omega} + \tilde{\epsilon})$$

which is seen to be a non-negative solution to the Hamilton-Jacobi inequality

$$P_{x}(x) \left( f(x, \omega_{d}) + gu \right) + \frac{1}{2} \frac{1}{\gamma^{2}} P_{x}(x) gg^{T} P_{x}(x) + \frac{1}{2} z^{T} z \leq 0.$$

This storage function is even positive definite outside the origin because all separate terms are non-negative and when  $\tilde{\epsilon} = -\tilde{\omega}$  it follows that  $\tilde{\eta}^2 = 1 - \tilde{\omega}^T \tilde{\omega} \neq 1$  and hence the first term is positive. Therefore the closed loop system is locally asymptotically stable.

To solve the singular  $\mathcal{H}_{\infty}$  problem we have followed the factorization approach as described in Subsection 5.2.7. Similarly we could use the results from Subsection 5.2.4, Theorem 5.21. These results also apply when we have a solution to the Hamilton-Jacobi inequality corresponding to the regular  $\mathcal{H}_{\infty}$  problem for the subsystem  $\Sigma_1$  instead of a solution to the equality.

### 6.2 Inverted pendulum on a cart

As a second example we shall consider the physical example of an inverted pendulum on a cart (see also [St 92]). We assume that the mass m of the pendulum is concentrated in the top. l is the length of the pendulum and M the mass of the cart. To describe the position, q and  $\varphi$  express the distance of the cart from some reference point, respectively the angle of the pendulum with respect to the vertical axis. The only input is the horizontal force u applied to the cart. We assume that the system is stiff and that the friction between the cart and the ground is linear with friction coefficient F. Finally g denotes the accelaration of gravity. We derive the following nonlinear model for this system:

$$(M+m)\ddot{q} + m\ddot{\varphi}\cos\varphi - ml(\dot{\varphi})^{2}\sin\varphi + F\dot{q} = u; \qquad (6.34)$$

$$l\ddot{\varphi} - g\sin\varphi + \ddot{q}\cos\varphi = 0. \qquad (6.35)$$



Figure 6.1: inverted pendulum on a cart

Furthermore we assume that  $l \neq 0$  and  $M \neq 0$ . Then the equations (6.34) and (6.35) can be rewritten to the following equations:

$$\ddot{q} = -\frac{mg\sin\varphi\cos\varphi}{M+m\sin^2\varphi} + \frac{ml(\dot{\varphi})^2\sin\varphi}{M+m\sin^2\varphi} - \frac{F\dot{q}}{M+m\sin^2\varphi}$$

$$+\frac{1}{M+m\sin^2\varphi}u; (6.36)$$

$$\ddot{\varphi} = \frac{g\sin\varphi}{l} + \frac{mg\sin\varphi\cos^2\varphi}{l(M+m\sin^2\varphi)} - \frac{m(\dot{\varphi})^2\sin\varphi\cos\varphi}{M+m\sin^2\varphi} + \frac{F\dot{q}\cos\varphi}{l(M+m\sin^2\varphi)} - \frac{\cos\varphi}{l(M+m\sin^2\varphi)}u.$$
(6.37)

We define the state  $x = (\begin{array}{cc} q & \dot{q} & \varphi & \dot{\varphi} \end{array})^T$ . Then the system is given by

$$\dot{x} = f(x) + g(x)u$$
 (6.38)

where

$$f(x) = \begin{pmatrix} \dot{q} \\ -\frac{mg\sin\varphi\cos\varphi}{M+m\sin^2\varphi} + \frac{ml(\dot{\varphi})^2\sin\varphi}{M+m\sin^2\varphi} - \frac{F\dot{q}}{M+m\sin^2\varphi} \\ \dot{\varphi} \\ \frac{g\sin\varphi}{l} + \frac{mg\sin\varphi\cos^2\varphi}{l(M+m\sin^2\varphi)} - \frac{m(\dot{\varphi})^2\sin\varphi\cos\varphi}{M+m\sin^2\varphi} + \frac{F\dot{q}\cos\varphi}{l(M+m\sin^2\varphi)} \end{pmatrix},$$

$$g(x) = \begin{pmatrix} 0 \\ \frac{1}{M+m\sin^2\varphi} \\ 0 \\ -\frac{\cos\varphi}{l(M+m\sin^2\varphi)} \end{pmatrix}.$$

An important question is which kind of disturbances we would like to guard against. Different kinds of disturbances will lead to different uncertainty models. We decided to guard against all fluctuations in the two differential equations (6.36) and (6.37), because we want to consider parameter uncertainties in the parameters m, F and M. Of course, this is only one of the options. We penalize the state by choosing the to-be-controlled variable z equal to the state x. Hence the perturbed model we will consider is

$$\Sigma \begin{cases} \dot{x} = f(x) + g(x)u + e(x)d \\ z = x \end{cases}$$

where the matrix e(x) is a constant matrix given by

$$e(x) = \begin{pmatrix} 0 & 0 \\ 1 & 0 \\ 0 & 0 \\ 0 & 1 \end{pmatrix}$$

and  $d = \begin{pmatrix} d_1 & d_2 \end{pmatrix}^T$  represents the two disturbances acting on the two differential equations (6.36) and (6.37).

We will take as nominal values for the mass of the pendulum m = 0.1 (kg), the mass of the cart M = 1 (kg), the length of the pendulum l = 1 (m) and the friction coefficient F = 0.1 (Ns/m). In the next subsection we will describe how the cheap control approach can be used to find a solution to this state feedback  $\mathcal{H}_{\infty}$  problem. Especially the choice of the parameter  $\varepsilon$  will be considered. In the second subsection we will show how the geometric approach can be applied to this problem. Finally in the Subsection 6.2.3 the feedbacks calculated using the two different approaches will be compared with respect to closedloop asymptotic stability and parameter robustness in the parameters m, F and M for two different values of the  $L_2$ -gain  $\gamma$ . This example is mainly meant to illustrate the methods and results from Chapter 3 and 5, and to show the issues to be considered. Therefore no a priori design goals have been formulated.

#### 6.2.1 Cheap control approach

We apply the theory of Chapter 3 to the system  $\Sigma$ . Solving the singular  $\mathcal{H}_{\infty}$  comes down to finding a constant  $\varepsilon > 0$  for which there exists a solution  $V \ge 0$  to the parameterized Hamilton-Jacobi equality

$$V_x(x)f(x) + \frac{1}{2}V_x(x)\left(\frac{1}{\gamma^2}e(x)e^T(x) - \frac{1}{\varepsilon}g(x)g^T(x)\right)V_x^T(x) + \frac{1}{2}x^Tx = 0.$$
(6.39)

Based on an article of Lukes ([Lu 69]) an iterative procedure for calculation of higher order polynomial solutions to the Hamilton-Jacobi equality (6.39) can be derived (see [Vr 94], [Dow 93]). Computation of these solutions is done using the formula manipulation computer package Mathematica. We have chosen to calculate polynomial solutions V of (6.39) up to order 4. Correspondingly we will calculate a feedback for the system  $\Sigma$  being an approximation of the feedback

$$u = -\frac{1}{\varepsilon}g^T(x)V_x^T(x)$$

considered in Subsection 3.2.2. A crucial point is the selection of the constant  $\varepsilon > 0$  as will be explained in the next part of this subsection.

#### The selection of $\varepsilon$

In the Mathematica procedure we first calculate the optimal  $\gamma^*$  for the linearized system. The  $L_2$ -bound  $\gamma$  for the nonlinear system is then chosen as

$$\gamma = \gamma^* + \Delta \gamma$$

where  $\Delta \gamma$  indicates the distance from the optimal value  $\gamma^*$ .

In most cases there is a link between the selection of  $\varepsilon$  and the value  $\gamma^*$  of the linearized system. Furthermore both  $\varepsilon$  and  $\Delta \gamma$  will probably influence the stability properties of the closed-loop system obtained from applying the approximation of the feedback. To give an idea of the influences of  $\varepsilon$  and  $\Delta \gamma$  on the closed-loop stability we have chosen three values of  $\Delta \gamma$  relative to  $\gamma^*$  and three values of  $\varepsilon$  as follows

$$\Delta \gamma_1 = 0.01 \gamma^*, \quad \Delta \gamma_2 = 0.05 \gamma^*, \quad \Delta \gamma_3 = 0.2 \gamma^*;$$
  
 $\varepsilon_1 = 0.05, \quad \varepsilon_2 = 0.4, \quad \varepsilon_3 = 1.$ 

For all 9 combinations of these parameters  $\Delta \gamma$  and  $\varepsilon$  we have calculated the maximal values of q(0) and  $\varphi(0)$  still leading to an asymptotically stable closed-loop behaviour. The results are shown in Table 6.1, 6.2 and 6.3.

3	$\gamma^*$	γ	$q_{\max}(0)$	$\varphi_{\max}(0)$
0.05	1.03	1.04	0.030	0.004
0.4	1.65	1.67	0.076	0.007
1	2.20	2.22	0.133	0.009

Table 6.1: stability results for  $\Delta \gamma_1 (= 0.01 \gamma^*)$ 

Е	$\gamma^*$	γ	$q_{\max}(0)$	$\varphi_{\max}(0)$
0.05	1.03	1.08	0.30	0.045
0.4	1.65	1.73	0.80	0.075
1	2.20	2.31	1.38	0.103

Table 6.2: stability results for  $\Delta \gamma_2$  (= 0.05 $\gamma^*$ )

From these results we see that the stability region increases when we choose the margin  $\Delta \gamma$  larger at the price of an higher  $L_2$ -gain  $\gamma$  for the closed-loop system. On the other hand for a specific choose of  $\Delta \gamma$  the  $L_2$ -gain  $\gamma$  can be

Е	$\gamma^*$	γ	$q_{\max}(0)$	$\varphi_{\max}(0)$
0.05	1.03	1.24	2.1	0.32
0.4	1.65	1.98	5.3	0.59
1	2.20	2.64	8.9	0.90

Table 6.3: stability results for  $\Delta \gamma_3 (= 0.2\gamma^*)$ 



Figure 6.2: inputs for  $\varepsilon = 0.05$  (dashed) and  $\varepsilon = 0.4$  (solid) ( $\Delta \gamma = 0.2 \gamma^*$ )

decreased by selecting  $\varepsilon$  smaller. The drawback of a smaller  $\varepsilon$  however is that this leads to an higher gain for the feedback as shown in Figure 6.2 where for  $\Delta \gamma = 0.2\gamma^*$  the inputs *u* are plotted for  $\varepsilon = 0.05$  and  $0.4 (x(0) = (1\ 0\ 0\ 0))$ .

Hence from the experiments we have done so far it is clear that given an  $L_2$ -gain  $\gamma$  for the closed-loop system the selection of  $\varepsilon$ , and correspondingly  $\Delta \gamma$ , is a choice between two possible design objectives, namely the domain of attraction and the gain of the feedback.

#### 6.2.2 Geometric approach

In this section we explain how the geometric approach can be applied to construct a solution to the state feedback  $\mathcal{H}_{\infty}$  problem. The approach followed in this subsection is based on Subsection 5.2.7. First we construct a state transformation as in Subsection 5.2.2. Therefore we consider the extended system

$$\Sigma_e \begin{cases} \dot{x} = f(x) + g(x)u + e(x)d \\ \dot{u} = v \\ \dot{d} = w \\ z = x \end{cases}$$

or written in a compact form

$$\Sigma_e \begin{cases} \dot{x}_e = f_e(x_e) + g_e(x_e)v + e_e(x_e)w \\ z = h_e(x_e) \end{cases}$$

where  $x_e = (x \ u \ d)^T$  is the state and  $f_e$ ,  $g_e$ ,  $e_e$  and  $h_e$  are given by

$$f_e(x_e) = \begin{pmatrix} f(x) + g(x)u + e(x)d \\ 0 \\ 0 \end{pmatrix}, \quad g_e(x_e) = \begin{pmatrix} 0 \\ 1 \\ 0 \end{pmatrix},$$
$$e_e(x_e) = \begin{pmatrix} 0 \\ 0 \\ I_2 \end{pmatrix}, \quad h_e(x_e) = x.$$

For this system  $\Sigma_e$  we calculate the minimal involutive conditioned invariant distribution containing the input vector field  $\frac{\partial}{\partial u}$ , denoted by  $\mathcal{S}_e^*$ , by applying the  $\mathcal{S}^*$  algorithm (see Subsection 5.2.2).

In the first step of the algorithm we obtain

$$S_{1e} = \operatorname{span}\left\{\frac{\partial}{\partial u}\right\} = \operatorname{span}\left\{\begin{pmatrix}0\\0\\0\\0\\1\\0\\0\end{pmatrix}\right\}$$

which is clearly contained in ker dx. Therefore we calculate in the second step

$$S_{2e} = S_{1e} + [f_e(x_e), S_{1e}]$$
  
= span  $\left\{ \frac{\partial}{\partial u}, g(x) \frac{\partial}{\partial x} \right\}$   
= span  $\left\{ \frac{\partial}{\partial u}, \frac{1}{M + m \sin^2 \varphi} \frac{\partial}{\partial \dot{q}} - \frac{\cos \varphi}{l(M + m \sin^2 \varphi)} \frac{\partial}{\partial \dot{\varphi}} \right\}$ 

$$= \operatorname{span} \left\{ \begin{pmatrix} 0 \\ 0 \\ 0 \\ 0 \\ 1 \\ 0 \\ 0 \end{pmatrix}, \begin{pmatrix} 0 \\ \frac{1}{M + m \sin^2 \varphi} \\ 0 \\ -\frac{\cos \varphi}{l(M + m \sin^2 \varphi)} \\ 0 \\ 0 \\ 0 \end{pmatrix} \right\}$$

Because  $\overline{S}_{2e} = S_{2e}$  and  $S_{2e} \cap \ker dx = S_{1e}$  we end up with  $S_e^* = S_{2e}$ . Clearly the distributions  $S_e^*$  and  $S_e^* \cap \ker dx$  are constant dimensional and satisfy the projectability conditions (see Subsection 5.2.2). The projections of the distributions  $S_e^*$  and  $S_e^* \cap \ker dx$  onto the state space of the original system  $\Sigma$ ,  $X_2 \oplus X_3$  and  $X_2$  respectivily, are given by:

$$X_{2} = 0;$$

$$X_{2} \oplus X_{3} = \operatorname{span} \left\{ \frac{1}{M + m \sin^{2} \varphi} \frac{\partial}{\partial \dot{p}} - \frac{\cos \varphi}{l(M + m \sin^{2} \varphi)} \frac{\partial}{\partial \dot{\varphi}} \right\}$$

$$= \operatorname{span} \left\{ \begin{pmatrix} 0 \\ \frac{1}{M + m \sin^{2} \varphi} \\ 0 \\ -\frac{\cos \varphi}{l(M + m \sin^{2} \varphi)} \end{pmatrix} \right\}.$$

Accordingly we define new coordinates

$$\begin{aligned} \xi_1 &= \left(\frac{l^2}{l^2 + \cos^2\varphi}\right)^{\frac{1}{2}} \left(\frac{\cos\varphi}{l}\dot{q} + \dot{\varphi}\right), \\ \xi_2 &= q, \\ \xi_3 &= \varphi, \\ \xi_4 &= \left(\frac{l^2}{l^2 + \cos^2\varphi}\right)^{\frac{1}{2}} \left(-\dot{q} + \frac{\cos\varphi}{l}\dot{\varphi}\right). \end{aligned}$$

This defines a globally defined state transformation  $\xi = S(x)$  since

$$\left|\frac{\partial S}{\partial x}(x)\right| = \begin{vmatrix} 0 & \left(\frac{l^2}{l^2 + \cos^2\varphi}\right)^{\frac{1}{2}} \frac{\cos\varphi}{l} & * & \left(\frac{l^2}{l^2 + \cos^2\varphi}\right)^{\frac{1}{2}} \\ 1 & 0 & 0 & 0 \\ 0 & 0 & 1 & 0 \\ 0 & -\left(\frac{l^2}{l^2 + \cos^2\varphi}\right)^{\frac{1}{2}} & * & \left(\frac{l^2}{l^2 + \cos^2\varphi}\right)^{\frac{1}{2}} \frac{\cos\varphi}{l} \end{vmatrix} = -1$$

for all x. Then we can choose the distribution  $X_1$  such that:

$$\begin{aligned} \mathcal{X}_1 &= \operatorname{span} \left\{ \frac{\partial}{\partial \xi_1}, \frac{\partial}{\partial \xi_2}, \frac{\partial}{\partial \xi_3} \right\}; \\ \mathcal{X}_2 &= 0; \\ \mathcal{X}_3 &= \operatorname{span} \left\{ \frac{\partial}{\partial \xi_4} \right\}. \end{aligned}$$

Now we calculate the expression for the system  $\Sigma$  in this new coordinates. For simplicity we introduce the following notation

$$n(\varphi) := \left(\frac{l^2}{l^2 + \cos^2 \varphi}\right)^{\frac{1}{2}}.$$

Then the system  $\Sigma$  is given by

.

$$\dot{\xi}_{1} = n(\xi_{3}) \frac{g \sin \xi_{3}}{l} + n^{3}(\xi_{3}) \frac{\sin \xi_{3}}{l} \xi_{1} \xi_{4} + n^{3}(\xi_{3}) \frac{\sin \xi_{3} \cos \xi_{3}}{l^{2}} \xi_{4}^{2} + n(\xi_{3}) \frac{\cos \xi_{3}}{l} d_{1} + n(\xi_{3}) d_{2}, \qquad (6.40)$$

$$\dot{\xi}_2 = n(\xi_3) \left( \frac{\cos \xi_3}{l} \xi_1 - \xi_4 \right),$$
 (6.41)

$$\dot{\xi}_3 = n(\xi_3) \left( \xi_1 + \frac{\cos \xi_3}{l} \xi_4 \right),$$
  
(6.42)

$$\dot{\xi}_{4} = n^{-1}(\xi_{3}) \frac{mg\sin\xi_{3}\cos\xi_{3}}{M+m\sin^{2}\xi_{3}} - n(\xi_{3}) \frac{ml\sin\xi_{3}\left(\xi_{1} + \frac{\cos\xi_{3}}{l}\xi_{4}\right)}{M+m\sin^{2}\xi_{3}} + \frac{F\left(\frac{\cos\xi_{3}}{l}\xi_{1} - \xi_{4}\right)}{M+m\sin^{2}\xi_{3}} + n(\xi_{3}) \frac{g\sin\xi_{3}\cos\xi_{3}}{l^{2}} -n^{3}(\xi_{3}) \frac{\sin\xi_{3}}{l}\left(\xi_{1} + \frac{\cos\xi_{3}}{l}\xi_{4}\right)\xi_{1} -n^{-1}(\xi_{3}) \frac{u}{M+m\sin^{2}\xi_{3}} - n(\xi_{3})d_{1} + n(\xi_{3}) \frac{\cos\xi_{3}}{l}d_{2}, \quad (6.43) \left(\sum_{n(\xi_{3})}^{\xi_{2}}\left(\sum_{n(\xi_{3})}^{\xi_{2}}\xi_{1} - \xi_{1}\right)\right)$$

$$z = \begin{pmatrix} n(\xi_3) \left( \frac{\cos \xi_3}{l} \xi_1 - \xi_4 \right) \\ \xi_3 \\ n(\xi_3) \left( \xi_1 + \frac{\cos \xi_3}{l} \xi_4 \right) \end{pmatrix}.$$
(6.44)

This system can be seen as an interconnection of two subsystems. The first system  $\Sigma_1$  has as state  $(\xi_1, \xi_2, \xi_3)$ , inputs  $(\xi_4, d_1, d_2)$  and output z. The second system  $\Sigma_2$  has state  $\xi_4$ , inputs  $(u, \xi_1, \xi_2, \xi_3, d_1, d_2)$  and output  $\xi_4$ . As described in Subsection 5.2.4 we start by solving the regular  $\mathcal{H}_{\infty}$  problem for the subsystem  $\Sigma_1$ . Note that the input  $\xi_4$  does not appear linearly in the equations for  $\Sigma_1$ . For simplicity of calculations we linearize the  $\xi_1$ -dynamics with respect to the input  $\xi_4$  around  $\xi_4 = 0$ . This comes down to neglecting the quadratic term in  $\xi_4$ . The (linearized) equations for  $\Sigma_1$ , with the state  $(\xi_1, \xi_2, \xi_3)$  denoted by  $\eta$ , are then given by

$$\Sigma_1 \begin{cases} \dot{\eta} = f_1(\eta) + g_1(\eta)\xi_4 + e_1(\eta)d \\ z = h_1(\eta) + k_1(\eta)\xi_4 \end{cases}$$

where

$$f_{1}(\eta) = \begin{pmatrix} n(\xi_{3}) \frac{g\sin\xi_{3}}{l} \\ n(\xi_{3}) \frac{\cos\xi_{3}}{l} \xi_{1} \\ n(\xi_{3})\xi_{1} \end{pmatrix}, \quad g_{1}(\eta) = \begin{pmatrix} n^{3}(\xi_{3}) \frac{\sin\xi_{3}}{l} \xi_{1} \\ -n(\xi_{3}) \\ n(\xi_{3}) \frac{\cos\xi_{3}}{l} \\ n(\xi_{3}) \frac{\cos\xi_{3}}{l} \end{pmatrix}, \quad g_{1}(\eta) = \begin{pmatrix} n^{3}(\xi_{3}) \frac{\sin\xi_{3}}{l} \\ -n(\xi_{3}) \\ n(\xi_{3}) \frac{\cos\xi_{3}}{l} \\ 0 \\ 0 \end{pmatrix}, \quad d = \begin{pmatrix} d_{1} \\ d_{2} \end{pmatrix}, \quad h_{1}(\eta) = \begin{pmatrix} \xi_{2} \\ n(\xi_{3}) \frac{\cos\xi_{3}}{l} \xi_{1} \\ \xi_{3} \\ n(\xi_{3}) \xi_{1} \end{pmatrix}, \quad k_{1}(\eta) = \begin{pmatrix} 0 \\ -n(\xi_{3}) \\ 0 \\ n(\xi_{3}) \frac{\cos\xi_{3}}{l} \\ 0 \\ n(\xi_{3}) \frac{\cos\xi_{3}}{l} \\ 0 \end{pmatrix}.$$

The system  $\Sigma_1$  satisfies the standard  $\mathcal{H}_{\infty}$  assumptions, namely  $k_1^T(\eta)h_1(\eta) = 0$ and  $k_1^T(\eta)k_1(\eta) = 1$ . Therefore solving the  $L_2$ -gain problem with constant  $\gamma$ for this system  $\Sigma_1$  comes down to finding a solution  $V \ge 0$  to the following Hamilton-Jacobi equality

$$V_{\eta}(\eta) f_{1}(\eta) + \frac{1}{2} h_{1}^{T}(\eta) h_{1}(\eta) \qquad (6.45)$$
  
+  $\frac{1}{2} V_{\eta}(\eta) \left( \frac{1}{\gamma^{2}} e_{1}(\eta) e_{1}^{T}(\eta) - g_{1}(\eta) g_{1}^{T}(\eta) \right) V_{\eta}^{T}(\eta) = 0.$ 

Similar as in Subsection 6.2.1 an approximate 4th order polynomial solution V to this Hamilton-Jacobi equality (6.45) will be calculated. The corresponding approximate solution  $\xi_4^*(\eta)$  to the regular state feedback  $\mathcal{H}_{\infty}$  problem for the

regular subsystem  $\Sigma_1$  will be used to define an auxiliary system  $\Sigma_P$  as done in Subsection 5.2.7. It can be shown that

$$\frac{1}{2} \left( \xi_4 - \xi_4^*(\eta) \right)^T \left( \xi_4 - \xi_4^*(\eta) \right) = V_\eta(\eta) f_1(\eta) + \frac{1}{2} h_1^T(\eta) h_1(\eta) \\ + \frac{1}{2} V_\eta(\eta) \left( \frac{1}{\gamma^2} e_1(\eta) e_1^T(\eta) - g_1(\eta) g_1^T(\eta) \right) V_\eta^T(\eta).$$

Now we define the new variable

$$q = \xi_4 - \xi_4^*(\eta). \tag{6.46}$$

From Subsection 5.2.7 it follows that the state feedback  $\mathcal{H}_{\infty}$  problem with gain  $\gamma$  is solvable for  $\Sigma$  if the same problem is solvable for the auxiliary system

$$\Sigma_{P} \begin{cases} \dot{\eta} = f_{1}(\eta) + \frac{1}{\gamma^{2}} e_{1}(\eta) e_{1}^{T}(\eta) V_{\eta}^{T}(\eta) + g_{1}(\eta) \left(q + \xi_{4}^{*}(\eta)\right) + e_{1}(\eta) d_{p} \\ \dot{q} = f_{q}(\eta, q) + g_{q}(\eta) u + e_{q}(\eta) d_{p} \\ z = q \end{cases}$$

where  $f_q$ ,  $g_q$ ,  $e_q$  follows from (6.43), (6.37) and  $\Sigma_1$  when we choose

$$d_p = d - \frac{1}{\gamma^2} e_1^T(\eta) V_{\eta}^T(\eta).$$

This singular  $\mathcal{H}_{\infty}$  control problem can be solved by using the feedback construction derived in [MRST 94] and [MT 95] (see also Chapter 4). The applied feedback is given by

$$u = -g_q^{-1}(\eta) f_q(\eta) - g_q^{-1}(\eta) q - \frac{1}{4} \frac{1}{\gamma^2} g_q^{-1}(\eta) q \left( e_q(\eta) e_q^T(\eta) \right)$$

which together with  $\Sigma_P$  and hence also together with  $\Sigma$  leads to a closed-loop system which has  $L_2$ -gain less than or equal to  $\gamma$ . In fact we had to make an extra 4th order approximation of the feedback which was applied to the original system to reduce the complexity of the final differential equation. Compared to the calculation of the feedback using the cheap control approach Mathematica used quite some time to calculate an approximate feedback using this geometric approach.

#### 6.2.3 Comparison of the two controllers

We will compare the feedbacks constructed by the cheap control approach on the one hand and the geometric approach on the other hand for two different values of  $\gamma$ . First we consider  $\gamma = 0.8$  which is close to the optimal  $\gamma^*$  for the linearized system when we let  $\varepsilon$  go to zero. Therefore we have to choose  $\varepsilon$ small to be able to find an approximate solution to the Hamilton-Jacobi equality using the iterative method described in [Vr 94]. In Table 6.5 we see that the geometric approach yields a greater region of attraction. Probably this is due to the fact that in the cheap control approach we still penalize the inputs which avoids the gain of the feedback to become (too) high. The parameter robustness of the two methods is almost the same. In Table 6.4 we see that both the closedloop systems corresponding to the two different feedbacks are very robust with respect to uncertainties in the parameters *m*, *F* and *M*. Some simulation results comparing the two methods are shown in Figure 6.3 and 6.4.

	m <sub>nom</sub>	m <sub>max</sub>	F <sub>nom</sub>	F <sub>max</sub>	<i>M</i> <sub>nom</sub>	<i>M</i> <sub>max</sub>
Cheap control approach	0.1	340	0.1	910	1	38
Geometric approach	0.1	260	0.1	920	1	100

Table 6.4: robustness with respect to parameter uncertainties in *m*, *F* and *M* for  $\gamma = 0.8 \ (\varepsilon = 0.0001, x(0) = (\frac{1}{10} \ 0 \ 0 \ 0))$ 

	$q_{\max}(0)$	$\varphi_{\max}(0)$
Cheap control approach	0.36	0.084
Geometric approach	4.0	0.68

Table 6.5: stability results for  $\gamma = 0.8$  ( $\varepsilon = 0.0001$ )

To evaluate these results for different values of the computed  $L_2$ -gain we considered a second value of  $\gamma$  being  $\gamma = 1.5$ . As seen before this leads to an increase of the region of attraction (Table 6.7). The response for non-zero initial condition ( $x(0) = (1\ 0\ 0\ 0)$ ) are shown in Figure 6.5. It can be seen that the response corresponding to the feedback calculated using the geometric approach is slower and the corresponding input is a bit smoother. For the robustness properties with respect to the parameters m, F and M some more realistic values are computed, as shown in Table 6.6.

Remarkable are the extreme differences in robustness between the two values of  $\gamma$ . These differences are not only due to the differences in the initial condition in the two experiments. For  $\gamma = 1.5$  the robustness properties are still very good. Variations up to 100 times the nominal values for the param-

.

	m <sub>nom</sub>	m <sub>max</sub>	Fnom	F <sub>max</sub>	<i>M</i> <sub>nom</sub>	M <sub>max</sub>
Cheap control approach	0.1	10	0.1	14	1	6.0
Geometric approach	0.1	2.5	0.1	9.0	1	2.1

Table 6.6: robustness with respect to parameter uncertainties in *m*, *F* and *M* for  $\gamma = 1.5$  ( $\varepsilon = 0.2$ ,  $x(0) = (1\ 0\ 0\ 0)$ )

	$q_{\max}(0)$	$\varphi_{\max}(0)$
Cheap control approach	1.3	0.16
Geometric approach	12	0.88

Table 6.7: stability results for  $\gamma = 1.5$  ( $\varepsilon = 0.2$ )

eters still lead to a stable closed-loop behaviour. For  $\gamma = 1.5$  we also see in Table 6.6 that the feedback constructed using the cheap control approach leads to better robustness properties than the feedback constructed with the geometric approach. Again it should be noted that this example is mainly meant to illustrate the two methods and to make some comparing remarks about possible differences. We did not pretend to make a feedback design which satisfies certain design objectives.



Figure 6.3: input, position and angle corresponding to parameter robustness with respect to *m* for the closed-loop system together with the cheap control feedback (solid) and the geometric feedback (dashed) against time (horizontal axis) for  $\gamma = 0.8$  (m = 250 instead of the nominal value m = 0.1)( $\varepsilon = 0.0001$ ,  $x(0) = (\frac{1}{10} \ 0 \ 0 \ 0)$ )



Figure 6.4: input, position and angle corresponding to parameter robustness with respect to *M* for the closed-loop system together with the cheap control feedback (solid) and the geometric feedback (dashed) against time (horizontal axis) for  $\gamma = 0.8$  (M = 35 instead of the nominal value M = 1)( $\varepsilon = 0.0001$ ,  $x(0) = (\frac{1}{10} \ 0 \ 0 \ 0)$ )

163



Figure 6.5: response for non-zero initial condition for the closed-loop system together with the cheap control feedback (solid) and the geometric feedback (dashed) against time (horizontal axis) for  $\gamma = 1.5$  ( $\varepsilon = 0.2$ ,  $x(0) = (1\ 0\ 0\ 0)$ )

## Chapter 7

# **Robust stabilization under gain-bounded uncertainties**

Several robustness problems can be described as a  $\mathcal{H}_{\infty}$  control problem. For instance, robust stabilization under additive perturbations, parameter uncertainties, or multiplicative perturbations, are all examples of robust control problems under gain-bounded uncertainties (nonlinear systems: [vdS 95], [ST 95], [IsTa 95]; linear systems: [XS 90], [XS 92]), which can be described as  $\mathcal{H}_{\infty}$ control problems. Also other types of uncertainty can be captured into the  $\mathcal{H}_{\infty}$ framework such as gain-bounded and Lipschitz bounded uncertainties as in the paper [Ng 96a], or uncertainties which satisfy a certain integral functional constraint ([Ng 96b]), and robust stability and robust performance problems (see [AsGu 93]). Most of these problems, however, give rise to regular  $\mathcal{H}_{\infty}$  control problems, for which we do not need the theory developed in this book.

In this chapter we apply the results derived in this book to two problems concerned with the stabilization of an uncertain system which do lead to singular  $\mathcal{H}_{\infty}$  problems. The results from the Chapters 3 and 5 are applied to systems with two different types of gain-bounded uncertainty:

- Parameter uncertainties;
- Multiplicative uncertainties.

#### 7.1 Parameter uncertainty

Consider the system

$$\dot{x} = f(x,\theta) + g(x)u \tag{7.1}$$

where *x* is the *n*-dimensional state vector in local coordinates and  $u \in \mathbb{R}^m$ . The matrix  $\theta \in \mathbb{C}^{q \times p}$  contains the uncertain parameters. So  $\theta$  is unknown and constant.

First we assume that we are able to measure the state completely. We assume the following linear dependency of f on the uncertain parameters  $\theta$ 

$$f(x,\theta) = f(x,\theta_{\text{nom}}) + e(x)\left(\theta - \theta_{\text{nom}}\right)h(x)$$
(7.2)

for some known *p*-dimensional vector h(x) and  $n \times q$ -matrix e(x) where  $\theta_{\text{nom}}$  denotes the nominal value of  $\theta$ . We will denote  $f(x) = f(x, \theta_{\text{nom}})$ . These kind of uncertainties are considered in [HiPr 86] for linear systems and are referred to as structured perturbations.



Figure 7.1: parameter perturbed system with feedback

Under assumption (7.2) the perturbed system (7.1) can be rewritten as

$$\Sigma^{\text{pert}} \begin{cases} \dot{x} = f(x) + g(x)u + e(x)d\\ z = h(x)\\ y = x \end{cases}$$
(7.3)

where d is given by  $d = \Delta \theta_z$  with  $\Delta \theta = \theta - \theta_{nom}$  the constant matrix specifying the parameter uncertainties.

Then the *robust stabilization problem* is to find a feedback such that the closed loop system is  $L_2$ -stable for the largest possible class of perturbations  $\Delta \theta$ , where "largest" refers to the  $L_2$ -norm of  $\Delta \theta$ , which, since  $\Delta \theta$  is constant, is the maximal singular value of  $\Delta \theta$ . This robust stabilization problem comes

166

down to finding a feedback (F)

$$u = \sigma(x) \tag{7.4}$$

such that the  $L_2$ -gain of the closed loop system (7.3) and (7.4) from d to z is minimized to  $\gamma^*$ . Using the small-gain theorem (see e.g. [DeVi 75]) this means that the closed-loop system is  $L_2$ -stable for all perturbations  $\Delta\theta$  with maximal singular value of  $\Delta\theta$  strictly less than  $\frac{1}{\nu^*}$ .

Note that  $\Delta \theta$  is a real matrix, while the small-gain theorem allows for timevarying  $\Delta \theta$  or complex  $\Delta \theta$ . Hence we actually obtain conservative bounds on the real perturbations  $\Delta \theta$  (see [HiPr 88], [HiPr 90], [ToRy 91]).

The problem of minimizing the  $L_2$ -gain from d to z = h(x) is a singular state feedback  $\mathcal{H}_{\infty}$  optimal nonlinear optimal control problem as studied in this monograph. For the solution to the suboptimal problem using a cheap control approach we can formulate the following result.

**Theorem 7.1** Suppose there exist a constant  $\varepsilon > 0$  and a solution  $V \ge 0$  to

$$V_{x}(x) f(x) + \frac{1}{2}h^{T}(x)h(x)$$

$$\frac{1}{2}V_{x}(x) \left[\frac{1}{\gamma^{2}}e(x)e^{T}(x) - \frac{1}{\varepsilon}g(x)g^{T}(x)\right]V_{x}^{T}(x) \leq 0$$

$$V(0) = 0$$
(7.5)

and the closed loop system with the feedback

$$u = -\frac{1}{\varepsilon}g^{T}(x)V_{x}^{T}(x)$$
(7.6)

is zero-state observable. Then the feedback (7.6) locally asymptotically stabilizes the origin of the closed-loop system (7.3), (7.6) for every perturbation  $\Delta \theta$ with maximal singular value less than  $\frac{1}{\gamma}$ .

**Remark 7.2** If the solution  $V \ge 0$  of (7.5) is also proper then the feedback (7.6) globally asymptotically stabilizes the closed loop system.

Under Assumption 3 (in Subsection 3.2.1) we can also formulate a converse result:

**Theorem 7.3** Suppose there exists a feedback (7.4) which locally stabilizes the closed loop system (7.3), (7.4). Assume that Assumption 3 (see Subsection 3.2.1) is satisfied. Then there exist a constant  $\varepsilon > 0$  and a solution  $V \ge 0$  to the Hamilton-Jacobi inequality (7.5).

168

The geometric approach to this singular state feedback  $\mathcal{H}_{\infty}$  control problem is described in Chapter 5. As in Subsection 5.2.2 we consider the following extended version of the system  $\Sigma^{pert}$ 

$$\Sigma_e^{\text{pert}} \begin{cases} \dot{x} = f(x) + g(x)u + e(x)d \\ \dot{u} = v \\ \dot{d} = w \\ z = h(x) \\ y = x \end{cases}$$

For this extended system  $\Sigma_e^{\text{pert}}$  we construct the minimal involutive conditioned invariant distribution containing the input vector fields, denoted as  $S_e^*$ , as described in Subsection 5.2.2. We assume that the distributions  $S_e^*$  and  $S_e^* \cap \text{kerd}h$  are constant dimensional and that

$$\left[\frac{\partial}{\partial d_j}, S_e^*\right] \subset S_e^*, \qquad j = 1, \dots, q.$$
(7.7)

Then as we have seen Section 5.2 there exist local coordinates such that the system  $\Sigma^{\text{pert}}$  is of the form

$$\Sigma^{\text{pert}} \begin{cases} \dot{x}_1 = f_1(x_1, x_3) + e_1(x_1, x_3)d \\ \dot{x}_2 = f_2(x_1, x_2, x_3) + g_2(x_1, x_2, x_3)u + e_2(x_1, x_2, x_3)d \\ \dot{x}_3 = f_3(x_1, x_2, x_3) + g_3(x_1, x_2, x_3)u + e_3(x_1, x_2, x_3)d \\ z = h(x_1, x_3) \end{cases}$$
(7.8)

In these new coordinates the perturbed system with the to be constructed feedback has the form shown in Figure 7.2, where the subsystem  $\Sigma_1^{\text{pert}}$  is given by

$$\Sigma_1^{\text{pert}} \begin{cases} \dot{x}_1 = f_1(x_1, x_3) + e_1(x_1, x_3)d \\ z = h(x_1, x_3) \end{cases}$$

and the subsystem  $\Sigma_2^{\text{pert}}$  by the  $x_2$  and  $x_3$  dynamics of (7.8). Then as in Chapter 5 the singular  $\mathcal{H}_{\infty}$  control problem can be split in two steps. First we consider



Figure 7.2: parameter perturbed system with feedback after transformation of the state

a regular  $L_2$ -gain control problem for the system  $\Sigma_1^{\text{pert}}$ . The pre-Hamiltonian corresponding to this problem for  $L_2$ -gain less than or equal to  $\overline{\gamma}$  is given by

$$K_{\overline{\gamma}}(x_1, p_1, d, x_3) = p_1^T (f_1(x_1, x_3, 0) + e_1(x_1, x_3)d) + \frac{1}{2}h^T(x_1, x_3)h(x_1, x_3) - \frac{1}{2}\frac{1}{\overline{\gamma}^2}d^Td.$$

The saddle point solution of this pre-Hamiltonian, given by  $x_3 = x_3^*(x_1, p_1)$ ,  $d^* = \frac{1}{\overline{y}^2} e_1(x_1, x_3^*(x_1, p_1)) e_1^T(x_1, x_3^*(x_1, p_1)) p_1$ , can be found by solving

$$\frac{\partial K_{\overline{Y}}}{\partial d} = 0, \qquad \frac{\partial K_{\overline{Y}}}{\partial x_3} = 0.$$

Substitution of this saddle point solution into the pre-Hamiltonian results in the Hamiltonian

$$H_{\overline{\gamma}}(x_1, p_1) = p_1^T f_1(x_1, x_3^*(x_1, p_1)) \\ + \frac{1}{2} \frac{1}{\overline{\gamma}^2} p_1^T e_1(x_1, x_3^*(x_1, p_1)) e_1^T(x_1, x_3^*(x_1, p_1)) p_1 \\ + \frac{1}{2} h^T(x_1, x_3^*(x_1, p_1)) h(x_1, x_3^*(x_1, p_1)).$$

Then the following results can be stated (see Theorem 5.22).

**Theorem 7.4** Consider the system  $\Sigma_e^{\text{pert}}$ . Suppose that the distributions  $S_e^*$  and  $S_e^* \cap \text{kerdh}$  for the system  $\Sigma_e^{\text{pert}}$  are constant dimensional, and that condition (7.7) holds. Furthermore assume there exists a solution  $V \ge 0$  to the Hamilton-Jacobi inequality

$$H_{\overline{\gamma}}(x_1, V_{x_1}^T(x_1)) \leq 0$$

for some constant  $\overline{\gamma} < \gamma$ . Then there exists a constant  $k^*$  such that the robust stabilization problem is solvable for all disturbances  $\Delta \theta$  with maximal singular value less than or equal to  $\frac{1}{\gamma}$  by the feedback

$$u = \beta(x_1, x_2, x_3, k)$$

for  $k > k^*$ , if the parameterized feedback

$$u = \beta(x_1, x_2, x_3, k)$$

solves the  $\mathcal{H}_{\infty}$  almost disturbance decoupling problem for the system  $\Sigma^{\text{pert}}$  with output  $x_3 - x_3^*(x_1, V_{x_1}^T(x_1))$ .

On the other hand the factorization approach from Subsection 5.2.7 can be used to rewrite the robust stabilization problem into the following form. Assume there exists a solution  $V \ge 0$ , V(0) = 0, to the Hamilton-Jacobi equality

$$H_{\gamma}(x_1, V_{x_1}^T(x_1)) = 0.$$

We consider the auxiliary system

$$\Sigma_{p}^{\text{pert}} \begin{cases} \dot{x}_{1} = f_{1}(x_{1}, x_{3}) + \frac{1}{\gamma^{2}} e_{1}(x_{1}, x_{3}) e_{1}^{T}(x_{1}, x_{3}) V_{x_{1}}^{T}(x_{1}) + e_{1}^{T}(x_{1}, x_{3}) d_{p} \\ \dot{x}_{2} = f_{2}(x_{1}, x_{2}, x_{3}) + \frac{1}{\gamma^{2}} e_{2}(x_{1}, x_{2}, x_{3}) e_{1}^{T}(x_{1}, x_{3}) V_{x_{1}}^{T}(x_{1}) \\ + g_{2}(x_{1}, x_{2}, x_{3}) u + e_{2}(x_{1}, x_{2}, x_{3}) d_{p} \\ \dot{x}_{3} = f_{3}(x_{1}, x_{2}, x_{3}) + \frac{1}{\gamma^{2}} e_{3}(x_{1}, x_{2}, x_{3}) e_{1}^{T}(x_{1}, x_{3}) V_{x_{1}}^{T}(x_{1}) \\ + g_{3}(x_{1}, x_{2}, x_{3}) u + e_{3}(x_{1}, x_{2}, x_{3}) d_{p} \\ z = h_{p}(x_{1}, x_{3}) \end{cases}$$

where

$$d_p = d - \frac{1}{\gamma^2} e_1^T(x_1, x_3) V_{x_1}^T(x_1)$$

and the new output equation  $h_p$  is such that

$$\frac{1}{2}h_{p}^{T}(x_{1}, x_{3})h_{p}(x_{1}, x_{3}) = V_{x_{1}}^{T}(x_{1})f_{1}(x_{1}, x_{3}) + \frac{1}{2}\frac{1}{\gamma^{2}}V_{x_{1}}^{T}(x_{1})e_{1}(x_{1}, x_{3})e_{1}^{T}(x_{1}, x_{3})V_{x_{1}}(x_{1}) \\
+ \frac{1}{2}h^{T}(x_{1}, x_{3})h(x_{1}, x_{3}).$$
(7.9)

170
Recall that a smooth output equation  $h_p$  locally exists if the Hessian of the right hand part of equation (7.9) has constant rank around the origin. The following result can be stated (see Subsection 5.2.7).

**Theorem 7.5** Assume there exists a solution  $V \ge 0$  to

$$H_{\gamma}(x_1, V_{x_1}^T(x_1)) = 0.$$

Consider the systems  $\Sigma_P^{\text{pert}}$  and  $\Sigma_P^{\text{pert}}$ . The following statements are equivalent:

- (i) the robust stabilization problem for the system  $\Sigma^{\text{pert}}$  is solvable for all perturbations  $\Delta \theta$  with maximal singular value less than or equal to  $\frac{1}{\gamma}$  with a differentiable storage function W + V,  $W \ge 0$ ;
- (ii) the robust stabilization problem for the system  $\Sigma_p^{\text{pert}}$  is solvable for all perturbations  $\Delta \theta$  with maximal singular value less than or equal to  $\frac{1}{\gamma}$  with a differentiable storage function  $W \ge 0$ .

. 1		

If the full state is not available for measurements we can try to solve the robust stabilization problem under parameter uncertainties using the worst case certainty equivalence principle described in Section 3.3. Consider the following system

$$\dot{x} = f(x,\theta) + g(x)u$$
  

$$y = c(x,\eta)$$
(7.10)

where the uncertainty is given by the constant matrix  $\theta$  and by the constant vector  $\eta$ . We assume that  $f(x, \theta)$  satisfies (7.2) and furthermore we assume that c can be described as:

$$c(x, \eta) = c(x, \eta_{\text{nom}}) + (\eta - \eta_{\text{nom}}) h(x)$$
(7.11)

where we denote  $\Delta \eta := \eta - \eta_{\text{nom}}$  and  $c(x) := c(x, \eta_{\text{nom}})$ . Then the perturbed system is

$$\Sigma^{\text{pert}} \begin{cases} \dot{x} = f(x) + g(x)u + e(x)d_1 \\ z = h(x) \\ y = c(x) + d_2 \end{cases}$$
(7.12)

where  $d_1 = \Delta \theta z$  and  $d_2 = \Delta \eta z$ .

Then the *robust stabilization problem* is to find a compensator such that the closed loop system is  $L_2$ -stable for the largest possible class of perturbations  $\Delta\theta$  and  $\Delta\eta$ , where again "largest" refers to the  $L_2$ -norm of  $\Delta\theta$  and  $\Delta\eta$ , which, since  $\Delta\theta$  and  $\Delta\eta$  are constant is the maximal singular value of  $\Delta\theta$  respectively  $\Delta\eta$ . This robust stabilization problem comes down to finding a compensator *C* 

172

$$C \begin{cases} \dot{\xi} = p(\xi, y), & p(0, 0) = 0\\ u = q(\xi, y), & q(0, 0) = 0 \end{cases}$$
(7.13)

such that the  $L_2$ -gain of the closed loop system (7.12) and (7.13) from  $d_1$ ,  $d_2$  to z is minimized.



Figure 7.3: parameter perturbed system with controller

The suboptimal measurement feedback  $\mathcal{H}_{\infty}$  control problem for the system  $\Sigma^{\text{pert}}$  is (as we have seen in Section 3.3) locally solvable if there exists a constant  $\varepsilon > 0$  and a solution  $V \ge 0$  to (7.5) with equality and a solution  $W \ge 0$  to (3.36) satisfying:

$$f - \frac{1}{\varepsilon} gg^T V_x^T + \frac{1}{\gamma^2} ee^T V_x^T \text{ is exponentially stable;} - \left(f + \frac{1}{\gamma^2} ee^T W_x^T\right) \text{ is exponentially stable;}$$
(7.14)  
$$W_{xx}(x) > V_{xx}(x), \quad \forall x.$$

Hence we can state the following result.

**Theorem 7.6** Suppose there exists a constant  $\varepsilon > 0$  and a solution  $V \ge 0$  to (7.5) with equality and a solution  $W \ge 0$  to (3.36) with satisfies (7.14). Assume that the certainty equivalence principle holds for  $\Sigma^{\text{pert}}$ . Then the controller

$$\dot{\xi} = f(\xi) - \frac{1}{\varepsilon} g(\xi) g^{T}(\xi) V_{\xi}^{T}(\xi) + \frac{1}{\gamma^{2}} e(\xi) e^{T}(\xi) V_{\xi}^{T}(\xi) + \gamma^{T} \left[ W_{\xi\xi}(\xi) - V_{\xi\xi}(\xi) \right]^{-1} \frac{\partial c^{T}}{\partial \xi} (\xi) (y(t) - c(\xi)), \quad (7.15)$$
$$u = -\frac{1}{\varepsilon} g^{T}(\xi) V_{\xi}^{T}(\xi)$$

locally asymptotically stabilizes the closed loop system (7.12), (7.15) for every perturbation  $\Delta\theta$ ,  $\Delta\eta$  having maximal singular values less than  $\gamma$ .

#### 7.2 Multiplicative perturbations

Consider the system

$$\Sigma \begin{cases} \dot{x} = f(x) + g(x)u\\ \bar{y} = h(x) \end{cases}$$
(7.16)

and suppose the output  $\bar{y}$  is perturbed by a disturbance d. We assume that this disturbance is the output of an arbitrary nonlinear system with input  $\bar{y}$ :

$$\Delta \begin{cases} \dot{\varphi} = \alpha(\varphi, \bar{y}) \\ d = \beta(\varphi, \bar{y}) \end{cases}$$
(7.17)

Then from the interconnection shown in Figure 7.4 we see that the perturbed system is given by

$$\Sigma^{\text{pert}} \begin{cases} \dot{x} = f(x) + g(x)u \\ y = h(x) + d \\ z = h(x) \end{cases}$$
(7.18)

Now the robust stabilization problem is, similar to the previous section, to find a compensator such that the closed loop system is  $L_2$ -stable for the largest possible class of perturbations  $\Delta$ , where again "largest" refers to the  $L_2$ -norm



Figure 7.4: multiplicative perturbed system with controller

of  $\Delta$ . This robust stabilization problem comes down to finding a compensator C

$$C\begin{cases} \dot{\xi} = p(\xi, y)\\ u = q(\xi, y) \end{cases}$$
(7.19)

such that the  $L_2$ -gain of the closed loop system (7.18), (7.19) from d to z is minimized.

In Section 3.3 we have seen that the suboptimal measurement feedback  $\mathcal{H}_{\infty}$  control problem for the system  $\Sigma^{\text{pert}}$  is locally solvable if there exists a constant  $\varepsilon > 0$  and a solution  $V \ge 0$  to

$$V_x(x)f(x) - \frac{1}{2\varepsilon}V_x(x)g(x)g^T(x)V_x^T(x) + \frac{1}{2}h^T(x)h(x) = 0$$
  
$$V(0) = 0 \quad (7.20)$$

and a solution  $W \ge 0$  to

$$W_x(x)f(x) + \frac{1}{2}(1-\gamma^2)h^T(x)h(x) = 0$$
  
W(0) = 0 (7.21)

such that:

$$f - \frac{1}{\varepsilon}gg^{T}V_{x}^{T} \text{ is exponentially stable;}$$
  
- f is exponentially stable; (7.22)  
$$W_{xx}(x) > V_{xx}(x), \quad \forall x.$$

Summarizing:

174

**Theorem 7.7** Suppose there exist a constant  $\varepsilon > 0$  and a solution  $V \ge 0$  to (7.20) and a solution  $W \ge 0$  to (7.21) which satisfies (7.22). Assume that the certainty equivalence principle holds for  $\Sigma^{\text{pert}}$ . Then the controller

$$\dot{\xi} = f(\xi) - \frac{1}{\varepsilon} g(\xi) g^{T}(\xi) V_{\xi}^{T}(\xi) + \gamma^{T} \left[ W_{\xi\xi}(\xi) - V_{\xi\xi}(\xi) \right]^{-1} \frac{\partial h^{T}}{\partial \xi} (\xi) (y(t) - h(\xi)), \quad (7.23)$$
$$u = -\frac{1}{\varepsilon} g^{T}(\xi) V_{\xi}^{T}(\xi)$$

locally asymptotically stabilizes the closed loop system (7.18), (7.17) and (7.23) for every perturbation system  $\Delta$  as in (7.17), having  $L_2$ -gain less than  $\gamma$ .

**Example 7.8** A nonlinear system  $\dot{x} = f(x) + g(x)u$ , y = h(x),  $x \in M$ ,  $u \in \mathbb{R}^m$ ,  $y \in \mathbb{R}^m$  is called *lossless* ([Wi 72]) if there exists a nonnegative function  $H: M \to \mathbb{R}$ , H(0) = 0, such that

$$H_x(x)f(x) = 0,$$
  

$$H_x(x)g(x) = h^T(x)$$

or, equivalently,  $\frac{d}{dt}H = u^T y$  (see also Remark 6.4). Assume that the system  $\Sigma$  is lossless. We note that

$$V(x) = \sqrt{\varepsilon} H(x)$$

is a solution of the Hamilton-Jacobi equality (7.20). Furthermore for  $\gamma = 1$ 

$$W(x) = bH(x)$$

is a solution to (7.21) for every  $b \ge 0$ . If we assume that  $H_{xx} > 0$  for all x we also have that the last conditions of (7.22) is satisfied when  $b > \sqrt{\varepsilon}$ . Hence when  $f - \frac{1}{\sqrt{\varepsilon}}gg^tH_x^T$  and -f are exponentially stable it follows that the controller

$$\dot{\xi} = f(\xi) - \frac{1}{\sqrt{\varepsilon}} g(\xi) h(\xi) + \frac{1}{b - \sqrt{\varepsilon}} H_{\xi\xi}^{-1}(\xi) \frac{\partial h^T}{\partial \xi}(\xi) (y(t) - h(\xi)), u = -\frac{1}{\sqrt{\varepsilon}} h(\xi)$$

locally asymptotically stabilizes the closed-loop system for every perturbation  $\Delta$  having  $L_2$ -gain less than 1. But the assumption that -f is exponentially stable is violated in this case because the system is lossless which corresponds to the fact that the linearized system has invariant zeros on the imaginary axis.

On the other hand, the static output feedback

$$u = -y$$

yields a closed-loop system which has  $L_2$ -gain less than or equal to 1. This can be seen by constructing the closed-loop system

$$\dot{x} = f(x) - g(x)h(x) - g(x)d$$
  
 $z = h(x)$ 
(7.24)

for which V(x) = H(x) is clearly a solution to the Hamilton-Jacobi inequality

$$V_x(x) (f(x) - g(x)h(x)) + \frac{1}{2}V_x(x)g(x)g^T(x)V_x^T(x) + \frac{1}{2}h^T(x)h(x) = 0.$$

Hence under local asymptotic stability of  $\dot{x} = f(x) - g(x)h(x)$  this static output feedback also solves the robust stabilization problem for all perturbations  $\Delta$  with  $L_2$ -gain less than or equal to 1.

### **Chapter 8**

## Conclusions

In this chapter we summarize the main results derived in this monograph, and afterwards we consider some open problems associated with the subjects covered.

#### 8.1 Summary

A standing assumption in most of the literature on the state space approach to linear or nonlinear  $\mathcal{H}_{\infty}$  theory is a certain regularity condition. When this regularity assumption is violated the problem generalizes to the singular  $\mathcal{H}_{\infty}$  problem. Two classes of singular  $\mathcal{H}_{\infty}$  problems can be distinguished. The first class appears when the direct feed through from the inputs to the to-be-controlled variables is not injective and the second class of singular  $\mathcal{H}_{\infty}$  problems appears when the direct feed through from disturbances to the measurements is not surjective. In this book we have presented two methods to solve the singular  $\mathcal{H}_{\infty}$ problem for nonlinear systems with respect to singularities from the first class. These methods are extensions of approaches to the singular linear  $\mathcal{H}_{\infty}$  problem derived at the end of the eighties. The first method has been considered by Khargonekar, Petersen and Zhou ([KPZ 87], [Pe 87a], [Pe 87b], [KPR 88], [ZK 88], [KPZ 90]), and is based on a regularization of the linear system such that the regular linear  $\mathcal{H}_{\infty}$  theory can be applied ("cheap control" approach). The second method has been derived by Stoorvogel, Trentelman ([StTr 90], [St 92]) and Scherer ([Sch 91]). This method uses results from geometric system theory to decompose the system, and to split the singular  $\mathcal{H}_{\infty}$  control problem into a regular  $\mathcal{H}_{\infty}$  problem and an almost disturbance decoupling problem.

In Chapter 3 the cheap control approach has been investigated to solve the

singular  $\mathcal{H}_{\infty}$  problem for nonlinear systems. We have considered nonlinear systems whose dynamical equations are affine in the inputs and the disturbances. Probably most of the results can be extended to general nonlinear systems which are not affine in the inputs and the disturbances. The solvability of the singular state feedback  $\mathcal{H}_{\infty}$  problem is characterized by the solvability of a parameterized Hamilton-Jacobi inequality which generalizes the parameterized Riccati equality from the linear theory. This Hamilton-Jacobi inequality also corresponds to a regular state feedback  $\mathcal{H}_{\infty}$  control problem for a regularized version of the system. This connection lead to some nice properties about the feedback solution to the singular  $\mathcal{H}_{\infty}$  problem. The connection with the linear state feedback  $\mathcal{H}_{\infty}$  problem for the linearization has also been treated. We have proved that the singular  $\mathcal{H}_{\infty}$  problem for the nonlinear system is locally solvable with an  $L_2$ -gain less than a certain constant  $\gamma$  if and only if the singular  $\mathcal{H}_{\infty}$  problem with gain less than  $\gamma$  is solvable for its linearization. Finally, the singular nonlinear measurement feedback  $\mathcal{H}_{\infty}$  problem has been solved using the worst case certainty equivalence principle. Necessary conditions generalizing the necessary and sufficient conditions for linear systems are derived. Also a set of sufficient conditions is given. At this moment we cannot expect anything, since even for regular systems a full characterization of the measurement feedback  $\mathcal{H}_{\infty}$  problem has not been derived yet.

In Chapter 4 the sufficient conditions for the solvability of the almost disturbance decoupling problem for single-input single-output systems derived in the papers [MRST 94] and [MT 95] are extended to multi-input multi-output systems. In fact the almost disturbance decoupling problem is a special case of a singular  $\mathcal{H}_{\infty}$  problem. The solvability of the almost disturbance decoupling problem is equal to the solvability of the singular  $\mathcal{H}_{\infty}$  control problem for arbitrary small  $L_2$ -gain  $\gamma > 0$ . We have given a set of sufficient conditions for the solvability of the almost disturbance decoupling problem for affine nonlinear systems without a direct feed through from the inputs to the outputs. This set of conditions is still far from being necessary, and in fact we give some other slightly different sets of sufficient conditions as well.

We partially extend the results from the geometric approach to the linear singular  $\mathcal{H}_{\infty}$  problem to general nonlinear systems. An important notion in this geometric approach to the singular state feedback  $\mathcal{H}_{\infty}$  problem is the minimal involutive conditioned invariant distribution containing the input vector fields for an extended system which includes the inputs and the disturbances as extra state components. This notion extends the notion of strongly controllable subspace in the linear theory, and is used to decompose the nonlinear system. This decomposition leads to several sets of sufficient conditions for the solvability

of the state feedback  $\mathcal{H}_{\infty}$  problem. These conditions are in terms of the solvability of a regular  $\mathcal{H}_{\infty}$  control problem for a subsystem of the original system and an almost disturbance decoupling problem for a complementary subsystem. Under certain extra assumption we proved that the solvability of the regular subproblem is also a necessary condition for the solvability of the singular  $\mathcal{H}_{\infty}$  problem.

A promising way to attack the singular  $\mathcal{H}_{\infty}$  problem is to define an auxiliary system based on the solvability of the Hamilton-Jacobi equality corresponding to the regular  $\mathcal{H}_{\infty}$  subproblem. There exists a very strong relation between the solvability of the singular  $\mathcal{H}_{\infty}$  problem for the original system and the solvability of the same problem for the auxiliary system. In the example considered in Chapter 6 it is shown that this method leads to satisfactory results. For a special class of nonlinear systems we can fully characterize the solvability of the singular problem for the complete system by the solvability of the regular problem for one of the subsystems, where the states of the complementary subsystem are considered as inputs for this subsystem.

The geometric approach is certainly more elegant and appealing than the cheap control method. In a sense the singular  $\mathcal{H}_{\infty}$  problem is characterized by a regular  $\mathcal{H}_{\infty}$  problem for a reduced order system. For nonlinear systems however a full characterization of the singular state feedback  $\mathcal{H}_{\infty}$  problem has not been derived yet. This is one of the main drawbacks of this geometric method compared with the cheap control approach.

The merits and drawbacks of the two methods are also shown in Chapter 6. As a first example the model of the orientation of a rigid body is considered. For this system we treated the tracking problem which is described as a singular  $\mathcal{H}_{\infty}$  problem. The problem is solved using the two different methods. The cheap control approach comes down to the search for a solution to the parameterized Hamilton-Jacobi inequality. Another solution is constructed using the geometric approach. The main advantage of the geometric method is that we only need to find a solution to a reduced order Hamilton-Jacobi inequality. By this fact the geometric approach can be easily extended to the general tracking problem of a rigid body model including motion of its center of mass (see [MD 95]). In the second example, the inverted pendulum on a cart, some computational issues are described.

#### 8.2 Open problems

We mention some open problems of interest.

#### (i) Almost disturbance decoupling problem

In order to obtain a full geometric characterization of the nonlinear  $\mathcal{H}_{\infty}$  problem, even in the state feedback case, a better understanding of the almost disturbance decoupling problem for nonlinear systems is instrumental. In the current approach a triangular structure on the disturbance vector fields is needed which is quite restrictive (see [MRST 94], [MT 95] and Chapter 4).

#### (*ii*) Measurement feedback $\mathcal{H}_{\infty}$ control problem

The dynamic output feedback  $\mathcal{H}_{\infty}$  problem is still largely open both in the regular and singular case. Necessary conditions have been found and there is some insight in the structure of possible, compensators but a full characterization and understanding of this problem has not been derived yet. One of the problems we have not investigated is the singular  $\mathcal{H}_{\infty}$  control problem with singular measurements in which case the direct feed through from disturbances to measurements is not surjective.

#### (iii) Nonlinear robust design methodology

The usefulness of nonlinear  $\mathcal{H}_{\infty}$  control needs further research. The application of the theory to design problems can lead to new insight and can contribute to a better understanding of some of the open problems. It is useful to look how frequency based notions such as the sensitivity function and the well known limits of performance for linear systems can be captured in the nonlinear theory.

#### (iv) Computational issues

Further research should be done on the computation of (approximate) solutions of the Hamilton-Jacobi inequalities used in the nonlinear  $\mathcal{H}_{\infty}$  theory.

#### (v) Factorization approach

The factorization approach considered in Subsection 5.2.7 is promising. Perhaps a better understanding of the almost disturbance decoupling problem can help to derive a full characterization of the singular state feedback  $\mathcal{H}_{\infty}$  problem using this factorization approach.

## Appendix A

# Notions from differential geometry

In this appendix an overview is given of concepts from differential geometry that are used in this book. The presentation of this material largely follows [Hu 91] and [NvdS 90].

#### Manifold

Consider the space  $\mathbb{R}^N$  with Euclidean norm  $\|\cdot\|$  defined by

$$||x|| = \left(\sum_{i=1}^{N} x_i^2\right)^{\frac{1}{2}}.$$

A subset  $U \subset \mathbb{R}^N$  is called an *open subset* of  $\mathbb{R}^N$  if for every  $\bar{x} \in U$  there exists an  $\varepsilon > 0$  such that  $\{x \in \mathbb{R}^N | \|x - \bar{x}\| < \varepsilon\} \subset U$ . For  $x \in \mathbb{R}^N$ , an open subset  $U \subset \mathbb{R}^N$  containing x is called a *neighborhood* of x. Let U be a (not necessarily open) subset of  $\mathbb{R}^N$ . A subset  $\overline{U} \subset U$  is called a *relatively open subset* of U if there exists an open subset  $\tilde{U}$  of  $\mathbb{R}^N$  such that  $\tilde{U} \cap U = \overline{U}$ .

Let U, V be subsets of  $\mathbb{R}^N$  and consider a mapping  $\Phi: U \to V$ .  $\Phi$  is called *continuous* if the inverse image of every open subset of V is an open subset of U. If U is open,  $\Phi$  is called  $C^k$  if it has continuous partial derivatives up to order k, and for  $k = \infty$  the mapping is also called *smooth*. If U is not open,  $\Phi$  is called  $C^k$  if for every  $\bar{x} \in U$  there exists a neighborhood  $\overline{U}$  of  $\bar{x}$  in  $\mathbb{R}^N$  and a  $C^k$  mapping  $\Psi: \overline{U} \to V$  such that  $\Psi$  equals  $\Phi$  on  $U \cap \overline{U}$ .  $\Phi$  is called a  $C^k$  diffeomorphism if  $\Phi^{-1}: V \to U$  exists and both  $\Phi$  and  $\Phi^{-1}$  are  $C^k$ .

A subset  $M \subset \mathbb{R}^N$  is called a *manifold* of dimension *n* if there exist an index set *I*, relatively open subsets  $U_i$  ( $i \in I$ ) of *M*, open subsets  $V_i$  ( $i \in I$ ) of  $\mathbb{R}^n$  and diffeomorphisms  $\phi_i : U_i \to V_i$  ( $i \in I$ ), such that  $\bigcup_{i \in I} U_i = M$ . A subset *N* of *M* is called a *submanifold* of *M* if *N* itself is a manifold.

#### Local coordinates

For every  $i \in I$ , the pair  $(U_i, \phi_i)$  is called a *local coordinate chart* on M.

For k = 1, ..., n, let  $r_k$  denote the natural *coordinate functions* on  $\mathbb{R}^n$ , i.e.,  $r_k(a_1, ..., a_n) = a_k$ . The functions  $x_i = r_i \circ \phi$  (i = 1, ..., n) are called *local coordinate functions* and the values  $x_1(p), ..., x_n(p)$  of a point  $p \in U$  are called the *local coordinates* of p.

#### Tangent space, tangent bundle

For  $a \in \mathbb{R}^n$ , the *tangent space*  $T_a \mathbb{R}^n$  at *a* is the set of tangent vectors to  $\mathbb{R}^n$  at *a*. The natural basis of  $T_a \mathbb{R}^n$  associated with the natural coordinate functions  $r_i$  is denoted by  $\{\frac{\partial}{\partial r_1}|_a, \ldots, \frac{\partial}{\partial r_n}|_a\}$ . Consider a manifold *M* of dimension *n* and let  $p \in M$ . Let  $(U, \phi)$  be a local coordinate chart around *p*, and  $x_1, \ldots, x_n$  local coordinates. For  $X \in T_{\phi(p)} \mathbb{R}^n$ , define  $\phi_{*p}^{-1} X := \frac{\partial \phi^{-1}}{\partial a}(\phi(p)) X$ . Define

$$\frac{\partial}{\partial x_i}\Big|_p := \phi_{*p}^{-1} \frac{\partial}{\partial r_i}\Big|_{\phi(p)}, \qquad i = 1, \dots, n.$$

Then the *tangent space*  $T_pM$  of M at p is defined as

$$T_p M = \operatorname{span} \left\{ \left. \frac{\partial}{\partial x_1} \right|_p, \ldots, \left. \frac{\partial}{\partial x_n} \right|_p \right\}.$$

Hence  $\{\frac{\partial}{\partial x_1}|_p, \ldots, \frac{\partial}{\partial x_n}|_p\}$  is a basis of  $T_pM$ . The elements of  $T_pM$  are called *tangent vectors* at p. The set

$$TM = \left\{ (p, X_p) | p \in M, X_p \in T_pM \right\}$$

is called the *tangent bundle* of M.

#### Vector field

A vector field on M is a mapping X that assigns to each  $p \in M$  a tangent vector  $X_p \in T_p M$ . X is called a  $C^k$  vector field if for each  $p \in M$  there exists a local

coordinate chart  $(U, \phi)$  around p and  $C^k$  functions  $X_1, \ldots, X_n$  such that for all  $\bar{p} \in U$  we have

$$X(x_1(\bar{p}),\ldots,x_n(\bar{p}))=\sum_{i=1}^n X_i(x_1(\bar{p}),\ldots,x_n(\bar{p}))\left.\frac{\partial}{\partial x_i}\right|_{\bar{p}}.$$

Throughout by vector fields we will mean  $C^k$  vector fields with k sufficiently large. The vector field X in local coordinates, is often identified with the *n*-dimensional column vector  $(X_1(x_1, \ldots, x_n), \ldots, X_n(x_1, \ldots, x_n))^T$ .

A smooth curve  $\sigma$  on M is a smooth mapping  $\sigma : (a, b) \to M$ , where (a, b) is an open interval of  $\mathbb{R}$ . For  $t \in (a, b)$ , let  $(U, \phi)$  be a local coordinate chart around  $\sigma(t) \in M$ . Define  $\dot{\sigma}(t) \in T_{\sigma(t)}M$  by the conventional limit

$$\dot{\sigma}(t) = \lim_{h \to 0} \frac{\sigma(t+h) - \sigma(t)}{h}$$

 $\sigma$  is called an *integral curve* of a given vector field X on M if  $\dot{\sigma}(t) = X(\sigma(t))$  for all  $t \in (a, b)$ . In local coordinates  $x_1, \ldots, x_n$  this means that  $\sigma(t) = (\sigma_1(t), \ldots, \sigma_n(t))$  is a solution of the set of differential equations

$$\begin{cases} \dot{\sigma}_1(t) = X_1(\sigma_1(t), \dots, \sigma_n(t)) \\ \vdots \\ \dot{\sigma}_n(t) = X_n(\sigma_1(t), \dots, \sigma_n(t)) \end{cases} \quad t \in (a, b)$$

where X is identified with the column vector  $(X_1, \ldots, X_n)^T$ . So, to a vector X given in local coordinates we associate in a one-to-one way the set of differential equations

$$\begin{cases} \dot{x}_{1}(t) = X_{1}(x_{1}(t), \dots, x_{n}(t)) \\ \vdots \\ \dot{x}_{n}(t) = X_{n}(x_{1}(t), \dots, x_{n}(t)) \end{cases}$$

also abbreviated as  $\dot{x} = X(x)$ . A submanifold  $N \subset M$  is called *invariant* for  $\dot{x} = X(x)$  if

$$X(x) \in T_p N, \qquad \forall p \in N.$$

By the existence and uniqueness theorem for smooth differential equations it follows that for any  $p \in M$  there exists an interval (a, b) of maximal length containing 0 and a unique interval curve  $\sigma(t), t \in (a, b)$  with  $\sigma(0) = p$ . If for every p we have  $(a, b) = (-\infty, \infty)$ , and so solutions are defined for all time t the vector field X is called *complete*.

#### **Coordinate transform**

Let now  $(U, \phi)$  and  $(V, \psi)$  be two overlapping coordinates charts yielding a *coordinate transformation* z = S(x), where x and z are local coordinates corresponding to  $(U, \phi)$  and  $(V, \psi)$  respectively, with  $S = \psi \circ \phi^{-1}$ . Let  $X(p) \in T_p M$ , with  $p \in U \cap V$ , be expressed in the basis corresponding to  $(U, \phi)$  respectively  $(V, \psi)$  as

$$X(p) = \sum_{i=1}^{n} \alpha_i(p) \left. \frac{\partial}{\partial x_i} \right|_p = \sum_{i=1}^{n} \beta_i(p) \left. \frac{\partial}{\partial z_i} \right|_p$$

then the coefficients  $\alpha_i(p)$  and  $\beta_i(p)$  are related as

$$\begin{pmatrix} \beta_1(p) \\ \vdots \\ \beta_n(p) \end{pmatrix} = \frac{\partial S}{\partial x}(x(p)) \begin{pmatrix} \alpha_1(p) \\ \vdots \\ \alpha_n(p) \end{pmatrix}.$$

#### Lie-derivative, Lie-bracket

A vector field X defines in any  $p \in M$  a tangent vector X(p). For  $f : M \to \mathbb{R}$  this yields in any  $p \in M$  the directional derivative X(p)(f). Hence by varying p we obtain a smooth function X(f) defined as

$$X(f)(p) := X(p)(f).$$

The function  $X(f): M \to \mathbb{R}$  will be called the *Lie-derivative* of f along X also denoted as  $L_X f$ .

For two vector fields X and Y on M, we define a new vector field, denoted as [X, Y] and called the *Lie-bracket* of X and Y defined by (in local coordinates)

$$[X, Y] = \sum_{j=1}^{n} \left( \sum_{i=1}^{n} \frac{\partial Y_j}{\partial x_i} X_i - \frac{\partial X_j}{\partial x_i} Y_i \right) \frac{\partial}{\partial x_j}.$$

#### **Cotangent bundle**

The *dual space* of the tangent space  $T_pM$ , denoted by  $T_p^*M$ , is called the cotangent space, i.e., the dual space  $V^*$  of a linear space V is the set of all linear functions on V. Elements of  $T_p^*M$  are called *cotangent vectors*. Let

$$\{\frac{\partial}{\partial x_1}|_p,\ldots,\frac{\partial}{\partial x_n}|_p\}$$

be a basis for  $T_pM$  corresponding to local coordinates  $x_1, \ldots, x_n$  on M, then we denote the dual basis of  $T_p^*M$  by  $dx_1|_p, \ldots, dx_n|_p$ . The set

$$T^*M = \left\{ (p, \omega) | p \in M, \, \omega \in T_p^*M \right\}$$

is called the *cotangent bundle* of M. A *covector field* (or *one-form*)  $\omega$  on M is a  $C^k$ -mapping that to each  $p \in M$  assigns a cotangent vector. If  $\omega \in T_p^*M$ , then the value of  $\omega$  at  $X \in T_pM$  is denoted by  $\langle \omega, X \rangle$ .

With every continuous function h on M we can associate a covector field dh by defining

$$\mathrm{d}h(x) = \sum_{i=1}^{n} \frac{\partial h}{\partial x_i}(x) \,\mathrm{d}x_i|_x$$

#### **Distributions**, Frobenius

A distribution  $\mathcal{D}$  on M is a mapping that assigns to each  $p \in M$  a linear subspace of  $T_p M$ . If  $X_1, \ldots, X_q$  is a set of vector fields on M, then their span, denoted by span  $\{X_1, \ldots, X_q\}$  is the distribution defined by

 $\operatorname{span}\left\{X_1,\ldots,X_q\right\}: p \mapsto \operatorname{span}\left\{X_1(p),\ldots,X_q(p)\right\}, \quad (p \in M). \quad (A.1)$ 

A distribution is called  $C^k$  if it is defined by (A.1) for  $C^k$  vector fields  $X_1, \ldots, X_q$ . Throughout by distributions we will mean  $C^k$  distributions with k sufficiently large. The sum and the intersection of two distributions  $\mathcal{D}_1$  and  $\mathcal{D}_2$  are defined as:

$$\begin{aligned} \mathcal{D}_1 + \mathcal{D}_2 : p &\mapsto & \mathcal{D}_1(p) + \mathcal{D}_2(p); \\ \mathcal{D}_1 \cap \mathcal{D}_2 : p &\mapsto & \mathcal{D}_1(p) \cap \mathcal{D}_2(p). \end{aligned}$$

A vector field X on M is said to belong to a distribution  $\mathcal{D}$ , denoted by  $X \in \mathcal{D}$ , if for every  $p \in M$  we have  $X(p) \in \mathcal{D}(p)$ . A distribution  $\mathcal{D}_1$  is said to be contained in a distribution  $\mathcal{D}_2$ , denoted by  $\mathcal{D}_1 \subset \mathcal{D}_2$ , if every vector field belonging to  $\mathcal{D}_1$  also belongs to  $\mathcal{D}_2$ . The *dimension* of a distribution  $\mathcal{D}$  at  $p \in M$  is the dimension of the linear subspace  $\mathcal{D}(p)$ . A distribution is called *constant dimensional* if the dimension of  $\mathcal{D}(p)$  does not depend on the point  $p \in M$ . If  $\mathcal{D}$  is a distribution of constant dimension, say k, then around any  $p \in M$  there exist k independent vector fields  $X_1, \ldots, X_k$  such that

$$\mathcal{D}(q) = \operatorname{span} \{X_1(q), \dots, X_k(q)\}, \qquad q \operatorname{near} p.$$

A distribution  $\mathcal{D}$  is called *involutive* if  $[X, Y] \in \mathcal{D}$  whenever  $X, Y \in \mathcal{D}$ . If  $\mathcal{D}$  is not involutive there always exists a smallest involutive distribution containing  $\mathcal{D}$ . This distribution is called the *involutive closure* of  $\mathcal{D}$  and is denoted by  $\overline{\mathcal{D}}$ .

With the covector field dh defined before we can define the distribution

 $\ker dh(x): p \mapsto \left\{ X(p) \in T_p M | \langle dh(x(p)), X(p) \rangle = 0 \right\}.$ 

This distribution is automatically involutive.

#### **Theorem A.1 Frobenius theorem** (local version)

Let  $\mathcal{D}$  be an involutive constant dimensional distribution on M. Then around any  $p \in M$  there exists a coordinate chart  $(U, \phi)$  with local coordinates denoted by  $x_1, \ldots, x_n$  such that:

$$\mathcal{D}(q) = \operatorname{span}\left\{ \left. \frac{\partial}{\partial x_1} \right|_q, \dots, \left. \frac{\partial}{\partial x_k} \right|_q \right\} = \ker\left\{ \left. \operatorname{d} x_{k+1} \right|_q, \dots, \left. \operatorname{d} x_n \right|_q \right\}, \qquad q \in U.$$

There also exists a global version of the Frobenius theorem ([Spi 70]).

# **Bibliography**

- [AbMa 78] R.A. Abraham and J.E. Marsden, *Foundations of mechanics*, 2nd edition, Benjamin/Cummings, Reading, Massachusetts, 1978.
- [AM 90] B.D.O. Anderson and J.B. Moore, *Optimal control: linear quadratic methods*, Prentice Hall, Englewood Cliffs, 1990.
- [As 94] A. Astolfi, 'Singular  $\mathcal{H}_{\infty}$  control', *Proceedings of the 33rd IEEE* Conference on Decision and Control, Orlando, pp. 2543-2548, 1994.
- [AsGu 93] A. Astolfi and L. Guzzella, 'Robust control of nonlinear systems an  $\mathcal{H}_{\infty}$ -approach', *IFAC conference*, Sidney, vol. 9, pp. 281-284, 1993.
- [AsLa 94] A. Astolfi and L. Lanari, 'Disturbance attenuation and set-point regulation of rigid robots via  $\mathcal{H}_{\infty}$  control', in *Proceedings of the 33rd IEEE Conference on Decision and Control*, Lake Buena Vista, FL, pp. 2578-2583, 1994.
- [BHW 93] J. Ball, J.W. Helton and M. Walker, ' $\mathcal{H}_{\infty}$  control for nonlinear systems with output feedback', *IEEE Transactions on Automatic Control*, vol. 38, pp. 546-559, 1993.
- [BB 90] T. Başar and P. Bernhard,  $\mathcal{H}_{\infty}$ -optimal control and related minimax design problems, Birkhauser, Boston, 1990.
- [BO 82] T. Başar and G.J. Olsder, *Dynamic noncooperative game theory*, Academic Press, London, 1982.
- [BaMa 92] G. Basile and G. Marro, *Controlled and conditioned invariants in linear system theory*, Prentice Hall, Englewood Cliffs, 1992.

[DaEg 96]	M. Dalsmo and O. Egeland, 'Tracking of rigid body motion via nonlinear $\mathcal{H}_{\infty}$ control', <i>Proceedings of the 13th IFAC World Congress</i> , San Francisco, California, pp. 395-400, 1996.
[DaEg 95]	M. Dalsmo and O. Egeland, 'State feedback $\mathcal{H}_{\infty}$ control of a rigid spacecraft', in <i>Proceedings of the 34th IEEE Conference on Decision and Control</i> , New Orleans, Louisiana, pp. 3968-3973, 1995.
[DaMa 96]	M. Dalsmo and W.C.A. Maas, 'Singular $\mathcal{H}_{\infty}$ suboptimal control for a class of two-block interconnected nonlinear systems', 35th <i>IEEE Conference on Decision and Control</i> , 1996.
[DeVi 75]	C.A. Desoer and M. Vidyasagar, <i>Feedback systems: Input-Output Properties</i> , Academic Press, New York, 1975.
[DBB 93]	G. Didinsky, T. Başar and P. Bernhard, 'Structural properties of minimax controllers for a class of differential games in nonlinear $H^{\infty}$ control', <i>Systems &amp; Control Letters</i> , vol. 21, pp. 433-441, 1993.
[Dow 93]	P. Dower, Nonlinear state feedback $\mathcal{H}_{\infty}$ control of the ball and beam system, final year project report, Department of Electrical and Computer Engineering, University of Newcastle, 1993.
[DGKF 89]	J.C. Doyle, K. Glover, P.P. Khargonekar and B.A. Francis, 'State-space solutions to standard $\mathcal{H}_2$ and $\mathcal{H}_\infty$ control problems', <i>IEEE Transactions on Automatic Control</i> , vol. 34, pp. 831-847, 1989.
[EDS 94]	O. Egeland, M. Dalsmo and O.J. Sørdalen, 'Feedback Control of a Nonholonomic Underwater Vehicle with Constant Desired Configuration', <i>International Journal of Robotics Research</i> , vol. 15, 1996.
[Fr 87]	B.A. Francis, A course in $\mathcal{H}_{\infty}$ control theory, Lecture notes in control and information sciences, vol. 88, Springer Verlag, Berlin, 1987.
[Ha 83]	M.L.J. Hautus, 'Strong detectability and observers', <i>Linear Algebra &amp; Applications</i> , vol. 50, pp. 353-368, 1983.

[HaSi 83]	M.L.J. Hautus and L.M. Silverman, 'System structure and sin- gular control', <i>Linear algebra and its applications</i> , vol. 50, pp. 369-402, 1983.
[Hij 80]	O.B. Hijab, <i>Minimum energy estimation</i> , Doctoral dissertation, University of California, Berkeley, 1980.
[HiMo 76]	D. Hill and P. Moylan, 'The stability of nonlinear dissipative systems', <i>IEEE Transactions on Automatic Control</i> , vol. 21, pp. 708-711, 1976.
[HiPr 86]	D. Hinrichsen and A.J. Pritchard, 'Stability radius for structured perturbations and the algebraic Riccati equation', <i>Systems &amp; Control Letters</i> , vol. 8, pp. 105-113, 1986.
[HiPr 88]	D. Hinrichsen and A.J. Pritchard, 'New robustness results for linear systems under real perturbations, <i>Proceedings 27th IEEE Conference on Decision and Control</i> , Austin, pp. 1375-1379, 1988.
[HiPr 90]	D. Hinrichsen and A.J. Pritchard, 'A note on some differences

- [HIPF 90] D. Hinrichsen and A.J. Pritchard, 'A note on some differences between real and complex stability radii', *Systems & Control Letters*, vol. 14, pp. 401-408, 1990.
- [Hug 86] P.C. Hughes, *Spacecraft attitude dynamics*, John Wiley, New York, 1986.
- [Hu 91] H.J.C. Huijberts, *Dynamic feedback in nonlinear synthesis problems*, CWI Tracts, vol. 101, Amsterdam, 1994.
- [IKGM 81] A. Isidori, A.J. Krener, C. Gori-Giorgi and S. Monaco, 'Nonlinear decoupling via feedback: a differential geometric approach', *IEEE Transactions on Automatic Control*, vol. 26, pp. 331-345, 1981.
- [Is 89] A. Isidori, *Nonlinear Control Systems*, Springer, Berlin, 2nd edition, 1989.
- [IsAs 92] A. Isidori and A. Astolfi, 'Disturbance attenuation and  $\mathcal{H}_{\infty}$  control via measurement feedback in nonlinear systems', *IEEE Transactions on Automatic Control*, vol. 37, pp. 1283-1293, 1992.

[IsKa 95]	A. Isidori and W. Kang, ' $\mathcal{H}_{\infty}$ control via measurement feedback for general nonlinear systems', <i>IEEE Transactions on Automatic</i> <i>Control</i> , vol. 40, pp. 466-472, 1995.
[IsTa 95]	A. Isidori and T.J. Tarn, 'Robust regulation for nonlinear systems with gain-bounded uncertainties', <i>IEEE Transactions on Automatic Control</i> , vol. 40, pp. 1744-1754, 1995.
[Ja 77]	D.H. Jacobson, <i>Extensions of linear-quadratic control, optimiza-</i> <i>tion and matrix theory</i> , Academic Press, New York, 1977.
[JJT 86]	H.Th. Jongen, P. Jonker and F. Twilt, <i>Nonlinear optimization in</i> $\mathbb{R}^n$ , Lang, Frankfurt am Main, 1986.
[Ka 95]	W. Kang, 'Nonlinear $\mathcal{H}_{\infty}$ control and its applications to rigid spacecraft', <i>IEEE Transactions on Automatic Control</i> , vol. 40, pp. 1281-1285, 1995.
[KPR 88]	P.P. Khargonekar, I.R. Petersen and M.A. Rotea, ' $\mathcal{H}_{\infty}$ -optimal control with state-feedback', <i>IEEE Transactions on Automatic Control</i> , vol. 33, pp. 786-788, 1988.
[KPZ 87]	P.P. Khargonekar, I.R. Petersen and K. Zhou, 'Robust stabilization of uncertain systems and $\mathcal{H}_{\infty}$ optimal control', internal report, 1987.
[KPZ 90]	P.P. Khargonekar, I.R. Petersen and K. Zhou, 'Robust stabilization of uncertain linear systems: quadratic stabilizability and $\mathcal{H}_{\infty}$ control theory', <i>IEEE Transactions on Automatic Control</i> , vol. 35, pp. 356-361, 1990.
[KS 72]	H. Kwakernaak and R. Sivan, <i>Linear optimal control systems</i> , Wiley-Interscience, New York, 1972.
[LeMa 67]	E.B. Lee and L. Markus, Foundations of optimal control theory, John Wiley & Sons, New York, 1967.
[Lu 69]	D.L. Lukes, 'Optimal regulation of nonlinear dynamical systems', <i>SIAM Journal on Control</i> , vol. 7, pp. 75-100, 1969.
[M 96]	W.C.A. Maas, Nonlinear $\mathcal{H}_{\infty}$ control: the singular case, Ph.D. Thesis, University of Twente, Enschede, The Netherlands, 1996.

[MD 95]	W.C.A. Maas and M. Dalsmo, 'Tracking of a rigid body in $SE(3)$ using singular $\mathcal{H}_{\infty}$ control', internal report, University of Twente, vol. 1301, 1995.
[MD 96]	W.C.A. Maas and M. Dalsmo, 'Singular $\mathcal{H}_{\infty}$ control for tracking a rigid body without cost weighting on the controls', <i>IEEE Transactions on Automatic Control</i> , submitted, 1996.
[MvdS 94]	W.C.A. Maas and A.J. van der Schaft, 'Singular nonlinear $\mathcal{H}_{\infty}$ optimal control by state feedback', <i>Proceedings of the 33rd IEEE Conference on Decision and Control</i> , Orlando, pp. 1415-1420, 1994.
[MvdS 96]	W.C.A. Maas and A.J. van der Schaft, 'Singular nonlinear $\mathcal{H}_{\infty}$ optimal control problem', <i>International Journal on Robust and Nonlinear Control</i> , vol. 6, pp. 669-689, 1996.
[MRS 89]	R. Marino, W. Respondek and A.J. van der Schaft, 'Almost dis- turbance decoupling for single-input single-output nonlinear sys- tems', <i>IEEE Transactions on Automatic Control</i> , vol. 34, pp. 1013-1017, 1989.
[MRS 94]	R. Marino, W. Respondek and A.J. van der Schaft, 'Equivalence of nonlinear systems to input-output prime forms', <i>SIAM Journal</i> <i>on Control and Optimization</i> , vol. 32, pp. 387-407, 1994.
[MRST 94]	R. Marino, W. Respondek, A.J. van der Schaft and P. Tomei, 'Nonlinear $\mathcal{H}_{\infty}$ almost disturbance decoupling', <i>Systems &amp; Control Letters</i> , vol. 23, pp. 159-168, 1994.
[MT 95]	R. Marino and P. Tomei, <i>Nonlinear control design; geometric, adaptive &amp; robust</i> , Information and system sciences series, Prentice Hall, London, 1995.

- [Ng 96a] S.K. Nguang, 'Global robust  $\mathcal{H}_{\infty}$  control of a class of nonlinear systems', to appear in *International Journal on Robust and Nonlinear Control*, 1996.
- [Ng 96b] S.K. Nguang, 'Robust nonlinear  $\mathcal{H}_{\infty}$  output feedback control', to appear in *IEEE Transactions on Automatic Control*, 1996.
- [NvdS 90] H. Nijmeijer and A.J. van der Schaft, *Nonlinear dynamical control systems*, Springer-Verlag, New York, 1990.

[Pe 87a]	I.R. Petersen, 'Disturbance attenuation and $\mathcal{H}_{\infty}$ optimization: a design method based on the algebraic Riccati equation', <i>IEEE Transactions on Automatic Control</i> , vol. 32, pp. 427-429, 1987.
[Pe 87b]	I.R. Petersen, 'A stabilization algorithm for a class of uncertain linear systems', <i>Systems &amp; Control Letters</i> , vol. 8, pp. 351-357, 1987.
[SBE 91]	C. Samson, M. Le Borgne and B. Espiau, <i>Robot control: The task function approach</i> , Oxford Engineering Science Series, vol. 22, Clarendon Press, London, 1991.
[vdS 85]	A.J. van der Schaft, 'On nonlinear observers', <i>IEEE Transac-</i> tions on Automatic Control, vol. 30, pp. 1254-1256, 1985.
[vdS 86]	A.J. van der Schaft, 'Optimal control and Hamiltonian input- output systems', <i>Algebraic and Geometric Methods in Nonlinear</i> <i>Control Theory</i> , eds. M. Fliess and M. Hazewinkel, pp. 389-407, 1986.
[vdS 87]	A.J. van der Schaft, 'On realization of nonlinear systems de- scribed by higher-order differential equations', <i>Mathematical</i> <i>Systems Theory</i> , vol. 19, pp. 239-275, 1987.
[vdS 91]	A.J. van der Schaft, 'On a state space approach to nonlinear $\mathcal{H}_{\infty}$ control', <i>Systems &amp; Control Letters</i> , vol. 16, pp. 1-8, 1991.
[vdS 92]	A.J. van der Schaft, ' $L_2$ -Gain analysis of nonlinear systems and nonlinear state feedback $\mathcal{H}_{\infty}$ control', <i>IEEE Transactions on Automatic Control</i> , vol. 37, pp. 770-784, 1992.
[vdS 93]	A.J. van der Schaft, 'Nonlinear state space $\mathcal{H}_{\infty}$ control theory', <i>Essays on Control: Perspectives in the theory and its Applica-</i> <i>tions</i> , Ed. H.L. Trentelman and J.C. Willems, Birkhäuser, pp. 153-190, 1993.
[vdS 95]	A.J. van der Schaft, 'Robust stabilization of nonlinear systems via stable kernel representations with $L_2$ -gain bounded uncertainty', <i>Systems &amp; Control Letters</i> , vol. 24, pp. 75-81, 1995.
[vdS 96]	A.J. van der Schaft, $L_2$ -gain and passivity techniques in nonlin- ear control, Lecture notes in control and information sciences, vol. 218, Springer Verlag, Berlin, 1996.

[Sch 91]	C. Scherer, The Riccati inequality and state space $\mathcal{H}_{\infty}$ control theory, Doctoral dissertation, University of Würzburg, Germany, 1991.
[ST 95]	T. Shen and K. Tamura, 'Robust $\mathcal{H}_{\infty}$ control of uncertain nonlinear system via state feedback', <i>IEEE Transactions on Automatic Control</i> , vol. 40, pp. 766-768, 1995.
[SW 94]	E. Sontag and Y. Wang, 'Notions equivalent to input-to-state sta- bility', <i>Proceedings of the 33rd IEEE Conference on Decision</i> <i>and Control</i> , Orlando, pp. 3438-3443, 1994.
[Spi 70]	M. Spivak, A comprehensive introduction to differential geome- try, vol. 1, Publish or Perish, Boston, 1970.
[StTr 90]	A.A. Stoorvogel and H.L. Trentelman, 'The quadratic matrix in- equality in singular $\mathcal{H}_{\infty}$ control with state feedback', <i>SIAM Jour-</i> <i>nal on Control &amp; Optimization</i> , vol. 28, pp. 1190-1208, 1990.
[St 91]	A.A. Stoorvogel, 'The singular $\mathcal{H}_{\infty}$ control problem with dynamic measurement feedback', <i>SIAM Journal on Control &amp; Optimization</i> , vol. 29, pp. 160-184, 1991.
[St 92]	A.A. Stoorvogel, The $\mathcal{H}_{\infty}$ control problem: a state space approach, Prentice Hall, Englewood Cliffs, 1992.
[TaAr 81]	M. Takegaki and S. Arimoto, 'A new feedback method for dy- namic control of manipulators', <i>Transactions ASME, Journal on</i> <i>Dyn. Systems, Meas. Control</i> , vol. 103, pp. 119-125, 1981.
[ToRy 91]	S. Townley and E.P. Ryan, 'On the supremally regulated complex stability radius as a real robustness measure', <i>Systems &amp; Control Letters</i> , vol. 16, pp. 249-259, 1991.
[Tr 86]	H.L. Trentelman, Almost invariant subspaces and high gain feedback, CWI Tracts, vol. 29, Amsterdam, 1986.
[Vi 93]	M. Vidyasagar, Nonlinear systems analysis, 2nd edition, Prentice Hall, Englewood Cliffs, 1993.
[Vr 94]	D. de Vries, Design of a nonlinear $\mathcal{H}_{\infty}$ controller for a 2-DOF robot, Master's thesis, Department of Applied Mathematics,

University of Twente, 1994.

- [WeDe 91] J.T-Y. Wen and K. Kreutz-Delgado, 'The attitude control problem', *IEEE Transactions on Automatic Control*, vol. 36, pp. 1148-1162, 1991.
- [Wi 71] J.C. Willems, 'Least squares stationary optimal control and the algebraic Riccati equation', *IEEE Transactions on Automatic Control*, vol. 16, pp. 621-634, 1971.
- [Wi 72] J.C. Willems, 'Dissipative dynamical systems, Part I: General theory', *Arch. Rat. Mech. Anal.*, vol. 45, pp. 321-351, 1972.
- [Wi 81] J.C. Willems, 'Almost invariant subspaces: an approach to high gain feedback design- Part I: almost controlled invariant subspaces', *IEEE Transactions on Automatic Control*, vol. 26, pp. 235-252, 1981.
- [Wi 82] J.C. Willems, 'Almost invariant subspaces: an approach to high gain feedback design- Part II: almost conditionally invariant subspaces', *IEEE Transactions on Automatic Control*, vol. 27, pp. 1071-1084, 1982.
- [Wo 79] W.M. Wonham, *Linear multivariable control: a geometric approach*, 2nd edition, Springer Verlag, Berlin, 1979.
- [XS 90] L. Xie and C.E. de Souza, 'Robust  $\mathcal{H}_{\infty}$  control for linear timeinvariant systems with norm-bounded uncertainty in the input matrix', *Systems & Control Letters*, vol. 14, pp. 389-396, 1990.
- [XS 92] L. Xie and C.E. de Souza, 'Robust  $\mathcal{H}_{\infty}$  control for linear systems with norm-bounded time-varying uncertainty', *IEEE Transactions on Automatic Control*, vol. 37, pp. 1188-1191, 1992.
- [Za 81] G. Zames, 'Feedback and optimal sensitivity: model reference transformations, multiplicative seminorms, and approximate inverses', *IEEE Transactions on Automatic Control*, vol. 26, pp. 301-320, 1981.
- [ZK 88] K. Zhou and P.P. Khargonekar, 'An algebraic Riccati equation approach to  $\mathcal{H}_{\infty}$  optimization', *Systems & Control Letters*, vol. 11, pp. 85-91, 1988.

# Index

Algebraic Riccati Equation (ARE), 31 almost disturbance decoupling problem, 4  $L_2$ -gain, 62  $\mathcal{H}_{\infty}$ , 62 available storage, 10

canonical projection, 15 certainty equivalence worst case, see worst case certainty equivalence characteristic indices disturbance, 63 strong control, 62 well defined, 63 cheap control approach, 3 complete, 183 conditioned invariant distribution, 92 minimal, 93 subspace, 81 minimal, 81, 82 continuous mapping, 181 controllability pencil, 86 coordinate function, 182 local, 182 coordinate transformation, 184 cotangent bundle, 184 cotangent vector, 184

decoupling matrix, 63

diffeomorphism, 181 differential dissipation inequality, 11 differential game, 2 zero sum, 2, 22 dissipation inequality differential, 11 integral, 11 nonlinear, 124 quadratic, 124 dissipative, 11 dissipativity, 2 distribution, 185 constant dimensional, 185 dimension, 185 involutive, 185 involutive closure, 185 disturbance characteristic indices, 63 dual space, 184 Euclidean norm, 181  $F_{\gamma}(P), 86$ 

feedback, 1 Frobenius, 186

gain  $L_2$ , see  $L_2$ -gain general  $\mathcal{H}_{\infty}$  problem, 21 geometric approach, 4 globally asymptotically stable, 13

 $\mathcal{H}_{\infty}$  (control) problem

general, 21 measurement feedback, 20 regular, 21 singular, 21 state feedback, 20  $\mathcal{H}_{\infty}$  almost disturbance decoupling problem, 62 Hamilton-Jacobi equation, 2 Hamilton-Jacobi inequality, 3, 11 Hamiltonian, 14 matrix, 14 pre-, 13 vector field, 14 G.H. Hardy, 2 hyperbolic, 14 inputs regular (part of), 35 singular (part of), 35 integral curve, 183 integral dissipation inequality, 11 invariant conditioned, see conditioned invariant distribution, 92 manifold, 14 stable, 14 subspace, 81

zeros on the im-axis, 25, 87 involutive, 185 involutive closure, 185

#### L<sub>2</sub>-gain

equal, 10 less than, 10 less than or equal, 10 local equal, 16 less than, 16 less than or equal, 15

L<sub>2</sub>-gain almost disturbance decoupling problem, 62  $L_2$ -gain control problem state feedback, 20  $L_{\nu}(P, s), 86$ Lie-bracket, 184 Lie-derivative, 184  $L_2$ -induced norm, 2 linear matrix inequality (LMI), 4 local coordinate chart, 182 local coordinates, 182 locally asymptotically stable, 13 locally stable, 12 lossless, 146 LQG-problem, 1 Lyapunov function, 24 manifold, 181

invariant, 14 stable invariant, 14 sub-, 181 Morse's Lemma, 125 multiplicative perturbations, 173 multiplicative uncertainty, 3

nonlinear dissipation inequality, 124 nonlinear measurement feedback  $\mathcal{H}_{\infty}$ optimal control problem, 20 nonlinear state feedback  $L_2$ -gain optimal control problem, 20 nonlinear state feedback  $\mathcal{H}_{\infty}$  optimal control problem, 20 normal rank, 86

observable zero-state, 13, 24 open subset, 181 relatively, 181

parameter uncertainty, 3, 165

pre-Hamiltonian, 13 projectable, 15 proper, 13 quadratic dissipation inequality, 124 quadratic matrix inequality, 86 regular  $\mathcal{H}_{\infty}$  problem, 3, 21 regularized system, 44 Riccati equation, 2 *S*\*, 93  $S^*$  algorithm, 93  $S^{*}(W), 81, 93$  $S^*(\text{im}G)$ , 82 saddle point, 22 condition, 22 singular  $\mathcal{H}_{\infty}$  problem, 3, 21 smooth curve, 183 smooth mapping, 181 stable asymptotically globally, 13 locally, 13 locally, 12 storage function, 11 available, 10 minimal, 11 strong control characteristic indices, 62 well defined, 63 strongly controllable subspace, 79, 80 submanifold, 181 supply rate, 11 system, 1 regularized, 44

 $T_a \mathbb{R}^n$ , 182  $T_p M$ , 182 tangent bundle, 182 tangent space, 182 tangent vector, 182 uncertainty multiplicative, 3, 173 parameter, 3, 165 vector field, 182 complete, 183 Hamiltonian, 14 independent, 184 worst case certainty equivalence principle, 54–55 solution, 55

zero sum differential game, 22 zero-dynamics, 78 zero-state observable, 13, 24

197

.

#### **CWI TRACTS**

- 1 D.H.J. Epema. Surfaces with canonical hyperplane sections. 1984.
- 2 J.J. Dijkstra. Fake topological Hilbert spaces and characterizations of dimension in terms of negligibility. 1984.
- 3 A.J. van der Schaft. System theoretic descriptions of physical systems. 1984.
- 4 J. Koene. Minimal cost flow in processing networks, a primal approach. 1984.
- 5 B. Hoogenboom. Intertwining functions on compact Lie groups. 1984.
- 6 A.P.W. Böhm. Dataflow computation. 1984.
- 7 A. Blokhuis. Few-distance sets. 1984.
- 8 M.H. van Hoorn. Algorithms and approximations for queueing systems. 1984.
- 9 C.P.J. Koymans. *Models of the lambda calculus.* 1984.
- 10 C.G. van der Laan, N.M. Temme. Calculation of special functions: the gamma function, the exponential integrals and error-like functions. 1984.
- 11 N.M. van Dijk. Controlled Markov processes; timediscretization. 1984.
- 12 W.H. Hundsdorfer. The numerical solution of nonlinear stiff initial value problems: an analysis of one step methods. 1985.
- 13 D. Grune. On the design of ALEPH. 1985.
- 14 J.G.F. Thiemann. Analytic spaces and dynamic programming: a measure theoretic approach. 1985.
- 15 F.J. van der Linden. Euclidean rings with two infinite primes. 1985.
- 16 R.J.P. Groothuizen. Mixed elliptic-hyperbolic partial differential operators: a case-study in Fourier integral operators. 1985.
- 17 H.M.M. ten Eikelder. Symmetries for dynamical and Hamiltonian systems. 1985.
- 18 A.D.M. Kester. Some large deviation results in statistics. 1985.
- 19 T.M.V. Janssen. Foundations and applications of Montague grammar, part 1: Philosophy, framework, computer science. 1986.
- 20 B.F. Schriever. Order dependence. 1986.
- 21 D.P. van der Vecht. Inequalities for stopped Brownian motion. 1986.
- 22 J.C.S.P. van der Woude. Topological dynamix. 1986.
- 23 A.F. Monna. Methods, concepts and ideas in mathematics: aspects of an evolution. 1986.
- 24 J.C.M. Baeten. Filters and ultrafilters over definable subsets of admissible ordinals. 1986.
- 25 A.W.J. Kolen. Tree network and planar rectilinear location theory. 1986.
- 26 A.H. Veen. The misconstrued semicolon: Reconciling imperative languages and dataflow machines. 1986.
- 27 A.J.M. van Engelen. Homogeneous zerodimensional absolute Borel sets. 1986.
- 28 T.M.V. Janssen. Foundations and applications of Montague grammar, part 2: Applications to natural language. 1986.
- 29 H.L. Trentelman. Almost invariant subspaces and high gain feedback. 1986.
- 30 A.G. de Kok. Production-inventory control models: approximations and algorithms. 1987.

- 31 E.E.M. van Berkum. Optimal paired comparison designs for factorial experiments. 1987.
- 32 J.H.J. Einmahl. Multivariate empirical processes. 1987.
- 33 O.J. Vrieze. Stochastic games with finite state and action spaces. 1987.
- P.H.M. Kersten. Infinitesimal symmetries: a computational approach. 1987.
   M.L. Eaton. Lectures on topics in probability in-
- equalities. 1987. 36 A.H.P. van der Burgh, R.M.M. Mattheij (eds.).
- Proceedings of the first international conference on industrial and applied mathematics (ICIAM 87). 1987.
- 37 L. Stougie. Design and analysis of algorithms for stochastic integer programming. 1987.
- 38 J.B.G. Frenk. On Banach algebras, renewal measures and regenerative processes. 1987.
- 39 H.J.M. Peters, O.J. Vrieze (eds.). Surveys in game theory and related topics. 1987.
- 40 J.L. Geluk, L. de Haan. Regular variation, extensions and Tauberian theorems. 1987.
- 41 Sape J. Mullender (ed.). The Amoeba distributed operating system: Selected papers 1984-1987. 1987.
- 42 P.R.J. Asveld, A. Nijholt (eds.). Essays on concepts, formalisms, and tools. 1987.
- 43 H.L. Bodlaender. Distributed computing: structure and complexity. 1987.
- 44 A.W. van der Vaart. Statistical estimation in large parameter spaces. 1988.
- 45 S.A. van de Geer. Regression analysis and empirical processes. 1988.
- 46 S.P. Spekreijse. Multigrid solution of the steady Euler equations. 1988.
- 47 J.B. Dijkstra. Analysis of means in some nonstandard situations. 1988.
- 48 F.C. Drost. Asymptotics for generalized chi-square goodness-of-fit tests. 1988.
- 49 F.W. Wubs. Numerical solution of the shallowwater equations. 1988.
- 50 F. de Kerf. Asymptotic analysis of a class of perturbed Korteweg-de Vries initial value problems. 1988.
- 51 P.J.M. van Laarhoven. Theoretical and computational aspects of simulated annealing. 1988.
- 52 P.M. van Loon. Continuous decoupling transformations for linear boundary value problems. 1988.
- 53 K.C.P. Machielsen. Numerical solution of optimal control problems with state constraints by sequential quadratic programming in function space. 1988.
- 54 L.C.R.J. Willenborg. Computational aspects of survey data processing. 1988.
- 55 G.J. van der Steen. A program generator for recognition, parsing and transduction with syntactic patterns. 1988.
- 56 J.C. Ebergen. Translating programs into delayinsensitive circuits. 1989.
- 57 S.M. Verduyn Lunel. Exponential type calculus for linear delay equations. 1989.
- 58 M.C.M. de Gunst. A random model for plant cell population growth. 1989.
- 59 D. van Dulst. Characterizations of Banach spaces not containing l<sup>1</sup>. 1989.
- 60 H.E. de Swart. Vacillation and predictability properties of low-order atmospheric spectral models. 1989.

- 61 P. de Jong. Central limit theorems for generalized multilinear forms. 1989.
- 62 V.J. de Jong. A specification system for statistical software. 1989.
- 63 B. Hanzon. Identifiability, recursive identification and spaces of linear dynamical systems, part I. 1989.
- 64 B. Hanzon. Identifiability, recursive identification and spaces of linear dynamical systems, part II. 1989.
- 65 B.M.M. de Weger. Algorithms for diophantine equations. 1989.
- 66 A. Jung. Cartesian closed categories of domains. 1989.
- 67 J.W. Polderman. Adaptive control & identification: Conflict or conflux?. 1989.
- 68 H.J. Woerdeman. *Matrix and operator extensions*. 1989.
- 69 B.G. Hansen. Monotonicity properties of infinitely divisible distributions. 1989.
- 70 J.K. Lenstra, H.C. Tijms, A. Volgenant (eds.). Twenty-five years of operations research in the Netherlands: Papers dedicated to Gijs de Leve. 1990.
- 71 P.J.C. Spreij. Counting process systems. Identification and stochastic realization. 1990.
- 72 J.F. Kaashoek. Modeling one dimensional pattern formation by anti-diffusion. 1990.
- 73 A.M.H. Gerards. Graphs and polyhedra. Binary spaces and cutting planes. 1990.
- 74 B. Koren. Multigrid and defect correction for the steady Navier-Stokes equations. Application to aerodynamics. 1991.
- 75 M.W.P. Savelsbergh. Computer aided routing. 1992.
- 76 O.E. Flippo. Stability, duality and decomposition in general mathematical programming. 1991.
- 77 A.J. van Es. Aspects of nonparametric density estimation. 1991.
- 78 G.A.P. Kindervater. Exercises in parallel combinatorial computing. 1992.
- 79 J.J. Lodder. Towards a symmetrical theory of generalized functions. 1991.
- 80 S.A. Smulders. Control of freeway traffic flow. 1996.
- 81 P.H.M. America, J.J.M.M. Rutten. A parallel object-oriented language: design and semantic foundations. 1992.
- 82 F. Thuijsman. Optimality and equilibria in stochastic games. 1992.
- 83 R.J. Kooman. Convergence properties of recurrence sequences. 1992.
- 84 A.M. Cohen (ed.). Computational aspects of Lie group representations and related topics. Proceedings of the 1990 Computational Algebra Seminar at CWI, Amsterdam. 1991.
- 85 V. de Valk. One-dependent processes. 1994.
- 86 J.A. Baars, J.A.M. de Groot. On topological and linear equivalence of certain function spaces. 1992.
- 87 A.F. Monna. The way of mathematics and mathematicians. 1992.
- 88 E.D. de Goede. Numerical methods for the threedimensional shallow water equations. 1993.

- 89 M. Zwaan. Moment problems in Hilbert space with applications to magnetic resonance imaging. 1993.
- 90 C. Vuik. The solution of a one-dimensional Stefan problem. 1993.
- 91 E.R. Verheul. Multimedians in metric and normed spaces. 1993.
- 92 J.L.M. Maubach. Iterative methods for non-linear partial differential equations. 1994.
- 93 A.W. Ambergen. Statistical uncertainties in posterior probabilities. 1993.
- 94 P.A. Zegeling. Moving-grid methods for timedependent partial differential equations. 1993.
- 95 M.J.C. van Pul. Statistical analysis of software reliability models. 1993.
- 96 J.K. Scholma. A Lie algebraic study of some integrable systems associated with root systems. 1993.
- 97 J.L. van den Berg. Sojourn times in feedback and processor sharing queues. 1993.
- 98 A.J. Koning. Stochastic integrals and goodness-offit tests. 1993.
  99 B.P. Sommeijer. Parallelism in the numerical inte-
- gration of initial value problems. 1993.
  100 J. Molenaar. Multigrid methods for semiconductor
- device simulation. 1993.
  101 H.J.C. Huijberts. Dynamic feedback in nonlinear
- synthesis problems. 1994. 102 J.A.M. van der Weide. Stochastic processes and
- point processes of excursions. 1994. 103 P.W. Hemker, P. Wesseling (eds.). Contributions
- to multigrid. 1994. 104 I.J.B.F. Adan. A compensation approach for queue-
- ing problems. 1994.
- 105 O.J. Boxma, G.M. Koole (eds.). Performance evaluation of parallel and distributed systems - solution methods. Part 1. 1994.
- 106 O.J. Boxma, G.M. Koole (eds.). Performance evaluation of parallel and distributed systems - solution methods. Part 2. 1994.
- 107 R.A. Trompert. Local uniform grid refinement for time-dependent partial differential equations. 1995.
- 108 M.N.M. van Lieshout. Stochastic geometry models in image analysis and spatial statistics. 1995.
- 109 R.J. van Glabbeek. Comparative concurrency se-
- mantics and refinement of actions. 1996.
- 110 W. Vervaat (ed.). Probability and lattices. 1996.111 I. Helsloot. Covariant formal group theory and some
- applications. 1995. 112 R.N. Bol. Loop checking in logic programming. 1995.
- 113 G.J.M. Koole. Stochastic scheduling and dynamic programming. 1995.
- 114 M.J. van der Laan. Efficient and inefficient estimation in semiparametric models. 1995.
- 115 S.C. Borst. Polling Models. 1996.
- 116 G.D. Otten. Statistical test limits in quality control. 1996.
- 117 K.G. Langendoen. Graph Reduction on Shared-Memory Multiprocessors. 1996.
- 118 W.C.A. Maas. Nonlinear  $\mathcal{H}_\infty$  control: the singular case. 1996.

#### MATHEMATICAL CENTRE TRACTS

1 T. van der Walt. Fixed and almost fixed points. 1963. 2 A.R. Bloemena. Sampling from a graph. 1964.

3 G. de Leve. Generalized Markovian decision processes, part 1: model and method. 1964.

4 G. de Leve. Generalized Markovian decision processes, part 11: probabilistic background. 1964.

5 G. de Leve, H.C. Tijms, P.J. Weeda. Generalized Markovian decision processes, applications. 1970.

6 M.A. Maurice. Compact ordered spaces. 1964.

W.R. van Zwet. Convex transformations of random variables. 1964

8 J.A. Zonneveld. Automatic numerical integration. 1964.

9 P.C. Baayen. Universal morphisms. 1964.

10 E.M. de Jager. Applications of distributions in mathematical physics. 1964.

11 A.B. Paalman-de Miranda. Topological semigroups. 1964. 12 J.A.Th.M. van Berckel, H. Brandt Corstius, R.J. Mokken, A. van Wijngaarden. Formal properties of newspaper Dutch. 1965

13 H.A. Lauwerier. Asymptotic expansions. 1966, out of print: replaced by MCT 54. 14 H.A. Lauwerier. Calculus of variations in mathematical physics. 1966.

15 R. Doornbos. Slippage tests. 1966.

16 J.W. de Bakker. Formal definition of programming languages with an application to the definition of ALGOL 60. 1967.

17 R.P. van de Riet. Formula manipulation in ALGOL 60, part 1. 1968.

18 R.P. van de Riet. Formula manipulation in ALGOL 60, part 2. 1968.

19 J. van der Slot. Some properties related to compactness. 1968.

20 P.J. van der Houwen. Finite difference methods for solving partial differential equations. 1968.

21 E. Wattel. The compactness operator in set theory and topology. 1968.

22 T.J. Dekker. ALGOL 60 procedures in numerical algebra, part 1. 1968.

23 T.J. Dekker, W. Hoffmann. ALGOL 60 procedures in numerical algebra, part 2. 1968.

24 J.W. de Bakker. Recursive procedures. 1971.

25 E.R. Paërl. Representations of the Lorentz group and projec-tive geometry. 1969.

26 European Meeting 1968. Selected statistical papers, part I. 1968.

27 European Meeting 1968. Selected statistical papers, part II. 1968.

28 J. Oosterhoff. Combination of one-sided statistical tests. 1969

29 J. Verhoeff. Error detecting decimal codes. 1969

30 H. Brandt Corstius. Exercises in computational linguistics. 1970

31 W. Molenaar. Approximations to the Poisson, binomial and hypergeometric distribution functions. 1970.

32 L. de Haan. On regular variation and its application to the weak convergence of sample extremes. 1970.
33 F.W. Steutel. Preservations of infinite divisibility under mixing and related topics. 1970.

34 I. Juhász, A. Verbeek, N.S. Kroonenberg. Cardinal func-tions in topology. 1971.

35 M.H. van Emden. An analysis of complexity. 1971. 36 J. Grasman. On the birth of boundary layers. 1971.

37 J.W. de Bakker, G.A. Blaauv, A.J.W. Duijvestijn, E.W. Dijkstra, P.J. van der Houwen, G.A.M. Kamsteeg-Kemper, F.E.J. Kruseman Aretz, W.L. van der Poel, J.P. Schaap-Kruseman, M.V. Wilkes, G. Zoutendijk. MC-25 Informatica Symposium. 1971.

38 W.A. Verloren van Themaat. Automatic analysis of Dutch compound words. 1972.

39 H. Bavinck. Jacobi series and approximation. 1972.

40 H.C. Tijms. Analysis of (s,S) inventory models. 1972.

41 A. Verbeek. Superextensions of topological spaces. 1972.

42 W. Vervaat. Success epochs in Bernoulli trials (with applica-tions in number theory). 1972.

43 F.H. Ruymgaart. Asymptotic theory of rank tests for independence. 1973.

44 H. Bart. Meromorphic operator valued functions. 1973.

45 A.A. Balkema. Monotone transformations and limit laws. 1973.

46 R.P. van de Riet. ABC ALGOL, a portable language for formula manipulation systems, part 1: the language. 1973.

Jormia manipulation systems, part 1: the language. 1975. 47 R.P. van de Riet. ABC ALGOL, a portable language for formula manipulation systems, part 2: the compiler. 1973. 48 F.E.J. Kruseman Aretz, P.J.W. ten Hagen, H.L. Oudshoorn. An ALGOL 60 compiler in ALGOL 60, text of the MC-compiler for the EL-X8. 1973.

49 H. Kok. Connected orderable spaces. 1974.

50 A. van Wijngaarden, B.J. Mailloux, J.E.L. Peck, C.H.A. Koster, M. Sintzoff, C.H. Lindsey, L.G.L.T. Meertens, R.G. Fisker (eds.). *Revised report on the algorithmic language ALGOL* 68, 1976.

51 A. Hordijk. Dynamic programming and Markov potential theory. 1974.

52 P.C. Baayen (ed.). Topological structures. 1974. 53 M.J. Faber. Metrizability in generalized ordered spaces.

1974

54 H.A. Lauwerier. Asymptotic analysis, part 1. 1974.

55 M. Hall, Jr., J.H. van Lint (eds.). Combinatorics, part 1: theory of designs, finite geometry and coding theory. 1974.

56 M. Hall, Jr., J.H. van Lint (eds.). Combinatorics, part 2: graph theory, foundations, partitions and combinatoria geometry, 1974.

ST M. Hall, Jr., J.H. van Lint (eds.). Combinatorics, part 3: combinatorial group theory. 1974.
 SW. Albers. Asymptotic expansions and the deficiency con-cept in statistics. 1975.

59 J.L. Mijnheer. Sample path properties of stable processes. 1975.

60 F. Göbel. Queueing models involving buffers. 1975. 63 J.W. de Bakker (ed.). Foundations of computer science. 1975.

64 W.J. de Schipper. Symmetric closed categories. 1975.

65 J. de Vries. Topological transformation groups, 1: a categor-ical approach. 1975.

66 H.G.J. Pijls. Logically convex algebras in spectral theory and eigenfunction expansions. 1976.

68 P.P.N. de Groen. Singularly perturbed differential operators of second order. 1976.

69 J.K. Lenstra. Sequencing by enumerative methods. 1977. 70 W.P. de Roever, Jr. Recursive program schemes: semantics and proof theory. 1976.

71 J.A.E.E. van Nunen. Contracting Markov decision processes. 1976.

72 J.K.M. Jansen. Simple periodic and non-periodic Lamé functions and their applications in the theory of conical waveguides. 1977.

73 D.M.R. Leivant. Absoluteness of intuitionistic logic. 1979.

74 H.J.J. te Riele. A theoretical and computational study of generalized aliquot sequences. 1976.

75 A.E. Brouwer. Treelike spaces and related connected topo-logical spaces. 1977. 76 M. Rem. Associons and the closure statements. 1976.

77 W.C.M. Kallenberg, Asymptotic optimality of likelihood ratio .ests in exponential families. 1978.

78 E. de Jonge, A.C.M. van Rooij. Introduction to Riesz spaces. 1977.

79 M.C.A. van Zuijlen. Empirical distributions and rank statistics. 1977.

80 P.W. Hemker. A numerical study of stiff two-point boundary

problems. 1977. 81 K.R. Apt, J.W. de Bakker (eds.). Foundations of computer science II, part 1. 1976.

82 K.R. Apt, J.W. de Bakker (eds.). Foundations of computer science II, part 2. 1976.

83 L.S. van Benthem Jutting. Checking Landau's "Grundlagen" in the AUTOMATH system. 1979.

4 H.L.L. Busard. The translation of the elements of Euclid from the Arabic into Latin by Hermann of Carinthia (?), books vii-xii. 1977.

85 J. van Mill. Supercompactness and Wallmann spaces. 1977. 86 S.G. van der Meulen, M. Veldhorst. Torrix I, a program-ming system for operations on vectors and matrices over arbi-trary fields and of variable size. 1978.

88 A. Schrijver. Matroids and linking systems. 1977.

89 J.W. de Roever. Complex Fourier transformation and ana-lytic functionals with unbounded carriers. 1978.

90 L.P.J. Groenewegen. Characterization of optimal strategies in dynamic games. 1981.

91 J.M. Geysel. Transcendence in fields of positive characteris-tic. 1979.

92 P.J. Weeda. Finite generalized Markov programming. 1979 93 H.C. Tijms, J. Wessels (eds.). Markov decision theory. 1977

94 A. Bijlsma. Simultaneous approximations in transcendental number theory, 1978.

95 K.M. van Hee. Bayesian control of Markov chains. 1978.

96 P.M.B. Vitányi. Lindenmayer systems: structure, languages, and growth functions. 1980.

97 A. Federgruen. Markovian control problems; functional equations and algorithms. 1984.

98 R. Geel. Singular perturbations of hyperbolic type. 1978. 99 J.K. Lenstra, A.H.G. Rinnooy Kan, P. van Emde Boas (eds.). Interfaces between computer science and operations research. 1978.

100 P.C. Baayen, D. van Dulst, J. Oosterhoff (eds.). Proceed-ings bicentennial congress of the Wiskundig Genootschap, part 1. 1979.

101 P.C. Baayen, D. van Dulst, J. Oosterhoff (eds.). Proceed-ings bicentennial congress of the Wiskundig Genootschap, part 2. 1979.

102 D. van Dulst. Reflexive and superreflexive Banach spaces.

103 K. van Harn. Classifying infinitely divisible distributions by functional equations. 1978.

104 J.M. van Wouwe. GO-spaces and generalizations of metri-zability. 1979.

105 R. Helmers. Edgeworth expansions for linear combinations of order statistics. 1982.

106 A. Schrijver (ed.). Packing and covering in combinatorics. 1979.

107 C. den Heijer. The numerical solution of nonlinear opera-tor equations by imbedding methods. 1979.

108 J.W. de Bakker, J. van Leeuwen (eds.). Foundations of computer science III, part 1. 1979. 109 J.W. de Bakker, J. van Leeuwen (eds.). Foundations of computer science III, part 2. 1979.

110 J.C. van Vliet. ALGOL 68 transput, part 1: historical review and discussion of the implementation model. 1979.

111 J.C. van Vliet. ALGOL 68 transput, part 11: an implemen-tation model. 1979. 112 H.C.P. Berbee. Random walks with stationary increments and renewai theory. 1979.

113 T.A.B. Snijders. Asymptotic optimality theory for testing problems with restricted alternatives. 1979. problem

114 A.J.E.M. Janssen. Application of the Wigner distribution to harmonic analysis of generalized stochastic processes. 1979. 115 P.C. Baayen, J. van Mill (eds.). Topological structures 11, root 1, 1975.

First A. Balyen, S. Van Will (eds.). Topological structures 11, part 1. 1979.
 116 P.C. Baayen, J. van Mill (eds.). Topological structures 11, part 2. 1979.

. 117 P.J.M. Kallenberg. Branching processes with continuous state space. 1979.

118 P. Groeneboom. Large deviations and asymptotic efficiencies. 1980.

119 F.J. Peters. Sparse matrices and substructures, with a novel implementation of finite element algorithms. 1980.

120 W.P.M. de Ruyter. On the asymptotic analysis of large scale ocean circulation. 1980.

121 W.H. Haemers. Eigenvalue techniques in design and graph theory. 1980.

122 J.C.P. Bus. Numerical solution of systems of nonlinear equations. 1980.

123 I. Yuhász. Cardinal functions in topology - ten years later.

124 R.D. Gill. Censoring and stochastic integrals. 1980.

125 R. Eising. 2-D systems, an algebraic approach. 1980. 126 G. van der Hoek. *Reduction methods in nonlinear pro-*gramming. 1980.

127 J.W. Klop. Combinatory reduction systems. 1980.

128 A.J.J. Talman. Variable dimension fixed point algorithms and triangulations. 1980.

129 G. van der Laan. Simplicial fixed point algorithms. 1980. 130 P.J.W. ten Hagen, T. Hagen, P. Klint, H. Noot, H.J. Sint, A.H. Veen. *ILP: intermediate language for pictures.* 

1980

131 R.J.R. Back. Correctness preserving program refinements: proof theory and applications. 1980.

132 H.M. Mulder. The interval function of a graph. 1980.

133 C.A.J. Klaassen. Statistical performance of location esti-mators. 1981.

134 J.C. van Vliet, H. Wupper (eds.). Proceedings interna-tional conference on ALGOL 68. 1981.

135 J.A.G. Groenendijk, T.M.V. Janssen, M.J.B. Stokhof (eds.). Formal methods in the study of language, part I. 1981. 136 J.A.G. Groenendijk, T.M.V. Janssen, M.J.B. Stokhof (eds.). Formal methods in the study of language, part II. 1981.

137 J. Telgen. Redundancy and linear programs. 1981. 138 H.A. Lauwerier. Mathematical models of epidemics. 1981.

139 J. van der Wal. Stochastic dynamic programming, succes-sive approximations and nearly optimal strategies for Markov decision processes and Markov games. 1981.

140 J.H. van Geldrop. A mathematical theory of pure exchange economies without the no-critical-point hypothesis. 1981.

141 G.E. Welters. Abel-Jacobi isogenies for certain types of Fano threefolds. 1981.

142 H.R. Bennett, D.J. Lutzer (eds.). Topology and order structures, part 1. 1981.

143 J.M. Schumacher. Dynamic feedback in finite- and infinite-dimensional linear systems. 1981.

144 P. Eijgenraam. The solution of initial value problems using interval arithmetic; formulation and analysis of an algorithm. 1981.

145 A.J. Brentjes. Multi-dimensional continued fraction algo-rithms. 1981.

146 C.V.M. van der Mee. Semigroup and factorization methods in transport theory. 1981.

147 H.H. Tigelaar. Identification and informative sample size. 1087

148 L.C.M. Kallenberg. Linear programming and finite Mar-kovian control problems, 1983.

149 C.B. Huijsmans, M.A. Kaashoek, W.A.J. Luxemburg, W.K. Vietsch (eds.). From A to Z, proceedings of a symposium in honour of A.C. Zaanen. 1982.

150 M. Veldhorst. An analysis of sparse matrix storage schemes, 1982.

151 R.J.M.M. Does. Higher order asymptotics for simple linear rank statistics. 1982.

152 G.F. van der Hoeven. Projections of lawless sequencies. 1982.

153 J.P.C. Blanc. Application of the theory of boundary value problems in the analysis of a queueing model with paired services. 1982.

Vices. 1962.
154 H.W. Lenstra, Jr., R. Tijdeman (eds.). Computational methods in number theory, part I. 1982.
155 H.W. Lenstra, Jr., R. Tijdeman (eds.). Computational methods in number theory, part II. 1982.

156 P.M.G. Apers. Query processing and data allocation in distributed database systems. 1983.

157 H.A.W.M. Kneppers. The covariant classification of two-dimensional smooth commutative formal groups over an alge-braically closed field of positive characteristic. 1983.

158 J.W. de Bakker, J. van Leeuwen (eds.). Foundations of computer science IV, distributed systems, part 1. 1983.

159 J.W. de Bakker, J. van Leeuwen (eds.). Foundations of computer science IV, distributed systems, part 2. 1983.

160 A. Rezus. Abstract AUTOMATH. 1983.

161 G.F. Helminck. Eisenstein series on the metaplectic group, an algebraic approach. 1983.

162 J.J. Dik. Tests for preference. 1983. 163 H. Schippers. Multiple grid methods for equations of the second kind with applications in fluid mechanics. 1983.

164 F.A. van der Duyn Schouten. Markov decision processes with continuous time parameter. 1983.

165 P.C.T. van der Hoeven. On point processes. 1983. 166 H.B.M. Jonkers. Abstraction, specification and implemen-tation techniques, with an application to garbage collection. 1983.

167 W.H.M. Zijm. Nonnegative matrices in dynamic program-ming. 1983.

168 J.H. Evertse. Upper bounds for the numbers of solutions of diophantine equations. 1983.

169 H.R. Bennett, D.J. Lutzer (eds.). Topology and order structures, part 2. 1983.